WASHINGTON, D.C. 20549

SCHEDULE TO (RULE 14d-100)

TENDER OFFER STATEMENT UNDER SECTION 14(d)(1) OR 13(e)(1)

OF THE SECURITIES EXCHANGE ACT OF 1934

ODWALLA, INC.

(Name of Subject Company (Issuer))

TCCC ACQUISITION CORP.

AND

THE COCA-COLA COMPANY

(OFFERORS)

(Names of Filing Persons (identifying status as offeror, issuer or other

person))

COMMON STOCK, NO PAR VALUE PER SHARE

(Title of Class of Securities)

676111107

(Cusip Number of Class of Securities)

THE COCA-COLA COMPANY

AND

TCCC ACQUISITION CORP.

C/O THE COCA-COLA COMPANY

ONE COCA-COLA PLAZA

ATLANTA, GEORGIA 30313

ATTENTION: CHIEF FINANCIAL OFFICER AND GENERAL COUNSEL

TELEPHONE: 404-676-2121

(Name, Address and Telephone Number of Person Authorized

to Receive Notices and Communications on Behalf of Bidders)

Copies To:

C. WILLIAM BAXLEY, ESQ.

ALANA L. GRIFFIN, ESQ.

KING & SPALDING

191 PEACHTREE STREET

ATLANTA, GEORGIA 30303-1763

TELEPHONE: (404) 572-4600

NOVEMBER 6, 2001

CALCULATION OF FILING FEE

<Table> <Caption>

\_\_\_\_\_\_

TRANSACTION VALUATION\*

AMOUNT OF FILING FEE

\$39,839

\$199,191,108 \_\_\_\_\_\_

</Table>

- $^{\star}$  For the purpose of calculating the fee only, this amount assumes the purchase of 13,060,712 shares of Common Stock, no par value per share, of Odwalla at a purchase price of \$15.25 per share. Such number includes all outstanding shares as of October 25, 2001, and assumes the exercise of all in-the-money stock options and warrants to purchase shares of Common Stock which are outstanding as of such date.
- [X] Check the box if any part of the fee is offset as provided by Rule 0-11(a)(2) and identify the filing with which the offsetting fee was previously paid. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.

<Table> <S>

<C>

Amount Previously Paid: \$39,839 Form or Registration No.: Schedule TO-C

Filing Party: TCCC Acquisition Corp. and The Coca-Cola Company

<C>

Date Filed: October 30, 2001

</Table>

[ ] Check the box if the filing relates solely to preliminary communications made before the commencement of a tender offer.

Check the appropriate boxes below to designate any transactions to which the statement relates:

[X] third-party tender offer subject to Rule 14d-1.

[ ] issuer tender offer subject to Rule 13e-4.

[ ] going-private transaction subject to Rule 13e-3.

[ ] amendment to Schedule 13D under Rule 13d-2.

Check the following box if the filing is a final amendment reporting the results of the tender offer:  $[\ ]$ 

The information required in the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

This Tender Offer Statement on Schedule TO relates to the offer by TCCC Acquisition Corp., a California corporation formerly known as Perry Phillip Corp. (the "Offeror"), and a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"), to purchase all of the outstanding shares of common stock, no par value per share (the "Shares"), of Odwalla, Inc., a California corporation ("Odwalla"), at a purchase price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest thereon, upon the terms and subject to the conditions set forth in the Offer to Purchase, dated November 6, 2001 (the "Offer to Purchase"), a copy of which is attached hereto as Exhibit (a) (1), and in the related Letter of Transmittal (which, together with the Offer to Purchase, as amended from time to time, constitute the "Offer"), a copy of which is attached hereto as Exhibit (a) (2).

The Offeror and TCCC have entered into Tender Agreements, dated as of October 29, 2001 (the "Tender Agreements"), with certain shareholders of Odwalla (the "Tendering Shareholders"), pursuant to which the Tendering Shareholders have agreed to tender an aggregate of 6,280,594 Shares owned by them (the "Committed Shares") pursuant to the Offer. In addition, certain of the Tendering Shareholders have agreed to vote 5,950,650 of such Committed Shares in favor of the Merger (as defined in the Offer to Purchase) and otherwise in the manner directed by TCCC in connection with the transaction. Additional information about the Tender Agreements is contained in Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement") of the Offer to Purchase.

The information in the Offer to Purchase, including all schedules and annexes thereto, is hereby expressly incorporated herein by reference in response to all the Items of this Schedule TO.

#### ITEM 1. SUMMARY TERM SHEET.

The information set forth in the Summary Term Sheet in the Offer to Purchase is incorporated herein by reference.

### TTEM 2. SUBJECT COMPANY INFORMATION.

The name of the subject company is Odwalla, Inc., a California corporation, which has its principal executive offices at 120 Stone Pine Road, Half Moon Bay, California 94019 and may be contacted by telephone at (650) 726-1888.

The exact title of the class of equity securities being sought in the Offer is the common stock, no par value per share, of Odwalla. The information set forth in the introduction (the "Introduction") of the Offer to Purchase is incorporated herein by reference. As of October 25, 2001, there were 11,103,321 Shares issued and outstanding.

The information concerning the principal market in which the Shares are traded and certain high and low sales prices for the Shares in such principal market set forth in Section 6 ("Price Range of Shares; Dividends") of the Offer to Purchase is incorporated herein by reference.

## ITEM 3. IDENTITY AND BACKGROUND OF FILING PERSON.

This Tender Offer Statement is filed by the Offeror and TCCC. Information regarding the name, business address, business telephone number, description of business, state or other place of organization of both the Offeror and TCCC is incorporated herein by reference from Section 8 ("Certain Information Concerning the Offeror and TCCC") of the Offer to Purchase.

Information regarding the name, business address, business telephone number, current principal occupation or employment as well as past material occupations, positions, offices or employment held during the last five years (including past addresses corresponding to each of them, if material only) for each of the executive officers and directors of the Offeror and TCCC is incorporated herein by reference from Annex I of the Offer to Purchase. Except as set forth in Annex I hereto, each of the executive officers and directors of the Offeror and TCCC is a citizen of the United States.

A statement regarding the absence of any criminal proceeding, conviction or violation of federal or state securities laws by any of the Offeror, TCCC or any current executive officer or director of the Offeror or

TCCC is set forth in Section 8 ("Certain Information Concerning the Offeror and TCCC") and is incorporated herein by reference.

## ITEM 4. TERMS OF THE TRANSACTION.

The information set forth in the Offer to Purchase is incorporated herein by reference.

#### ITEM 5. PAST CONTACTS, TRANSACTIONS, NEGOTIATIONS AND AGREEMENTS.

During the past two years, there have been no transactions that would be required to be disclosed under this Item 5 between any of the Offeror or TCCC or, to the best knowledge of the Offeror and TCCC, any of the persons listed on Annex I to the Offer to Purchase, and Odwalla or any of its executive officers, directors or affiliates.

The information set forth in the Introduction, Section 8 ("Certain Information Concerning the Offeror and TCCC") and Section 10 ("Background of the Offer; Contacts with Odwalla") and Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement") of the Offer to Purchase and in Exhibits (d)(1), (d)(2), (d)(3), (d)(4) and (d)(5) of this Schedule TO is incorporated herein by reference.

#### ITEM 6. PURPOSES OF THE TRANSACTION AND PLANS OR PROPOSALS.

The information set forth in the Introduction, Section 10 ("Background of the Offer; Contacts with Odwalla"), Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement"), Section 12 ("Purpose of the Offer; the Merger; Plans for Odwalla") and Section 13 ("Dividends and Distributions") of the Offer to Purchase is incorporated herein by reference. The information set forth in Section 14 ("Effect of the Offer on the Market for the Shares, Stock Exchange Listing and Exchange Act Registration") of the Offer to Purchase is incorporated herein by reference.

#### ITEM 7. SOURCE AND AMOUNT OF FUNDS OR OTHER CONSIDERATION.

The information set forth in Section 9 ("Source and Amount of Funds") of the Offer to Purchase is incorporated herein by reference.

#### ITEM 8. INTEREST IN SECURITIES OF THE SUBJECT COMPANY.

The information set forth in the Introduction and Section 8 ("Certain Information Concerning the Offeror and TCCC") of the Offer to Purchase and Annex I to the Offer to Purchase is incorporated herein by reference.

### ITEM 9. PERSONS/ASSETS RETAINED, EMPLOYED, COMPENSATED OR USED.

The information set forth in the Introduction and Section 17 ("Fees and Expenses") of the Offer to Purchase is incorporated herein by reference.

## ITEM 10. FINANCIAL STATEMENTS.

Not applicable.

### ITEM 11. ADDITIONAL INFORMATION.

The information set forth in Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement"), Section 14 ("Effect of the Offer on the Market for the Shares, Stock Exchange Listing and Exchange Act Registration"), Section 16 ("Certain Legal Matters and Regulatory Approvals") and Section 18 ("Miscellaneous") of the Offer to Purchase is incorporated herein by reference.

The information set forth in the entire text of each of (i) the Offer to Purchase and (ii) the Letter of Transmittal is incorporated herein by reference.

### ITEM 12. EXHIBITS.

<Table>
<Caption>
EXHIBIT
NO.

# O. DESCRIPTION

- <S> <C>
- (a) (1) Offer to Purchase dated November 6, 2001.
- (a) (2) Form of Letter of Transmittal.
- (a) (3) Form of Letter from the Information Agent to Brokers,
  Dealers, Commercial Banks, Trust Companies and Other
- (a) (4) Form of Letter to Clients for use by Brokers, Dealers, Commercial Banks, Trust Companies and Other Nominees.
- (a)(5) Form of Notice of Guaranteed Delivery.
- (a) (6) Guidelines for Certification of Taxpayer Identification Number on substitute Form W-9.
- (a)(7) Instructions for Form W-8BEN.
- (a) (8) Summary Advertisement as published on November 6, 2001.
- (a) (9) Press Release jointly issued by Odwalla and TCCC on October
- (a) (10) Press Release issued by TCCC on November 6, 2001.

- (b) None
- (d) (1) Agreement and Plan of Merger, dated October 29, 2001, by and among The Coca-Cola Company, Perry Phillip Corp. and Odwalla, Inc. (Incorporated by reference from Appendix A to the Offer to Purchase filed as Exhibit (a) (1) hereto.)
- (d) (2) Form of Tender Agreement with voting agreement, dated October 29, 2001, by and among the Tendering Shareholders, The Coca-Cola Company and Perry Phillip Corp. (Incorporated by reference from Appendix B to the Offer to Purchase filed as Exhibit (a) (1) hereto.)
- (d) (3) Form of Tender Agreement without voting agreement, dated October 29, 2001, by and among the Tendering Shareholders, The Coca-Cola Company and Perry Phillip Corp. (Incorporated by reference from Appendix C to the Offer to Purchase filed as Exhibit (a) (1) hereto.)
- (d) (4) Stock Option Agreement dated as of October 29, 2001, by and among Odwalla, Inc., The Coca-Cola Company and Perry Phillip Corp. (Incorporated by reference from Appendix D to the Offer to Purchase filed as Exhibit (a)(1) hereto.)
- (d)(5) Employment Agreement, made October 29, 2001 by and between Odwalla, Inc. and D. Stephen C. Williamson.
- (g) Not applicable.
- (h) Not applicable.

</Table>

ITEM 13. INFORMATION REQUIRED BY SCHEDULE 13E-3.

Not applicable.

#### SIGNATURE

After due inquiry and to the best of my knowledge and belief, I certify that the information set forth in this Statement is true, complete and correct.

THE COCA-COLA COMPANY

By: /s/ DAVID M. TAGGART

\_\_\_\_\_

Name: David M. Taggart

Title: Vice President and Treasurer

TCCC ACQUISITION CORP.

By: /s/ SHAWN SUGARMAN

-----

Name: Shawn Sugarman

Title: Chief Financial Officer

Date: November 6, 2001

 $\begin{array}{c} \hbox{Offer To Purchase For Cash} \\ \hbox{All Outstanding Shares Of Common Stock} \end{array}$ 

οf

Odwalla, Inc.

\$15.25 Net Per Share by

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT, SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

-----

THE OFFER (THE "OFFER") IS BEING MADE IN CONNECTION WITH THE AGREEMENT AND PLAN OF MERGER (THE "MERGER AGREEMENT"), DATED AS OF OCTOBER 29, 2001, BY AND AMONG THE COCA-COLA COMPANY ("TCCC"), TCCC ACQUISITION CORP. (FORMERLY KNOWN AS PERRY PHILLIP CORP., THE "OFFEROR") AND ODWALLA, INC. ("ODWALLA"). THE OFFER IS CONDITIONED UPON, AMONG OTHER THINGS, (I) THERE BEING VALIDLY TENDERED AND NOT WITHDRAWN PRIOR TO THE EXPIRATION OF THE OFFER SUCH NUMBER OF SHARES OF COMMON STOCK OF ODWALLA (THE "SHARES") THAT WOULD CONSTITUTE AT LEAST NINETY AND ONE-TENTH PERCENT (90.1%) OF THE SHARES THEN OUTSTANDING, INCLUDING FOR PURPOSES OF SUCH CALCULATION SHARES THAT ARE ISSUABLE UPON EXERCISE OF VESTED OPTIONS AND OUTSTANDING WARRANTS BUT EXCLUDING SHARES THAT ARE ISSUABLE UPON THE EXERCISE OF OPTIONS AND WARRANTS THAT ARE CANCELLED PURSUANT TO THE MERGER AGREEMENT (THE FOREGOING BEING REFERRED TO HEREIN AS "FULLY DILUTED BASIS"), (II) ANY WAITING PERIOD UNDER THE HSR ACT (AS DEFINED HEREIN) APPLICABLE TO THE PURCHASE OF SHARES PURSUANT TO THE OFFER HAVING EXPIRED OR HAVING BEEN TERMINATED PRIOR TO THE EXPIRATION OF THE OFFER, AND (III) THE SATISFACTION OR WAIVER OF CERTAIN OTHER TERMS AND CONDITIONS. SEE SECTION 15.

IF MORE THAN 50% OF THE OUTSTANDING SHARES AND LESS THAN 90.1% OF THE SHARES THEN OUTSTANDING ON A FULLY DILUTED BASIS ARE TENDERED PURSUANT TO THE OFFER AND NOT WITHDRAWN, THE OFFEROR AND TCCC MAY, UNDER CERTAIN CIRCUMSTANCES DESCRIBED BELOW, EITHER EXERCISE THE OPTION PURSUANT TO THE STOCK OPTION AGREEMENT DESCRIBED HEREIN OR REDUCE THE NUMBER OF SHARES SUBJECT TO THE OFFER TO A NUMBER EQUAL TO 49.9% OF THE SHARES THEN OUTSTANDING.

IN CONNECTION WITH THE MERGER AGREEMENT, THE OFFEROR AND TCCC ENTERED INTO TENDER AGREEMENTS DATED AS OF OCTOBER 29, 2001 (THE "TENDER AGREEMENTS"), WITH CERTAIN SHAREHOLDERS OF ODWALLA (THE "TENDERING SHAREHOLDERS") WHO OWN APPROXIMATELY 57% OF THE SHARES THAT ARE OUTSTANDING. PURSUANT TO THE TENDER AGREEMENTS, SUCH SHAREHOLDERS HAVE AGREED TO TENDER SUCH SHARES PURSUANT TO THE OFFER.

-----

THE BOARD OF DIRECTORS OF ODWALLA HAS UNANIMOUSLY APPROVED AND ADOPTED THE MERGER AGREEMENT AND THE OPTION AGREEMENT REFERRED TO HEREIN AND THE TRANSACTIONS CONTEMPLATED THEREBY, APPROVED THE OFFER AND THE MERGER (AS DEFINED HEREIN) AND DETERMINED THAT THE TERMS OF THE OFFER, THE MERGER AND THE OPTION AGREEMENT ARE FAIR TO AND IN THE BEST INTERESTS OF ODWALLA'S SHAREHOLDERS AND RECOMMENDS THAT ALL SHAREHOLDERS ACCEPT THE OFFER AND TENDER THEIR SHARES TO THE OFFEROR.

# IMPORTANT

Any shareholder desiring to tender all or any portion of such shareholder's Shares should either (1) complete and sign the accompanying Letter of Transmittal (or a facsimile thereof) in accordance with the instructions in the Letter of Transmittal and deliver the Letter of Transmittal with the Shares and all other required documents to the Depositary (as hereinafter defined), (2) follow the procedure for book-entry transfer set forth in Section 3 of this Offer to Purchase or (3) request such shareholder's broker, dealer, commercial bank, trust company or other nominee to effect the transaction for such shareholder. Shareholders having Shares registered in the name of a broker, dealer, commercial bank, trust company or other nominee must contact such broker, dealer, commercial bank, trust company or other nominee if they desire to tender their Shares.

Any shareholder who desires to tender Shares and whose certificates representing such Shares are not immediately available, or who cannot comply with the procedure for book-entry transfer on a timely basis or who cannot deliver all required documents to the Depositary, in each case prior to the expiration of the Offer, must tender Shares pursuant to the guaranteed delivery procedure set forth in Section 3 of this Offer to Purchase.

Questions and requests for assistance may be directed to the Dealer Manager or to the Information Agent at their respective addresses and telephone numbers set forth on the back cover of this Offer to Purchase. Additional copies of this Offer to Purchase, the Letter of Transmittal, the Notice of Guaranteed Delivery and other related materials may be obtained from the Information Agent or from brokers, dealers, commercial banks, trust companies or other nominees.

-----

#### The Dealer Manager for the Offer is:

#### MORGAN STANLEY

November 6, 2001

#### TABLE OF CONTENTS

<Table> <Caption>

		PAGE
<s></s>		<c></c>
SUMMAI	RY TERM SHEET	1
INTRO	DUCTION	6
THE T	ENDER OFFER	10
1.	Terms of the Offer; Expiration Date	10
2.	Acceptance for Payment and Payment for Shares	11
3.	Procedure for Accepting the Offer and Tendering	
	Shares	12
4.	Withdrawal Rights	15
5.	Certain United States Federal Income Tax	
	Consequences	16
6.	Price Range of Shares; Dividends	17
7.	Certain Information Concerning Odwalla	17
8.	Certain Information Concerning the Offeror and TCCC	23
9.	Source and Amount of Funds	24
10.	Background of the Offer; Contacts with Odwalla	25
11.	The Merger Agreement; the Option Agreement; the Tender	
	Agreements; and the Employment Agreement	29
12.	Purpose of the Offer; the Merger; Plans for Odwalla	40
13.	Dividends and Distributions	42
14.	Effect of the Offer on the Market for the Shares,	
	Stock Exchange Listing and Exchange Act	
	Registration	43
15.	Certain Conditions of the Offer	44
16.	Certain Legal Matters and Regulatory Approvals	46
17.	Fees and Expenses	48
18.		49
	ICertain Information Concerning the Directors and	- 1
	cutive Officers of the Offeror and TCCC	I-1
	IISchedule BChapter 13 of the California General	TT 1
-	poration LawDissenters' Rights	II-1
	dix AAgreement and Plan of Merger dix BForm of Tender Agreement with Voting Agreement	
	dix BForm of Tender Agreement with voting Agreement  dix CForm of Tender Agreement without Voting	
	eement	
	dix DStock Option Agreement	
<td>± 3</td> <td></td>	± 3	

i

### SUMMARY TERM SHEET

TCCC Acquisition Corp. (formerly known as Perry Phillip Corp., the "Offeror") is offering to purchase all of the outstanding shares of common stock of Odwalla, Inc. ("Odwalla") for \$15.25 per share, net to the seller in cash, less any required withholding taxes and without interest thereon (the "Offer"). Through a question and answer format, this Summary Term Sheet is designed to explain to you, the shareholders of Odwalla, a number of important terms of the proposed transaction. This Summary Term Sheet serves only as an introduction, and we urge you to read carefully the remainder of the Offer to Purchase and the accompanying Letter of Transmittal in order to fully educate yourself on the details of the proposed transaction. Cross-referenced text refers to sections within the Offer to Purchase, unless otherwise noted.

## WHO IS OFFERING TO BUY THE COMMON STOCK OF ODWALLA?

Our name is TCCC Acquisition Corp. We are a California corporation formerly known as Perry Phillip Corp. formed for the purpose of making a cash tender offer for all of the outstanding shares of common stock of Odwalla. We are a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"). See "Introduction."

TCCC is the largest manufacturer, distributor and marketer of soft drink concentrates and syrups in the world. Finished beverage products bearing TCCC's trademarks, sold in the United States since 1886, are now sold in nearly 200

countries and include the leading soft drink products in most of these countries

TCCC's business is nonalcoholic beverages, principally soft drinks, but also a variety of noncarbonated beverages, including juice and juice-drink products. TCCC is one of numerous competitors in the commercial beverages market. Of the approximately 48 billion beverage servings of all types consumed worldwide every day, beverages bearing TCCC's trademarks account for more than one billion.

TCCC was incorporated in September 1919 under the laws of the State of Delaware and succeeded to the business of a Georgia corporation with the same name that had been organized in 1892.

WHAT ARE THE CLASSES AND AMOUNTS OF SECURITIES SOUGHT IN THE OFFER? HOW MUCH ARE YOU OFFERING TO PAY AND WHAT IS THE FORM OF PAYMENT?

We are offering to purchase all of the outstanding shares of common stock of Odwalla for \$15.25 per share, net to you, in cash less any required withholding taxes and without interest. See "Introduction."

If you tender your shares in the Offer, you will not be obligated to pay brokerage fees or commissions or, except as otherwise provided in Instruction 6 of the Letter of Transmittal, stock transfer taxes with respect to the sale of your shares. See "Introduction."

#### WHAT IS THE PURPOSE OF THE OFFER?

The purpose of the tender offer is to enable TCCC to acquire control of, and the entire equity interest in, Odwalla. Following the Offer, we intend to acquire any remaining equity interest in Odwalla that was not acquired during the tender offer. See "Introduction" and Section 12 ("Purpose of the Offer; the Merger; Plans for Odwalla").

#### WHAT ARE THE MOST SIGNIFICANT CONDITIONS TO THE OFFER?

We are not obligated to purchase any shares that you validly tender unless the number of shares validly tendered and not withdrawn before the expiration date of the Offer represents, in the aggregate, at least ninety and one tenth percent (90.1%) of the shares of Odwalla's common stock, including for purposes of such calculation shares that are issuable upon exercise of vested options and outstanding warrants but excluding shares that are issuable upon the exercise of options and warrants that are cancelled pursuant to the Merger Agreement (the foregoing being referred to herein as "Fully Diluted Basis"). For a complete description of all of the conditions of the Offer, see Section 15 ("Certain Conditions of the Offer"). If more than 50% of the outstanding shares and less than 90.1% of the shares then outstanding on a Fully Diluted Basis are validly

tendered pursuant to the Offer and not withdrawn, the Offeror and TCCC may, under certain circumstances described in this document, either exercise the option agreement described in this document and purchase enough shares of Odwalla common stock such that the Offeror would own at least 90.1% of all Odwalla common stock on a Fully Diluted Basis or reduce the number of shares subject to the Offer to a number equal to 49.9% of the shares then outstanding. For a complete description of all of the conditions of the Offer, see Section 15 ("Certain Conditions of the Offer"). If the number of shares subject to the Offer is reduced, we will reduce the amount of shares we will purchase from you by the same proportion we reduce the amount purchased from all other tendering shareholders. If the subsequent merger between the Offeror and Odwalla occurs (the "Merger"), you will receive the Offer Price as the merger consideration, subject to your right, if any, under California law to dissent and demand the fair market value of your shares, for your shares which were either not tendered or tendered but not purchased in the Offer. See Section 12 ("Purpose of the Offer; the Merger; Plans for Odwalla").

We are also not obligated to purchase any shares which you validly tender if, among other things, the applicable waiting period under applicable antitrust laws has not expired or been terminated. See Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement").

We are also not obligated to purchase any shares you validly tender if any other conditions set forth in Section 15 ("Certain Conditions of the Offer") and discussed in Section 1 ("Terms of the Offer; Expiration Date") is not satisfied or waived.

## HOW LONG DO I HAVE TO DECIDE WHETHER TO TENDER IN THE OFFER?

Our offer to purchase your shares expires at 12:00 midnight, San Francisco, California time, on Thursday, December 6, 2001. This is called the initial expiration date. See Section 1 ("Terms of the Offer; Expiration Date"). If you cannot deliver everything that is required in order to make a valid tender by that time, you may be able to use a guaranteed delivery procedure which is described in Section 3 of this Offer to Purchase.

CAN WE EXTEND THE OFFER PAST THE INITIAL EXPIRATION DATE AND UNDER WHAT

Yes. We can and may extend the Offer past the initial expiration date. If we do so, you will be able to tender your shares until 12:00 midnight, New York time, on the new expiration date.

Several terms define the circumstances in which we can extend the Offer, including: (1) if any conditions to the Offer have not been satisfied or waived, or (2) if any rule, requirement, interpretation or position of the Securities and Exchange Commission requires us to extend the Offer. Furthermore, if certain conditions in the Merger Agreement are not satisfied as of any scheduled expiration date of the Offer, Odwalla may request that we extend the Offer, subject to applicable termination rights. See Section 1 ("Terms of the Offer; Expiration Date").

HOW DO I FIND OUT IF THE OFFEROR EXTENDS THE OFFER?

We will announce an extension no later than 9:00~a.m., New York time, on the business day after a scheduled expiration date by issuing a press release to the Dow Jones News Service. See Section 1 ("Terms of the Offer; Expiration Date").

HOW DO I GET PAID FOR MY TENDERED SHARES?

We will pay for the shares accepted for payment by depositing the purchase price with EquiServe Trust Company, N.A. (who is the depositary in the Offer (the "Depositary")). The Depositary will transmit to you the payment for all shares accepted for payment. See Section 2 ("Acceptance for Payment and Payment for Shares").

2

#### HOW DO I TENDER MY SHARES?

To tender your shares in the Offer, you must:

- complete and sign the accompanying Letter of Transmittal (or a manually signed facsimile of the Letter of Transmittal) in accordance with the instructions on the Letter of Transmittal and mail or deliver it together with your share certificates, and any other required documents to the Depositary;
- -- tender your shares pursuant to the procedure for book-entry transfer set forth in Section 3 ("Procedure for Accepting the Offer and Tendering Shares"); or
- -- if your share certificates are not immediately available or if you cannot deliver your share certificates and any other required documents to the Depositary prior to the expiration of the offer, or you cannot complete the procedure for delivery by book-entry transfer on a timely basis, you may still tender your shares if you comply with the guaranteed delivery procedures described in Section 3 ("Procedure for Accepting the Offer and Tendering Shares").

# UNTIL WHAT TIME CAN I WITHDRAW MY PREVIOUSLY TENDERED SHARES?

You can withdraw your tendered shares at any time on or prior to a scheduled expiration date. After the Offer expires, the tender is irrevocable unless we have not accepted for payment your shares by January 5, 2002. At and after this date, you can withdraw your tendered shares unless and until we accept them for payment. See Section 4 ("Withdrawal Rights").

## HOW DO I WITHDRAW PREVIOUSLY TENDERED SHARES?

To withdraw your shares, you must deliver to the Depositary written or facsimile transmission notice of withdrawal that specifies your name, the number of shares being withdrawn and the name of the registered holder of the shares, if different from the person who tendered the shares. See Section 4 ("Withdrawal Rights").

WHAT ARE THE TAX CONSEQUENCES OF THE SALE OF SHARES TO THE OFFEROR THROUGH THE

The sale of shares to us through the Offer is a taxable transaction for United States federal income tax purposes, and may also be taxable under applicable state, local, foreign and other tax laws. In general, if you are a United States person, as defined under U.S. federal income tax law, you will recognize gain or loss equal to the difference between the amount of cash that you receive from us for the shares and the tax basis of your shares.

We encourage you to consult with your own tax advisor about the particular effect the Offer will have on you. See Section 5 ("Certain United States Federal Income Tax Consequences").

WHAT IS THE MARKET VALUE OF MY SHARES AS OF A RECENT DATE?

On October 29, 2001, the last full trading day before we announced that we entered into the Merger Agreement, the closing price per share of Odwalla's common stock on the Nasdaq National Market System was \$11.83 per share. On November 5, 2001, the last full trading day before we commenced the Offer, the closing price per share of Odwalla's common stock on Nasdaq was \$15.15 per share.

We encourage you to obtain a current market quotation for your shares before deciding whether to tender your shares. See Section 6 ("Price Range of Shares; Dividends") for recent high and low closing prices for the shares.

WHAT IS THE TOTAL AMOUNT OF FUNDS THAT WILL BE REQUIRED TO CONSUMMATE THE PROPOSED TRANSACTION?

We estimate that the total amount of funds required to purchase all outstanding shares (on a Fully Diluted Basis) pursuant to the Offer, to repay outstanding indebtedness and to pay fees and expenses and other obligations related to the Offer and the Merger will be approximately \$192.5 million.

3

#### DOES THE OFFEROR HAVE THE FINANCIAL RESOURCES TO MAKE PAYMENT?

Yes. We will obtain all necessary funds through capital contributions or advances by TCCC. TCCC will have sufficient funds from cash on hand to fully fund the Offer and the Merger. See Section 9 ("Source and Amount of Funds").

IS THE OFFEROR'S FINANCIAL CONDITION RELEVANT TO MY DECISION ON WHETHER TO TENDER SHARES IN THE OFFER?

We do not think our financial condition is relevant to your decision whether to tender shares and accept the Offer because: (1) the Offer consists solely of cash, (2) the Offer is not subject to any financing condition, (3) TCCC will have adequate cash available, and (4) the Offer is for all outstanding shares of Odwalla's common stock.

WHAT DOES ODWALLA'S BOARD OF DIRECTORS THINK OF THE OFFER AND MERGER?

On October 29, 2001, the board of directors of Odwalla unanimously determined that the Offer, the Option (as defined herein) and the Merger were fair to you and in your best interests.

Odwalla's board of directors recommends that you accept the Offer and tender your shares and/or vote to approve the Merger. See "Introduction" and Section 12 ("Purpose of the Offer; the Merger; Plans for Odwalla").

## HAVE ANY SHAREHOLDERS ALREADY AGREED TO TENDER THEIR SHARES?

Yes. Shareholders holding 6,280,594 shares of common stock of Odwalla (the "Committed Shares") have entered into tender agreements in which they have agreed to tender their shares in the Offer. Certain of these shareholders owning 5,950,650 shares of common stock of Odwalla have agreed to vote in favor of the Merger. See "Introduction" and Section 10 ("Background of the Offer; Contacts with Odwalla").

# IF THE OFFEROR CONSUMMATES THE OFFER, WHAT HAPPENS TO ODWALLA AFTER THE OFFER?

If the Offeror purchases at least 90.1% of the total shares outstanding on a Fully Diluted Basis in the Offer, it intends to cause a merger to occur between the Offeror and Odwalla in which shareholders who have not previously tendered their shares will also receive \$15.25 in cash, less any required withholding taxes and without interest, subject to their right, if any, under California law to dissent and demand the fair market value of their shares. If less than 90.1% of the shares then outstanding on a Fully Diluted Basis but more than 50% of the then outstanding shares are tendered in the Offer, the Offeror may either (a) purchase from Odwalla pursuant to the option agreement entered into on October 29, 2001 among TCCC, the Offeror and Odwalla enough shares so that the Offeror would own after completion of the Offer at least 90.1% of the total shares outstanding on a Fully Diluted Basis and immediately thereafter cause the Merger to occur (the "Option") or (b) purchase 49.9% of the shares then outstanding pursuant to the Offer and Odwalla will seek the approval of the holders of a majority of the Odwalla shareholders to consummate the merger of the Offeror with and into Odwalla. We and certain of the shareholders who have signed tender agreements representing 5,950,650 of the Committed Shares have agreed to vote in favor of the Merger.

Once the Merger takes place, Odwalla will no longer be publicly owned. Odwalla will become a wholly owned subsidiary of TCCC. In such case, Odwalla's common stock will no longer be traded through the Nasdaq or on any other securities exchange. See Introduction and Section 14 ("Effect of the Offer on the Market for the Shares, Stock Exchange Listing and Exchange Act Registration").

This Offer to Purchase also constitutes notice to you pursuant to Section

1110 of the California General Corporation Law (the "CGCL") that if we acquire at least 90.1% of the outstanding shares of Odwalla on a Fully Diluted Basis, we will cause the Merger to become effective without any further notice to Odwalla's shareholders. See Appendix A (Agreement and Plan of Merger).

4

IF I DECIDE NOT TO TENDER BUT THE OFFER IS SUCCESSFUL, OR IF NOT ALL OF THE SHARES I TENDERED ARE PURCHASED WHAT WILL HAPPEN TO MY SHARES?

If the Offer is successful and the Merger occurs, shareholders either who do not tender or who did not have all of the shares that they tendered purchased in the Offer will receive the merger consideration per share of Odwalla common stock described below, subject to their right, if any, under California law to dissent and demand the fair market value of their shares. See Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement"), Section 14 ("Effect of the Offer on the Market for the Shares, Stock Exchange Listing and Exchange Act Registration") and Section 16 ("Certain Legal Matters and Regulatory Approvals").

WHAT WILL BE PAID TO SHAREHOLDERS OF ODWALLA IN THE MERGER AND WHAT IS THE FORM OF PAYMENT?

Shareholders of Odwalla who exchange their shares of common stock in the Merger will receive the same amount they would have received had they tendered their shares in the Offer -- \$15.25 per share, net in cash less any required withholding taxes and without interest.

WHO CAN I TALK TO IF I HAVE QUESTIONS ABOUT THE OFFER?

Shareholders can call Innisfree M&A Incorporated, the information agent for the Offer, at 888-750-5834. In addition, shareholders can call Morgan Stanley & Co. Incorporated, the dealer manager for the Offer, at 212-761-3539.

5

TO THE SHAREHOLDERS OF ODWALLA:

#### INTRODUCTION

TCCC Acquisition Corp., a California corporation (formerly known as Perry Phillip Corp., the "Offeror") and a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"), hereby offers to purchase all of the outstanding shares of common stock, no par value (the "Shares"), of Odwalla, Inc., a California corporation ("Odwalla"), at a purchase price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest thereon (the "Offer Price"), upon the terms and subject to the conditions set forth in this Offer to Purchase and in the related Letter of Transmittal (which, together with any amendments or supplements hereto or thereto, collectively constitute the "Offer").

Tendering shareholders will not be obligated to pay brokerage fees or commissions or, except as set forth in the Letter of Transmittal, stock transfer taxes on the transfer and sale of Shares pursuant to the Offer. The Offeror will pay all fees and expenses of EquiServe Trust Company, N.A., which is acting as the depositary (in such capacity, the "Depositary"), Morgan Stanley & Co. Incorporated, which is acting as the dealer manager (in such capacity, the "Dealer Manager") and Innisfree M&A Incorporated, which is acting as the information agent (in such capacity, the "Information Agent"), incurred in connection with the Offer.

THE BOARD OF DIRECTORS OF ODWALLA (THE "BOARD OF DIRECTORS") HAS UNANIMOUSLY APPROVED THE OFFER AND THE MERGER (AS DEFINED HEREIN) AND APPROVED AND ADOPTED THE MERGER AGREEMENT AND THE OPTION AGREEMENT (EACH AS DEFINED HEREIN) AND THE TRANSACTIONS CONTEMPLATED THEREBY, INCLUDING THE OFFER, THE OPTION (AS DEFINED HEREIN) AND THE MERGER, AND DETERMINED THAT THE TERMS OF THE OFFER, THE OPTION AND THE MERGER, ARE FAIR TO AND IN THE BEST INTERESTS OF THE SHAREHOLDERS OF ODWALLA, AND RECOMMENDS THAT ALL HOLDERS OF SHARES ACCEPT THE OFFER AND TENDER THEIR SHARES PURSUANT TO THE OFFER.

The Board of Directors has received the written opinion dated October 29, 2001 of Dresdner Kleinwort Wasserstein, Inc. ("DrKW"), financial advisor to Odwalla, to the effect that, as of such date and based upon and subject to certain matters stated in such opinion, the Offer Price to be received in the Offer and the Merger by the holders of Shares (other than TCCC and its affiliates) is fair, from a financial point of view, to such holders. A copy of such opinion is attached to Odwalla's Solicitation/Recommendation Statement on Schedule 14D-9 (the "Schedule 14D-9") which is being distributed to the shareholders of Odwalla concurrently herewith. Holders of Shares are urged to read the opinion carefully in its entirety.

THE OFFER IS CONDITIONED UPON, AMONG OTHER THINGS, THERE BEING VALIDLY TENDERED AND NOT WITHDRAWN PRIOR TO THE EXPIRATION DATE (AS DEFINED HEREIN) AT LEAST NINETY AND ONE-TENTH PERCENT (90.1%) OF THE SHARES THEN OUTSTANDING, INCLUDING FOR PURPOSES OF SUCH CALCULATION SHARES THAT ARE ISSUABLE UPON

EXERCISE OF VESTED OPTIONS AND OUTSTANDING WARRANTS BUT EXCLUDING SHARES THAT ARE ISSUABLE UPON THE EXERCISE OF OPTIONS AND WARRANTS OF ODWALLA THAT ARE CANCELLED PURSUANT TO THE MERGER AGREEMENT (THE FOREGOING BEING REFERRED TO HERIN AS "FULLY DILUTED BASIS") (THE "MINIMUM CONDITION"). THE OFFER IS ALSO SUBJECT TO THE EXPIRATION OR TERMINATION OF ALL APPLICABLE WAITING PERIODS UNDER THE HART-SCOTT-RODINO ANTITRUST IMPROVEMENTS ACT OF 1976, AS AMENDED (THE "HSR ACT") AND THE SATISFACTION OR WAIVER OF CERTAIN OTHER TERMS AND CONDITIONS. SEE SECTIONS 1 AND 15.

The Offer is being made pursuant to an Agreement and Plan of Merger, dated October 29, 2001 (the "Merger Agreement"), among TCCC, the Offeror and Odwalla. The Merger Agreement provides, among other things, for the making of the Offer by the Offeror, and further provides that, following the completion of the Offer, upon the terms and subject to the conditions of the Merger Agreement and in accordance with Section 1110 of the California General Corporation Law (the "CGCL"), the Offeror will be merged with and into Odwalla (the "Merger") and each Share then outstanding (other than Shares owned by TCCC, the Offeror, their direct or indirect subsidiaries and dissenting shareholders, if any) will be converted into the Offer Price less any required withholding taxes and without interest thereon. Following the Merger, Odwalla will continue as the surviving corporation (the "Surviving Corporation") and become a wholly owned subsidiary of TCCC, and the separate corporate existence of the Offeror will cease.

6

The purpose of the Offer is for TCCC, indirectly through the Offeror, to acquire a voting interest in Odwalla as the first step in a business combination. In the event the Minimum Condition is not satisfied on any scheduled expiration date of the Offer, at the request of Odwalla, the Offeror shall, and TCCC shall cause the Offeror to take any of the actions set forth in clauses (x), (y), or (z) (provided that the Offeror shall have complete discretion concerning which action to take): (x) extend the Offer pursuant to the provisions of the Merger Agreement, (y) amend the Offer in contemplation of the exercise of the Option Agreement (to the extent the Option Agreement is exercisable at such time) to reduce the Minimum Condition to that number of Shares (the "Option Exercise Minimum Number") equal to the number of Shares which, when combined with the number of shares issued upon exercise of the Option Agreement, equals 90.1% of the Shares then outstanding on a Fully Diluted Basis or (z) amend the Offer to provide that, if (i) the Minimum Condition is not satisfied at the next scheduled expiration date of the Offer (after giving effect to the issuance of any Shares theretofore acquired by TCCC or the Offeror) and (ii) the number of Shares tendered pursuant to the Offer and not withdrawn as of such next scheduled expiration date is more than 50% of the then outstanding Shares, the Offeror shall waive the Minimum Condition and amend the Offer to reduce the number of Shares subject to the Offer to 49.9% of the Shares then outstanding (the "Revised Minimum Number") and, subject to the prior satisfaction or waiver of the other conditions of the Offer, purchase, on a pro rata basis, the Revised Minimum Number of Shares (it being understood that the Offeror shall not in any event be required to accept for payment, or pay for, any Shares less than the Revised Minimum Number of Shares that are tendered pursuant to the Offer and not withdrawn at the expiration date).

The consummation of the Merger is subject to the satisfaction or waiver of certain conditions, including the approval and adoption of the Merger Agreement by the requisite vote of the shareholders of Odwalla, if required by the CGCL. Under the CGCL, if TCCC or the Offeror acquires, pursuant to the Offer, the Option Agreement or otherwise, at least 90% of the Shares then outstanding, it will be able to effect the Merger without a vote of the shareholders. In such event, TCCC, the Offeror and Odwalla have agreed in the Merger Agreement to take, subject to the satisfaction or waiver of the conditions set forth in the Merger Agreement, all necessary and appropriate action to cause the Merger to be effective as soon as practicable after the acceptance for payment and purchase of Shares pursuant to the Offer, without a meeting of shareholders of Odwalla, in accordance with Section 1110 of the CGCL (the "Short-Form Merger"). Under the CGCL, the merger consideration paid to the shareholders of Odwalla may not be cash if TCCC and the Offeror own, directly or indirectly, more than 50% but less than 90% of the then outstanding Shares, unless either all the shareholders of Odwalla consent to the Merger or the Commissioner of Corporations of the State of California approves, after a hearing, the terms and conditions of the Merger and the fairness thereof. If, pursuant to the Offer, the Option Agreement or otherwise, the Offeror does not acquire Shares that, taken together with Shares owned by TCCC, represent at least 90.1% of the Shares then outstanding on a Fully Diluted Basis as of any scheduled expiration date of the Offer, and the Offeror instead amends the Offer to reduce the number of Shares subject to the Offer to the Revised Minimum Number, then the Offeror, together with TCCC, would own 49.9% of the Shares then outstanding upon consummation of the Offer, and would thereafter solicit the approval of the Merger and the Merger Agreement by a majority vote of the shareholders of Odwalla. Under such circumstances, a significantly longer period of time will be required to effect the Merger. For a description of the conditions set forth in the Merger Agreement and the CGCL as it relates to this transaction, see Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement"), Section 12 ("Purpose of the Offer; the Merger; Plans for Odwalla") and Section 15 ("Certain Conditions of the Offer").

This Offer to Purchase also constitutes notice pursuant to Section 1110 of the CGCL that if the Offeror, TCCC or any other subsidiary of TCCC acquires at least 90.1% of the outstanding Shares on a Fully Diluted Basis, TCCC or the Offeror will cause the Short-Form Merger to become effective without any further notice to shareholders of Odwalla. See Appendix A (Agreement and Plan of Merger).

Concurrently with the execution of the Merger Agreement, and as a condition and inducement to TCCC's and the Offeror's entering into the Merger Agreement, Odwalla entered into a Stock Option Agreement, dated as of October 29, 2001 (the "Option Agreement"), with TCCC and the Offeror. Pursuant to the Option Agreement, Odwalla granted to the Offeror an irrevocable option (the "Top-Up Stock Option")

7

to purchase that number of Shares (the "Top-Up Option Shares") equal to the number of Shares that, when added to the number of Shares owned by TCCC and the Offeror immediately following consummation of the Offer, will constitute 90.1% of the Shares then outstanding on a Fully Diluted Basis (assuming the issuance of the Top-Up Option Shares) at a purchase price per Top-Up Option Share equal to the Offer Price, subject to the terms and conditions set forth in the Option Agreement, including, without limitation, that the Top-Up Stock Option shall not be exercisable if the number of Shares that would otherwise be issued thereunder would exceed the number of authorized Shares available for issuance. If the Top-Up Stock Option is exercised by the Offeror (resulting in the Offeror and TCCC owning 90.1% on a Fully Diluted Basis or more of the Shares then outstanding), the Offeror will be able to effect a short-form merger under the CGCL, subject to the terms and conditions of the Merger Agreement. For a description of the Option Agreement, see Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement").

In connection with the Merger Agreement, the Offeror and TCCC entered into Tender Agreements, dated October 29, 2001 (the "Tender Agreements"), with each of the following shareholders: Bain Capital Fund VI, L.P.; BCIP Associates II; BCIP Trust Associates II, BCIP Associates II-B; BCIP Trust Associates II-B; BCIP Associates III-B; BCIP Associates II-B; BCIP Associates II-B; BCIP Associates II-B; BCIP Associates III-B; BCIP Associates III-B; BCIP Associates II-B; BCIP Associates II

Tender Agreements. Concurrently with the execution and delivery of the Merger Agreement, TCCC, the Offeror and each of the Tendering Shareholders entered into the Tender Agreements. Pursuant to the Tender Agreements, the Tendering Shareholders agreed to tender into the Offer the Committed Shares. The Tender Agreements also provide that U.S. Equity Partners, L.P.; U.S. Equity Partners (Offshore), L.P.; Catterton-Simon Partners III, L.P.; Bain Capital Fund VI, L.P.; BCIP Associates II; D. Stephen C. Williamson; Douglas Levin; Michael Carter; Julie Carter; and Robert Carter (the "Voting Shareholders"), representing in the aggregate 5,950,650 shares, irrevocably appoint the Offeror as their proxy to vote their portion of such Committed Shares in connection with the transaction in the following manner: (i) for the adoption and approval of the Merger Agreement and the Merger and (ii) in any manner as TCCC, in its sole discretion, may see fit with respect to any extraordinary corporate transaction (other than the Merger), such as a merger, consolidation, business combination, tender or exchange offer, reorganization, recapitalization, liquidation, sale or transfer of a material amount of the assets or securities of Odwalla or any of its subsidiaries (other than pursuant to the Merger) or any other change of control involving Odwalla or any of its subsidiaries, including, but not limited to, any Competing Acquisition Proposal, as defined herein. Notwithstanding the foregoing grant to TCCC of the irrevocable proxy, if TCCC elects not to exercise its rights to vote the securities pursuant to the irrevocable proxy, each Voting Shareholder agreed to vote the securities during the term of the Tender Agreement (i) in favor of or give its consent to, as applicable, a proposal to adopt and approve the Merger Agreement and the Merger as described in the Tender Agreements, or (ii) in the manner directed by TCCC if the issue on which the Voting Shareholder is requested to vote is a matter described in clause (ii) of the preceding sentence, in each case at any annual, special or other meeting or action of the shareholders of Odwalla, in lieu of a meeting or otherwise. The Merger Agreement and the Tender Agreements are more fully described in Section 11 ("The Merger Agreement; the Option Agreement; the Tender Agreements; and the Employment Agreement"). The Committed Shares that are subject to voting agreements represent approximately 54% of the Shares outstanding.

The Merger Agreement provides that, promptly after the Offeror acquires Shares which represent at least the Minimum Condition, the Option Exercise Minimum Number or the Revised Minimum Number, the Offeror will be entitled to designate such number of directors, rounded up to the next whole number, on the

Board of Directors of Odwalla (the "Post-Acceptance Board"), subject to compliance with Section 14(f) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), as will make the number of Odwalla's directors designated by the Offeror equal to the product of (1) the total number of available seats on the Board of Directors and (2) the percentage that the number of Shares owned by the Offeror and TCCC bears to the total number of Shares issued and outstanding provided that, prior to or upon the purchase by the Offeror of any Shares in the Offer, Odwalla shall increase the number of directors on the Board of Directors from seven to eight and shall maintain the number of directors at eight until the Effective Time (as defined herein). Prior to the Effective Time, Odwalla shall use its reasonable best efforts to ensure that the Post-Acceptance Board includes four directors in office as of October 29, 2001 (the "Continuing Directors"). If the number of Continuing Directors is reduced below four, then any remaining Continuing Directors will be entitled to designate such persons to fill such vacancies, who will be considered Continuing Directors. If no Continuing Directors remain, the other directors shall designate four persons to fill the vacancies and such persons shall be considered Continuing Directors. Odwalla shall, at such time, take all actions necessary to cause the Offeror's designees to be so elected by its existing Board of Directors. In addition, Odwalla shall use its commercially reasonable efforts to cause to be delivered to TCCC resignations of all the directors of Odwalla's subsidiaries to be effective upon the Merger. In addition, Odwalla will use its reasonable best efforts to cause individuals designated by the Offeror to constitute the same percentage as such individuals represented on the Post-Acceptance Board of each committee of such Post-Acceptance Board (other than any committee of such Post-Acceptance Board established to take action under the Merger Agreement).

Odwalla has represented to TCCC that as of October 25, 2001 there (i) were 11,103,321 Shares issued and outstanding, (ii) 1,737,699 Shares reserved for issuance upon the exercise of outstanding options plus an additional 188,886 shares authorized and reserved for future issuance under the Odwalla stock option plans, and (iii) warrants to purchase 31,806 Shares reserved for future issuance pursuant to the exercise of the warrants. Based upon the foregoing, the Offeror believes that approximately 10,004,093 Shares constitute 90.1% of the outstanding Shares on a Fully Diluted Basis assuming all of the holders of Company Options and the Company Warrants (each as hereinafter defined) agree to exchange such Company Options and Company Warrants for the Option Consideration and Warrant Consideration (each as hereinafter defined), as the case may be, pursuant to the terms of the Merger Agreement (or approximately 11,139,730 Shares if none of the holders of Company Options and Company Warrants agree to exchange such Company Options and Company Warrants for the Option Consideration and Warrant Consideration, as the case may be, pursuant to the terms of the Merger Agreement) and approximately 5,551,661 Shares constitute 50% of the outstanding Shares.

If the Minimum Condition and the other conditions to the Offer are satisfied and the Offer is consummated (or if the Offer is consummated using the Option Exercise Minimum Number and the Top-Up Stock Option is exercised), the Offeror will own a number of Shares to cause the Merger to occur without a vote of the shareholders of Odwalla, pursuant to Section 1110 of the CGCL. As of the effective time of the Merger (the "Effective Time"), each Share outstanding immediately prior to the Effective Time (other than Shares owned by TCCC or any other direct or indirect wholly owned subsidiary of TCCC, which shall be canceled or Shares held by dissenting shareholders, if any) will be converted into the right to receive from the Surviving Corporation the Offer Price. If, however, after consummation of the Offer, the Offeror owns less than 90% of the then outstanding Shares, a vote of Odwalla's shareholders will be required under Section 1201 of the CGCL to approve the Merger, and a significantly longer period of time will be required to effect the Merger.

THIS OFFER TO PURCHASE AND THE RELATED LETTER OF TRANSMITTAL CONTAIN IMPORTANT INFORMATION WHICH SHOULD BE READ BEFORE ANY DECISION IS MADE WITH RESPECT TO THE OFFER.

9

## THE TENDER OFFER

1. TERMS OF THE OFFER; EXPIRATION DATE. Upon the terms and subject to the conditions of the Offer (including, if the Offer is extended or amended, the terms and conditions of such extension or amendment), the Offeror hereby offers to purchase all of the Shares at a price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest thereon, and will accept for payment and pay for all Shares validly tendered on or prior to the Expiration Date and not properly withdrawn as permitted by Section 4. The term "Expiration Date" means 12:00 midnight, San Francisco, California time, on December 6, 2001, unless the Offeror (subject to the terms and conditions of the Merger Agreement) shall have extended the period during which the Offer is open, in which event the term "Expiration Date" shall mean the latest time and date at which the Offer, as so extended by the Offeror, shall expire.

MINIMUM CONDITION, THE EXPIRATION OR TERMINATION OF ALL WAITING PERIODS APPLICABLE TO THE ACQUISITION OF THE SHARES IMPOSED BY THE HSR ACT AND CERTAIN OTHER TERMS AND CONDITIONS. SEE SECTION 15 ("CERTAIN CONDITIONS OF THE OFFER"). The Merger Agreement and the Offer may be terminated by the Offeror and TCCC if certain events occur. Subject to the provisions of the Merger Agreement and the applicable rules and regulations of the Securities and Exchange Commission (the "Commission"), the Offeror and TCCC expressly reserve the right to modify the terms and conditions of the Offer. Subject to the provisions of the Merger Agreement, including the provisions of the Merger Agreement described in the next paragraph, and the applicable rules and regulations of the Commission, if by the Expiration Date any or all of such conditions to the Offer have not been satisfied, the Offeror reserves the right (but shall not be obligated) to (i) terminate the Offer and return all tendered Shares to tendering shareholders, (ii) waive such unsatisfied conditions and purchase all Shares validly tendered. or (iii) extend the Offer and, subject to the terms of the Offer (including the rights of shareholders to withdraw their Shares), retain the Shares which have been tendered, until the termination of the Offer, as extended.

Under the terms of the Merger Agreement, the Offeror may not, without the consent of Odwalla, (i) reduce the number of Shares subject to the Offer, (ii) reduce the Offer Price, (iii) impose any other conditions to the Offer other than the Offer Conditions or modify the Offer Conditions (other than to waive any Offer Conditions to the extent permitted by the Merger Agreement) in a manner adverse to the holders of the Shares, (iv) except as provided in the Merger Agreement, extend the Offer, (v) change the form of consideration payable in the Offer, or (vi) amend any other term of the Offer in a manner adverse to the Odwalla shareholders. Notwithstanding the foregoing, the Offeror may, without the consent of Odwalla, (i) extend the Offer, if at the scheduled or extended expiration date of the Offer any of the Offer Conditions shall not be satisfied or waived, until such time as such conditions are satisfied or waived, subject in each case to any right of TCCC, the Offeror or Odwalla to terminate the Merger Agreement pursuant to the terms thereof or (ii) extend the Offer for any period required by any rule, regulation, interpretation or position of the Commission or the staff thereof applicable to the Offer. Further, if the Minimum Condition is not satisfied on any scheduled expiration date of the Offer, at the request of Odwalla, subject to any right of the Offeror or TCCC to terminate the Merger Agreement, the Offeror shall, and TCCC shall cause the Offeror to take any of the actions set forth in clauses (x), (y), or (z) (provided that the Offeror shall have complete discretion concerning which action to take): (x) extend the Offer pursuant to the provisions of the Merger Agreement, (y) amend the Offer in contemplation of the exercise of the Option Agreement (to the extent the Option Agreement is exercisable at such time) to reduce the Minimum Condition to the Option Exercise Minimum Number or (z) amend the Offer to provide that, if (i) the Minimum Condition is not satisfied at the next scheduled Expiration Date of the Offer (after giving effect to the issuance of any Shares theretofore acquired by TCCC or the Offeror) and (ii) the number of Shares tendered pursuant to the Offer and not withdrawn as of such next scheduled Expiration Date is more than 50% of the then outstanding Shares, the Offeror shall waive the Minimum Condition and amend the Offer to reduce the number of Shares subject to the Offer to the Revised Minimum Number and, subject to the prior satisfaction or waiver of the other conditions of the Offer, purchase, on a pro rata basis, the Revised Minimum Number of Shares (it being understood that the Offeror shall not in any event be required to accept for payment, or pay for, any Shares less than the Revised Minimum Number of Shares are tendered pursuant to the Offer and not withdrawn at the Expiration Date).

1.0

Any extension of the period during which the Offer is open, delay in acceptance for payment or payment, termination or amendment of the Offer will be followed, as promptly as practicable, by public announcement thereof, such announcement in the case of an extension to be issued not later than 9:00 a.m., New York time, on the next business day after the previously scheduled Expiration Date in accordance with the public announcement requirements of rules  $14d-4\,(c)$ ,  $14d-6\,(d)$  and 14e-1 under the Exchange Act. Without limiting the obligation of the Offeror under such rule or the manner in which the Offeror may choose to make any public announcement, the Offeror currently intends to make announcements by issuing a press release to the Dow Jones News Service and making any appropriate filing with the Commission.

If, subject to the terms of the Merger Agreement, the Offeror makes a material change in the terms of the Offer or the information concerning the Offer, or if it waives a material condition of the Offer, the Offeror will disseminate additional tender offer materials and extend the Offer if and to the extent required by Rules 14d-4(c), 14d-6(d) and 14e-1 under the Exchange Act or otherwise. The minimum period during which a tender offer must remain open following material changes in the terms of the Offer or the information concerning the Offer, other than a change in price or a change in percentage of securities sought, will depend upon the facts and circumstances, including the materiality of the changes. In the Commission's view, an offer should remain open for a minimum of five business days from the date the material change is first published, sent or given to shareholders, and, if material changes are made with respect to information that is significant, such as the Offer Price and the percentage of securities sought, a minimum ten business day period may be required to allow for adequate dissemination to shareholders and investor

response. With respect to a change in price or a change in percentage of securities sought, a minimum ten business day period is generally required to allow for adequate dissemination to shareholders and investor response.

Odwalla has provided the Offeror with Odwalla's shareholder list and security position listings for the purpose of disseminating the Offer to holders of Shares. This Offer to Purchase and the related Letter of Transmittal and other relevant materials will be mailed by the Offeror to record holders of Shares whose names appear on Odwalla's shareholder list and will be furnished for subsequent transmittal to beneficial owners of Shares to brokers, dealers, commercial banks, trust companies and similar persons whose names, or the names of whose nominees, appear on the shareholder list or, if applicable, who are listed as participants in a clearing agency's security position listing, for subsequent transmittal to beneficial owners of Shares.

2. ACCEPTANCE FOR PAYMENT AND PAYMENT FOR SHARES. Upon the terms and subject to the conditions of the Offer (including, if the Offer is extended or amended, the terms and conditions of any such extension or amendment), the Offeror will accept for payment and will pay for all Shares validly tendered and not properly withdrawn in accordance with Section 4 prior to the Expiration Date as soon as practicable (and in any event within five business days) after the satisfaction or waiver of the conditions of the Offer set forth in Section 15, including without limitation, the expiration or termination of the waiting period applicable to the acquisition of Shares pursuant to the Offer under the HSR Act. In addition, subject to applicable rules of the Commission, the Offeror expressly reserves the right, in its sole discretion, to delay acceptance for payment of or payment for Shares pending receipt of any other regulatory approvals specified in Section 16 or to comply in whole or in part with applicable law. Any such delays will be effected in compliance with Rule 14e-1(c) under the Exchange Act.

For information with respect to approvals required to be obtained prior to the consummation of the Offer, including under the HSR Act, see Section 16.

In all cases, payment for Shares tendered and accepted for payment pursuant to the Offer will be made only after timely receipt by the Depositary of (i) certificates evidencing such Shares ("Share Certificates") or timely confirmation (a "Book-Entry Confirmation") of a book-entry transfer of such Shares into the Depositary's account at The Depository Trust Company (the "Book-Entry Transfer Facility") pursuant to the procedures set forth in Section 3, (ii) the appropriate Letter of Transmittal, properly completed and duly executed (or a facsimile thereof), with any required signature guarantees, or an Agent's Message (as defined below) in connection with a book-entry transfer and (iii) any other documents required by the Letter of Transmittal.

11

The term "Agent's Message" means a message transmitted by the Book-Entry Transfer Facility to, and received by, the Depositary and forming a part of a Book-Entry Confirmation, which states that the Book-Entry Transfer Facility has received an express acknowledgment from the participant in the Book-Entry Transfer Facility tendering the Shares that such participant has received and agrees to be bound by the terms of the Letter of Transmittal and that the Offeror may enforce such agreement against such participant.

Shareholders who cannot comply on a timely basis with the foregoing procedures for acceptance of the Offer may deposit Share Certificates pursuant to the procedures set forth below for guaranteed delivery.

For purposes of the Offer, the Offeror will be deemed to have accepted for payment (and thereby purchased) Shares validly tendered and not properly withdrawn if and when the Offeror gives oral notice followed by written notice to the Depositary of the Offeror's acceptance for payment of such Shares pursuant to the Offer. Upon the terms and subject to the conditions of the Offer, payment for Shares accepted for payment pursuant to the Offer will be made by deposit of the purchase price therefor with the Depositary, which will act as agent for tendering shareholders for the purpose of receiving payments from the Offeror and transmitting such payments to tendering shareholders whose Shares have been accepted for payment. If for any reason whatsoever acceptance for payment of or payment for any Shares tendered pursuant to the Offer is delayed or the Offeror is unable to accept for payment or pay for Shares tendered pursuant to the Offer, then without prejudice to the Offeror's rights set forth herein, the Depositary may nevertheless, on behalf of the Offeror and subject to Rule 14e-1(c) under the Exchange Act, retain tendered Shares and such Shares may not be withdrawn except to the extent that the tendering shareholder is entitled to and duly exercises withdrawal rights as described in Section 4 ("Withdrawal Rights"). UNDER NO CIRCUMSTANCES WILL INTEREST ON THE PURCHASE PRICE FOR SHARES BE PAID BY THE OFFEROR, REGARDLESS OF ANY EXTENSION OF THE OFFER OR ANY DELAY IN MAKING SUCH PAYMENT.

If any tendered Shares are not accepted for payment for any reason pursuant to the terms and conditions of the Offer, or if Share Certificates are submitted for more Shares than are tendered, Share Certificates evidencing unpurchased or untendered Shares will be returned without expense to the tendering shareholder (or, in the case of Shares tendered by book-entry transfer into the Depositary's

account at the Book-Entry Transfer Facility pursuant to the procedures set forth in Section 3, such Shares will be credited to an account maintained at the Book-Entry Transfer Facility), as promptly as practicable following the expiration, termination or withdrawal of the Offer.

If, prior to the Expiration Date, the Offeror increases the price being paid for Shares accepted for payment pursuant to the Offer, such increased consideration will be paid to all shareholders whose Shares are purchased pursuant to the Offer, whether or not such Shares were tendered prior to such increase in consideration.

The Offeror reserves the right to transfer or assign, in whole or from time to time in part, to one or more of its affiliates the right to purchase all or any portion of the Shares tendered pursuant to the Offer, but any such transfer or assignment will not relieve the Offeror of its obligations under the Offer and will in no way prejudice the rights of tendering shareholders to receive payment for Shares validly tendered and accepted for payment pursuant to the Offer.

#### 3. PROCEDURE FOR ACCEPTING THE OFFER AND TENDERING SHARES.

Valid Tenders. Except as set forth below, for Shares to be validly tendered pursuant to the Offer, the Letter of Transmittal (or a facsimile thereof), properly completed and duly executed, together with any required signature guarantees, or an Agent's Message in connection with a book-entry delivery of Shares, and any other documents required by the Letter of Transmittal, must be received by the Depositary at one of its addresses set forth on the back cover of this Offer to Purchase on or prior to the Expiration Date. In addition either (i) Share Certificates evidencing tendered Shares must be received by the Depositary at such address or such Shares must be tendered pursuant to the procedure for book-entry transfer described below and a Book-Entry Confirmation must be received by the Depositary, in each case on or prior to the Expiration Date or

12

(ii) the guaranteed delivery procedures described below must be complied with. No alternative, conditional or contingent tenders will be accepted.

THE METHOD OF DELIVERY OF SHARE CERTIFICATES, THE LETTER OF TRANSMITTAL AND ALL OTHER REQUIRED DOCUMENTS, INCLUDING DELIVERY THROUGH THE BOOK-ENTRY TRANSFER FACILITY, IS AT THE OPTION AND SOLE RISK OF THE TENDERING SHAREHOLDER, AND THE DELIVERY WILL BE DEEMED MADE ONLY WHEN ACTUALLY RECEIVED BY THE DEPOSITARY (INCLUDING, IN THE CASE OF BOOK-ENTRY TRANSFER, BY BOOK-ENTRY CONFIRMATION). IF DELIVERY IS BY MAIL, REGISTERED MAIL WITH RETURN RECEIPT REQUESTED, PROPERLY INSURED, IS RECOMMENDED. IN ALL CASES, SUFFICIENT TIME SHOULD BE ALLOWED TO ENSURE TIMELY DELIVERY.

Book-Entry Transfer. The Depositary will make a request to establish an account with respect to the Shares at the Book-Entry Transfer Facility for purposes of the Offer. Any financial institution that is a participant in the system of the Book-Entry Transfer Facility may make book-entry delivery of Shares by causing the Book-Entry Transfer Facility to transfer such Shares into the Depositary's account at the Book-Entry Transfer Facility in accordance with the Book-Entry Transfer Facility's procedures for such transfer. However, although delivery of Shares may be effected through book-entry transfer at the Book-Entry Transfer Facility, the Letter of Transmittal (or a facsimile thereof), properly completed and duly executed, together with any required signature guarantees, or an Agent's Message in connection with a book-entry transfer, and any other documents required by the Letter of Transmittal, must in any case be received by the Depositary at one of its addresses set forth on the back cover of this Offer to Purchase on or prior to the Expiration Date, or the tendering shareholder must comply with the guaranteed delivery procedures described below.

DELIVERY OF DOCUMENTS TO THE BOOK-ENTRY TRANSFER FACILITY IN ACCORDANCE WITH THE BOOK-ENTRY TRANSFER FACILITY'S PROCEDURES DOES NOT CONSTITUTE DELIVERY TO THE DEPOSITARY.

Signature Guarantees. Signatures on Letters of Transmittal must be guaranteed by a firm which is a bank, broker, dealer, credit union, savings association or other entity which is a member in good standing of the Securities Transfer Agents Medallion Program or by any other bank, broker, dealer, credit union, savings organization or other entity which is an "eligible guarantor institution" as such term defined in Rule 17Ad-15 under the Exchange Act (each of the foregoing being referred to as an "Eligible Institution"), except in cases where Shares are tendered (i) by a registered holder of Shares who has not completed the box labeled "Special Payment Instructions" on the Letter of Transmittal or (ii) for the account of an Eligible Institution.

If the Share Certificates are registered in the name of a person other than the signer of the Letter of Transmittal, or if payment is to be made, to a person other than the registered holder, the tendered Share Certificate must be endorsed or accompanied by appropriate stock powers, in either case, signed exactly as the name of the registered holder appears on such certificates, with

the signatures on such certificates or stock powers guaranteed in the manner described above. See Instructions 1 and 5 of the Letter of Transmittal.

If Share Certificates are forwarded separately to the Depositary, a properly completed and duly executed Letter of Transmittal (or a facsimile thereof) must accompany each such delivery.

Guaranteed Delivery. If a shareholder desires to tender Shares pursuant to the Offer and such shareholder's Share Certificates evidencing such Shares are not immediately available or such shareholder cannot deliver the Share Certificates and all other required documents to the Depositary on or prior to the Expiration Date, or such shareholder cannot complete the procedure for delivery by book-entry transfer on a timely basis, such Shares may nevertheless be tendered, provided that all of the following conditions are satisfied:

- (i) such tender is made by or through an Eligible Institution;
- (ii) a properly completed and duly executed Notice of Guaranteed Delivery substantially in the form made available by the Offeror is received by the Depositary, as provided below, on or prior to the Expiration Date; and
- (iii) the Share Certificates (or a Book-Entry Confirmation), representing all tendered Shares, in proper form for transfer, together with the Letter of Transmittal (or a facsimile thereof) properly completed and duly executed, with any required signature guarantees (or, in the case of a book-

13

entry transfer, an Agent's Message) and any other documents required by the Letter of Transmittal are received by the Depositary within three trading days after the date of execution of such Notice of Guaranteed Delivery. A "trading day," for purposes of the preceding sentence, is any day on which the Nasdaq Stock Market's National Market System (the "Nasdaq") is open for business.

The Notice of Guaranteed Delivery may be delivered by hand or transmitted by facsimile transmission or mail to the Depositary and must include a guarantee by an Eligible Institution and a representation that the shareholder owns the Shares tendered within the meaning of, and that the tender of the Shares effected thereby complies with, Rule 14e-4 under the Exchange Act, each in the form set forth in such Notice of Guaranteed Delivery.

Notwithstanding any other provision hereof, payment for Shares accepted for payment pursuant to the Offer will in all cases be made only after timely receipt by the Depositary of (i) Share Certificates for, or of Book-Entry Confirmation of the delivery of, such Shares, (ii) a properly completed and duly executed Letter of Transmittal (or a facsimile thereof), together with any required signature guarantees (or, in the case of a book-entry transfer, an Agent's Message), and (iii) any other documents required by the Letter of Transmittal. Accordingly, payment might not be made to all tendering shareholders at the same time and will depend upon when Share Certificates or Book-Entry Confirmations of such Shares are received into the Depositary's account at the Book-Entry Transfer Facility. UNDER NO CIRCUMSTANCES WILL INTEREST BE PAID ON THE PURCHASE PRICE OF THE SHARES TO BE PAID BY THE OFFEROR, REGARDLESS OF ANY EXTENSION OF THE OFFER OR ANY DELAY IN MAKING ANY PAYMENT.

Backup Federal Income Tax Withholding and Substitute Form W-9. Under the "backup withholding" provisions of Federal income tax law, the Depositary may be required to withhold 30.5% of the amount of any payments of cash pursuant to the Offer or the Merger. In order to avoid backup withholding, each shareholder that is a United States person as defined under U.S. federal income tax law surrendering Shares in the Offer must, unless an exemption applies, provide the payor of such cash with such shareholder's correct taxpayer identification number ("TIN") on a substitute Form W-9 and certify, under penalties of perjury, that such TIN is correct and that such shareholder is not subject to backup withholding. If a shareholder does not provide its correct TIN or fails to provide the certifications described above, the Internal Revenue Service ("IRS") may impose a penalty on such shareholder and payment of cash to such shareholder pursuant to the Offer may be subject to backup withholding of 30.5%.

All shareholders that are United States persons as defined under U.S. federal income tax law surrendering Shares pursuant to the Offer should complete and sign the substitute Form W-9 included in the Letter of Transmittal to provide the information and certification necessary to avoid backup withholding (unless an applicable exemption exists and is proved in a manner satisfactory to the Depositary). Certain shareholders (including among others all corporations and certain foreign individuals and entities) are not subject to backup withholding. Foreign shareholders should complete and sign the enclosed Form W-8BEN, Certificate of Foreign Status in order to avoid backup withholding. See Instruction 9 of the Letter of Transmittal.

Determination of Validity. All questions as to the validity, form, eligibility (including time of receipt) and acceptance for payment of any tender of Shares will be determined by the Offeror in its sole discretion, which

determination shall be final and binding on all parties. The Offeror reserves the absolute right to reject any and all tenders determined by it not to be in proper form or the acceptance for payment of which may in the opinion of the Offeror be unlawful. The Offeror also reserves the absolute right to waive any of the conditions of the Offer subject to applicable law and the limitations set forth in the Merger Agreement, or any defect or irregularity in any tender of Shares of any particular shareholder whether or not similar defects or irregularities are waived in the case of other shareholders. The Offeror's interpretation of the terms and conditions of the Offer (including the Letter of Transmittal and the Instructions to the Letter of Transmittal) will be final and binding on all parties. No tender of Shares will be deemed to have been validly made until all defects and irregularities have been cured or waived. None of the Offeror, TCCC, any of their affiliates or

14

assigns, the Dealer Manager, the Depositary, the Information Agent or any other person will be under any duty to give notification of any defects or irregularities in tenders or incur any liability for failure to give any such notification.

Other Requirements. By executing the Letter of Transmittal as set forth above (including through delivery of an Agent's Message), a tendering shareholder irrevocably appoints designees of the Offeror as such shareholder's agent, attorneys-in-fact and proxies, each with full power of substitution, in the manner set forth in the Letter of Transmittal, to exercise all voting and other rights of the shareholder as each such attorney and proxy or his substitute shall in his sole judgment deem proper, with respect to all of the Shares tendered by such shareholder and accepted for payment by the Offeror (and any and all other Shares or other securities issued or issuable in respect of such Shares on or after the date of this Offer to Purchase). All such powers of attorney and proxies will be considered irrevocable and coupled with an interest in the tendered Shares. This appointment is effective when, and only to the extent that, the Offeror accepts for payment the Shares in accordance with the terms of the Offer. Upon acceptance for payment, all prior powers of attorney, proxies and written consents granted by the shareholder at any time with respect to such Shares or other securities or rights will, without further action, be revoked and no subsequent powers of attorney or proxies may be given or written consent executed by such shareholder (and, if given or executed, will not be deemed effective). The designees of the Offeror will, with respect to the Shares and other securities or rights, be empowered to exercise all voting and other rights of such shareholder as they in their sole judgment deem proper in respect of any annual or special meeting of Odwalla's shareholders, or any adjournment or postponement thereof, any actions by written consent in lieu of any such meeting or otherwise. In order for Shares to be deemed validly tendered, immediately upon the Offeror's payment for such Shares, the Offeror must be able to exercise full voting and other rights with respect to such Shares and the other securities or rights issued or issuable in respect of such Shares, including voting at any meeting of shareholders (whether annual or special or whether or not adjourned) in respect of such Shares.

A tender of Shares pursuant to any one of the procedures described above will constitute the tendering shareholder's acceptance of the terms and conditions of the Offer, as well as the tendering shareholder's representation and warranty that (i) such shareholder has the full power and authority to tender, sell, assign and transfer the tendered Shares (and any and all other Shares or other securities issued or issuable in respect of such Shares), and (ii) when the same are accepted for payment by the Offeror, the Offeror will acquire good and unencumbered title thereto, free and clear of all liens, restrictions, charges and encumbrances and not subject to any adverse claims. A tender of Shares pursuant to any of the procedures described above will constitute the tendering shareholder's acceptance of the terms and conditions of the Offer, as well as the tendering shareholder's representation and warranty to the Offeror that (a) such shareholder has a net long position in such Shares being tendered within the meaning of Rule 14e-4 under the Exchange Act and (b) the tender of such Shares complies with Rule 14e-4 under the Exchange Act. It is a violation of Rule 14e-4 under the Exchange Act for a person, directly or indirectly, to tender Shares for such person's own account unless, at the time of tender, the person so tendering (i) has a net long position equal to or greater than the amount of (x) Shares tendered or (y) other securities immediately convertible into or exchangeable or exercisable for the Shares tendered and such person will acquire such Shares for tender by conversion, exchange or exercise and (ii) will cause such Shares to be delivered in accordance with the terms of the Offer. Rule 14e-4 provides a similar restriction applicable to the tender or quarantee of a tender on behalf of another person. The Offeror's acceptance for payment of Shares tendered pursuant to the Offer will constitute a binding agreement between the tendering shareholder and the Offeror upon the terms and subject to the conditions of the

4. WITHDRAWAL RIGHTS. Except as otherwise provided in this Section 4, tenders of Shares made pursuant to the Offer are irrevocable. Shares tendered pursuant to the Offer may be withdrawn at any time prior to the Expiration Date and, unless theretofore accepted for payment pursuant to the Offer, may also be withdrawn at any time after January 5, 2002. If acceptance of any Shares

tendered is delayed for any reason or if the Offeror is unable to accept for payment or pay for Shares tendered pursuant to the Offer, then, without prejudice to the Offeror's rights under the Offer, the Depositary may, on behalf of the Offeror, retain tendered Shares, and such Shares may not be withdrawn except to the extent that tendering shareholders are entitled to

1 0

and duly exercise withdrawal rights as set forth in this Section 4. Any such delay in acceptance for payment will be accomplished by extension of the Offer to the extent required by law.

For withdrawal of Shares tendered pursuant to the Offer to be effective, a written or facsimile transmission notice of withdrawal must be timely received by the Depositary at one of its addresses set forth on the back cover of this Offer to Purchase. The facsimile number for the Depositary is 781-575-4826 or 781-575-4827. Any notice of withdrawal must specify the name of the person who tendered the Shares to be withdrawn, the number of Shares to be withdrawn and (if certificates have been tendered) the name in which the certificates are registered, if different from that of the person who tendered the Shares. If certificates have been delivered or otherwise identified to the Depositary, then, prior to the physical release of such certificates, the shareholder must submit the serial numbers shown on the certificates evidencing the Shares to be withdrawn to the Depositary and, unless such Shares have been tendered for the account of an Eligible Institution, the signatures on the notice of withdrawal must be Medallion guaranteed by an Eligible Institution. If Shares have been tendered pursuant to the procedures for book-entry transfer set forth in Section 3, any notice of withdrawal must also specify the name and number of the account at the Book-Entry Transfer Facility to be credited with the withdrawn Shares and must otherwise comply with such Book-Entry Transfer Facility's procedures.

All questions as to the form and validity (including time of receipt) of notices of withdrawal will be determined by the Offeror, in its sole discretion, and its determination will be final and binding on all parties. None of the Offeror, TCCC, the Dealer Manager, the Depositary, the Information Agent or any other person will be under any duty to give notification of any defects or irregularities in any notice of withdrawal or incur any liability for failure to give any such notification.

Any Shares properly withdrawn will be deemed not validly tendered for purposes of the Offer, but may be tendered at any subsequent time prior to the Expiration Date by following any of the procedures described in Section 3.

5. CERTAIN UNITED STATES FEDERAL INCOME TAX CONSEQUENCES. The following is a summary of certain United States federal income tax consequences of the Offer and the Merger to beneficial owners of Shares whose Shares are purchased pursuant to the Offer or whose Shares are converted to cash in the Merger. The discussion is for general information only and does not purport to consider all aspects of United States federal income taxation that might be relevant to beneficial owners of Shares. The discussion is based on current provisions of the Internal Revenue Code of 1986, as amended (the "Code"), proposed, temporary and final regulations promulgated thereunder and administrative and judicial interpretations thereof, all of which are subject to change possibly on a retroactive basis. The discussion applies only to beneficial owners of Shares in whose hands Shares are capital assets within the meaning of Section 1221 of the Code, and may not apply to Shares received pursuant to the exercise of employee stock options or otherwise as compensation, or to certain types of beneficial owners of Shares (such as insurance companies, tax-exempt organizations, mutual funds and broker-dealers) who might be subject to special rules. This discussion does not discuss the United States federal income tax consequences to a beneficial owner of Shares who, for United States federal income tax purposes, is a non-resident alien individual, a foreign corporation, a foreign partnership or a foreign estate or trust, nor does it consider the effect of any foreign, state or local tax laws.

BECAUSE INDIVIDUAL CIRCUMSTANCES MAY DIFFER, EACH BENEFICIAL OWNER OF SHARES SHOULD CONSULT SUCH BENEFICIAL OWNER'S OWN TAX ADVISOR TO DETERMINE THE APPLICABILITY OF THE RULES DISCUSSED BELOW TO SUCH BENEFICIAL OWNER AND THE PARTICULAR TAX EFFECTS TO SUCH BENEFICIAL OWNER OF THE OFFER AND THE MERGER, INCLUDING THE APPLICATION AND EFFECT OF STATE, LOCAL AND OTHER TAX LAWS.

The receipt of cash for Shares pursuant to the Offer or the Merger will be a taxable transaction for United States federal income tax purposes. In general, for United States federal income tax purposes, a beneficial owner of Shares will recognize gain or loss equal to the difference (if any) between the beneficial owner's adjusted tax basis in the Shares sold pursuant to the Offer or converted to cash in the Merger and the amount of cash received therefor. In general, such gain or loss will be capital gain or loss and will be long-term capital

gain or loss if the beneficial owner held the Shares for more than one year as of the date of sale (in the case of the Offer) or the Effective Time (in case of the Merger). The excess of net long-term capital gains over net short-term capital losses is currently taxed at a maximum rate of 20% for noncorporate

taxpayers.

Payments in connection with the Offer or the Merger might be subject to "backup withholding" at a rate of 30.5%, unless a beneficial owner of Shares (a) is a corporation or comes within certain exempt categories and, when required, demonstrates this fact or (b) provides a correct TIN to the payor, certifies as to no loss of exemption from backup withholding and otherwise complies with applicable requirements of the backup withholding rules. A beneficial owner who does not provide a correct TIN may be subject to penalties imposed by the Internal Revenue Service. Any amount paid as backup withholding does not constitute an additional tax and will be creditable against the beneficial owner's United States federal income tax liability. Each beneficial owner of Shares should consult with his or her own tax advisor as to his or her qualification for exemption from backup withholding and the procedure for obtaining such exemption. Those tendering their Shares in the Offer may prevent backup withholding by completing the substitute Form W-9 included in the Letter of Transmittal. Similarly, those who convert their Shares into cash in the Merger may prevent backup withholding by completing a substitute Form W-9 and submitting it to the paying agent for the Merger.

TCCC and the Offeror will be entitled to deduct and withhold from the consideration otherwise payable pursuant to the Merger Agreement to any holder of Shares such amounts as TCCC or the Offeror is required to deduct and withhold with respect to the making of such payment. To the extent that amounts are so withheld by TCCC or the Offeror, such withheld amounts shall be treated for all purposes of the Merger Agreement as having been paid to the holder of the Shares in respect of which such deduction and withholding was made by TCCC or the

6. PRICE RANGE OF SHARES; DIVIDENDS. According to Odwalla's Annual Report on Form 10-K for the fiscal year ended September 2, 2000 (the "Form 10-K"), the Shares have been traded on Nasdaq under the symbol "ODWA." The following table sets forth, for the periods indicated, the high and low sales prices per Share on Nasdaq as reported by the Dow Jones News Service. Odwalla did not pay any cash dividends during such periods.

<Table> <Caption>

	HIGH	LOW
<\$>	<c></c>	<c></c>
FISCAL YEAR ENDED SEPTEMBER 2, 2000:		
First Quarter	\$ 7.625	\$ 4.313
Second Quarter	\$ 9.50	\$ 4.625
Third Quarter	\$ 9.125	\$ 5.25
Fourth Quarter	\$ 8.00	\$ 5.875
FISCAL YEAR ENDED SEPTEMBER 1, 2001:		
First Quarter	\$11.50	\$ 6.25
Second Quarter	\$10.625	\$ 8.563
Third Quarter	\$10.59	\$ 8.75
Fourth Quarter	\$11.00	\$ 7.30
FISCAL YEAR ENDING AUGUST 31, 2002:		
First Quarter (through November 5, 2001)	\$15.21	\$ 5.51

  |  |On October 29, 2001, the last full trading day prior to announcement of the Merger Agreement, the last reported closing sales price of the Shares on Nasdaq was \$11.83 per Share. On November 5, 2001, the last full trading day before commencement of the Offer, the last reported closing sales price of the Shares on Nasdaq was \$15.15 per Share. SHAREHOLDERS ARE URGED TO OBTAIN A CURRENT MARKET QUOTATION FOR THE SHARES.

Odwalla has not declared or paid any dividends on the Shares since its initial public offering.

7. CERTAIN INFORMATION CONCERNING ODWALLA. Odwalla began operations in September 1980 and was incorporated in California in September 1985.

17

Odwalla's business is to provide easy access to great tasting nourishment. Odwalla is the nation's leading branded super-premium beverage company, delivering nourishment coast to coast with the Odwalla and Samantha lines of more than 45 all natural juices, smoothies, dairy-free shakes, spring water and natural food bars. Odwalla's beverage product lines appeal to many consumers because of the superior taste of minimally processed beverages and greater nutritional value compared to juice from concentrate or with artificial flavors. Odwalla products are currently sold in over 30 states. Odwalla's nourishing food bars are available in additional states and can also be purchased online through Odwalla's website.

Set forth below is certain selected financial information with respect to Odwalla excerpted from the information contained in Odwalla's Annual Reports on Form 10-K for the fiscal years ended September 2, 2000 and August 28, 1999 (the "Odwalla 10-Ks") and Odwalla's Quarterly Reports on Form 10-Q for the periods

ended June 2, 2001 and May 27, 2000 (the "Odwalla 10-Qs"). More comprehensive financial information is included in the Odwalla 10-Ks and Odwalla 10-Qs and other documents filed by Odwalla with the Commission. The following summary is qualified in its entirety by reference to the Odwalla 10-Ks, the Odwalla 10-Qs and such other documents and all the financial information (including any related notes) contained therein. The financial statements include the accounts of Odwalla and Fresh Samantha, Inc., Fresh Samantha Juice Bars, Inc. (collectively, "Fresh Samantha") and Odwalla Canada, Inc. The accounts of Fresh Samantha are included since May 2, 2000, the date of acquisition. The Odwalla 10-Ks, the Odwalla 10-Qs and such other documents are available for inspection and copies thereof are obtainable in the manner set forth below under "Available Information."

ODWALLA, INC.

# SELECTED FINANCIAL INFORMATION (IN THOUSANDS)

<Table> <Caption>

	FISCAL YEAR			WEEKS ENDED,		
	2000	1999	1998	JUNE 2, 2001	MAY 27, 2000	
				(UNAUD	DITED)	
<\$>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	
Statement of Operations Data:						
Net Sales	\$93,450	\$68,042	\$59 <b>,</b> 088	\$97 <b>,</b> 975	\$60,541	
<pre>Income (loss) from operations</pre>	1,036	(2,361)	(1,241)	(161)	187	
Net income (loss)	3,655	(2,042)	(1,379)	(169)	3,121	
Balance Sheet Data:						
Current assets	\$28,318	\$20,337	\$13,866	\$28,340	\$26,269	
Total assets	89,354	35 <b>,</b> 305	29,350	90,194	78,573	
Total shareholders' equity	60 <b>,</b> 672	14,449	16,445	60,720	60,138	

THE TRUE NAME OF THE

On November 6, 2001, Odwalla issued the following press release, according to published sources:

(HALF MOON BAY, CALIF., NOVEMBER 6, 2001) -- Odwalla, Inc. (NASDAQ: ODWA) today announced financial results for its thirteen-week fourth quarter ended September 1, 2001. Fourth quarter net sales were \$30.6 million, a 1% increase over the same quarter in the prior year on a thirteen-week comparative basis. Odwalla's fourth quarter of fiscal 2000 was a fourteen-week period and reported net sales of \$32.9 million.

"In the fourth quarter, we continued to maintain a strong balance sheet and had substantial growth in EBITDA compared to the prior year," said Stephen Williamson, Chairman and CEO. "However, our sales growth rates declined in some parts of the country due to the slowing of the economy. Our business of providing people with easy access to great tasting nourishment is as powerful as ever. We look forward to the acquisition of Odwalla by The Coca-Cola Company under the terms of the merger agreement we signed with The Coca-Cola Company last week. The Coca-Cola Company believes in our vision, our products and our people. We believe the acquisition will allow us to continue Odwalla's entrepreneurial spirit and bringing our vision to more people with the support and experience that Coca-Cola can provide."

18

Gross profit for the fourth quarter was 50.4% of net sales compared to 50.6% for the same period last year. The prior quarter had fourteen weeks compared to the thirteen weeks of the current year, allowing fixed costs to be spread over a larger sales base in the prior year. The fixed cost impact was partially offset by improved variable costs resulting from investments made in the Dinuba production facility.

Sales and distribution costs for the fourth quarter were 36.6% of net sales compared to 33.4% for the same period last year. The increase reflects continued investments in our overall sales and distribution organization, resulting primarily from the merger with Fresh Samantha, Inc. in May 2000. The prior quarter had fourteen weeks compared to the thirteen weeks of the current year, allowing fixed costs to be spread over an additional week in the prior year. We expect to continue to invest in our sales and distribution organization in fiscal 2002.

Marketing and general and administrative expenses for the quarter were \$3.5 million or 11.5% of net sales compared to \$4.2 million or 12.9% of net sales for the same period last year. The decrease results from transitional costs last year associated with the acquisition of Fresh Samantha in May 2000, a reduction in expenses this quarter in discretionary areas, and the impact in the prior year of fourteen weeks of activity compared to thirteen weeks this year.

Amortization of intangible assets acquired with the Fresh Samantha merger

resulted in a \$577,000 expense this quarter compared to \$581,000 for the same quarter last year.

Operating income for the quarter, excluding amortization of intangible assets from the Fresh Samantha acquisition, was \$713,000 compared to \$1.4 million in the same period last year. Net income before taxes for the quarter was \$222,000 or \$0.02 per basic and diluted share compared to a net income before taxes of \$843,000 or \$0.08 per basic and diluted share for the same period last year. The effective tax rate in the current year results from the accounting for the Fresh Samantha acquisition, which was structured as a tax-free reorganization for income tax purposes. As a result, there remains a permanent difference between the financial accounting basis and the income tax reporting basis of goodwill.

Net sales for fiscal 2001 were \$128.3 million or 37.2% higher than the \$93.5 million of net sales last year, the increase due to the acquisition of Fresh Samantha and the growth of Odwalla branded products. Gross profit was 51.9% compared to 49.2% last year. The increase was due primarily to the end of the impact of the December 1998 California citrus freeze and the benefits from the investments in the Dinuba production facility, both reflected over the past two years.

Sales and distribution costs for fiscal 2001 were 35.5% of net sales compared to 34.0% for last year, the increase primarily due to investment in our organization and higher costs in our East Coast markets. Marketing and general and administrative expenses were \$15.2 million compared to \$12.4 million last year, the increase primarily due to systems infrastructure improvements and the acquisition of Fresh Samantha. Amortization of intangible assets acquired with the Fresh Samantha merger resulted in a \$2.3 million expense for fiscal 2001 compared to \$711,000 last year, reflecting the completion of the merger in May 2000. Net income, excluding the restructuring and other charges recorded in the second quarter of this year, was \$1.7 million or \$0.15 per basic share and diluted share for this year, compared to net income of \$603,000 or \$0.09 per basic share and \$0.08 per diluted share last year, excluding, net of tax, the insurance settlement proceeds, the cost of converting the Series A preferred stock and the preferred stock dividend. After considering the restructuring and other charges and the unusual effective tax rate discussed above, the net loss for fiscal 2001 was \$434,000 or \$.04 per share.

Odwalla, Inc. (NASDAQ: ODWA), the nation's leading branded super-premium beverage company, delivers nourishment coast to coast with the Odwalla and Samantha lines of all-natural juices, smoothies, dairy-free regular and chocolate milk, dairy-free shakes, spring water and natural food bars. To learn more about the Odwalla and Samantha brands, please visit us at www.odwalla.com and at www.freshsamantha.com.

This press release contains forward-looking statements, including statements regarding the investment in Odwalla's facilities and organization and the completion of the transaction with The Coca-Cola Company that involve certain risks and uncertainties. Our actual results could differ materially from those anticipated in these forward-looking statements as a result of factors and risks, including without limitation, those factors and risks regarding our products, distribution and business expansion, described in documents filed periodically with the Securities and Exchange Commission including our last annual report on Form 10-K, filed in November 2000, and changes in economic and political conditions. The annual report and other documents

19

contain and identify important factors that could cause the actual results to differ materially from those contained in our projections or forward-looking statements. The completion of the acquisition by The Coca-Cola Company is subject to the satisfaction or waiver of the various conditions to the acquisition set forth in the Agreement and Plan of Merger, dated as of October 29, 2001, among Odwalla, The Coca-Cola Company and TCCC Acquisition Corp. (the "Merger Agreement").

This press release is neither an offer to purchase nor a solicitation of an offer to sell shares of Odwalla in connection with the acquisition of Odwalla common stock pursuant to the terms of the tender offer by The Coca-Cola Company and TCCC Acquisition Corp. under the terms of the Merger Agreement, or otherwise. At the time the tender offer is commenced, The Coca-Cola Company will file a tender offer statement with the Securities and Exchange Commission (the "SEC") and Odwalla will file a solicitation/recommendation statement with respect to the tender offer. Odwalla shareholders are advised to read the tender offer statement regarding the acquisition of Odwalla by The Coca-Cola Company referenced in this press release, and the related solicitation/recommendation statement. The tender offer statement (including an offer to purchase, letter of transmittal and related tender documents) and the solicitation/recommendation statement will contain important information which should be read carefully before any decision is made with respect to the tender offer. These documents will be made available to all shareholders of Odwalla at no expense to them. These documents will also be available to all Odwalla shareholders at no charge on the SEC's web site at www.sec.gov.

# ODWALLA, INC.

# CONSOLIDATED STATEMENTS OF OPERATIONS (IN THOUSANDS, EXCEPT PER SHARE DATA)

<Table> <Caption>

<caption></caption>				
	14 WEEKS SEPT. 2, 2000	13 WEEKS SEPT. 1, 2001	53 WEEKS SEPT. 2, 2000	52 WEEKS SEPT. 1, 2001
<\$>	<c></c>	<c></c>	<c></c>	<c></c>
Net sales	\$32,909	\$30,585	\$93,450	\$128,260
Cost of sales	16,248	15,174	47,477	61,723
Gross profit	16,661 	15,411	45,973 	66 <b>,</b> 537
Operating expenses				
Sales and distribution	10,989	11,186	31,815	45,574
Marketing	936	846	2 <b>,</b> 795	3 <b>,</b> 672
General and administrative	3,306	2,666	9,616	11,518
Amortization of intangible assets from Fresh Samantha acquisition	581	577	711	2,308
Total operating expenses before restructuring				
and other charges	15,812	15 <b>,</b> 275	44,937	63 <b>,</b> 072
Income from operations before restructuring and				
other charges	849	136	1,036	3,465
Restructuring and other charges				3,490
Income (loss) from operations  Proceeds from insurance settlement, net of legal	849	136	1,036	(25)
fees			5,458	
convert to common stock			(1,587)	
Interest and other income (expense), net	(6)	86	(112)	(211)
Income (loss) before income taxes	843	222	4,795	(236)
Income tax expense	(309)	(487)	(1,140)	(198)
Net income (loss)  Preferred stock dividend	534 	(265) 	3,655 (568)	(434)
Net income (loss) applicable to common				
shareholders	\$ 534 ======	\$ (265) ======	\$ 3,087 ======	\$ (434) ======
Net income (loss) applicable to common shareholders after stock dividend				
Basic	\$ 0.05	\$ (0.02)	\$ 0.44	\$ (0.04)
Diluted	\$ 0.05	\$ (0.02)	\$ 0.43	\$ (0.04)
Shares used in per share amounts	======	======	======	======
Basic	11,032	11,079	7,074	11,058
Diluted	====== 11,115	====== 11 <b>,</b> 079	 7,134	11,058
	======	======	======	

  |  |  |  |21

# ODWALLA, INC.

# CONDENSED CONSOLIDATED BALANCE SHEETS (IN THOUSANDS)

<Table> <Caption>

	SEPTEMBER 2, 2000	SEPTEMBER 1, 2001
<\$>	<c></c>	<c></c>
Current assets		
Cash, cash equivalents and short-term investments	\$ 5 <b>,</b> 392	\$ 4,364
Trade accounts receivable	11,599	12,605
Inventories	6 <b>,</b> 705	6,913
Prepaid expenses and other	2,357	3,077
Deferred tax asset	2,265	2,725
Total current assets	28,318	29,684
Plant, property and equipment, net	20,011	23,789
Total other assets	41,025	37,612

	\$89,354	\$91,085
Current liabilities		
Accounts payable	\$ 9,139	\$ 9,322
Other accrued liabilities	4,926	4,651
Line of credit	1,950	1,987
Current maturities of capital lease obligations and		
long-term debt	591	1,632
Total current liabilities	16,606	17,592
Long-term liabilities		
Capital lease obligations and long-term debt and other		
liabilities, less current maturities	1,780	3,107
Deferred tax liability	10,296	9,906
Total liabilities	28,682	30,605
Shareholders' equity	60,672	60,480
	\$89,354	\$91,085
	======	======

#### </Table>

Odwalla Projections. Prior to entering into the Merger Agreement, TCCC conducted a due diligence review of Odwalla and in connection with such review received certain non-public information provided by Odwalla, including certain projected financial information (the "Projections") for the fiscal years 2002, 2003 and 2004. The Projections include expectations of (i) net sales of \$153.2 million, \$188.7 million and \$233.9 million, (ii) gross profit of \$83.1 million, \$104.1 million and \$125.3 million, (iii) earnings before interest and taxes of \$8.4 million, \$15.2 million and \$23.3 million and (iv) earnings before interest, taxes, depreciation and amortization of \$16.9 million, \$24.2 million and \$32.9 million, each for the fiscal years 2002, 2003 and 2004, respectively. Odwalla does not in the ordinary course publicly disclose projections, and the Projections were not prepared with a view to public disclosure. In addition, the Projections were prepared prior to the events in New York City and Washington, D.C. on September 11, 2001 and the aftermath of such events. None of Odwalla, TCCC or the Offeror intends to, and specifically declines any obligation to, update or otherwise revise the Projections to reflect circumstances existing since their preparation or to reflect the occurrence of unanticipated events, even if any or all of the Projections are shown to be in error. Also, none of Odwalla, TCCC or the Offeror intends to, and specifically declines any obligation to, update or revise the Projections to reflect changes in general economic or industry conditions. Odwalla has advised TCCC and the Offeror that the Projections reflect significant assumptions and subjective judgments by Odwalla's management regarding industry performance and general business and economic conditions. The Projections do not give effect to the Offer or the potential combined operations of TCCC and Odwalla. The Projections are included for the limited purpose of giving the holders of the Shares access to the material financial projections prepared by Odwalla's management that were made available to TCCC and the Offeror in connection with the Merger Agreement and the Offer. The inclusion of the Projections herein should not be regarded as an indication that TCCC, the Offeror or Odwalla considered or consider the Projections to be reliable predictions of future events and the Projections should not be relied upon as such.

The Projections were prepared for the limited purpose of managing the operating plan of Odwalla for the fiscal year 2001. While presented with numerical specificity, the Projections are based upon a variety of

22

estimates and hypothetical assumptions which may not be accurate, may not be realized, and are also inherently subject to significant business, economic and competitive uncertainties and contingencies, all of which are difficult to predict, and most of which are beyond the control of Odwalla, TCCC or the Offeror. Furthermore, it is expected that there will be differences between actual and projected results. Accordingly, there can be no assurance that any of the Projections will be realized and the actual results may vary materially from those shown above. In addition, the Projections were not prepared in accordance with guidelines of the Commission, generally accepted accounting principles or the quidelines established by the American Institute of Certified Public Accountants regarding projections or forecasts, and neither Odwalla's nor TCCC's independent accountants have examined or compiled any of the Projections or expressed any conclusion or provided any other form of assurance with respect to the Projections and accordingly assume no responsibility for, and disclaim any association with, the Projections. The Projections should be read in conjunction with the historical financial information of Odwalla included above and in the reports and other documents of Odwalla that may be obtained from the offices of the Commission in the manner set forth below under "Available Information." None of TCCC, the Offeror or any other person assumes any responsibility for the accuracy, completeness or validity of the foregoing Projections.

Certain matters discussed and statements made herein may constitute forward-looking statements within the meaning of the Securities Act of 1933, as

amended (the "Securities Act"), and the Exchange Act, as amended by the Private Securities Litigation Reform Act of 1995. Such forward-looking statements are not guarantees of future performance and involve risks and uncertainties and actual results may differ materially from those contemplated by such forward-looking statements. Forward-looking statements include the information set forth in "Odwalla Projections." Forward-looking statements also include those preceded by, followed by or that include the words "believes", "expects", "anticipates" or similar expressions. Such statements should be viewed with caution

Available Information. Odwalla is subject to the informational filing requirements of the Exchange Act and in accordance therewith is obligated to file periodic reports, proxy statements and other information with the Commission relating to its business, financial condition and other matters. Information as of particular dates concerning Odwalla's directors and officers, their remuneration, options granted to them, the principal holders of Odwalla's securities and any material interest of such persons in transactions with Odwalla is required to be disclosed in such proxy statements and distributed to Odwalla's shareholders and filed with the Commission. Such reports, proxy statements and other information should be available for inspection at the public reference facilities of the Commission located at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549, and may also be available for inspection and copying at prescribed rates. Such reports, proxy statements and other information may also be obtained at the Web site that the Commission maintains at http://www.sec.gov. Copies of this material may also be obtained by mail, upon payment of the Commission's customary fees, from the Commission's principal office at 450 Fifth Street, N.W., Washington, D.C. 20549. Except as otherwise noted in this Offer to Purchase, all of the information with respect to Odwalla set forth in this Offer to Purchase has been derived from publicly available information.

8. CERTAIN INFORMATION CONCERNING THE OFFEROR AND TCCC. The Offeror, a California corporation and a wholly owned subsidiary of TCCC, was newly organized in connection with the Offer and has not carried on any activities to date other than those incident to its formation and commencement of the Offer. The Offeror and TCCC have their principal executive offices at One Coca-Cola Plaza, Atlanta, Ga. 30313. The telephone number for the Offeror and TCCC at such offices is (404) 676-2121. TCCC owns 100% of the outstanding capital stock of the Offeror.

TCCC is the largest manufacturer, distributor and marketer of soft drink concentrates and syrups in the world. Finished beverage products bearing TCCC's trademarks, sold in the United States since 1886, are now sold in nearly 200 countries and include the leading soft drink products in most of these

TCCC's business is nonalcoholic beverages, principally soft drinks, but also a variety of noncarbonated beverages, including juice and juice-drink products. TCCC is one of numerous competitors in the commercial beverages market. Of the approximately 48 billion beverage servings of all types consumed worldwide every day, beverages bearing TCCC's trademarks account for more than one billion.

23

TCCC was incorporated in September 1919 under the laws of the State of Delaware and succeeded to the business of a Georgia corporation with the same name that had been organized in 1892.

The name, citizenship, business address, principal occupation or employment, and five-year employment history of each of the directors and executive officers of the Offeror and TCCC and certain other information are set forth in Annex I hereto. Except as described in this Offer to Purchase and in Annex I hereto, none of TCCC, the Offeror or, to the best knowledge of such corporations, any of the persons listed on Annex I to the Offer of Purchase has during the last five years (i) been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (ii) been a party to any judicial or administrative proceeding (except for matters that were dismissed without sanction or settlement) that resulted in a judgment, decree or final order enjoining the person from future violations of, or prohibiting activities subject to, federal or state securities laws or finding any violation of such laws.

Except as provided in the Merger Agreement, the Option Agreement, the Tender Agreements or as otherwise described in this Offer to Purchase, (i) none of the Offeror, TCCC nor, to the best knowledge of the Offeror and TCCC, any of the persons listed in Annex I to this Offer to Purchase or any associate or majority owned subsidiary of the Offeror, TCCC or any of the persons so listed, beneficially owns or has any right to acquire any Shares and (ii) none of the Offeror, TCCC nor, to the best knowledge of the Offeror and TCCC, any of the persons or entities referred to above nor any director, executive officer or subsidiary of any of the foregoing has effected any transaction in the Shares during the past 60 days.

Except as provided in the Merger Agreement, the Option Agreement or the

Tender Agreements and as otherwise described in this Offer to Purchase, none of the Offeror, TCCC nor, to the best knowledge of the Offeror and TCCC, any of the persons listed in Annex I to this Offer to Purchase, has any agreement, arrangement, understanding, whether or not legally enforceable, with any other person with respect to any securities of Odwalla, including, but not limited to, the transfer or voting of such securities, joint venture, loan or option arrangements, puts or calls, guaranties of loans, guaranties against loss or the giving or withholding of proxies, consents or authorizations. Except as set forth in this Offer to Purchase, neither the Offeror nor TCCC nor, to the best knowledge of the Offeror and TCCC, any of the persons listed on Annex I hereto, has had any transaction with Odwalla or any of its executive officers, directors or affiliates that is required to be reported under the rules and regulations of the Commission applicable to the Offer. Except as set forth in this Offer to Purchase, there have been no negotiations, transactions or material contacts between any of Offeror, TCCC, or any of their respective subsidiaries or, to the best knowledge of the Offeror and TCCC, any of the persons listed in Annex I to this Offer to Purchase, on the one hand, and Odwalla or its affiliates, on the other hand, concerning a merger, consolidation or acquisition, tender offer for or other acquisition of any class of Odwalla's securities, an election of Odwalla's directors or a sale or other transfer of a material amount of assets of Odwalla.

Available Information. TCCC is subject to the informational filing requirements of the Exchange Act and in accordance therewith is obligated to file periodic reports, proxy statements and other information with the Commission relating to its business, financial condition and other matters. Information as of particular dates concerning TCCC's directors and officers, their remuneration, options granted to them, the principal holders of TCCC's securities and any material interest of such persons in transactions with TCCC is required to be disclosed in such proxy statements and distributed to TCCC's shareholders and filed with the Commission. Such reports, proxy statements and other information should be available for inspection at the public reference facilities of the Commission located in Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549, and may also be available for inspection and copying at prescribed rates. Such reports, proxy statements and other information may also be obtained at the Web site that the Commission maintains at http://www.sec.gov. Copies of this material may also be obtained by mail, upon payment of the Commission's customary fees, from the Commission's principal office at 450 Fifth Street, N.W., Washington, D.C. 20549.

9. SOURCE AND AMOUNT OF FUNDS. The Offer is not conditioned upon any financing arrangements. The Offeror estimates that the total amount of funds required to purchase all outstanding Shares (on a Fully Diluted Basis) pursuant to the Offer, to repay outstanding indebtedness and to pay fees and expenses and other obligations related to the Offer and the Merger will be approximately \$192.5 million. The Offeror plans to obtain all funds needed for the Offer and the Merger through capital contributions or advances made by

24

TCCC. TCCC plans to obtain the funds for such capital contributions or advances from its available cash and from working capital.

While the foregoing represents the current intention of TCCC and the Offeror with respect to such funds, such financial arrangements may change depending on such factors as TCCC and the Offeror may deem appropriate.

### 10. BACKGROUND OF THE OFFER; CONTACTS WITH ODWALLA.

In November 2000, at TCCC's request, a representative of TCCC met with Stephen Williamson, the Chairman of the Board and Chief Executive Officer of Odwalla, and Jim Steichen, Chief Financial Officer of Odwalla, and held informal discussions. During this brief meeting, TCCC and Odwalla discussed their future business plans in a general manner. At the conclusion of the meeting, Mr. Williamson indicated that TCCC could contact Odwalla at a future date if TCCC believed it was in either party's best interest.

In January 2001 and in March 2001, representatives of TCCC made inquiries of Mr. Williamson to learn more about Odwalla. Mr. Williamson met with and had informal discussions with representatives of TCCC. During those discussions, TCCC indicated that it was TCCC's belief that bringing Odwalla into the TCCC family would greatly enhance the probability of achieving Odwalla's potential. The TCCC representatives explored generally with Mr. Williamson whether Odwalla would be interested in a possible strategic relationship with TCCC, although no specific terms or conditions were discussed. Mr. Williamson stated that Odwalla was not interested in pursuing a strategic transaction and preferred to pursue independently its business objectives. Mr. Williamson indicated, though, that the management and Board of Directors of Odwalla would give due consideration to any proposal deemed to be in the best interest of the Odwalla shareholders. Also, during this period, TCCC engaged Morgan Stanley & Co. Incorporated ("Morgan Stanley") to act as its financial advisor in connection with any potential transaction with Odwalla.

In June 2001, several directors and officers of Odwalla met with representatives of several investment banks to discuss strategic options  $\frac{1}{2}$ 

available to Odwalla and to consider whether Odwalla should retain an investment bank.

In early August 2001, representatives of TCCC approached DrKW, which was one of the investment banks with which Odwalla previously had discussions regarding Odwalla's strategic options. TCCC's representatives expressed an interest in discussing a possible cash acquisition of Odwalla for a price per outstanding Share in the range of \$12.00 to \$13.00. Following discussions between DrKW and Stephen Williamson, who had consulted with members of the Board of Directors, DrKW indicated to representatives of TCCC that Odwalla was unwilling to entertain discussions regarding a possible transaction between TCCC and Odwalla at the range of values suggested by TCCC. DrKW, however, indicated that Odwalla might be willing to entertain further discussions with TCCC if TCCC were interested in a possible acquisition of Odwalla at a value range in excess of \$15 per Share.

In late August 2001, discussions renewed. After extensive discussions, representatives of TCCC indicated to DrKW that TCCC would increase its preliminary indication of interest to \$15.50 for each outstanding Share, subject to satisfactory completion by TCCC of due diligence, the negotiation of a mutually satisfactory Merger Agreement and related documentation and satisfaction of mutually acceptable closing conditions. TCCC stated that its willingness to consider a possible transaction at this price was absolutely conditioned upon Odwalla's willingness to negotiate exclusively with TCCC for a period of time to complete a transaction and upon the willingness of the large shareholders of Odwalla to provide their unconditional support of the transaction. During these discussions, DrKW sought to encourage TCCC to increase the price reflected in TCCC's preliminary indication of interest above \$15.50 per Share, but TCCC was unwilling to do so.

In late August 2001, Odwalla received a draft indication of interest and a draft confidentiality agreement from TCCC. From August 31 through September 4, 2001, legal counsel to Odwalla, Morrison & Foerster, LLP, reviewed the terms of the draft indication of interest and the draft confidentiality agreement, and negotiated the terms of the draft confidentiality agreement with legal counsel to TCCC, King & Spalding.

On September 4, 2001, the Board of Directors met via teleconference to discuss the TCCC indication of interest. Representatives of DrKW reviewed the draft indication of interest from TCCC for the benefit of the

2.5

Board of Directors. The Board of Directors authorized its senior management to enter into discussions with TCCC, negotiate a confidentiality agreement with TCCC and proceed with appropriate due diligence efforts.

On September 5, 2001, the Board of Directors met via teleconference to discuss the terms of the proposed confidentiality agreement with TCCC, the form of which had been negotiated over the past week (the "Confidentiality Agreement"), and the process, timing, structure and terms of a potential transaction with TCCC. The Board of Directors considered and reviewed with representatives from DrKW and Morrison & Foerster the various provisions of the proposed Confidentiality Agreement and the draft indication of interest from TCCC. The Board of Directors also discussed the provisions of the Confidentiality Agreement including, among other provisions, a thirty day exclusivity period during which Odwalla and principal shareholders of Odwalla (the "Principal Shareholders") would agree, subject to the terms and conditions of the Confidentiality Agreement, not to solicit, encourage or engage in any discussion or negotiations with, or provide any information to, or otherwise cooperate with, encourage or assist, any person or entity regarding any acquisition transaction. After a lengthy discussion, the Board of Directors authorized the senior management of Odwalla to enter into the Confidentiality Agreement with TCCC in substantially the form presented to the Board of Directors.

On September 7, 2001, TCCC submitted a written non-binding indication of interest to the Board of Directors proposing a transaction in which Odwalla shareholders would receive \$15.50 in cash per Share, subject to satisfactory completion by TCCC of due diligence, the negotiation of a mutually satisfactory Merger Agreement and related documentation and satisfaction of mutually acceptable closing conditions. Also on September 7, TCCC, Odwalla and the Principal Shareholders entered into the Confidentiality Agreement with a thirty day exclusivity period ending October 7, 2001.

Following September 7, 2001 and throughout the month of September 2001, TCCC proceeded with its financial, legal and operational due diligence review of Odwalla. Due to delays caused by the aftermath of terrorist attacks in New York and Washington D.C. on September 11, 2001, TCCC requested, and Odwalla and the Principal Shareholders granted, an extension of the exclusivity period, ultimately through October 31, 2001 in order to continue negotiations and complete due diligence. Between October 1 and 26, 2001, TCCC conducted additional financial, legal and operational due diligence.

On September 13, 2001, King & Spalding delivered a proposed Merger

Agreement and, on September 14, 2001, delivered a proposed form of voting agreement to Morrison & Foerster. Morrison & Foerster and legal counsel to the Principal Shareholders began reviewing the Merger Agreement and form of voting agreement. During the period from September 13, 2001 to September 25, 2001, Morrison & Foerster discussed the proposed Merger Agreement and form of voting agreement with counsel to the Principal Shareholders and senior management of Odwalla.

On September 25, 2001, Morrison & Foerster delivered revised drafts of the proposed Merger Agreement and form of voting agreement to King & Spalding which, among other things, rejected the provisions that limited the Board of Directors' ability to consider alternative offers. Odwalla objected to the limitations in the TCCC draft, especially in combination with the proposed form of voting agreement, which would require the Principal Shareholders and certain other large shareholders to support the transaction with TCCC even if a more attractive offer to purchase Odwalla were made. The revised draft delivered by Morrison & Foerster on behalf of Odwalla provided that the Board of Directors could consider alternative offers and provide information to third parties in connection with alternative proposals received by Odwalla. If a more attractive offer were made following the exercise of those rights, Odwalla could also elect to terminate the Merger Agreement, which termination would, in turn, trigger the termination of the proposed form of voting agreement. The draft provided by Morrison & Foerster also limited the conditions to TCCC's obligations to close and the ability of TCCC to terminate the Merger Agreement, in each case, in order to provide Odwalla with more certainty that the proposed transaction, if agreed to, would be consummated.

On September 27, 2001 Morrison & Foerster and King & Spalding commenced negotiations of the proposed Merger Agreement and form of voting agreement.

On October 5, 2001, King & Spalding delivered a revised draft of the proposed Merger Agreement to Morrison & Foerster. The revised draft again limited the Board of Directors' ability to consider alternative

2.6

proposals or provide information in connection with an alternative proposal, and provided that Odwalla could not terminate the Merger Agreement if a more attractive offer was made. King & Spalding expressed TCCC's view that TCCC would not agree to pursue the transaction unless the Principal Shareholders and certain other large shareholders were committed to support the transaction with TCCC even if a more attractive offer to purchase Odwalla were made. TCCC also required more flexibility in its right to terminate the Merger Agreement, especially in connection with events or developments at Odwalla prior to the consummation of the Merger.

On October 12, 2001, Morrison & Foerster delivered revised drafts of the proposed Merger Agreement and voting agreement to King & Spalding, which revisions included structuring the acquisition of Odwalla as a tender offer followed by a second-step merger. Under the tender offer structure proposed, although Odwalla would retain certain rights with respect to considering alternative proposals, the rights of termination of the parties were limited, and TCCC could consummate the acquisition of Odwalla more quickly. The proposed structure thus alleviated the concerns of both parties in connection with the risks associated with not consummating the transaction once it was publicly

On October 17, 2001, King & Spalding delivered a revised draft of the proposed Merger Agreement to Morrison & Foerster. The revised draft did not reflect a tender offer structure and again included limitations on the Board of Directors' ability to consider alternative offers and did not allow Odwalla to terminate the Merger Agreement if the Board determined an alternative offer was superior. The revised draft also continued to provide TCCC with broad termination rights. King & Spalding emphasized to Morrison & Foerster that the premium offered to Odwalla justified the terms dictated by TCCC and that TCCC would not proceed without limitations on Odwalla's ability to consider other offers and the unconditional support of the transaction by the Principal Shareholders and certain other large shareholders.

On Friday, October 19, 2001, Beverage Digest published an article that speculated that TCCC was in talks to acquire Odwalla. The Wall Street Journal published a similar article on Monday, October 22, 2001. Neither Odwalla nor TCCC commented on the rumors. On October 22, 2001, the price of the Shares rose \$3.25 from its closing price on October 19, 2001 to close at \$10.05.

During the week of October 22, 2001, DrKW received communications from representatives of two investment banks, each representing an interested potential acquiror of Odwalla. DrKW indicated to the investment banks that DrKW was not in a position at that time to discuss the matter with the investment banks. Neither investment bank nor their principals made any further inquiries or made any oral or written proposals to DrKW or to Odwalla.

On October 24, 2001, King & Spalding delivered a revised draft Merger Agreement, stock option agreement and form of tender agreement to be entered into by the Principal Shareholders and certain other large shareholders to

Morrison & Foerster reflecting a tender offer structure followed by a second-step merger. The revised draft did not alter TCCC's position as to Odwalla's ability to consider alternative actions or TCCC's ability to terminate the Merger Agreement.

On October 25, 2001, the Board of Directors met via teleconference, with representatives of DrKW and Morrison & Foerster in attendance. Representatives of DrKW and Morrison & Foerster reviewed and discussed the principal issues in the proposed Merger Agreement and related documentation and the respective positions of Odwalla and TCCC as to each principal issue. Representatives of DrKW discussed the two inquiries from potential acquirors and indicated that the communications were only expressions of interest and that no terms were discussed. The Board of Directors considered the potential value and risks associated with exploring these inquiries after the expiration of the exclusivity period. The Board of Directors considered the recent rise in the price of the Shares to between \$10 and \$11 per Share which followed the Beverage Digest and The Wall Street Journal articles referring to Odwalla's discussions with TCCC. Morrison & Foerster discussed the fiduciary duties of the Board of Directors in considering the proposed transaction with TCCC. After discussion, the Board of Directors authorized senior management to continue negotiations with TCCC.

Late on October 25, 2001, Morrison & Foerster delivered revised drafts of the Merger Agreement, the tender agreement and the stock option agreement to King & Spalding which again proposed, among other

2

changes, broader rights for the Board of Directors to consider alternative offers and limited TCCC's ability to terminate the Merger Agreement prior to consummation of the proposed tender offer or merger.

On the morning of October 26, 2001, representatives of Odwalla, Morrison & Foerster and DrKW held a conference call with representatives of TCCC and King & Spalding during which due diligence concerns and the principal unresolved terms of the transaction were discussed.

Subsequently, on October 26, 2001, the Board of Directors met twice via teleconference, with representatives of DrKW and Morrison & Foerster in attendance at each telephonic meeting. During the first meeting the Board of Directors discussed with representatives from DrKW and Morrison & Foerster the results of the conference call that morning with representatives from TCCC and its advisors and counsel. Representatives from DrKW reported that after the morning call with representatives of TCCC, its counsel and financial advisor, DrKW received a call from Morgan Stanley in which Morgan Stanley reported that, based upon TCCC's due diligence and the effects on general economic conditions resulting from the events in New York and Washington, D.C., on September 11, 2001, and the aftermath of such events, TCCC was only willing to pay \$14.50 per share. Morgan Stanley also reiterated that TCCC's willingness to consider a possible transaction at this price was absolutely conditioned upon, among other things, the willingness of the Principal Shareholders and certain other large shareholders to provide their unconditional support of the transaction. Representatives from Morrison & Foerster reported that in calls with King & Spalding, it appeared that TCCC would be willing to compromise on several of the provisions of the agreements under discussion if the price could be agreed upon and if the Principal Shareholders and certain other large shareholders agreed to provide their unconditional support of the transaction.

The Board of Directors had a lengthy discussion of the reduction in purchase price and compromise terms proposed by King & Spalding on behalf of TCCC. The Board of Directors determined that the largest concern was with respect to the reduction in the purchase price. Strengthening its ability to consider alternative proposals was another concern. A third concern was TCCC's position that the Principal Shareholders and certain other large shareholders should pay to TCCC any additional proceeds received by the Principal Shareholders and certain other large shareholders from a sale of Odwalla to an alternative acquiror to the extent that the price received by the Principal Shareholders and certain other large shareholders was greater than the price offered by TCCC. The Board of Directors authorized representatives of DrKW to contact representatives of either TCCC or TCCC's financial advisor to address the Board of Directors' concerns.

On the afternoon of October 27, 2001, the Board of Directors met via teleconference, with representatives of DrKW and Morrison & Foerster in attendance. DrKW and Morrison & Foerster provided the Board of Directors with a report on the negotiations with TCCC. The Board of Directors directed DrKW and Morrison & Foerster to negotiate a price of at least \$15.25 per share, eliminate provisions sought by TCCC, including expanded rights of termination, and expand to the fullest extent possible the Board of Directors' ability to consider competing proposals that might emerge after entering into the proposed agreements with TCCC.

From October 27, 2001 through October 29, 2001, senior management of Odwalla, together with representatives from DrKW and Morrison & Foerster, met with representatives of TCCC, Morgan Stanley and King & Spalding in Atlanta,

Georgia to complete negotiations and drafting of the Merger Agreement and related agreements. After extensive negotiations, TCCC agreed to a purchase price of \$15.25 per share in cash. In addition, TCCC agreed to allow Odwalla broader rights to consider alternative transactions and to limit TCCC's ability to terminate the Merger Agreement. Odwalla's ability to terminate the Merger Agreement remained limited, and the Principal Shareholders and certain other large shareholders agreed to provide TCCC their unconditional support of the transaction.

On October 29, 2001, the Board of Directors met via teleconference, with representatives of DrKW and Morrison & Foerster in attendance. Morrison & Foerster provided the Board of Directors with a report on the revised terms and conditions to the proposed Merger Agreement and related agreements. DrKW presented its financial analyses and stated its oral opinion, later confirmed in writing as of the same date, that, as of that date, the \$15.25 per Share cash consideration to be received by the Odwalla shareholders (other than TCCC, the Offeror and their respective subsidiaries) pursuant to the Offer and the Merger was fair to the Shareholders from a financial point of view. The Board of Directors, after discussion, by the unanimous vote of the directors, determined that the Merger Agreement, the Option Agreement, the Tender Agreements and the

2.8

transactions contemplated thereby, including the Offer and the Merger are advisable and fair to, and in the best interest of, the shareholders, approved the Offer and the Merger and the entry by Odwalla into the definitive agreements, and recommended that the shareholders of Odwalla accept the Offer and tender their Shares to the Offeror. Also on October 29, 2001, TCCC's board of directors formally approved the proposed acquisition at a price of \$15.25 per share.

On the evening of October 29, 2001, Odwalla, the Offeror and TCCC entered into the Merger Agreement and the Stock Option Agreement and TCCC entered into the Tender Agreements with each of the Tendering Shareholders, which included the Principal Shareholders and certain other large shareholders. On October 30, 2001, Odwalla and TCCC issued a press release announcing the transaction.

On November 6, 2001, TCCC and the Offeror commenced the Offer.

11. THE MERGER AGREEMENT; THE OPTION AGREEMENT; THE TENDER AGREEMENTS; AND THE EMPLOYMENT AGREEMENT. The following is a summary of the material terms of the Merger Agreement, the Option Agreement, the Tender Agreements and the employment agreement of D. Stephen C. Williamson, which summary is qualified in its entirety by reference to the Merger Agreement, the two forms of Tender Agreements, and the Stock Option Agreement which are attached hereto as Appendix A, B, C and D, respectively.

The Offer. The Merger Agreement provides for the commencement of the Offer no later than November 8, 2001. The obligation of the Offeror to accept for payment the Shares tendered pursuant to the Offer is subject to (i) the Minimum Condition, (ii) any waiting period under the HSR Act applicable to the purchase of the Shares pursuant to the Offer having expired or having been terminated prior to the expiration of the Offer and (iii) the satisfaction or waiver of certain other terms and conditions of the Offer. See Section 15 ("Certain Conditions to the Offer"). The Minimum Condition of the Offer is that at the expiration of the Offer, a number at least equal to ninety and one-tenth percent (90.1%) of the Shares then outstanding on a Fully Diluted Basis, together with the Shares owned by TCCC, the Offeror and their direct and indirect subsidiaries, shall have been validly tendered and not withdrawn.

The Offeror expressly reserves the right to modify the terms of the Offer, except that, without the consent of Odwalla, the Offeror shall not, (i) reduce the number of Shares subject to the Offer, (ii) reduce the Offer Price, (iii) impose any other conditions to the Offer other than the Offer Conditions or modify the Offer Conditions (other than to waive any Offer Conditions to the extent permitted by the Merger Agreement) in a manner adverse to the holders of the Shares, (iv) except as provided in the Merger Agreement, extend the Offer, (v) change the form of consideration payable in the Offer, or (vi) amend any other term of the Offer in a manner adverse to the Odwalla shareholders. The rights reserved by the Offeror in this paragraph are in addition to the Offeror's rights to terminate the Offer as further described in this Section 11.

The Merger. The Merger Agreement provides that subject to the terms and conditions thereof (and including those described in Section 15) and in accordance with the CGCL, the Offeror shall be merged with and into Odwalla at the Effective Time pursuant to the CGCL. Following the Merger, the separate corporate existence of the Offeror shall cease and Odwalla shall continue as the Surviving Corporation and shall succeed to and assume all the rights and obligations of the Offeror in accordance with the CGCL.

Pursuant to the Merger Agreement, as of the Effective Time, each Share issued and outstanding immediately prior to the Effective Time (other than shares owned by TCCC, the Offeror, their direct or indirect subsidiaries and dissenting shareholders, if any) shall be canceled, and converted automatically into the right to receive an amount in cash equal to the Offer Price or any

higher price paid for each of the outstanding Shares payable, without interest, less any required withholding taxes, upon surrender of the certificate formerly representing such Share in the manner described in the Merger Agreement.

In connection and simultaneously with the Offer, Odwalla shall use its reasonable best efforts to ensure that each holder of outstanding options (whether vested or unvested) to acquire any Shares (the "Company Options") granted under Odwalla's Stock Option Plan adopted in 1993, Odwalla's 1994 Non-Employee Directors' Stock Option Plan and Odwalla's Amended and Restated 1997 Stock Option/Stock Issuance Plan (collectively, the "Odwalla Stock Option Plans") shall execute an agreement to exchange immediately prior to the consummation of the Offer such holder's Company Options to Odwalla for an amount in cash

29

determined by multiplying (A) the excess, if any, of the Offer Price over the applicable exercise price per share of the Company Option (regardless of the exercise price) by (B) the number of Shares subject to the Company Options (whether vested or unvested) held by such holder (such amount, the "Option Consideration") less any amounts withheld to satisfy applicable tax obligations. Upon the expiration date of the Offer, each holder of Company Options immediately prior to the consummation of the Offer, who has exchanged such Company Options shall become entitled to the Option Consideration, and all rights of such holder associated with the Company Options shall be terminated and canceled. As of the Effective Time, either (i) each Company Option not so exchanged immediately prior to the consummation of the Offer shall be canceled, and in consideration for such cancellation the holder shall become entitled to receive an amount in cash equal to the Option Consideration, subject to any amounts withheld to satisfy applicable tax obligations, or (ii) Odwalla shall use its reasonable best efforts to ensure that each holder of a Company Option shall execute an agreement which provides that such Company Option shall be canceled, and in consideration for such cancellation the holder thereof shall become entitled to receive an amount in cash equal to the product of (x) the number of Shares subject to the Company Options, whether vested or unvested, held by such holder and (y) the excess of the Merger Consideration per Share over the per Share exercise price of each such Company Option, subject to any amounts withheld to satisfy applicable tax obligations.

Under the terms of the Tender Agreements with each of D. Stephen C. Williamson, James R. Steichen and Theodore R. Leaman III (the "Executives"), TCCC may cause immediately prior to the consummation of the Offer such Executive to exercise any of the Executive's vested Company Options. Pursuant to the Merger Agreement, if TCCC requires any of the Executives to exercise Company Options to purchase Shares immediately prior to the consummation of the Offer pursuant to such Executive's Tender Agreement, Odwalla will loan to the Executive such funds as may be necessary to permit any of such Executives to exercise such Company Options. If Odwalla becomes obligated to advance funds as described in the immediately preceding sentence and Odwalla fails to advance such funds to any Executive, TCCC shall advance such funds to any Executive on commercially reasonable terms.

Pursuant to the Merger Agreement, Odwalla shall use its reasonable best efforts to provide that each of the issued and outstanding warrants to purchase Shares (the "Company Warrants") shall be exercised in full immediately prior to the consummation of the Offer. In addition, Odwalla shall, by written notice to each holder of Company Warrants, offer to pay such holder upon the consummation of the Offer, in exchange for the cancellation of such holder's Company Warrants (regardless of exercise price) upon the consummation of the Offer, an amount in cash determined by multiplying (A) the excess, if any, of the Offer Price over the applicable exercise price per Share of the Company Warrant by (B) the number of Shares such holder could have purchased had such holder exercised such Company Warrant in full immediately prior to the consummation of the Offer (such amount, the "Warrant Consideration"), subject to any amounts withheld to satisfy applicable tax obligations.

The Merger Agreement also provides that, at the Effective Time, the Articles of Incorporation of Odwalla, as in effect immediately prior to the Effective Time, shall be the Articles of Incorporation of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable law or such Articles of Incorporation. At the Effective Time, the Bylaws of Odwalla, as in effect immediately prior to the Effective Time, shall be the Bylaws of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable law, the Articles of Incorporation of the Surviving Corporation or such Bylaws.

The Merger Agreement provides that (i) the directors of the Offeror at the Effective Time shall be the directors of the Surviving Corporation, until the earlier of their resignation or removal or until their respective successors are duly elected or appointed and qualified, as the case may be, and (ii) the officers of the Offeror at the Effective Time shall be the officers of the Surviving Corporation, until the earlier of their resignation or removal or until their respective successors are duly elected and qualified, as the case may be.

Shareholder Meeting; Proxy Statement. The Merger Agreement provides, if the approval and adoption of the Merger Agreement by Odwalla's shareholders are required by law, Odwalla will, as soon as practicable following the consummation of the Offer, duly call, give notice of, convene and hold a shareholders' meeting for the purpose of considering the approval of the Merger Agreement and the transactions contemplated

30

thereby. The Board of Directors of Odwalla will recommend to the shareholders the adoption or approval of the Merger Agreement and the Merger, will solicit proxies in favor of the Merger Agreement and the Merger and will take all other actions necessary or, in the reasonable judgment of TCCC, helpful to secure the vote or consent of such holders required by the CGCL or the Merger Agreement, to effect the Merger and shall not withdraw such recommendation. In connection with such meeting, Odwalla will promptly prepare and file with the Commission and will thereafter mail to its shareholders as promptly as practicable a proxy statement of Odwalla (the "Proxy Statement") and all other proxy materials for such meeting. Notwithstanding the foregoing, if the Offeror acquires at least 90% of the outstanding Shares, the parties shall take all necessary and appropriate action to cause the Merger to become effective as soon as practicable after the expiration of the Offer without a meeting of Odwalla's shareholders in accordance with Section 1110 of the CGCL.

Board Recommendation; Competing Acquisition Proposals. Except as expressly permitted by the Merger Agreement, neither the Board of Directors of Odwalla nor any committee thereof may (i) withdraw, qualify or modify, or propose publicly to withdraw, qualify or modify, in a manner adverse to TCCC, the approval or recommendation by such Board of Directors or any such committee of the Offer, the Merger, or the Merger Agreement, (ii) endorse, approve, recommend or submit to the Odwalla shareholders, or propose publicly to endorse, approve, recommend or submit to the Odwalla shareholders, any transaction involving a Competing Acquisition Proposal (as hereinafter defined) or (iii) cause Odwalla to enter into any letter of intent, agreement in principle or acquisition agreement or other similar agreement (an "Acquisition Agreement") relating to any Competing Acquisition Proposal. Notwithstanding the foregoing, if prior to the consummation of the Offer, the Board of Directors of Odwalla determines in good faith that it has received a Superior Proposal (as hereinafter defined) in compliance with the solicitation provisions contained in the Merger Agreement and after taking into account the opinion of outside counsel with respect to its fiduciary duties to Odwalla's shareholders under applicable California law, the Board of Directors of Odwalla may (except as described in this and the following sentences) inform its shareholders that it no longer believes that the Merger is advisable and no longer recommends approval (a "Subsequent Adverse Determination") and enter into an Acquisition Agreement with respect to a Superior Proposal, but only at a time that is more than three (3) business days following TCCC's receipt of written notice advising TCCC that Odwalla's Board of Directors intends to make a Subsequent Adverse Determination; provided that the Board of Directors may not at any time withdraw its approval of the Offer, the Merger or the Merger Agreement or seek to abandon the Offer, the Merger or the Merger Agreement. Such notice shall specify the material terms and conditions of such Superior Proposal (and include a copy of such proposal with all accompanying documentation), identify the person making such Superior Proposal and state that the Board of Directors intends to make a Subsequent Adverse Determination. During this three business day period, Odwalla shall provide an opportunity for TCCC to propose adjustments to the terms and conditions of the Merger Agreement to enable Odwalla to proceed with its recommendation to its shareholders without a Subsequent Adverse Determination. A "Competing Acquisition Proposal" means any proposal from a third party with respect to a merger, consolidation, share exchange, tender offer or similar transaction involving Odwalla or any subsidiary of Odwalla, or any purchase or other acquisition of 20% or more of the assets of Odwalla or any subsidiary of Odwalla or any purchase or other acquisition of any equity interest in Odwalla or any subsidiary of Odwalla. A "Superior Proposal" means any bona fide proposal with respect to a merger, consolidation, share exchange, tender offer, business combination or similar transaction involving Odwalla or any subsidiary of Odwalla, or any purchase or other acquisition of 70% or more of the assets of Odwalla or any subsidiary of Odwalla, taken as a whole, or any purchase or other acquisition of more than 50% of the equity interests in the Odwalla in each case, which (w) is fully financed without any financing condition, (x) does not contain a "right of first refusal" or "right of first offer" with respect to any proposal that TCCC may make, (y) does not contain any "due diligence" condition and (z) the Board of Directors of Odwalla determines in good faith after taking into advice from its independent financial advisor is more favorable to the Odwalla shareholders than the transactions contemplated by the Merger Agreement from a financial point of view taking into account any proposed changes to such transactions that may be proposed by TCCC in response to such proposal. Notwithstanding any other provision of the Merger Agreement, Odwalla shall submit the Merger Agreement to the Odwalla shareholders whether or not the Board of Directors of Odwalla makes a Subsequent Adverse Determination provided that Odwalla, concurrently with the submission of the Merger Agreement and the

Proposals if it has complied with the provisions governing such matters in the Merger Agreement.

Other Offers. The Merger Agreement provides that Odwalla will not, nor shall Odwalla authorize or permit any of its subsidiaries, officers, directors or employees or any financial advisors, auditors, attorneys, lenders or other advisors or representatives to directly or indirectly, (a) solicit, initiate or encourage the submission of any Competing Acquisition Proposal or (b) participate in or encourage any discussion or negotiations regarding, or furnish to any person any non-public information with respect to, enter into any acquisition agreement with respect to, or take any other action to facilitate any inquiries or the making of, any proposal that constitutes, or may reasonably be expected to lead to, any Competing Acquisition Proposal; provided, however, that the foregoing shall not prohibit the Board of Directors of Odwalla, prior to the consummation of the Offer, from furnishing information to, or entering into or participating in discussions or negotiations with, any person or entity that makes an unsolicited Competing Acquisition Proposal only if, and only to the extent that, (A) the Board of Directors of Odwalla, after taking into account advice from independent outside legal counsel and financial advisors determines in good faith that such action is required for the Board of Directors of Odwalla to comply with its fiduciary obligations to the Odwalla shareholders under applicable California law, (B) prior to taking such action, Odwalla receives from such person or entity an executed agreement in reasonably customary form relating to the confidentiality of information to be provided to such person or entity and including standstill provisions no less favorable to Odwalla than those contained in the Confidentiality Agreement and (C) the Board of Directors of Odwalla concludes in good faith, after taking into account the advice from its independent financial advisor, that the Competing Acquisition Proposal is a Superior Proposal. Odwalla shall provide immediate oral and written notice to TCCC of (a) the receipt of any such Competing Acquisition Proposal or any inquiry which could reasonably be expected to lead to any Competing Acquisition Proposal, (b) the material terms and conditions of such Competing Acquisition Proposal or inquiry, (c) the identity of such person or entity making any such Competing Acquisition Proposal or inquiry and (d) Odwalla's intention to furnish information to, or enter into discussions or negotiations with, such person or entity. Odwalla shall continue to keep TCCC fully informed of the status and details of any such Competing Acquisition Proposal or inquiry.

Nothing contained in the Merger Agreement shall prohibit Odwalla from referring a third party to the provisions in the Merger Agreement governing Competing Acquisition Proposals, complying with Rule 14e-2(a) promulgated under the Exchange Act or making any disclosure to Odwalla's shareholders if, in the good faith judgment of the Board of Directors of Odwalla, based on advice from its outside counsel, failure so to disclose would violate applicable law; provided, however, that, except as expressly provided in the Merger Agreement, neither Odwalla nor its Board of Directors nor any committee thereof shall withdraw, qualify or modify, or propose to withdraw, qualify or modify, its position with respect to the Merger Agreement or in connection with the Offer or the Merger, or approve or recommend, or propose to approve or recommend, a Competing Acquisition Proposal.

Designation of Directors. The Merger Agreement provides that, promptly after the Offeror acquires Shares which represent at least the Minimum Condition, the Option Exercise Minimum Number or the Revised Minimum Number, the Offeror will be entitled to designate, rounded up to the next whole number, such number of directors on the Post-Acceptance Board, subject to compliance with Section 14(f) of the Exchange Act, as will make the number of Odwalla's directors designated by the Offeror equal to the product of (1) the total number of available seats on the Board of Directors and (2) the percentage that the number of Shares owned by the Offeror and TCCC bears to the total number of Shares issued and outstanding provided that, prior to or upon the purchase by the Offeror of any Shares in the Offer, Odwalla shall increase the number of directors on the Board of Directors from seven to eight and shall maintain the number of directors at eight until the Effective Time. Prior to the Effective Time, Odwalla shall use its reasonable best efforts to ensure that the Post-Acceptance Board includes four Continuing Directors. If the number of Continuing Directors is reduced below four, then any remaining Continuing Directors will be entitled to designate such persons to fill such vacancies, who will be considered Continuing Directors. If no Continuing Directors remain, the other directors shall designate four persons to fill the vacancies and such persons will be considered Continuing Directors. Odwalla shall, at such time, take all action necessary to cause the Offeror's designees to be so elected by its existing Board of Directors. Odwalla has agreed to take all action necessary to enable the

32

Offeror's designees to be elected or appointed to the Board of Directors including, without limitation increasing the number of directors and seeking and accepting the resignation of incumbent directors. In addition, Odwalla shall use its commercially reasonable efforts to cause to be delivered to TCCC resignations of all of the directors of Odwalla's subsidiaries to be effective upon consummation of the Merger.

Odwalla shall use its reasonable best efforts to cause individuals designated by Offeror to serve on the Board of Directors to constitute the same percentage as such individuals represent on the Post-Acceptance Board of each committee of the Board of Directors (other than any committee of the Board of Directors established to take action under the Merger Agreement).

Following the election or appointment of Offeror's designees pursuant to the Merger Agreement and prior to the Effective Time, TCCC and Offeror shall not cause Odwalla to take any action with respect to any amendment, or waiver of any term or condition, of the Merger Agreement, the Articles of Incorporation of Odwalla or the Bylaws of Odwalla, any termination or rescission of the Merger Agreement by Odwalla, any extension by Odwalla of the time for the performance of any of the obligations or other acts of Offeror or TCCC or waiver or assertion of any of Odwalla's rights under the Merger Agreement, or any other consent or action by the Post-Acceptance Board with respect to the Merger Agreement or the Offer (including any action which would cause Odwalla to breach the Merger Agreement), without the concurrence of a majority of the Continuing Directors and the concurrence of a majority of the directors who are not Continuing Directors.

Access to Information; Confidentiality. Until the Effective Time, Odwalla shall, and shall cause its officers, directors, employees, auditors, attorneys, financial advisors, lenders and other agents (collectively, the "Representatives") to afford the Representatives of TCCC and the Offeror reasonable access at all reasonable times and upon reasonable notice to the officers, employees, agents, properties, offices and other facilities, books and records of Odwalla and its subsidiaries, and shall furnish TCCC and the Offeror with all financial, operating and other data and information as TCCC or the Offeror, through its Representatives, may reasonably request. Odwalla shall furnish to TCCC and the Offeror monthly financial and operating data and information for each four or five weeks fiscal period in the form currently prepared by Odwalla within seventeen (17) days following the end of each such period.

All information obtained by TCCC or the Offeror concerning Odwalla and its subsidiaries in connection with the transactions contemplated by the Merger Agreement shall be kept confidential in accordance with the Confidentiality Agreement, dated as of September 7, 2001, as amended, between TCCC and Odwalla (the "Confidentiality Agreement").

Directors' and Officers' Indemnification and Insurance. The Merger Agreement provides that, for six (6) years from and after the Effective Time, TCCC will cause the Surviving Corporation to indemnify and hold harmless all past and present officers and directors of Odwalla and of its subsidiaries for acts or omissions occurring at or prior to the Effective Time to the same extent such persons are indemnified by Odwalla pursuant to its Articles of Incorporation or Bylaws or agreements in effect on October 29, 2001.

TCCC has agreed to maintain (or to cause the Surviving Corporation to maintain), for an aggregate period of at least six (6) years from the Effective Time, the current directors' and officers' insurance and indemnification policies ("D&O Insurance") that provide coverage for events occurring prior to the Effective Time or, coverage containing terms that are at least as favorable as the current D&O Insurance; provided, however, that the Surviving Corporation shall not be required to expend more than an amount per year equal to two hundred percent (200%) of current annual premiums paid by Odwalla for such insurance.

Conduct of Business Pending the Merger. Pursuant to the Merger Agreement, from and after the date thereof until the Effective Time or the termination of the Merger Agreement unless TCCC shall otherwise expressly agree in writing (which agreement shall not be unreasonably withheld or delayed):

(i) the business of Odwalla and its subsidiaries shall be conducted only in, and Odwalla and its subsidiaries shall not take any action except in, the ordinary course of business and in a manner consistent with prior practice; (ii) Odwalla and its subsidiaries shall use their reasonable best efforts to preserve intact their business organizations, keep available the services of their current officers and employees and preserve the current relationships of Odwalla and its subsidiaries with customers,

33

suppliers, distributors and other persons with which Odwalla or its subsidiaries has material business relations; and (iii) Odwalla and its subsidiaries will comply in all material respects with all applicable laws and regulations wherever their business is conducted, including, without limitation, the timely filing of all reports, forms or other documents with the Commission required pursuant to the Securities Act or the Exchange Act.

-- Odwalla shall not, nor shall it permit any of its subsidiaries to,
(i) declare or pay any dividends on or make other distributions
(whether in cash, stock or property) in respect of any of its capital

stock, except for dividends payable to Odwalla by a wholly owned subsidiary; (ii) split, combine or reclassify any of its capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; (iii) repurchase, redeem or otherwise acquire any shares of its capital stock; (iv) issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock or any securities convertible into any such shares of its capital stock, or any rights, warrants or options to acquire any such shares or convertible securities or any stock appreciation rights, phantom stock plans or stock equivalents, other than the issuance of Shares upon (A) the exercise of Company Options outstanding as of the date of the Merger Agreement or awarded in conformity with the provisions described in clause (v) below, and (B) the exercise of warrants outstanding as of the date of the Merger Agreement; (v) award or grant, or authorize or propose the award or grant of, any unvested Company Options, other than the award or grant of Company Options to employees (other than executive officers of Odwalla) in accordance with the terms of Odwalla Stock Option Plans and in the ordinary course of business and consistent with past practices for up to no more than 100,000 Shares in the aggregate for all employees; (vi) modify or adjust any outstanding options to acquire Shares; or (vii) take any action that would, or could reasonably be expected to, result in any of the conditions set forth in the Merger Agreement governing the conduct of business pending the Merger not being satisfied.

Odwalla shall not, nor shall it permit any of its subsidiaries to, (i) amend its articles of incorporation (including any certificate of designations attached thereto) or bylaws or other equivalent organizational documents; (ii) create, assume or incur any indebtedness for borrowed money or guaranty any such indebtedness of another person or mortgage or pledge any of its assets or properties, other than in connection with (A) existing lines of credit, (B) indebtedness owing to, or guaranties of indebtedness owing to, Odwalla, (C) leasing contracts entered into in the ordinary course of business with payments of less than \$50,000 in the aggregate on a monthly basis, (D) indebtedness incurred to make payments specifically provided or contemplated by the Merger Agreement or (E) accounts payable incurred in the ordinary course of business consistent with past practice; (iii) make any loans or advances to any other person other than loans or advances between any subsidiaries of Odwalla or between Odwalla, and any of its subsidiaries and other than advances of ordinary business expenses or to employees in the ordinary course of business consistent with past practice in principal amounts of not more than \$10,000; (iv) merge or consolidate with any other entity in any transaction, (v) sell any business or assets, other than sales of assets and other than sales of products and inventory in the ordinary course of business consistent with past practices) having a value of less than \$200,000 individually and \$500,000 in the aggregate to the extent such sales are in the ordinary course of business and consistent with past practices; (vi) change its accounting policies except as required by generally accepted accounting principles or applicable law; (vii) make any change in employment terms for any of its directors or officers, except as expressly contemplated by the Merger Agreement; (viii) alter, amend or create any obligations with respect to compensation, severance, benefits, change of control payments or any other payments to employees, directors or affiliates of Odwalla or its subsidiaries or enter into any new, or amend any existing, employment agreements, except (A) as required by applicable law or (B) severance agreements containing terms consistent with Odwalla's policies and practices in amounts not to exceed \$15,000 individually or \$150,000 in the aggregate other than severance agreements with certain key employees; (ix) make any change to the Odwalla benefit plans, except as required by applicable law; (x) amend or cancel or agree to the amendment or cancellation of any material contract; (xi) pay, loan or advance (other than the payment of compensation, directors' fees or reimbursement of expenses in the ordinary course of business) any

34

amount to, or sell, transfer or lease any properties or assets (real, personal or mixed, tangible or intangible) to, or enter into any agreement with, any of its officers or directors or any "affiliate" or "associate" of any of its officers or directors; (xii) form or commence the operations of any business or any corporation, partnership, joint venture, business association or other business organization or division thereof; (xiii) make any tax election or settle or compromise any tax liability involving amounts in excess of \$25,000 individually or \$200,000 in the aggregate; or (xiv) pay, discharge, settle or satisfy any claims litigation, liabilities or obligations (whether absolute, accrued, asserted or unasserted, contingent or otherwise) involving amounts in excess of \$75,000

individually or \$200,000 in the aggregate. Odwalla also shall not amend, modify, terminate or rescind, or waive any provision of, or otherwise alter in any way the employment agreement (of even date of the date of the Merger Agreement) between Odwalla and D. Stephen C. Williamson prior to the Effective Time without the prior written consent of TCCC.

In connection with the continued operation of Odwalla and its subsidiaries between the date of the Merger Agreement and the Effective Time or the termination of the Merger Agreement, Odwalla will confer in good faith on a regular and frequent basis with one or more representatives of TCCC designated to Odwalla regarding operational matters and the general status of ongoing operations promptly and will notify TCCC of any event or occurrence that has had or may reasonably be expected to have a Company Material Adverse Effect (as defined below). Odwalla acknowledges that TCCC does not and will not waive any rights it may have under the Merger Agreement as a result of such consultations. Odwalla shall not take any action that would, or that could reasonably be expected to, result in any of the representations and warranties of Odwalla set forth in the Merger Agreement becoming untrue.

Representations and Warranties. The Merger Agreement contains various customary representations and warranties of the parties thereto including, but not limited to, representations and warranties by Odwalla concerning: Odwalla's organization, standing and power; capitalization; the Board of Directors' approval of the Merger Agreement and the Offer and the transactions contemplated by the Merger Agreement; required filings and consents; Commission filings and financial statements; absence of certain changes or events; compliance with law; tax matters; absence of litigation; employee benefit plans; worker safety and environmental matters; labor matters; required shareholder vote; financial advisor opinion; and brokers. Some of the representations are qualified by a material adverse effect clause. For purposes of the Merger Agreement, "Company Material Adverse Effect" shall mean, with respect to Odwalla, any change, event or effect that, when taken together with all other adverse changes, events or effects that have occurred, (i) is materially adverse to the business, operations, properties, condition (financial or otherwise), assets, liabilities (including, without limitation, contingent liabilities) of Odwalla and its subsidiaries taken as a whole or (ii) prevents or delays beyond May 31, 2002 the consummation of the Offer or the Merger; provided, however, that a Company Material Adverse Effect shall not include (x) in and of itself failure by Odwalla to meet analysts' earnings forecasts or estimates or changes in the market price of the Shares in and of itself or (y) any effect resulting from any change in general economic conditions, including any change in general economic conditions due to any act of war, terrorism or threat of war or terrorism, which changes do not disproportionately affect Odwalla and its subsidiaries (provided that it is understood and agreed that the foregoing exclusion shall not exclude any change, event or effect that would otherwise be a Company Material Adverse Effect caused by any act of war or terrorism or threat of war or terrorism directed at the beverage industry which disproportionately affects Odwalla and its subsidiaries as compared to the beverage industry as a whole) or (z) any effect resulting solely in and of itself from the public announcement of the Merger Agreement or the transactions contemplated by the Merger Agreement or the consummation of such transactions (other than any litigation or claims the substance of which is not related to the public announcement of the Merger Agreement or the transactions contemplated by the Merger Agreement or the consummation of such transactions).

Conditions of the Merger. Under the Merger Agreement, the respective obligations of each party to effect the Merger shall be subject to the following conditions:

(a) The Offeror shall have previously accepted for payment and paid for Shares pursuant to the Offer, except that neither TCCC, the Offeror nor Odwalla shall be entitled to invoke this condition if, in

35

breach of its obligations under the Merger Agreement, it shall have been the cause of the failure of the Offeror to purchase pursuant to the Offer Shares validly tendered and not withdrawn;

- (b) The Merger Agreement and the Merger shall have been approved and adopted by the requisite vote of the Odwalla shareholders, if any, as required by the CGCL, the Odwalla Articles of Incorporation and the Odwalla Bylaws;
- (c) No temporary restraining order, preliminary or permanent injunction or other order issued by any court of competent jurisdiction or other legal prohibition preventing the consummation of the Merger shall be in effect, provided that the party relying on this condition shall have complied in all material respects with its obligations to perform certain obligations under the Merger Agreement; and
- (d) All actions by or in respect of or filings with any governmental entity required to permit the consummation of the Offer and the Merger shall have been obtained or made (including the expiration or

termination of any applicable waiting period under the HSR Act).

Termination Events. The Merger Agreement may be terminated at any time prior to the Effective Time, whether before or after any approval of the matters presented in connection with the Merger by the shareholders of Odwalla:

- (a) By mutual written consent of TCCC and Odwalla;
- (b) By any of TCCC, Offeror or Odwalla if any court of competent jurisdiction or other governmental entity shall have issued an order, decree, ruling or taken any other action permanently restraining, enjoining or otherwise prohibiting the Merger and such order, decree, ruling or other action shall have become final and nonappealable, except if the party relying on such order, decree or ruling or other action has not complied in all material respects with its obligations to perform certain obligations under the Merger Agreement;
- (c) By any of TCCC, Offeror or Odwalla if (x) as a result of the failure of any of the conditions of the Offer the Offer shall have terminated or expired in accordance with its terms without Offeror, in accordance with the Merger Agreement, having accepted for payment any Shares pursuant to the Offer within the time period for acceptance specified by the Agreement, or (y) Offeror, in accordance with the Agreement, shall not have accepted for payment any Shares pursuant to the Offer by May 31, 2002; provided, however, that the right to terminate the Merger Agreement shall not be available to any party whose failure to fulfill any obligation under the Merger Agreement has been the primary cause of, or resulted in, the expiration or termination of the Offer on or before such date;
- (d) By TCCC or the Offeror if a Competing Acquisition Proposal shall have been consummated or if the Board of Directors of Odwalla shall have made a Subsequent Adverse Determination;
- (e) By Odwalla, if TCCC or Offeror shall breach in any material respect any of their respective representations, warranties or obligations under the Merger Agreement and, within twenty (20) days after written notice of such breach to TCCC from Odwalla, such breach shall not have been cured in all material respects or waived by Odwalla;
- (f) By any of Odwalla, TCCC or Offeror, if the Merger Agreement and the Merger shall fail to be approved and adopted by the Odwalla shareholders at a duly held Shareholders' Meeting or at any adjournment or postponement thereof; provided, however, that neither TCCC nor Offeror may terminate the Merger Agreement under this provision if the Shares it is entitled to vote (whether by ownership, proxy or otherwise) shall not have been voted in favor of the Merger Agreement and the Merger;
- (g) By TCCC or Offeror, if Odwalla shall breach any of its representations, warranties, covenants or other obligations hereunder which would give rise to the failure of any of the following conditions: (A) the representations and warranties of Odwalla in the Merger Agreement shall be true and correct in all respects as of the date of the Merger Agreement and as of the expiration of the Offer without giving effect to any exceptions for "material" or "materiality" or "Company Material

36

Adverse Effect", except where the failure of such representations and warranties to be true and correct, individually or in the aggregate, does not have and is not reasonably likely to have a Company Material Adverse Effect; (B) Odwalla shall have performed in all material respects all covenants, agreements and obligations required to be performed by it under the Merger Agreement; (C) the directors of Odwalla shall have resigned as provided in the Merger Agreement; and (D) an officer of Odwalla shall have delivered to TCCC and the Offeror a certificate to the effect that each of the foregoing conditions are satisfied in all respects, and, within twenty (20) days after written notice of such breach to Odwalla from TCCC, such breach shall not have been cured, or waived by TCCC or Offeror;

(h) By TCCC or Offeror, if, for any reason (i) the representations and warranties regarding Odwalla's capitalization or (ii) the representations and warranties of any Tendering Shareholder in such Shareholder's Tender Agreement regarding ownership of Shares (clause (i) and clause (ii) collectively being referred to as the "Capital Stock Representations") shall not be true and correct in all respects; provided, that neither TCCC nor Offeror may terminate the Merger Agreement under this provision if, after giving effect to the failure of any Capital Stock Representations to be true and correct, the number of Shares owned by the Tendering Shareholders which is subject to the Tender Agreements shall be equal to or greater than 54.0% of the sum of (x) the issued and outstanding Shares and (y) the number of

Shares issuable upon the exercise of vested Company Options, and the number of Shares owned by the Tendering Shareholders which is subject to the voting provisions contained in the Tender Agreements shall be equal to or greater than 50.1% of the sum of (x) the issued and outstanding Shares and (y) the number of Shares issuable upon the exercise of vested Company Options.

Odwalla shall pay TCCC a fee of \$8.0 million (the "Termination Fee"), payable in same-day funds, if TCCC or the Offeror terminates the Merger Agreement in accordance with the provisions described paragraph (d) above. In addition, within one year after termination of the Merger Agreement (other than a termination pursuant to the provisions described paragraph (b) or paragraph (c) above (but only if the termination pursuant to paragraph (c) results from a failure to satisfy the following Offer Conditions: at or prior to the expiration date of the Offer, (x) any applicable waiting period under the HSR Act shall have been expired or terminated, and (y) all other actions by or in respect of or other filings with any governmental entity required to permit the consummation of the Offer and the Merger shall have been obtained or made other than in the case of clause (y), such actions or filings the failure of which to make or obtain would not reasonably be expected to have a Company Material Adverse Effect) or the provisions described paragraph (e) above) Odwalla shall enter into any agreement relating to a Competing Acquisition Proposal with a person or entity other than TCCC or the Offeror or a Competing Acquisition Proposal with a person or entity other than TCCC or the Offeror shall otherwise be consummated, then immediately prior to, and as a condition of, the entering into of such agreement or the consummation of such transaction, as the case may be, Odwalla shall pay to TCCC upon demand the Termination Fee, payable in same-day funds, to reimburse TCCC for its time, expense, lost opportunity costs and other costs and damages associated with pursuing the Merger; provided that no such amount shall be payable if the Termination Fee shall have been paid in accordance with the first sentence of this paragraph.

Fees and Expenses. Except as specifically provided, all fees and expenses incurred in connection with the transactions contemplated by the Merger Agreement shall be paid by the party incurring such fees or expenses.

Option Agreement. Under the Option Agreement, Odwalla granted to the Offeror an irrevocable Top-Up Stock Option to purchase that number of Top-Up Option Shares equal to the number of Shares that, when added to the number of Shares owned by Offeror and TCCC immediately following consummation of the Offer, will constitute 90.1% of the Shares then outstanding on a Fully Diluted Basis (assuming the issuance of the Top-Up Option Shares), calculated in accordance with the Option Agreement at a purchase price per Top-Up Option Share equal to the Offer Price. However, the Top-Up Stock Option will not be exercisable if the number of Shares subject thereto exceeds the number of authorized Shares available for issuance.

37

Subject to the terms and conditions of the Option Agreement, the Top-Up Stock Option may be exercised by the Offeror, at its election, in whole, but not in part, at any one time after the occurrence of a Top-Up Exercise Event (as defined below) and prior to the Top-Up Termination Date (as defined below). A "Top-Up Exercise Event" will occur for purposes of the Option Agreement upon the Offeror's acceptance for payment pursuant to the Offer of Shares constituting, together with Shares owned directly or indirectly by TCCC, more than 50% of the outstanding Shares but less than 90% of the Shares then outstanding on a Fully Diluted Basis. Except as provided in the last sentence of this paragraph, the "Top-Up Termination Date" will occur for purposes of the Option Agreement upon the earliest to occur of: (i) the Effective Time; (ii) the date which is 20 business days after the occurrence of a Top-Up Exercise Event; (iii) the termination of the Merger Agreement; and (iv) and the date on which the Offeror reduces the Minimum Condition to the Revised Minimum Number and accepts for payment the Revised Minimum Number of Shares. Nevertheless, even if the Top-Up Termination Date has occurred, the Offeror will be entitled to purchase the Top-Up Option Shares if it has exercised the Top-Up Stock Option in accordance with the terms of the Option Agreement prior to such occurrence.

The obligation of Odwalla to deliver Top-Up Option Shares upon the exercise of the Top-Up Stock Option is subject to the following conditions: (a) any applicable waiting period under the HSR Act and any applicable non-United States laws regulating competition, antitrust, investment or exchange controls relating to the issuance of the Top-Up Option Shares will have expired or been terminated prior to the expiration of the Offer; (b) no provision of any applicable law or regulation and no judgment, injunction, order or decree shall prohibit the exercise of the Top-Up Stock Option or the delivery of the Top-Up Option Shares in respect of any such exercise; and (c) delivery of the Top-Up Option Shares would not require the approval of Odwalla's shareholders pursuant to the rules of the National Association of Securities Dealers.

Tender Agreements. Concurrently with the execution and delivery of the Merger Agreement, TCCC, the Offeror and each of the Tendering Shareholders entered into the Tender Agreements. Pursuant to the Tender Agreements, the Tendering Shareholders agreed to tender into the Offer the Committed Shares currently owned by the Tendering Shareholders. The Tender Agreements also

provide that the Voting Shareholders representing in the aggregate 5,950,650 Shares, irrevocably appoint the Offeror as their proxy to vote their portion of such Committed Shares in connection with the transaction in the following manner: (i) for the adoption and approval of the Merger Agreement and the Merger and (ii) in any manner as TCCC, in its sole discretion, may see fit with respect to any extraordinary corporate transaction (other than the Merger), such as a merger, consolidation, business combination, tender or exchange offer, reorganization, recapitalization, liquidation, sale or transfer of a material amount of the assets or securities of Odwalla or any of its subsidiaries (other than pursuant to the Merger) or any other change of control involving Odwalla or any of its subsidiaries, including, but not limited to, any Competing Acquisition Proposal. Notwithstanding the foregoing grant to TCCC of the irrevocable proxy, if TCCC elects not to exercise its rights to vote the securities pursuant to the irrevocable proxy, the Voting Shareholders agreed to vote the securities during the term of the Tender Agreement (i) in favor of or give its consent to, as applicable, a proposal to adopt and approve the Merger Agreement and the Merger as described in the Tender Agreements, or (ii) in the manner directed by TCCC if the issue on which the Voting Shareholders are requested to vote is a matter described in clause (ii) of the preceding sentence, in each case at any annual, special or other meeting or action of the shareholders of Odwalla, in lieu of a meeting or otherwise. Pursuant to the Merger Agreement, if TCCC requires any of D. Stephen C. Williamson, James R. Steichen and/or Theodore R. Leaman III (the "Executives", each an "Executive") to exercise Company Options to purchase Shares immediately prior to the consummation of the Offer pursuant to such Executive's Tender Agreement, Odwalla will loan to the Executive such funds as may be necessary to permit any of such Executives to exercise such Company Options. If Odwalla becomes obligated to advance funds pursuant to the immediately preceding sentence and Odwalla fails to advance such funds to any Executive, TCCC shall advance such funds to any Executive on commercially reasonable terms.

The Tender Agreements provide that the Tendering Shareholders (i) except as consented to in writing by TCCC in its sole discretion, will not, directly or indirectly, sell, transfer, assign, pledge, hypothecate, tender, encumber or otherwise dispose of or limit their right to vote in any manner any of the Committed Shares, or agree to do any of the foregoing, and (ii) will not take any action which would have the effect of preventing or

38

disabling the Tendering Shareholders from performing their obligations under the Tender Agreements. Notwithstanding the foregoing, in connection with any transfer not involving or relating to any Competing Acquisition Proposal, the Tendering Shareholders may transfer any or all of the Committed Shares as follows: (i) in the case of a Tendering Shareholder that is an entity, to any subsidiary, partner or member of the Tendering Shareholder and (ii) in the case of an individual Tendering Shareholder, to such Tendering Shareholder's descendants or any trust for any of their benefits or to a charitable trust; provided, however, that in any such case, prior to and as a condition to the effectiveness of such transfer, (x) each person or entity to which any of such Committed Shares or any interest in any of such Committed Shares is or may be transferred (a) shall have executed and delivered to TCCC and the Offeror a counterpart to the applicable Tender Agreement pursuant to which such person or entity shall be bound by all of the terms and provisions of the applicable Tender Agreement, and (b) shall have agreed in writing with TCCC and the Offeror to hold such Shares or interest in such Shares subject to all of the terms and provisions of the applicable Tender Agreements, and (y) the applicable Tender Agreement shall be the legal, valid and binding agreement of such person, enforceable against such person in accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created by the applicable Tender Agreement is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles.

In addition, during the term of the Tender Agreements, neither the Tendering Shareholders nor any person acting as an agent of the Tendering Shareholders or otherwise on the Tendering Shareholders' behalf shall, directly or indirectly, (a) solicit, initiate or encourage the submission of any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Committed Shares or (b) participate in or encourage any discussion or negotiations regarding, or furnish to any person any non-public information with respect to, enter into any agreement with respect to, or take any other action to facilitate any inquiries or the making of any proposal that constitutes, or may reasonably be expected to lead to, any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Committed Shares, in any case, from, to or with any person other than TCCC or the Offeror. The Tendering Shareholders agreed to immediately cease and cause to be terminated any existing activities, discussions or negotiations with any parties with respect to any of the foregoing. The Tendering Shareholders also agreed to notify the Offeror immediately if any party contacts the Tendering Shareholders following the date of the Tender Agreements (other than the Offeror or an affiliate or associate of the Offeror) concerning any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of the Committed Shares.

The Tender Agreements terminate upon the earlier of (1) the termination of the Merger Agreement pursuant to Sections 8.1(a), (b), (c), (d) or (e) of the Merger Agreement or (2) six months after the termination of the Merger Agreement pursuant to Sections 8.1 (f), (g) or (h) of the Merger Agreement.

Employment Agreement. Concurrently with the execution of the Merger Agreement, Odwalla entered into a new employment agreement with D. Stephen C. Williamson, the Chief Executive Officer and Chairman of the Board of Odwalla (the "New Employment Agreement"). The New Employment Agreement will commence at the Effective Time (as defined in the Merger Agreement) and will supersede Mr. Williamson's current employment agreement with Odwalla, dated December 21, 1999, which is more fully described in the section entitled "Certain Relationships and Related Transactions" in the Information Statement attached as Annex I to the Schedule 14D-9.

The initial term of the New Employment Agreement extends through December 31, 2002, and thereafter is subject to annual automatic renewals through December 31, 2005, unless either party provides 60 days written notice to the other party in advance of a renewal period. Under the New Employment Agreement, Mr. Williamson will receive an annual salary of \$450,000, subject to annual review by the Board of Directors, and will be eligible for a special performance award of 20,000 TCCC stock options for achieving certain revenue and operating income targets. Odwalla shall also provide Mr. Williamson customary fringe benefits provided to similarly situated employees at Odwalla.

Upon the expiration of the New Employment Agreement, Odwalla shall pay Mr. Williamson his base salary then in effect for a period of one year following the date of the expiration, offset by any severance

39

payment Mr. Williamson may otherwise receive. Odwalla may terminate the New Employment Agreement with written notice for cause without liability or further obligation. If Mr. Williamson is not terminated for cause or Mr. Williamson terminates the New Employment Agreement for good reason, Odwalla will (1) pay Mr. Williamson his base salary then in effect for a period of one year following the date of the expiration, offset by any severance payment Mr. Williamson may otherwise receive, (2) reimburse Mr. Williamson for the cost of acquiring health benefits for a period of one year, and (3) negotiate with Mr. Williamson the treatment of any special performance award. In the event that Mr. Williamson's severance and other benefits constitute a parachute payment under the Federal tax law and would be subject to an excise tax, then Mr. Williamson's benefits will either be delivered in full or delivered to such lesser extent as to avoid an excise tax, whichever results in Mr. Williamson receiving the greatest amount on an after-tax basis.

Under the terms of the New Employment Agreement, Mr. Williamson agreed that he will not, while employed by Odwalla and for a period of two years following the expiration or termination of his employment, solicit, interfere with or endeavor to entice away from TCCC or any of its subsidiaries any other employee of TCCC. Additionally, Mr. Williamson agreed that he will not at any time while employed by Odwalla and for a period of one year following the expiration or termination of his employment engage in the manufacture, sale, or distribution of non-alcoholic beverages in the United States.

During the term of the New Employment Agreement and at all times thereafter, Mr. Williamson has agreed to keep in confidence and not publish, use or disclose to others, without Odwalla's prior written consent, any trade secrets or other confidential information related to TCCC or TCCC's business.

12. PURPOSE OF THE OFFER; THE MERGER; PLANS FOR ODWALLA. The purpose of the Offer is to acquire control of, and the entire equity interest in, Odwalla. The Offer is being made pursuant to the Merger Agreement. As promptly as practicable following the purchase of Shares pursuant to the Offer and after satisfaction or waiver of all conditions to the Merger set forth in the Merger Agreement, the Offeror intends to acquire the remaining equity interest in Odwalla not acquired in the Offer by consummating the Merger.

Vote Required to Approve the Merger. The Board of Directors of Odwalla has approved and adopted the Offer, the Merger and the Merger Agreement in accordance with the CGCL. If the Minimum Condition or the Option Exercise Minimum Number, as applicable, and the other conditions to the Offer are satisfied and the Offer is consummated, Offeror will own a number of Shares to cause the Merger to occur without a vote of the shareholders of Odwalla, pursuant to Section 1110 of the CGCL. The Minimum Condition requires that there shall have been validly tendered and not properly withdrawn, together with the Shares owned by TCCC, the Offeror and their direct and indirect subsidiaries, at least ninety and one-tenth percent (90.1%) of the Shares then outstanding on a Fully Diluted Basis. If however, the Minimum Condition or the Option Exercise Minimum Number, as applicable, is not satisfied but the Revised Minimum Number and the other conditions are satisfied, the Board of Directors will be required to submit the Merger Agreement to Odwalla's shareholders for approval at a shareholders' meeting convened for that purpose in accordance with the CGCL. Pursuant to the Merger Agreement, Odwalla has represented that the execution and delivery of the Merger Agreement and the Option Agreement by Odwalla and the

consummation by Odwalla of the transactions contemplated thereby have been duly authorized by all necessary corporate action on the part of Odwalla, subject to the approval and adoption of the Merger by the shareholders of Odwalla in accordance with the CGCL. In addition, Odwalla has represented that the affirmative vote of the holders of a majority of the outstanding Shares is the only vote of the holders of any of Odwalla's capital stock necessary in connection with the consummation of the Merger. Therefore, unless the Merger is consummated in accordance with the short-form merger provisions under the CGCL described below (in which case no action by the shareholders of Odwalla will be required to consummate the Merger), the only remaining corporate action of Odwalla will be the approval and adoption of the Merger Agreement and the transactions contemplated thereby by the affirmative vote of the holders of a majority of the Shares. The Merger Agreement provides that TCCC will vote all Shares beneficially owned by it in favor of the adoption of the Merger Agreement at Odwalla shareholder's meeting. If the Offeror acquires the Revised Minimum Number of Shares, it would have the ability to ensure approval of the Merger by the shareholders of Odwalla with the approval of a de minimis number of remaining outstanding Shares.

40

Under the CGCL, the merger consideration paid to Odwalla's shareholders may not be cash if Offeror or TCCC owns, directly or indirectly, more than 50% but less than 90% of the then outstanding Shares unless either all the shareholders consent to the Merger or the Commissioner of Corporations of the State of California approves, after a hearing, the terms and conditions of the Merger and the fairness thereof. If such shareholder consent or Commissioner of Corporations approval is not obtained, the CGCL requires that the consideration received in the Merger consist only of non-redeemable common stock of TCCC. The purpose of the Offer is to obtain 90.1% or more of the Shares then outstanding on a Fully Diluted Basis and thus to enable TCCC and Offeror to acquire all the equity of Odwalla for consideration consisting solely of cash.

THIS OFFER TO PURCHASE ALSO CONSTITUTES NOTICE UNDER SECTION 1110 OF THE CGCL THAT IF THE OFFEROR, TCCC OR ANY OTHER SUBSIDIARY OF TCCC ACQUIRES AT LEAST 90.1% OF THE OUTSTANDING SHARES ON A FULLY DILUTED BASIS, TCCC OR THE OFFEROR SHALL CAUSE THE SHORT-FORM MERGER TO BECOME EFFECTIVE WITHOUT ANY FURTHER NOTICE TO ODWALLA'S SHAREHOLDERS. SEE APPENDIX A (AGREEMENT AND PLAN OF MERGER).

THE OFFER IS CONDITIONED UPON THE MINIMUM CONDITION, THE OPTION EXERCISE MINIMUM NUMBER OR THE REVISED MINIMUM NUMBER BEING SATISFIED.

THIS OFFER TO PURCHASE DOES NOT CONSTITUTE A SOLICITATION OF A PROXY, CONSENT OR AUTHORIZATION FOR OR WITH RESPECT TO THE ANNUAL MEETING OR ANY SPECIAL MEETING OF ODWALLA'S SHAREHOLDERS OR ANY ACTION IN LIEU THEREOF. ANY SUCH SOLICITATION WHICH THE OFFEROR MAY MAKE WILL BE MADE ONLY PURSUANT TO SEPARATE PROXY MATERIALS IN COMPLIANCE WITH THE REQUIREMENTS OF SECTION 14 (A) OF THE EXCHANGE ACT.

Dissenters' Rights. Holders of Shares do not have dissenters' rights as a result of the Offer. However, if the Merger is consummated, certain holders of Shares who fully comply with and meet all the requirements of Chapter 13 of the CGCL ("Qualifying Shareholders"), may have certain rights to dissent and to require Odwalla to purchase their Shares for cash at "fair market value." Additionally, Qualifying Shareholders will be entitled to exercise dissenters' rights under the CGCL only if the holders of five percent or more of the outstanding Shares properly file demands for payment or if the Shares held by such holders are subject to any restriction on transfer imposed by Odwalla or by any law or regulation ("Restricted Shares"). Accordingly, if the holders of five percent or more of the Shares properly file demands for payment in compliance with Chapter 13 of the CGCL, all other Qualifying Shareholders will be entitled to require Odwalla to purchase their Shares for cash at their fair market value if the Merger is consummated. If the holders of fewer than five percent of the Shares properly file demands for payment in compliance with Chapter 13 of the CGCL but any holder of Restricted Shares properly files such a demand, only such holder or holders of Restricted Shares shall be entitled to require Odwalla to purchase their Shares as described in the preceding sentence. In addition, if immediately prior to the Effective Time, the Shares are not listed on a national securities exchange certified by the California Commission of Corporations or listed on the National Market System of the Nasdaq Stock Market, holders of Shares may exercise dissenters' rights as to any or all of their Shares entitled to such rights.

Under the CGCL, the "fair market value" of the Shares may be one agreed to by Odwalla and the Qualifying Shareholder or judicially determined, depending on the circumstances. The "fair market value" is determined as of the day before the first announcement of the terms of the proposed Merger, excluding any appreciation or depreciation as a result of the Merger and subject to adjustments. The value so determined could be more or less than the Offer Price.

If the Merger is not completed, no Qualifying Shareholder will be entitled to have Odwalla purchase such holder's Shares under Chapter 13 of the CGCL. If a shareholder and Odwalla do not agree on whether that shareholder is a Qualifying Shareholder, or if a Qualifying Shareholder and Odwalla fail to agree on the fair market value of Shares and neither Odwalla nor the Qualifying Shareholder

files a complaint or intervenes in a pending action within six months after Odwalla mails the required notice that shareholders have approved the Merger, that shareholder does not have (or will cease to have) rights as a dissenting shareholder. After a shareholder files a demand to exercise dissenters' rights, that shareholder may not withdraw the demand without Odwalla's consent.

4

The foregoing discussion of the rights of Qualifying Shareholders does not purport to be a complete statement of the procedures to be followed by shareholders desiring to exercise any available dissenters' rights and is qualified in its entirety by reference to Chapter 13 of the CGCL, which is set forth in Annex II to this Offer to Purchase and incorporated herein by reference.

FAILING TO FOLLOW THE STEPS REQUIRED BY CHAPTER 13 OF THE CGCL FOR PERFECTING DISSENTER'S RIGHTS MAY RESULT IN THE LOSS OF SUCH RIGHTS.

Rule 13e-3. The Commission has adopted Rule 13e-3 under the Exchange Act which is applicable to certain "going private" transactions and which may under certain circumstances be applicable to the Merger following the purchase of Shares pursuant to the Offer in which the Offeror seeks to acquire any remaining Shares. Rule 13e-3 should not be applicable to the Merger if the Merger is consummated within one year after the expiration or termination of the Offer and the price paid in the Merger is not less than the per Share price paid pursuant to the Offer. However, if the Offeror is deemed to have acquired control of Odwalla pursuant to the Offer and if the Merger is consummated more than one year after completion of the Offer or an alternative acquisition transaction is effected whereby shareholders of Odwalla receive consideration less than that paid pursuant to the Offer, in either case at a time when the Shares are still registered under the Exchange Act, the Offeror may be required to comply with Rule 13e-3 under the Exchange Act. If applicable, Rule 13e-3 would require, among other things, that certain financial information concerning Odwalla and certain information relating to the fairness of the Merger or such alternative transaction and the consideration offered to minority shareholders in the Merger or such alternative transaction, be filed with the Commission and disclosed to shareholders prior to consummation of the Merger or such alternative transaction. The purchase of a substantial number of Shares pursuant to the Offer may result in Odwalla being able to terminate its Exchange Act registration. See Section 14 ("Effect of the Offer on the Market for the Shares, Stock Exchange Listing and Exchange Act Registration"). If such registration were terminated, Rule 13e-3 would be inapplicable to any such future Merger or such alternative transaction.

Plans for Odwalla. TCCC will continue to evaluate the business and operations of Odwalla during the pendency of the Offer and after the consummation of the Offer and the Merger. TCCC intends to seek additional information about Odwalla during this period. Thereafter, TCCC intends to review such information as part of a comprehensive review of Odwalla's business, operations, capitalization and management with a view to optimizing Odwalla's potential contribution to TCCC's business.

Except as indicated in this Offer to Purchase, TCCC does not have any current plans or proposals which relate to or would result in any of the following: an extraordinary corporate transaction, such as a merger, reorganization or liquidation involving Odwalla or any of its subsidiaries; a sale or transfer of a material amount of assets of Odwalla or any of its subsidiaries; any change in the present Board of Directors or management of Odwalla; any material change in Odwalla's present capitalization or dividend policy; or any other material change in Odwalla's corporate structure or business. Notwithstanding the foregoing, promptly after the Offeror acquires a majority of the Shares, the Offeror will be entitled to designate such number of directors on the Board of Directors of Odwalla as will make the percentage of Odwalla's directors designated by the Offeror equal to the percentage of aggregate voting power of the Shares held by TCCC or any of its subsidiaries. In addition, assuming the Offeror's nominees are appointed as directors of Odwalla and so long as there are holders of Shares other than TCCC or any of its subsidiaries, TCCC expects that the Board of Directors of Odwalla would not declare dividends on the Shares.

13. DIVIDENDS AND DISTRIBUTIONS. The Merger Agreement provides that between the date of the Merger Agreement and the earlier of the Effective Time or the termination of the Merger Agreement in accordance with its terms, unless TCCC otherwise expressly agrees in writing (which agreement shall not be unreasonably withheld or delayed), Odwalla shall not, nor shall Odwalla permit any of its subsidiaries to, (i) declare or pay any dividends on or make other distributions (whether in cash, stock or property) in respect of any of its capital stock, except for dividends payable to Odwalla by a wholly owned subsidiary of Odwalla; (ii) split, combine or reclassify any of its capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; (iii) repurchase, redeem or otherwise acquire any shares of its capital stock; (iv) issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock or any securities convertible into any such shares of

its capital stock, or any rights, warrants or options to acquire any such shares or convertible securities or any stock appreciation rights, phantom stock plans or stock equivalents, other than the issuance of Shares upon (A) the exercise of Company Options outstanding as of the date of the Merger Agreement or awarded in conformity with clause; (v) below, and (B) the exercise of warrants outstanding as of the date of the Merger Agreement; (v) award or grant, or authorize or propose the award or grant of, any unvested Company Options, other than the award or grant of Company Options to employees (other than executive officers of Odwalla) in accordance with the terms of the Company Stock Option Plans and in the ordinary course of business and consistent with past practices for up to no more than 100,000 Shares in the aggregate for all employees issued to employees who have not prior to the date of the Merger Agreement received option grants in calendar year 2001; (vi) modify or adjust any outstanding options to acquire Shares; or (vii) take any action that would, or could reasonably be expected to, result in any of the conditions to closing set forth in the Merger Agreement not being satisfied.

14. EFFECT OF THE OFFER ON THE MARKET FOR THE SHARES, STOCK EXCHANGE LISTING AND EXCHANGE ACT REGISTRATION. The purchase of Shares pursuant to the Offer will reduce the number of Shares that might otherwise trade publicly and will reduce the number of holders of Shares, which could adversely affect the liquidity and market value of the remaining Shares held by the public. Following the purchase of Shares pursuant to the Offer, at least ninety and one-tenth percent (90.1%) of the Shares then outstanding on a Fully Diluted Basis or 49.9% of the outstanding Shares will be owned by the Offeror.

Depending on the number of Shares purchased pursuant to the Offer, the Shares may no longer meet the requirements of the Nasdag National Market for continued listing. To maintain such designation, a security must substantially meet one of two maintenance standards. The first maintenance standard requires that (i) there be at least 750,000 publicly held shares; (ii) the publicly held shares have a market value of at least \$5 million; (iii) the issuer have net tangible assets of at least \$4 million or stockholder's equity of \$10 million; (iv) there be at least 400 shareholders of round lots; (v) the minimum bid price per share must be at least \$1.00; and (vi) there be at least two registered and active market makers. The second maintenance standard requires that (i) the issuer have either (A) a market capitalization of at least \$50 million or (B) total assets and total revenue of at least \$50 million each for the most recently completed fiscal year or two of the last three most recently completed fiscal years; (ii) there be at least 1,100,000 shares publicly held; (iii) the publicly held shares have a market value of at least \$15 million; (iv) the minimum bid price per share be at least \$3.00; (v) there be at least 400 shareholders of round lots; and (vi) there be at least four registered and active market makers.

If these standards for continued listing for Nasdaq National Market are not met, the Shares might nevertheless continue to be included in the Nasdaq SmallCap Market. Inclusion in the Nasdaq SmallCap Market, however, would require that (i) there be at least 300 round lot holders, (ii) there be at least 500,000 publicly held shares, (iii) the publicly held shares have a market value of at least \$1 million, (iv) there be at least two registered and active market makers, of which one may be entering stabilizing bids and (v) the issuer has either (A) net tangible assets of at least \$2 million or stockholder's equity of \$2.5 million, (B) market capitalization of at least \$35 million or (C) net income of at least \$500,000 in the most recently completed fiscal year or in two of the last three most recently completed fiscal years. Shares held directly or indirectly by directors, officers or beneficial owners of more than 10% of the Shares are not considered as being publicly held for the purpose of determining whether either of the Nasdaq Stock Market listing criteria are met.

On September 27, 2001 the Nasdaq Stock Market implemented a moratorium on the minimum bid and public float requirements for continued listing on the Nasdaq National Market and the Nasdaq SmallCap Market. The temporary relief is available until January 2, 2002. Until that date, companies will not be cited for failing to meet the above described minimum bid and public float requirements for continued listing.

If the purchase of Shares pursuant to the Offer causes the Shares to no longer meet the requirements for continued inclusion in Nasdaq National Market or the Nasdaq SmallCap Market as a result of a reduction in the number or market value of publicly held Shares or the number of round lot holders or otherwise, as the case may be, the market for Shares could be adversely affected. It is possible that the Shares would continue to trade in the over-the-counter market and that price quotations would be reported by other sources. The extent of the public market therefor and the availability of such quotations would depend, however, upon such

registration under the Exchange Act as described below and other factors. The Offeror cannot predict whether the reduction in the number of Shares that might otherwise trade publicly would have an adverse or beneficial effect on the market price for or marketability of the Shares or whether it would cause future market prices to be greater or less than the Offer Price.

The Shares are currently registered under the Exchange Act. The purchase of Shares pursuant to the Offer may result in the Shares becoming eligible for deregistration under the Exchange Act. Registration of the Shares may be terminated upon application of Odwalla to the Commission if the Shares are not listed on a national securities exchange and there are fewer than 300 record holders. The termination of the registration of the Shares under the Exchange Act would substantially reduce the information required to be furnished by Odwalla to holders of the Shares and would make certain provisions of the Exchange Act, such as the short-swing profit recovery provisions of Section 16(b), the requirement of furnishing a proxy statement in connection with shareholders' meetings and the requirements of Rule 13e-3 under the Exchange Act with respect to "going private" transactions, no longer applicable to the Shares. Furthermore, "affiliates" of Odwalla and persons holding "restricted securities" of Odwalla may be deprived of the ability to dispose of the securities pursuant to Rule 144 under the Securities Act.

The Shares are currently "margin securities" under the rules of the Board of Governors of the Federal Reserve System (the "Federal Reserve Board"), which has the effect, among other things, of allowing brokers to extend credit on the collateral of such Shares for the purpose of buying, carrying, or trading in securities. Depending upon factors similar to those described above with respect to listing and market quotations, it is possible that, following the Offer, the Shares might no longer constitute "margin securities" for the purposes of the Federal Reserve Board's margin regulations and therefore could no longer be used as collateral for purpose credits made by brokers. In any event, the Shares will cease to be "margin securities" if registration of the Shares under the Exchange Act is terminated.

TCCC currently intends to seek delisting of the Shares from Nasdaq and the termination of the registration of the Shares under the Exchange Act as soon after completion of the Offer as the requirements for such delisting and termination are met. If Nasdaq listing and the Exchange Act registration of the Shares are not terminated prior to the Merger, then the Shares will be delisted from Nasdaq and the registration of the Shares under the Exchange Act will be terminated following the consummation of the Merger.

- 15. CERTAIN CONDITIONS OF THE OFFER. Notwithstanding any other provision of the Offer, and in addition to (and not in limitation of) the Offeror's rights to extend and/or amend the Offer at any time in its sole discretion (subject to the provisions of the Merger Agreement), the Offeror's obligations under the Merger Agreement pursuant to the Offer shall be subject to the following conditions, and if such conditions are not satisfied, the Offeror may delay the acceptance for payment of or, subject to any applicable rules and regulations of the Commission, including Rule 14e-1(c) under the Exchange Act, the payment for, any tendered Shares, and may terminate or amend the Offer as to any Shares not then paid for:
  - -- at or prior to the expiration date of the Offer, the number of Shares validly tendered and not withdrawn, together with any Shares then owned by TCCC or the Offeror, shall satisfy the Minimum Condition or, if applicable pursuant to the provisions of the Merger Agreement, the Option Exercise Minimum Number or the Revised Minimum Number;
  - -- at or prior to the expiration date of the Offer, (a) any applicable waiting period under the HSR Act shall have expired or been terminated, and (b) all other actions by or in respect of or other filings with any governmental entity required to permit the consummation of the Offer and the Merger shall have been obtained or made other than in the case of clause (b), such actions or filings the failure of which to make or obtain would not reasonably be expected to have a Company Material Adverse Effect;
  - -- the Shareholders' Rights Agreement dated as of May 2, 2000 among Odwalla and certain shareholders shall have been terminated;
  - -- immediately prior to the consummation of the Offer all unpaid fines, penalties and interest owed to the United States government under Odwalla's settlement agreement with the United States

44

government dated July 23, 1998 (the "Settlement Fee") shall have been paid, unless such failure to make payment of the Settlement Fee is caused by the failure of TCCC to meet its obligation to loan Odwalla the amount necessary to pay the Settlement Fee; and Odwalla shall have provided to the Food and Drug Administration's San Francisco District Office not less than 10 days prior notice of the consummation of the Offer; and

- -- at any time after the date of the Merger Agreement and prior to acceptance for payment of Shares pursuant to the Offer, none of the following events or conditions shall occur or exist:
  - there shall be instituted, pending or threatened any action or proceeding by any governmental authority: (i) challenging or seeking to permanently restrain, enjoin or otherwise prohibit the Offer or the Merger; (ii) seeking to restrain or prohibit TCCC's or the Offeror's full rights of ownership or operation of any portion of the business or assets of Odwalla, or to compel TCCC or the Offeror to dispose of or hold separate all or any portion of the business or assets of Odwalla; (iii) seeking to impose limitations on the ability of TCCC or the Offeror effectively to exercise full rights of ownership of the Shares acquired pursuant to the Offer and the Merger, including, without limitation, the right to vote any Shares acquired or owned by TCCC or the Offeror on all matters properly presented to Odwalla's shareholders; (iv) seeking to require divestiture by TCCC or the Offeror of any Shares; or (v) that otherwise would reasonably be expected to have a Company Material Adverse Effect:
  - -- there shall be enacted, enforced, promulgated, in force, or deemed applicable to the Offer or the Merger by any governmental authority any statute, rule, regulation, judgment, order or injunction (other than the application of the routine waiting period provisions of the HSR Act) that has, directly or indirectly, resulted, or is reasonably likely to, directly or indirectly, result in any of the consequences referred to in the paragraph above;
- -- an event shall have occurred that has resulted in, or would be reasonably expected to result in, a Company Material Adverse Effect; or
- any court of competent jurisdiction or other governmental entity shall have issued an order, decree, ruling or take any other action restraining, enjoining or otherwise prohibiting the Offer or the Merger provided, that with respect to shareholder litigation, unless such order, decree, ruling or other action shall be final and nonappealable, this condition in and of itself shall not be a ground for termination and shall only have the effect of delaying the consummation of the Offer during the effective period of such order, decree, ruling or other action;
- there shall have occurred (i) any general suspension of trading in, or limitation on prices for, securities on any national securities exchange or in the over-the-counter market, (ii) the declaration of any banking moratorium or any suspension of payments in respect of banks or any material limitation (whether or not mandatory) on the extension of credit by lending institutions in the United States, (iii) the commencement of a war, material armed hostilities or other material international or national calamity directly or indirectly involving the United States that has a significant adverse effect on the functioning of the financial markets in the United States, or (iv) in the case of any of the foregoing existing at the time of execution of the Agreement, a material acceleration or worsening thereof, provided that this condition in and of itself shall not be a ground for termination and shall only have the effect of delaying the consummation of the Offer during the effective period of such declaration or suspension or other event;
- -- (A) the representations and warranties of Odwalla in the Merger Agreement shall not be true and correct in all respects as of the date of the Merger Agreement and as of the expiration of the Offer without giving effect to any exceptions for "material" or "materiality" or "Company Material Adverse Effect", except where the failure of such representations and warranties to be true and correct, individually or in the aggregate, does not have and is not reasonably likely to have a Company Material Adverse Effect; (B) Odwalla shall not have performed in all material respects all covenants, agreements and obligations required to be performed by it under the Merger Agreement; (C) the directors of Odwalla shall have not resigned as provided in the Merger

45

Agreement; or (D) an officer of Odwalla shall not have delivered to TCCC and the Offeror a certificate to the effect that each of the foregoing conditions are satisfied in all respects; or

-- the Merger Agreement shall have been terminated in accordance with its terms or amended in accordance with its terms to provide for such termination or amendment of the Offer.

circumstances giving rise to any such condition (including any action or omission by TCCC or the Offeror), in whole or in part at any time and from time to time in their sole discretion. The failure by TCCC or the Offeror at any time to exercise any of the foregoing rights shall not be deemed a waiver of any such right; the waiver of any such right with respect to any particular facts and circumstances shall not be deemed a waiver with respect to any other facts and circumstances; and each such right shall be deemed an ongoing right and may be asserted at any time and from time to time.

## 16. CERTAIN LEGAL MATTERS AND REGULATORY APPROVALS.

General. Except as set forth below, based upon its examination of publicly available filings by Odwalla with the Commission and other publicly available information concerning Odwalla, neither the Offeror nor TCCC is aware of any licenses or other regulatory permits that appear to be material to the business of Odwalla and its subsidiaries, taken as a whole, that might be adversely affected by the Offeror's acquisition of Shares (and the indirect acquisition of the stock of Odwalla's subsidiaries) as contemplated herein, or of any filings, approvals or other actions by or with any domestic (Federal or state), foreign or supranational governmental authority or administrative or regulatory agency that would be required prior to the acquisition of Shares (or the indirect acquisition of the stock of Odwalla's subsidiaries) by the Offeror pursuant to the Offer as contemplated herein. Should any such approval or other action be required, it is the Offeror's present intention to seek such approval or action. However, the Offeror does not presently intend to delay the purchase of Shares tendered pursuant to the Offer pending the receipt of any such approval or the taking of any such action (subject to the Offeror's right to delay or decline to purchase Shares if any of the conditions in Section 15 shall have occurred). There can be no assurance that any such approval or other action, if needed, would be obtained without substantial conditions or that adverse consequences might not result to the business of Odwalla, TCCC or the Offeror or that certain parts of the businesses of Odwalla, TCCC or the Offeror might not have to be disposed of or held separate or other substantial conditions complied with in order to obtain such approval or other action or in the event that such approval was not obtained or such other action was not taken, any of which could cause the Offeror to elect to terminate the Offer without purchasing the Shares thereunder. The Offeror's obligation under the Offer to accept for payment and pay for Shares is subject to certain conditions, including conditions relating to the legal matters discussed in this Section 16.

Antitrust. Under the HSR Act and the rules promulgated thereunder by the Federal Trade Commission (the "FTC"), the acquisition of Shares under the Offer may not be consummated until notifications have been given and certain information has been furnished to the FTC and the Antitrust Division of the Department of Justice (the "Antitrust Division") and the specified waiting period requirements have been satisfied. The FTC has deemed that TCCC filed notification and report forms under the HSR Act with the FTC and the Antitrust Division on November 2, 2001. Accordingly, the waiting period under the HSR Act will expire at 11:59 p.m., Eastern time, on November 19, 2001 unless TCCC or Odwalla receives a request for additional documentary material, or the Antitrust Division and the FTC terminate the waiting period prior thereto. In practice, complying with a request for additional information or material can take a significant amount of time. In addition, if the Antitrust Division or the FTC raises substantive issues in connection with a proposed transaction, the parties may engage in negotiations with the relevant governmental agency concerning possible means of addressing those issues and may agree to delay consummation of the transaction while such negotiations continue.

The Antitrust Division and the FTC frequently scrutinize the legality under the antitrust laws of transactions such as the Offeror's proposed acquisition of Odwalla. At any time before or after the Offeror's acquisition of Shares pursuant to the Offer, the Antitrust Division or the FTC could take such action under the antitrust laws as either deems necessary or desirable in the public interest, including seeking to enjoin the purchase of Shares pursuant to the Offer or the consummation of the Merger, or seeking the divestiture of Shares acquired by the Offeror or the divestiture of substantial assets of Odwalla or its subsidiaries or TCCC

46

or its subsidiaries. Private parties may also bring legal action under the antitrust laws under certain circumstances. Additionally, at any time before or after the Offeror's acquisition of Shares pursuant to the Offer and notwithstanding that the HSR Act waiting period may have expired, any state could take such action under the antitrust laws as it deems necessary or desirable in the public interest. Such action could include seeking to enjoin the purchase of Shares pursuant to the Offer or the consummation of the Merger, or seeking the divestiture of Shares acquired by the Offeror or the divestiture of substantial assets of Odwalla or its subsidiaries or TCCC or its subsidiaries. There can be no assurance that a challenge to the Offer, the consummation of the Merger or the tender of the Shares pursuant to the Tender Agreements on antitrust grounds will not be made, or, if such a challenge is made, of the result thereof.

If any waiting period under the HSR Act applicable to the Offer has not

expired or been terminated prior to the Expiration Date, the Offeror will not be obligated to proceed with the Offer or the purchase of any Shares not theretofore purchased pursuant to the Offer. See Section 15 ("Certain Conditions of the Offer").

State Anti-Takeover Laws--California. Odwalla is incorporated under the laws of the State of California. Section 1203 of the CGCL provides that if a tender offer is made to some or all of a corporation's shareholders by an "interested party," an affirmative opinion in writing as to the fairness of the consideration to the shareholders of such corporation is required to be delivered to the shareholders at the time that the tender offer is first made in writing to the shareholders. However, if the tender offer is commenced by publication and tender offer materials are subsequently mailed or otherwise distributed to the shareholders, the opinion may be omitted in the publication if the opinion is included in the materials distributed to the shareholders. For purposes of Section 1203, the term "interested party" includes, among other things, a person who is a party to the transaction and (A) directly or indirectly controls the corporation that is the subject of the tender offer or proposal, (B) is, or is directly or indirectly controlled by, an officer or director of the subject corporation or (C) is an entity in which a material financial interest is held by any director or executive officer of the subject corporation. While none of Odwalla, TCCC or Offeror believes that the Offer constitutes a transaction that falls within the provisions of Section 1203, an independent financial advisor, DrKW, has been retained by Odwalla to provide a fairness opinion with respect to the Offer and has provided such opinion to

Under the CGCL, the Merger consideration paid to the shareholders of Odwalla may not be cash if the Offeror or TCCC owns directly or indirectly more than 50% but less than 90% of the then outstanding Shares, unless either all the shareholders of Odwalla consent or the Commissioner of Corporations of the State of California approves, after a hearing, the terms and conditions of the Merger and the fairness thereof.

If the Minimum Condition is not satisfied on any scheduled Expiration Date of the Offer, the Offeror may, in its sole discretion, either (x) extend the Offer pursuant to the provisions of the Merger Agreement, (y) amend the Offer in contemplation of the exercise of the Option Agreement (to the extent the Option Agreement is exercisable at such time) to reduce the Minimum Condition to the Option Exercise Minimum Number, or (z) amend the Offer to provide that, in the event (i) the Minimum Condition is not satisfied at the next scheduled expiration date of the Offer (after giving effect to the issuance of any Shares theretofore acquired by TCCC or Offeror) and (ii) the number of Shares tendered pursuant to the Offer and not withdrawn as of such next scheduled expiration date is more than 50% of the then outstanding Shares, Offeror shall waive the Minimum Condition and amend the Offer to reduce the number of Shares subject to the Offer to the Revised Minimum Number and, subject to the prior satisfaction or waiver of the other conditions of the Offer, purchase, on a pro rata basis, the Revised Minimum Number of shares (it being understood that Offeror shall not in any event be required to accept for payment, or pay for, any Shares less than the Revised Minimum Number of Shares are tendered pursuant to the Offer and not withdrawn at the expiration date).

State Takeover Laws--Other. A number of states have adopted takeover laws and regulations which purport to varying degrees to be applicable to attempts to acquire securities of corporations which are incorporated in such states or which have or whose business operations have substantial economic effects in such states, or which have substantial assets, security holders, principal executive offices or principal places of business therein. In 1982, the Supreme Court of the United States, in Edgar v. Mite Corp., invalidated on constitutional grounds the Illinois Business Takeovers Act, which as a matter of state securities law made takeovers of corporations meeting certain requirements more difficult, and the reasoning in such decision is

47

likely to apply to certain other state takeover statutes. However, in 1987, in CTS Corp. v. Dynamics Corp. of America, the Supreme Court of the United States held that the State of Indiana could, as a matter of corporate law and in particular those aspects of corporate law concerning corporate governance, constitutionally disqualify a potential acquiror from voting on the affairs of a target corporation without the prior approval of the remaining shareholders, provided that such laws were applicable only under certain conditions. Subsequently, in TLX Acquisition Corp. v. Telex Corp., a federal district court in Oklahoma ruled that the Oklahoma statutes were unconstitutional insofar as they applied to corporations incorporated outside Oklahoma in that they would subject such corporations to inconsistent regulations. Similarly, in Tyson Foods, Inc. v. McReynolds, a federal district court in Tennessee ruled that four Tennessee takeover statutes were unconstitutional as applied to corporations incorporated outside Tennessee. This decision was affirmed by the United States Court of Appeals for the Sixth Circuit.

Except as described herein, the Offeror has not attempted to comply with any state takeover statutes in connection with the Offer. The Offeror reserves the right to challenge the validity or applicability of any state law allegedly

applicable to the Offer and nothing in this Offer to Purchase nor any action taken in connection herewith is intended as a waiver of that right. If any state takeover statute is found applicable to the Offer, the Offeror might be unable to accept for payment or purchase Shares tendered pursuant to the Offer or be delayed in continuing or consummating the Offer. In such case, the Offeror may not be obligated to accept for purchase or pay for any Shares tendered. See Section 15 ("Certain Conditions of the Offer").

Margin Regulations. The Shares are currently "margin securities," as such term is defined under the rules of the Board of Governors of the Federal Reserve System (the "Federal Reserve Board"), which has the effect, among other things, of allowing brokers to extend credit on the collateral of such securities. Depending upon factors similar to those described above regarding listing and market quotations, following the Offer, it is possible that the Shares might no longer constitute "margin securities" for purposes of the margin regulations of the Federal Reserve Board, in which event such Shares could no longer be used as collateral for loans made by brokers. In addition, if registration of the Shares under the Exchange Act were terminated, the Shares would no longer constitute "margin securities."

17. FEES AND EXPENSES. Except as set forth below, neither the Offeror nor TCCC, nor any officer, director, shareholder, agent or other representative of the Offeror or TCCC will pay any fees or commissions to any broker, dealer or other person for soliciting tenders of Shares pursuant to the Offer. Brokers, dealers, commercial banks and trust companies and other nominees will, upon request, be reimbursed by the Offeror for customary mailing and handling expenses incurred by them in forwarding materials to their customers.

Morgan Stanley is acting as Dealer Manager in connection with the Offer and has provided certain financial advisory services to TCCC in connection with the proposed acquisition of Odwalla. TCCC has agreed to pay Morgan Stanley customary compensation for such services. TCCC has also agreed to reimburse Morgan Stanley for reasonable expenses incurred, including fees and expenses of legal counsel or advisors should they be engaged by Morgan Stanley, and to indemnify Morgan Stanley and certain related persons against certain liabilities and expenses in connection with the engagement of Morgan Stanley, including certain liabilities under the federal securities laws. Morgan Stanley and its affiliates have from time to time rendered, and continue to render, various investment banking services to TCCC and have received, and may receive, fees for the rendering of such services. In addition, at any time, Morgan Stanley and its affiliates may actively trade Shares for its own account or for the account of customers and, accordingly, may at any time hold a long or short position in Shares.

The Offeror has retained Innisfree M&A Incorporated as Information Agent and EquiServe Trust Company, N.A. as Depositary in connection with the Offer. The Information Agent and the Depositary will receive reasonable and customary compensation for their services hereunder and reimbursement for their reasonable out-of-pocket expenses. The Depositary will also be indemnified by the Offeror against certain liabilities in connection with the Offer. The Information Agent may contact holders of Shares by mail, facsimile or personal interviews and may request brokers, dealers and other nominee shareholders to forward materials relating to the Offer to beneficial owners of Shares.

The Offeror will not pay any fees or commissions to any broker or dealer or any other person for soliciting tenders of Shares pursuant to the Offer (other than to the Dealer Manager, the Information Agent and the

48

Depositary). Brokers, dealers, commercial banks and trust companies will, upon request, be reimbursed by the Offeror for customary mailing and handling expenses incurred by them in forwarding offering materials to their customers.

18. MISCELLANEOUS. The Offer is being made solely by this Offer to Purchase and the related Letter of Transmittal and is being made to holders of Shares. The Offeror is not aware of any jurisdiction where the making of the Offer is prohibited by any administrative or judicial action pursuant to any valid state statute. If the Offeror becomes aware of any valid state statute prohibiting the making of the Offer or the acceptance of Shares pursuant thereto, the Offeror will make a good faith effort to comply with any such state statute. If, after such good faith effort, the Offeror cannot comply with any such state statute, the Offer will not be made to (nor will tenders be accepted from or on behalf of) the holders of Shares in such state. In any jurisdiction where the securities, blue sky or other laws require the Offer to be made by a licensed broker or dealer, the Offer shall be deemed to be made on behalf of the Offeror by the Dealer Manager or by one or more registered brokers or dealers licensed under the laws of such jurisdiction.

Pursuant to Rule 14d-3 of the General Rules and Regulations under the Exchange Act, TCCC and Offeror have filed with the Commission the Schedule TO, together with exhibits, furnishing certain additional information with respect to the Offer. The Schedule TO and any amendments thereto, including exhibits, may be inspected at, and copies may be obtained from, the same places and in the same manner as set forth in Section 7 ("Certain Information Concerning Odwalla") (except that they will not be available at the regional offices of the

NO PERSON HAS BEEN AUTHORIZED TO GIVE ANY INFORMATION OR TO MAKE ANY REPRESENTATION ON BEHALF OF THE OFFEROR OR TCCC NOT CONTAINED HEREIN OR IN THE LETTER OF TRANSMITTAL, AND IF GIVEN OR MADE, SUCH INFORMATION OR REPRESENTATION MUST NOT BE RELIED UPON AS HAVING BEEN AUTHORIZED.

November 6, 2001

49

ANNEX T

# CERTAIN INFORMATION CONCERNING THE DIRECTORS AND EXECUTIVE OFFICERS OF THE OFFEROR AND TCCC

1. DIRECTORS AND EXECUTIVE OFFICERS OF THE OFFEROR. Set forth below are the name, current business address, citizenship, present principal occupation or employment history (covering a period of not less than five years) of each executive officer and director of Offeror. The business address of each such director and executive officer is: c/o The Coca-Cola Company, One Coca-Cola Plaza, Atlanta, Georgia 30313. All directors and executive officers listed below are citizens of the United States.

DIRECTORS OF OFFEROR

<Table> <Caption>

PRESENT PRINCIPAL OCCUPATION OR EMPLOYMENT AND MATERIAL POSITIONS HELD DURING PAST FIVE YEARS \_\_\_\_\_\_ <S> <C> Donald W. Short..... Vice President of TCCC; President and Chief Executive Officer of The Minute Maid Company ("TMMC"), a division of TCCC; Chief Executive Officer of Global Health & Wellness Business of TCCC from January 2001 until October 2001; President of the Africa and Middle East Group from January 2000 until January 2001; President of the India Division of TCCC from January 1997 until December 1999; Senior Vice President, Retail Marketing for Coca-Cola (Japan) Co. Ltd., a subsidiary of TCCC, from 1993 until 1997. Sole Director of the Board of Directors of the Offeror.

</Table>

EXECUTIVE OFFICERS OF OFFEROR

<Table> <Caption> PRESENT PRINCIPAL OCCUPATION OR EMPLOYMENT AND MATERIAL POSITIONS HELD DURING PAST FIVE YEARS \_\_\_\_\_\_ -----<C> Assistant Vice President of TCCC and Director of Mergers and Paul Etchells..... Acquisitions; Group Mergers & Acquisitions Manager for Asia for TCCC from 1998 to September 2001; Managing Director of Swire Beverages, a key bottler with operations in China, Hong Kong, Taiwan, and the United States, from 1996 until he joined The Coca-Cola Company in 1998. President of the Offeror. Mr. Etchells is a citizen of the United Kingdom. Shawn Sugarman..... Vice President and General Manager, Bottler Juice Brands Team and Business Development--TMMC from March 2001 to present; Vice President, Business Strategy--TMMC from January 2001 until February 2001; Vice President & Managing Director, Single Serve Business Team--TMMC from August 2000 until December 2000; President and CEO--TMMC from July 1998 until July 2000; Vice President and Controller--TMMC from May 1996 until June 1998. Chief Financial Officer of the Offeror. Vice President and General Counsel of TMMC. Steven Hauser..... Secretary of the Offeror. </Table> I-1

<Table> <Caption>

</Table>

2. DIRECTORS AND EXECUTIVE OFFICERS OF TCCC. Set forth below are the name, current business address, citizenship, present principal occupation or employment history (covering a period of not less than five years) of each executive officer and director of TCCC. Unless otherwise indicated, the business address of each such director and executive officer is: c/o The Coca-Cola Company, One Coca-Cola Plaza, Atlanta, Georgia 30313. Except as set forth below, all directors and executive officers listed below are citizens of the United States.

TCCC from December 5, 1999 until

DIRECTORS OF TCCC

<Table> <Caption>

NAME	POSITIONS HELD DURING PAST FIVE YEARS	ADDRESS
3	CC> Chairman of the Board of Directors and Chief Executive Officer of TCCC; President and Chief Operating Officer of	<c></c>

PRINCIPAL OCCUPATION OR EMPLOYMENT AND MATERIAL

February 17, 2000; Senior Vice President

of TCCC from 1991 until December 5, 1999. Mr. Daft is a citizen of Australia. Herbert A. Allen..... President and Chief Executive Officer of Allen & Company Allen & Company Incorporated, a privately held investment banking firm, for more than the past five years;

Director of Convera Corporation. Ronald W. Allen..... Consultant to and advisory director of Delta Air Lines, Inc., a major U.S. air transportation company; Chairman of the Board of Directors and Chief Executive Officer of Delta Air Lines, Inc. from 1987 to July 1997.

Cathleen P. Black..... President of Hearst Magazines, a unit of Hearst Magazines The Hearst Corporation, a major media and communications company; Director of International Business Machines Corporation and Women.com Networks, Inc.

Warren E. Buffett..... Chairman of the Board of Directors and Chief Executive Officer of Berkshire Hathaway Inc., a diversified holding company; Director of The Gillette Company and Director of The Washington Post Company.

Incorporated 711 Fifth Avenue New York, NY 10022

Monarch Tower Suite 1745 3424 Peachtree Road, N.E. Atlanta, GA 30326

959 8th Avenue New York, NY 10019

Berkshire Hathaway Inc. 1440 Kiewit Plaza Omaha, NE 68131

</Table>

T-2

<Table> <Caption>

# PRINCIPAL OCCUPATION

	NAME	OR EMPLOYMENT AND MATERIAL POSITIONS HELD DURING PAST FIVE YEARS	ADDRESS
<s></s>		<c></c>	<c></c>
Susan B.	King	President of The Leadership Initiative, a support corporation of Duke University charged with the establishment of undergraduate college leadership programs, since September 1999; From January 1995 to September 1999, served as Leader in Residence, Hart Leadership Program, Stanford Institute of Public Policy, Duke University, a program for the development and advancement of leadership and management skills in the public and private sectors; Director of Guidant Corporation.	Box 90545
Donald F	. McHenry	±	1320 19th Street, N.W. Suite 410

consulting firm; Director of AT&T Corporation, FleetBoston Financial Corporation, International Paper Company

Sam Nunn	Spalding since the United Stat through 1996; D Health Systems, Corporation, Ge Internet Securi National Service	line plc. law firm of King & January 1997; A member of es Senate from 1972 irector of Community Inc., Dell Computer neral Electric Company, ty Systems, Inc., e Industries, Inc., nta, Inc., Texaco, Inc.	King & Spalding 191 Peachtree Street Atlanta, GA 30303-1763
Paul F. Oreffice	and Total Syste Retired in 1992 of Directors an of The Dow Chem diversified che	m Services, Inc. as Chairman of the Board d Chief Executive Officer ical Company, a mical, metals, plastics	<del>_</del>
James D. Robinson III	Officer of RRE information tec investment firm Ventures, L.P.; Byorum & Partne private firm sp advisory and in activities in L Bristol-Myers S Technology Part Inc., First Dat	irman and Chief Executive Investors, LLC, a private hnology venture ; General Partner of RRE Chairman of Violy, rs Holdings, LLC, a ecializing in financial vestment banking atin America; Director of quibb Company, Cambridge ners (Massachusetts),	22nd Floor 126 East 56th Street

 1 |  |  ||  |  | I-3 |  |
(Oup cions	PRINC	IPAL OCCUPATION		
NAME		YMENT AND MATERIAL DURING PAST FIVE YEARS	ADDRESS	
Peter V. Ueberroth		airman of The Contrarian	The Contrarian Group, Inc.	
	\_	business management an of Ambassadors	Suite 111 1071 Camelback Street	
	International,	Inc.; Director of Hilton ion and Co-Chairman of	Newport Beach, CA 92660	
James B. Williams	Director and Chairman of the Executive Committee of SunTrust Banks, Inc., a		SunTrust Banks, Inc. P.O. Box 4418 Atlanta, GA 30302	
EXECUTIVE OFFICERS OF T	EXECUTIVE OFFICERS OF THE COCA-COLA COMPANY			
		DDTNG====	DARTON OD EMPLOMENT AND	
NAME			PATION OR EMPLOYMENT AND HELD DURING PAST FIVE YEARS	
~~Rrian G Dyson~~		``` Vice Chairman and Chief Or ```	nerating Officer. Joined TCCC in	
Vice Chairman and Chief Operating Officer; Joined TCCC in 1959 and worked for many years in South America, the Caribbean and Mexico; President of Coca-Cola USA from 1978 until 1983; President of Coca-Cola North America from 1983 until 1986; President and Chief Executive Officer of Coca-Cola Enterprises, Inc. from 1988 until 1994				
Mr. Dyson is a citizen of Argentina.  Executive Vice President and President and Chief Operatin Officer, Coca-Cola Ventures; In 1996, became President an Chief Operating Officer of Turner Broadcasting System, In In 1994, joined TBS, Inc. as President of Turner Broadcasting Sales, Inc.		and President and Chief Operating es; In 1996, became President and f Turner Broadcasting System, Inc.; as President of Turner		
Mary Minnick Executive Vice President and President and Chief Operating Officer, Coca-Cola Asia Group; From 1996-1997, Vice President and Director, Middle and Far East Marketing President, South Pacific Division from 1997-2000; President Coca-Cola (Japan) Co., Ltd. 2000-2001; Ms. Minnick began has career at TCCC in 1983, and spent ten years working in Coca-Cola USA, both in Fountain Sales and in the Bottle/Car Division.				
Alexander R.C. Allan..... Executive Vice President and President and Chief Operating Officer, Europe, Eurasia and Middle East Group; Mr. Allan, a native of Johannesburg, joined the Coca-Cola system in 1968; Formerly President of the Middle East Division (renamed Middle East & North Africa Division in 1998); In October 1999, appointed President of the Middle & Far East Group. </Table> I-4<Table> <Caption> PRINCIPAL OCCUPATION OR EMPLOYMENT AND MATERIAL POSITIONS HELD DURING PAST FIVE YEARS NAME. \_\_\_\_\_\_ <C> James E. Chestnut..... Executive Vice President, Operations Support; Joined TCCC in 1972 as financial analyst for The Coca-Cola Export Corporation in London; In 1989, became Chief Financial Officer of Coca-Cola (Japan) Company Limited; Vice President and Controller of TCCC from 1993 until 1994; Elected Chief Financial Officer of TCCC in 1994; Director of Coca-Cola FEMSA and Coca-Cola Enterprises, Inc. Mr. Chestnut is a citizen of the United Kingdom. Carl Ware..... Executive Vice President, Public Affairs; Appointed Vice President, Special Markets, Coca-Cola USA in 1979; Appointed Vice President, Urban Affairs of TCCC in 1982; Elected Senior Vice President and Director, Corporate External Affairs in 1986 and became Deputy Group President of Northeast Europe/Africa Group of International Business Sector in 1991; Appointed President of the Africa Group in 1993. Deval L. Patrick..... Executive Vice President and General Counsel; In 1999, joined Texaco, Inc. as Vice President and General Counsel; Partner with the law firm of Day Berry & Howard LLP since 1997; Assistant Attorney General of the United States and Chief of the U.S. Justice Department's Civil Rights Division from 1994 until 1997, where he was responsible for enforcing federal laws prohibiting discrimination. Senior Vice President and Chief Financial Officer; In July Gary P. Fayard..... 1994, elected Vice President and Controller of TCCC; Partner with Ernst & Young prior to April 1994. Senior Vice President and Chief Marketing Officer; Joined Stephen C. Jones..... Coca-Cola Canada in 1986 as Brand Manager for Sprite; In

Coca-Cola Canada in 1986 as Brand Manager for Sprite; In 1998, joined Coca-Cola USA as Brand Manager for diet Coke and Sprite; In 1994, appointed Senior Vice President, Consumer Marketing for Coca-Cola (Japan) Co. Ltd. ("CCJC"), and was named Deputy Division Manager and Executive Vice President of CCJC in 1997; Appointed President and Chief Executive Officer of The Minute Maid Company in October 1999.

</Table>

I-5

ANNEX II

#### SCHEDULE B

#### CHAPTER 13 OF THE CALIFORNIA GENERAL CORPORATION LAW

#### DISSENTERS' RIGHTS

SEC.1300. REORGANIZATION OR SHORT-FORM MERGER; DISSENTING SHARES; CORPORATE PURCHASE AT FAIR MARKET VALUE; DEFINITIONS

- (a) If the approval of the outstanding shares (Section 152) of a corporation is required for a reorganization under subdivisions (a) and (b) or subdivision (e) or (f) of Section 1201, each shareholder of the corporation entitled to vote on the transaction and each shareholder of a subsidiary corporation in a short-form merger may, by complying with this chapter, require the corporation in which the shareholder holds shares to purchase for cash at their fair market value the shares owned by the shareholder which are dissenting shares as defined in subdivision (b). The fair market value shall be determined as of the day before the first announcement of the terms of the proposed reorganization or short-form merger, excluding any appreciation or depreciation in consequence of the proposed action, but adjusted for any stock split, reverse stock split, or share dividend which becomes effective thereafter.
- (b) As used in this chapter, "dissenting shares" means shares which come within all of the following descriptions:
- (1) Which were not immediately prior to the reorganization or short-form merger either

- (A) listed on any national securities exchange certified by the Commissioner of Corporations under subdivision (o) of Section 25100 or
- (B) listed on the National Market System of the Nasdaq Stock Market, and the notice of meeting of shareholders to act upon the reorganization summarizes this section and Sections 1301, 1302, 1303 and 1304; provided, however, that this provision does not apply to any shares with respect to which there exists any restriction on transfer imposed by the corporation or by any law or regulation; and provided, further, that this provision does not apply to any class of shares described in subparagraph (A) or (B) if demands for payment are filed with respect to 5 percent or more of the outstanding shares of that class.
- (2) Which were outstanding on the date for the determination of shareholders entitled to vote on the reorganization and (A) were not voted in favor of the reorganization or, (B) if described in subparagraph (A) or (B) of paragraph (1) (without regard to the provisos in that paragraph), were voted against the reorganization, or which were held of record on the effective date of a short-form merger; provided, however, that subparagraph (A) rather than subparagraph (B) of this paragraph applies in any case where the approval required by Section 1201 is sought by written consent rather than at a meeting.
- (3) Which the dissenting shareholder has demanded that the corporation purchase at their fair market value, in accordance with Section 1301.
- (4) Which the dissenting shareholder has submitted for endorsement, in accordance with Section 1302.
- (c) As used in this chapter, "dissenting shareholder" means the recordholder of dissenting shares and includes a transferee of record.
- SEC.1301. NOTICE TO HOLDERS OF DISSENTING SHARES IN REORGANIZATIONS; DEMAND FOR PURCHASE; TIME; CONTENTS
- (a) If, in the case of a reorganization, any shareholders of a corporation have a right under Section 1300, subject to compliance with paragraphs (3) and (4) of subdivision (b) thereof, to require the corporation to purchase their shares for cash, such corporation shall mail to each such shareholder a notice of

#### T T – 1

the approval of the reorganization by its outstanding shares (Section 152) within 10 days after the date of such approval, accompanied by a copy of Sections 1300, 1302, 1303, 1304 and this section, a statement of the price determined by the corporation to represent the fair market value of the dissenting shares, and a brief description of the procedure to be followed if the shareholder desires to exercise the shareholder's right under such sections. The statement of price constitutes an offer by the corporation to purchase at the price stated any dissenting shares as defined in subdivision (b) of Section 1300, unless they lose their status as dissenting shares under Section 1309.

- (b) Any shareholder who has a right to require the corporation to purchase the shareholder's shares for cash under Section 1300, subject to compliance with paragraphs (3) and (4) of subdivision (b) thereof, and who desires the corporation to purchase such shares shall make written demand upon the corporation for the purchase of such shares and payment to the shareholder in cash of their fair market value. The demand is not effective for any purpose unless it is received by the corporation or any transfer agent thereof (1) in the case of shares described in clause (i) or (ii) of paragraph (1) of subdivision (b) of Section 1300 (without regard to the provisos in that paragraph), not later than the date of the shareholders' meeting to vote upon the reorganization, or (2) in any other case within 30 days after the date on which the notice of the approval by the outstanding shares pursuant to subdivision (a) or the notice pursuant to subdivision (i) of Section 1110 was mailed to the shareholder.
- (c) The demand shall state the number and class of the shares held of record by the shareholder which the shareholder demands that the corporation purchase and shall contain a statement of what such shareholder claims to be the fair market value of those shares as of the day before the announcement of the proposed reorganization or short-form merger. The statement of fair market value constitutes an offer by the shareholder to sell the shares at such price.
- SEC. 1302. SUBMISSION OF SHARE CERTIFICATES FOR ENDORSEMENT; UNCERTIFICATED SECURITIES

Within 30 days after the date on which notice of the approval by the outstanding shares or the notice pursuant to subdivision (i) of Section 1110 was mailed to the shareholder, the shareholder shall submit to the corporation at its principal office or at the office of any transfer agent thereof, (a) if the

shares are certificated securities, the shareholder's certificates representing any shares which the shareholder demands that the corporation purchase, to be stamped or endorsed with a statement that the shares are dissenting shares or to be exchanged for certificates of appropriate denomination so stamped or endorsed or (b) if the shares are uncertificated securities, written notice of the number of shares which the shareholder demands that the corporation purchase. Upon subsequent transfers of the dissenting shares on the books of the corporation, the new certificates, initial transaction statement, and other written statements issued therefor shall bear a like statement, together with the name of the original dissenting holder of the shares.

- SEC. 1303. PAYMENT OF AGREED PRICE WITH INTEREST; AGREEMENT FIXING FAIR MARKET VALUE; FILING; TIME OF PAYMENT
- (a) If the corporation and the shareholder agree that the shares are dissenting shares and agree upon the price of the shares, the dissenting shareholder is entitled to the agreed price with interest thereon at the legal rate on judgments from the date of the agreement. Any agreements fixing the fair market value of any dissenting shares as between the corporation and the holders thereof shall be filed with the secretary of the corporation.
- (b) Subject to the provisions of Section 1306, payment of the fair market value of dissenting shares shall be made within 30 days after the amount thereof has been agreed or within 30 days after any statutory or contractual conditions to the reorganization are satisfied, whichever is later, and in the case of certificated securities, subject to surrender of the certificates therefor, unless provided otherwise by agreement.

II-2

- SEC. 1304. ACTION TO DETERMINE WHETHER SHARES ARE DISSENTING SHARES OR FAIR MARKET VALUE; LIMITATION; JOINDER; CONSOLIDATION; DETERMINATION OF ISSUES; APPOINTMENT OF APPRAISERS
- (a) If the corporation denies that the shares are dissenting shares, or the corporation and the shareholder fail to agree upon the fair market value of the shares, then the shareholder demanding purchase of such shares as dissenting shares or any interested corporation, within six months after the date on which notice of the approval by the outstanding shares (Section 152) or notice pursuant to subdivision (i) of Section 1110 was mailed to the shareholder, but not thereafter, may file a complaint in the superior court of the proper county praying the court to determine whether the shares are dissenting shares or the fair market value of the dissenting shares or both or may intervene in any action pending on such a complaint.
- (b) Two or more dissenting shareholders may join as plaintiffs or be joined as defendants in any such action and two or more such actions may be consolidated.
- (c) On the trial of the action, the court shall determine the issues. If the status of the shares as dissenting shares is in issue, the court shall first determine that issue. If the fair market value of the dissenting shares is in issue, the court shall determine, or shall appoint one or more impartial appraisers to determine, the fair market value of the shares.
- SEC. 1305. REPORT OF APPRAISERS; CONFIRMATION; DETERMINATION BY COURT; JUDGMENT; PAYMENT; APPEAL; COSTS
- (a) If the court appoints an appraiser or appraisers, they shall proceed forthwith to determine the fair market value per share. Within the time fixed by the court, the appraisers, or a majority of them, shall make and file a report in the office of the clerk of the court. Thereupon, on the motion of any party, the report shall be submitted to the court and considered on such evidence as the court considers relevant. If the court finds the report reasonable, the court may confirm it.
- (b) If a majority of the appraisers appointed fail to make and file a report within 10 days from the date of their appointment or within such further time as may be allowed by the court or the report is not confirmed by the court, the court shall determine the fair market value of the dissenting shares.
- (c) Subject to the provisions of Section 1306, judgment shall be rendered against the corporation for payment of an amount equal to the fair market value of each dissenting share multiplied by the number of dissenting shares which any dissenting shareholder who is a party, or who has intervened, is entitled to require the corporation to purchase, with interest thereon at the legal rate from the date on which judgment was entered.
- (d) Any such judgment shall be payable forthwith with respect to uncertificated securities and, with respect to certificated securities, only upon the endorsement and delivery to the corporation of the certificates for the shares described in the judgment. Any party may appeal from the judgment.
- (e) The costs of the action, including reasonable compensation to the appraisers to be fixed by the court, shall be assessed or apportioned as the

court considers equitable, but, if the appraisal exceeds the price offered by the corporation, the corporation shall pay the costs (including in the discretion of the court attorneys' fees, fees of expert witnesses and interest at the legal rate on judgments from the date of compliance with Sections 1300, 1301 and 1302 if the value awarded by the court for the shares is more than 125 percent of the price offered by the corporation under subdivision (a) of Section 1301).

SEC. 1306. PREVENTION OF IMMEDIATE PAYMENT; STATUS AS CREDITORS; INTEREST

To the extent that the provisions of Chapter 5 prevent the payment to any holders of dissenting shares of their fair market value, they shall become creditors of the corporation for the amount thereof together with interest at the legal rate on judgments until the date of payment, but subordinate to all other creditors in any liquidation proceeding, such debt to be payable when permissible under the provisions of Chapter 5.

TT-1

#### SEC. 1307. DIVIDENDS ON DISSENTING SHARES

Cash dividends declared and paid by the corporation upon the dissenting shares after the date of approval of the reorganization by the outstanding shares (Section 152) and prior to payment for the shares by the corporation shall be credited against the total amount to be paid by the corporation therefor.

SEC. 1308. RIGHTS OF DISSENTING SHAREHOLDERS PENDING VALUATION; WITHDRAWAL OF

Except as expressly limited in this chapter, holders of dissenting shares continue to have all the rights and privileges incident to their shares, until the fair market value of their shares is agreed upon or determined. A dissenting shareholder may not withdraw a demand for payment unless the corporation consents thereto.

SEC. 1309. TERMINATION OF DISSENTING SHARE AND SHAREHOLDER STATUS

Dissenting shares lose their status as dissenting shares and the holders thereof cease to be dissenting shareholders and cease to be entitled to require the corporation to purchase their shares upon the happening of any of the following:

- (a) The corporation abandons the reorganization. Upon abandonment of the reorganization, the corporation shall pay on demand to any dissenting shareholder who has initiated proceedings in good faith under this chapter all necessary expenses incurred in such proceedings and reasonable attorneys' fees.
- (b) The shares are transferred prior to their submission for endorsement in accordance with Section 1302 or are surrendered for conversion into shares of another class in accordance with the articles.
- (c) The dissenting shareholder and the corporation do not agree upon the status of the shares as dissenting shares or upon the purchase price of the shares, and neither files a complaint or intervenes in a pending action as provided in Section 1304, within six months after the date on which notice of the approval by the outstanding shares or notice pursuant to subdivision (i) of Section 1110 was mailed to the shareholder.
- (d) The dissenting shareholder, with the consent of the corporation, withdraws the shareholder's demand for purchase of the dissenting shares.
- SEC. 1310. SUSPENSION OF RIGHT TO COMPENSATION OR VALUATION PROCEEDINGS; LITIGATION OF SHAREHOLDERS' APPROVAL

If litigation is instituted to test the sufficiency or regularity of the votes of the shareholders in authorizing a reorganization, any proceedings under Sections 1304 and 1305 shall be suspended until final determination of such litigation.

SEC. 1311. EXEMPT SHARES

This chapter, except Section 1312, does not apply to classes of shares whose terms and provisions specifically set forth the amount to be paid in respect to such shares in the event of a reorganization or merger.

- SEC. 1312. RIGHT OF DISSENTING SHAREHOLDER TO ATTACK, SET ASIDE OR RESCIND MERGER OR REORGANIZATION; RESTRAINING ORDER OR INJUNCTION; CONDITIONS
- (a) No shareholder of a corporation who has a right under this chapter to demand payment of cash for the shares held by the shareholder shall have any right at law or in equity to attack the validity of the reorganization or short-form merger, or to have the reorganization or short-form merger set aside or rescinded, except in an action to test whether the number of shares required

to authorize or approve the reorganization have been legally voted in favor thereof; but any holder of shares of a class whose terms and provisions specifically set forth the amount to be paid in respect to them in the event of a reorganization or short-form merger is entitled to payment in accordance with those terms and provisions or, if the principal terms of the

T T - 4

reorganization are approved pursuant to subdivision (b) of Section 1202, is entitled to payment in accordance with the terms and provisions of the approved reorganization.

- (b) If one of the parties to a reorganization or short-form merger is directly or indirectly controlled by, or under common control with, another party to the reorganization or short-form merger, subdivision (a) shall not apply to any shareholder of such party who has not demanded payment of cash for such shareholder's shares pursuant to this chapter; but if the shareholder institutes any action to attack the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded, the shareholder shall not thereafter have any right to demand payment of cash for the shareholder's shares pursuant to this chapter. The court in any action attacking the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded shall not restrain or enjoin the consummation of the transaction except upon 10 days' prior notice to the corporation and upon a determination by the court that clearly no other remedy will adequately protect the complaining shareholder or the class of shareholders of which such shareholder is a member.
- (c) If one of the parties to a reorganization or short-form merger is directly or indirectly controlled by, or under common control with, another party to the reorganization or short-form merger, in any action to attack the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded, (1) a party to a reorganization or short-form merger which controls another party to the reorganization or short-form merger shall have the burden of proving that the transaction is just and reasonable as to the shareholders of the controlled party, and (2) a person who controls two or more parties to a reorganization shall have the burden of proving that the transaction is just and reasonable as to the shareholders of any party so controlled.

II-5

APPENDIX A

PAGE

AGREEMENT AND PLAN OF MERGER

AMONG

THE COCA-COLA COMPANY,

PERRY PHILLIP CORP.

AND

ODWALLA, INC.

DATED OCTOBER 29, 2001

AGREEMENT AND PLAN OF MERGER

TABLE OF CONTENTS

# <Table> <Caption>

<s></s>	<c></c>	<c></c>
ARTICLE I	THE OFFER	A-1
Section 1.1.	The Offer	A-1
Section 1.2.	Company Action	A-3
Section 1.3.	SEC Actions	A-3
Section 1.4.	Directors	A-4
ARTICLE II	THE MERGER	A-5
Section 2.1.	The Merger	A-5
Section 2.2.	Effective Time; Closing	A-5
Section 2.3.	Effect of the Merger	A-6
Section 2.4.	Conversion of Company Common Stock	A-6
Section 2.5.	Dissenting Shares	A-6
Section 2.6.	Stock Option Plans	A-7
Section 2.7.	Company Warrants	A-7
Section 2.8.	Surrender of Shares of Company Common Stock; Stock Transfer	
	Books	A-8
Section 2.9.	Shareholders Meeting	A-10
Section 2.10.	Subsequent Actions	A-11

ARTICLE III Section 3.1. Section 3.2.	THE SURVIVING CORPORATION.  Articles of Incorporation.  Bylaws.	A-11 A-11 A-11
Section 3.3.	Directors and Officers	A-11
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	A-11
Section 4.1.	Organization and Standing	A-11
Section 4.2.	Capitalization	A-12
Section 4.3.	Authority for Agreement	A-12
Section 4.4.	No Conflict	A-13
Section 4.5.	Required Filings and Consents	A-14
Section 4.6.	Compliance	A-14
Section 4.7.	SEC Filings, Financial Statements	A-14
Section 4.8. Section 4.9.	Absence of Certain Changes or Events	A-15 A-15
Section 4.9.	TaxesAssets.	A-15 A-16
Section 4.10.	Change of Control Agreements	A-16 A-16
Section 4.11.	Litigation	A-16
Section 4.13.	Contracts and Commitments	A-17
Section 4.14.	Employee Benefit Plans	A-17
Section 4.15.	Labor and Employment Matters	A-18
Section 4.16.	Environmental Compliance and Disclosure	A-20
Section 4.17.	Intellectual Property Rights	A-21
Section 4.18.	Products	A-22
Section 4.19.	Brokers	A-22
Section 4.20.	Trade Secrets	A-22
Section 4.21.	Insurance Policies	A-22
Section 4.22.	Notes and Accounts Receivable	A-23

			i	
	1			
		PAGE		
Section 4.23.	Transactions with Affiliates	A-23		
Section 4.24.	No Existing Discussions	A-23		
Section 4.25. Section 4.26.	Shareholder Rights Agreement	A-23 A-23		
Section 4.20.	major suppriers, customers and bistributors	A 23		
ARTICLE V	REPRESENTATIONS AND WARRANTIES OF PARENT AND SUB	A-24		
Section 5.1.	Organization and Standing	A-24		
Section 5.2.	Sub	A-24		
Section 5.3.	Authority for Agreement	A-24		
Section 5.4.	No Conflict	A-24		
Section 5.5.	Required Filings and Consents	A-25		
Section 5.6.	Brokers	A-25		
Section 5.7.	Financing	A-25		
ARTICLE VI	COVENANTS	A-25		
Section 6.1.	Conduct of the Business Pending the Merger	A-25		
Section 6.2.	Access to Information; Confidentiality	A-27		
Section 6.3.	Notification of Certain Matters	A-27		
Section 6.4.	Reasonable Efforts; Further Assurances; Cooperation	A-27		
Section 6.5.	Board Recommendations	A-28		
Section 6.6.	Shareholder Litigation	A-29		
Section 6.7.	Indemnification	A-29		
Section 6.8.	Public Announcements	A-30		
Section 6.9.	Competing Acquisition Proposals	A-30		
Section 6.10.	Undertakings of Parent	A-31		
Section 6.11.	Director Resignations	A-31		
Section 6.12.	Shareholders' Rights Agreement	A-31		
Section 6.13.	Employee Benefits	A-31		
Section 6.14. Section 6.15.	Environmental Assessments	A-31 A-32		
Section 6.15. Section 6.16.	Conveyance Taxes	A-32 A-32		
Section 6.17.	Shareholder Rights Plan	A-32 A-32		
Section 6.18.	Actions by Company Board	A-32		
Section 6.19.	Loans to Permit Exercise of Company Options	A-32		
ARTICLE VII Section 7.1.	CONDITIONS	A-33 A-33		
ARTICLE VIII	TERMINATION, AMENDMENT AND WAIVER	A-33		
Section 8.1.	Termination	A-33		
Section 8.2.	Effect of Termination	A-34		
Section 8.3.	Amendments	A-35		
Section 8.4.	Waiver	A-35		
ARTICLE IX	GENERAL PROVISIONS	A-35		
Section 9.1.	No Third Party Beneficiaries	A-35		
Section 9.2.	Entire Agreement	A-35		

Section 9.3.	Succession and Assignment	A-35
Section 9.4.	Counterparts	A-35
Section 9.5.	Headings	A-35

  |  |ii

<Table> <Caption>

		PAGE
<s></s>	<c></c>	<c></c>
Section 9.6.	Governing Law	A-35
Section 9.7.	Severability	A-35
Section 9.8.	Specific Performance	A-36
Section 9.9.	Construction	A-36
Section 9.10.	Non-Survival of Representations and Warranties and	
	Agreements	A-36
Section 9.11.	Certain Definitions	A-36
Section 9.12.	Fees and Expenses	A-37
Section 9.13.	Notices	A-37
Section 9.14.	Waiver of Jury Trial	A-38
ANNEX I	CONDITIONS TO THE OFFER	I-1
EXHIBIT A 		

  | A-1 |iii

#### AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") dated October 29, 2001 by and among The Coca-Cola Company, a Delaware corporation ("Parent"), Perry Phillip Corp., a California corporation ("Sub") and wholly owned subsidiary of Parent, and Odwalla, Inc., a California corporation (the "Company").

#### WITNESSETH:

WHEREAS, the parties to this Agreement desire to effect the acquisition of the Company by Parent pursuant to a tender offer (the "Offer") by Sub for any and all the issued and outstanding shares (the "Shares") of Common Stock, no par value, of the Company ("Company Common Stock") followed by a merger (the "Merger") of Sub with and into the Company with the Company as the surviving corporation in accordance with the California General Corporation Law (the "CGCL");

WHEREAS, as an inducement to Parent and Sub to enter into this Agreement, Parent, Sub and the Company have entered into a Stock Option Agreement (the "Option Agreement"), pursuant to which the Company has granted to Sub an option to purchase newly issued shares of Company Common Stock under certain circumstances;

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent entering into this Agreement, each of the persons and entities set forth on Exhibit A hereto (collectively, the "Principal Shareholders") has entered into a tender and/or voting agreement, dated as of the date hereof (collectively, the "Tender Agreements"), pursuant to which, among other things, each of the Principal Shareholders has agreed to tender its shares of Company Common Stock to Sub in the Offer;

WHEREAS, the Board of Directors of Parent and Sub have each unanimously approved this Agreement, the Option Agreement, the Offer, the Merger and the transactions contemplated hereby upon the terms and subject to the conditions set forth herein;

WHEREAS, the Board of Directors of the Company has unanimously determined that this Agreement, the Option Agreement, the Offer, the Merger and the transactions contemplated hereby, subject to the terms and conditions of this Agreement, are fair to and in the best interests of the holders of the Company Common Stock (the "Company Shareholders"); and

WHEREAS, the Board of Directors of the Company has unanimously resolved to recommend that the Company Shareholders approve this Agreement, the Option Agreement, the Offer, the Merger and the transactions contemplated hereby to which the Company is a party.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements contained in this Agreement and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE I

(a) Subject to the provisions of this Agreement, Parent shall cause Sub, no later than November 8, 2001, to commence (within the meaning of Rule 14d-2 under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) the Offer, at a price of \$15.25 per share, net to the seller in cash, without interest (or at such higher price as Sub in its sole discretion elects to offer) (the "Offer Price"), but subject to any withholding required by law. The obligation of Parent and Sub to accept and pay for Shares tendered shall be subject only to the condition that there shall be validly tendered prior to the expiration date of the Offer and not withdrawn a number of Shares which, when added to the shares of Company Common Stock owned by Parent, represent at least 90.1% of the Shares issued and outstanding on a fully diluted basis (including for

A-1

purposes of such calculation all shares of Company Common Stock issuable upon exercise of all Company Options (as defined in Section 2.6) that vest (or upon consummation of the Offer will vest) (but excluding shares of Company Common Stock that are issuable upon the exercise of Company Options that are cancelled pursuant to Section 2.6(a) hereof or upon the exercise of Company Warrants that are cancelled pursuant to Section 2.7 hereof) and the exercise of all outstanding Company Warrants (as defined in Section 2.7)) (the "Minimum Condition") and to the other conditions set forth in Annex I (collectively, the "Offer Conditions"). Parent and Sub expressly reserve the right to waive any of the Offer Conditions (except that Parent and Sub may not waive the Minimum Condition except with the consent of the Company or as and to the extent provided in this Agreement), to increase the price per share payable in the Offer and to make any other change or changes in the terms or conditions of the Offer, including without limitation extending the expiration date, except that, without the consent of the Company, Parent and Sub shall not (i) reduce the number of Shares subject to the Offer, (ii) reduce the Offer Price, (iii) impose any other conditions to the Offer other than the Offer Conditions or modify the Offer Conditions (other than to waive any Offer Conditions to the extent permitted by this Agreement) in a manner adverse to the holders of the Company Common Stock, (iv) except as provided in Section 1.1(b), extend the Offer, (v) change the form of consideration payable in the Offer, or (vi) amend any other term of the Offer in a manner adverse to the Company Shareholders.

- (b) Subject to the terms and conditions hereof, the Offer shall expire at midnight, San Francisco time, on the date twenty business days (as defined under the Exchange Act) after the date the Offer is commenced, provided, that Sub may, without the consent of the Company, (i) extend the Offer, if at the scheduled or extended expiration date of the Offer any of the Offer Conditions shall not be satisfied or waived, until such time as such conditions are satisfied or waived, subject in each case to any right of Parent, Sub or the Company to terminate this Agreement pursuant to the terms hereof or (ii) extend the Offer for any period required by any rule, regulation, interpretation or position of the Securities and Exchange Commission (the "SEC") or the staff thereof applicable to the Offer. If, at any scheduled expiration date of the Offer, the conditions set forth in paragraph 5(g) of Annex I are satisfied, but any of the conditions set forth in paragraphs 2 or 5(a), (b), (c), (d), (e) or (f) of Annex I are not satisfied, then at the request of the Company, Sub shall extend the Offer from time to time, subject to any right of Parent, Sub or the Company to terminate this Agreement pursuant to the terms hereof.
- (c) Notwithstanding any other provision contained herein, in the event the Minimum Condition is not satisfied on any scheduled expiration date of the Offer, at the request of the Company, subject to any right of Parent or Sub to terminate this Agreement pursuant to the terms hereof, Sub shall, and Parent shall cause Sub to, take any of the actions set forth in the following clauses (x), (y) or (z) (provided that Sub shall have complete discretion as to which actions among the three clauses to take) as follows: (x) extend the Offer pursuant to Section 1.1(b), (y) amend the Offer in contemplation of the exercise of the Option Agreement (to the extent the Option Agreement is exercisable at such time) to reduce the Minimum Condition to that number of shares (the "Option Exercise Minimum Number") equal to the number of shares which, when combined with the number of shares issued upon exercise of the Option Agreement, equals 90.1% of the shares of Company Common Stock on a fully diluted basis (but excluding shares of Company Common Stock that are issuable upon the exercise of Company Options that are cancelled pursuant to Section 2.6(a) hereof or upon the exercise of Company Warrants that are cancelled pursuant to Section 2.7 hereof) or (z) amend the Offer to provide that, in the event (i) the Minimum Condition is not satisfied at the next scheduled expiration date of the Offer (after giving effect to the issuance of any shares of Company Common Stock theretofore acquired by Parent or Sub) and (ii) the number of shares of Company Common Stock tendered pursuant to the Offer and not withdrawn as of such next scheduled expiration date is more than 50% of the then outstanding shares of Company Common Stock, Sub shall waive the Minimum Condition and amend the Offer to reduce the number of shares of Company Common Stock subject to the Offer to 49.9% of the shares of Company Common Stock then outstanding (the "Revised Minimum Number") and, subject to the prior satisfaction or waiver of the other conditions of the Offer, purchase, on a pro rata basis, the Revised Minimum Number of shares (it being understood that Sub shall not in any event be required to accept for payment, or pay for, any shares of Company Common Stock

A - 2

(d) Sub shall, on the terms and subject to the prior satisfaction or waiver of the conditions of the Offer, accept for payment shares of Company Common Stock validly tendered as soon as practicable (and in any event within five business days) after such satisfaction or waiver of all conditions of the Offer, and pay for accepted shares of Company Common Stock as promptly thereafter as reasonably practicable, and in any event in compliance with Rule 14e-1(c) under the Exchange Act. Parent shall provide, or cause to be provided to Sub on a timely basis the funds necessary to pay for any shares of Company Common Stock Sub accepts or becomes obligated to accept for payment.

Section 1.2. Company Action.

- (a) The Company hereby approves and consents to the Offer and to the inclusion in the Offer and the related documents thereto of the recommendation of the Board of Directors of the Company described in Section  $4.3\,(b)$  and the opinion of the Independent Advisor described in Section  $4.3\,(c)$ .
- (b) The Company will promptly, and in any event within three (3) business days after the execution of this Agreement, instruct its transfer agent to furnish Sub with a list of its shareholders, mailing labels and any available listing or computer file containing the names and addresses of all record holders of outstanding shares of Company Common Stock and lists of securities positions of outstanding shares of Company Common Stock held in stock depositories, and to provide to Sub such additional information (including, without limitation, updated lists of shareholders, mailing labels and lists of securities positions) and such other assistance as Parent or Sub or any of their respective agents may reasonably request in connection with the Offer and Merger. Subject to the requirements of applicable Law (as defined in Section 4.4), and except for such steps as are necessary to disseminate the Offer Documents (as defined below) and any other documents necessary to consummate the Merger, Parent and Sub shall, and shall cause their agents to, hold in confidence and otherwise in conformity with the provisions of the Confidentiality Agreement dated September 7, 2001, as amended (the "Confidentiality Agreement"), the information contained in any such labels, listings and files, use such information only in connection with the Offer and the Merger and, if this Agreement shall be terminated, upon request, promptly deliver to the Company all copies of such information then in their possession or control.

#### Section 1.3. SEC Actions.

- (a) On the date of commencement of the Offer, Parent and Sub shall file with the SEC a Tender Offer Statement on Schedule TO with respect to the Offer (including all supplements and amendments thereto, the "Schedule TO"), which shall contain an offer to purchase, a related letter of transmittal and summary advertisement (such Schedule TO and all other documents required to be filed by Parent and Sub with the SEC in connection with the Offer and the Merger, are collectively referred to as the "Offer Documents").
- (b) On or as soon as practicable after the date the Offer Documents are filed with the SEC, the Company shall file with the SEC a Solicitation/Recommendation Statement on Schedule 14D-9 (including all supplements and amendments thereto, the "Schedule 14D-9"). The Schedule 14D-9 shall (except as specifically provided in Section 6.5 hereof) contain the unanimous recommendation of the Board of Directors of the Company in favor of the Offer and the Merger.
- (c) (i) Parent and Sub will take all steps necessary to ensure that the Offer Documents, and the Company will take all steps necessary to ensure that the Schedule 14D-9 and all other documents required to be filed by the Company with the SEC in connection with the Offer and the Merger (collectively, the "Company Disclosure Documents"), comply or complies in all material respects with the provisions of applicable federal securities laws and, on the date filed with the SEC and on the date first published, sent or given to the Company's shareholders, do not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, provided, that Parent and Sub make no representation with respect to information furnished by the Company for inclusion in the Offer Documents and provided further that the Company makes no representation with respect to information furnished by Parent or Sub for inclusion in the Company Disclosure Documents.

A-3

(ii) The Company will ensure that the information with respect to the Company that the Company furnishes to Parent in writing specifically for inclusion in the Offer Documents does not, at the time of the filing of the Offer Documents, at the time of any distribution thereof and at the time of the

consummation of the Offer, contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Parent and Sub will ensure that the information with respect to Parent and Sub that (x) Parent or Sub furnishes to the Company in writing specifically for inclusion in the Company Disclosure Documents, (y) is incorporated in the Company Disclosure Documents by reference to any of the Offer Documents (other than any information set forth in any of the Offer Documents that is furnished by the Company for inclusion therein), or (z) is set forth in the Schedule TO (other than any information set forth in the Schedule TO that is furnished by the Company for inclusion therein), does not, at the time of the filing of the Offer Documents, at the time of any distribution thereof and at the time of the consummation of the Offer, contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Each of Parent and Sub, on the one hand, and the Company, on the other hand, will promptly correct any information provided by it for use in the Offer Documents and the Company Disclosure Documents, as the case may be, if and to the extent that it shall have become false and misleading in any material respect.

- (iii) Each of Parent and Sub will take all steps necessary to cause the Offer Documents, and the Company will take all steps necessary to cause the Company Disclosure Documents, in each case including all amendments thereto, to be filed with the SEC and to be disseminated to holders of the outstanding shares of Company Common Stock as and to the extent required by applicable federal securities laws.
- (iv) Each of the Company, on the one hand, and Parent and Sub on the other hand, will give the other, and their respective counsel, the opportunity to review and provide comments with respect to the Company Disclosure Documents and the Offer Documents, as the case may be, before they are filed with the SEC, in each case including all amendments thereto. In addition, each such party will provide the other such parties and their counsel with any comments, whether written or oral, which it may receive from time to time from the SEC or its staff with respect to the Company Disclosure Documents or the Offer Documents promptly after the receipt of such comments.

#### Section 1.4. Directors.

(a) Effective upon the purchase by Sub of such number of shares of Company Common Stock as shall constitute satisfaction of the Minimum Condition, the Option Exercise Minimum Number or the Revised Minimum Number, as the case may be, and subject to compliance with Section 14(f) of the Exchange Act, Sub shall be entitled, at its option, to designate the number of directors, rounded up to the next whole number, on the Company Board for the period following such purchase (the "Post-Acceptance Board") that equals the product of (i) the total number of available seats on the Post-Acceptance Board (giving effect to the election of any additional directors, the resignation of any existing directors and/or the increase in the number of members of the Company's Board of Directors pursuant to this Section 1.4) and (ii) the percentage that the number of shares of Company Common Stock owned by Sub (including shares of Company Common Stock accepted for payment) and Parent bears to the total number of shares of Company Common Stock issued and outstanding; provided that, prior to or upon the purchase by Sub of Company Common Stock in the Offer, the Company shall increase the number of directors on the Company's Board of Directors from seven to eight and shall maintain the number of directors at eight until the Effective Time. The Company shall take all action necessary to cause Sub's designees to be elected or appointed to the Post-Acceptance Board, including, without limitation, increasing the number of directors and seeking and accepting resignations of incumbent directors. The Company will use its reasonable best efforts to cause individuals designated by Sub to constitute the same percentage as such individuals represent on the Post-Acceptance Board of each committee of such Post-Acceptance Board (other than any committee of such Post-Acceptance Board established to take action under this Agreement). Notwithstanding the foregoing, the Company shall use its reasonable best efforts to ensure that prior to the Effective Time, the Post-Acceptance Board shall include at least four directors in office as of the date hereof (any such director remaining in office being a "Continuing Director"). In the event that the number of Continuing Directors shall be reduced below four for any reason whatsoever, any remaining

A-4

Continuing Directors (or Continuing Director, if there shall only be one remaining) shall be entitled to designate persons to fill such vacancies who shall be deemed to be Continuing Directors for purposes of this Agreement. If no Continuing Directors remain, the other directors shall designate four persons to fill such vacancies and such persons shall be deemed to be Continuing Directors for purposes of this Agreement. The provisions of this Section 1.4(a) are in addition to and shall not limit any rights which Parent or Sub or any of their Affiliates may have as a holder or beneficial owner of shares of Company Common Stock as a matter of Law with respect to the election of directors or otherwise.

(b) To the fullest extent permitted by applicable Law, upon any request by Sub following the purchase by Sub of such number of Shares as shall constitute

satisfaction of the Minimum Condition, the Option Exercise Minimum Number or the Revised Minimum Number, as the case may be, the Company shall promptly take all actions required in order to fulfill its obligations under this Section 1.4, including without limitation, in the case of satisfaction of the Minimum Condition, all actions required pursuant to Section 14(f) of the Exchange Act and Rule 14f-1 promulgated thereunder, which shall include without limitation filing with the SEC and transmitting to the record shareholders of the Company such information with respect to the Company and its officers and directors and Sub's designees as is necessary to enable Sub's designees to be elected to the Post-Acceptance Board. Parent or Sub will supply to the Company any information with respect to itself and such nominees, officers, directors and Affiliates required by such Section 14(f) and Rule 14f-1, and Parent and Sub jointly and severally represent to the Company that such information will not, at the time of the filing with the SEC of any document required to be filed pursuant to this Section 1.4, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order that the statements made therein, in light of the circumstances under which they were made, are not misleading.

(c) Following the election or appointment of Sub's designees pursuant to Section 1.4(a) and prior to the Effective Time, Parent and Sub shall not cause the Company to take any action with respect to any amendment, or waiver of any term or condition, of this Agreement, the Articles of Incorporation of the Company or the Bylaws of the Company, any termination or rescission of this Agreement by the Company, any extension by the Company of the time for the performance of any of the obligations or other acts of Sub or Parent or waiver or assertion of any of the Company's rights hereunder, or any other consent or action by the Board of Directors of the Company with respect to this Agreement or the Offer (including any action which would cause the Company to breach this Agreement), without the concurrence of a majority of the Continuing Directors and the concurrence of a majority of the directors who are not Continuing Directors.

#### ARTICLE II

#### THE MERGER

Section 2.1. The Merger. Upon the terms and subject to the conditions of this Agreement, and in accordance with the CGCL, at the Effective Time (as defined in Section 2.2), Sub shall be merged with and into the Company. As a result of the Merger, the separate corporate existence of Sub shall cease and the Company shall continue as the surviving corporation following the Merger (the "Surviving Corporation"). The corporate existence of the Company, with all its purposes, rights, privileges, franchises, powers and objects, shall continue unaffected and unimpaired by the Merger and, as the Surviving Corporation, it shall be governed by the Laws of the State of California.

Section 2.2. Effective Time; Closing. As promptly as practicable (and in any event within three (3) business days) after the conditions set forth in Article VII hereof are satisfied or waived (other than those conditions that by their nature are to be satisfied at the Closing), the parties hereto shall cause the Merger to be consummated by filing an agreement of merger pursuant to Section 1103 of the CGCL (the "Agreement of Merger"), with the Secretary of State of the State of California and by making all other filings or recordings required under the CGCL in connection with the Merger, in such form as is required by, and executed in accordance with the relevant provisions of, the CGCL. The Merger shall become effective at such time as the Agreement of Merger is duly filed with the Secretary of State of the State of California, or at such other time as the parties hereto agree shall be specified in the Agreement of Merger (the date and time the Merger

A-5

becomes effective, the "Effective Time"). On the date of such filing, a closing (the "Closing") shall be held at 1:00 p.m., local time, at the offices of King & Spalding, 191 Peachtree Street, Atlanta, Georgia 30303, or at such other time and location as the parties hereto shall otherwise agree.

Section 2.3. Effect of the Merger. At the Effective Time, the effect of the Merger, shall be as provided in the Agreement of Merger, the applicable provisions of the CGCL and this Agreement. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time all the property, rights, privileges, powers and franchises of the Company and Sub shall vest in the Surviving Corporation, and all debts, liabilities, obligations, restrictions, disabilities and duties of the Company and Sub shall become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving Corporation.

# Section 2.4. Conversion of Company Common Stock.

- (a) At the Effective Time, by virtue of the Merger and without any action on the part of Sub, the Company or the holders of any of the following securities:
  - (i) Each share of Company Common Stock issued and outstanding

immediately prior to the Effective Time (other than shares canceled pursuant to Section 2.4(a)(ii) and Dissenting Shares (as defined in Section 2.5), if any) shall be canceled and, subject to Section 2.5, shall by virtue of the Merger and without any action on the part of the holder thereof be converted automatically into the right to receive an amount in cash equal to the Offer Price or any higher price paid for each outstanding share of Company Common Stock in the Offer payable, without interest, to the holder of such share of Company Common Stock, upon surrender of the certificate that formerly evidenced such share of Company Common Stock in the manner provided in Section 2.8 (together with amounts payable under Section 2.6(b), the "Merger Consideration"); and

- (ii) Each share of Company Common Stock issued and outstanding immediately prior to the Effective Time that is owned by Parent or Sub or any of their direct or indirect subsidiaries and each share of Company Common Stock that is owned by the Company as treasury stock shall be canceled and retired and cease to exist and no payment or distribution shall be made with respect thereto.
- (b) At the Effective Time, by virtue of the Merger and without any action on the part of Sub, the Company or the holders of any of the following securities, each share of common stock, no par value per share, of Sub issued and outstanding immediately prior to the Effective Time shall be converted into and become one validly issued, fully paid and nonassessable share of common stock, no par value per share, of the Surviving Corporation and all such shares together shall constitute the only outstanding shares of capital stock of the Surviving Corporation.
- (c) At the Effective Time, all shares of the Company Common Stock converted pursuant to Section 2.4(a)(i) shall no longer be outstanding and shall automatically be canceled and retired and cease to exist, and each holder of a certificate ("Certificate") representing any such shares of Company Common Stock shall cease to have any rights with respect thereto, except the right to receive the Merger Consideration in accordance with Section 2.4(a)(i).

#### Section 2.5. Dissenting Shares.

(a) If and only to the extent that any holders of shares of Company Common Stock are entitled under Section 1300(b)(1) of the CGCL to be paid the "fair market value" of such shares of Company Common Stock, then notwithstanding anything in this Agreement to the contrary, shares of Company Common Stock that are issued and outstanding immediately prior to the Effective Time and which are held by Company Shareholders who have demanded and perfected their demands for appraisal of such shares of Company Common Stock in the time and manner provided in Section 1300 of the CGCL and, as of the Effective Time, have neither effectively withdrawn nor lost their rights to such appraisal and payment under the CGCL (the "Dissenting Shares") shall not be converted as described in Section 2.4(a)(i), but shall, by virtue of the Merger, be entitled to only such rights as are granted by Section 1300 of the CGCL; provided, however, that if such holder shall have failed to perfect or shall have effectively withdrawn or lost his, her or its right to appraisal and payment under the CGCL, such holder's shares of Company Common Stock shall thereupon be

A - 6

deemed to have been converted, at the Effective Time, as described in Section 2.4(a) (i), into the right to receive the Merger Consideration set forth in such provisions, without any interest thereon.

(b) The Company shall give Parent (i) prompt notice of any demands for appraisal pursuant to Section 1300 of the CGCL received by the Company, withdrawals of such demands, and any other instruments served pursuant to the CGCL and received by the Company and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the CGCL. The Company shall not, except with the prior written consent of Parent or as otherwise required by applicable Law, make any payment with respect to any such demands for appraisal or offer to settle or settle any such demands.

# Section 2.6. Stock Option Plans.

(a) In connection and simultaneously with the Offer, the Company shall use its reasonable best efforts to ensure that each holder of outstanding options (whether vested or unvested) to acquire Company Common Stock (the "Company Options") granted under the Company's Stock Option Plan adopted in 1993, the Company's 1994 Non-Employee Directors' Stock Option Plan and the Company's Amended and Restated 1997 Stock Option/Stock Issuance Plan (collectively, the "Company Stock Option Plans") shall execute an agreement to exchange immediately prior to the consummation of the Offer such holder's Company Options to the Company for an amount in cash determined by multiplying (A) the excess, if any, of the Offer Price over the applicable exercise price per share of the Company Option (regardless of the exercise price) by (B) the number of shares of Company Common Stock subject to the Company Options (whether vested or unvested) held by such holder (such amount, the "Option Consideration"). Upon the expiration date of the Offer, each holder of Company Options immediately prior to the consummation of the Offer, who has exchanged such Company Options in accordance

with this Section 2.6 shall become entitled to the Option Consideration, and all rights of such holder associated with the Company Options shall be terminated and canceled. The obligation of the Company to accept and pay for the Company Options so exchanged shall be subject to the satisfaction, or waiver by Sub, of the conditions set forth in Annex I. On or prior to the expiration date of the Offer, the Purchaser shall, or shall cause Sub to, lend to the Company on commercially reasonable terms an amount equal to the aggregate amount of the Option Consideration. The Company shall be entitled to deduct and withhold from the Option Consideration otherwise payable to any holder of Company Options such amount that the Company is required to deduct and withhold with respect to the making of such payment under the Code, the rules and regulations promulgated thereunder, or any provision of state, local or foreign tax Law. The Company shall promptly pay or cause to be paid any amounts withheld pursuant to this Section 2.6 for applicable foreign, federal, state and local taxes to the appropriate Governmental Entity on behalf of such holders of Company Options.

(b) As of the Effective Time, either (i) each Company Option not so exchanged immediately prior to the consummation of the Offer shall be canceled, and in consideration for such cancellation the holder thereof shall become entitled to receive an amount in cash (subject to Section 2.8(f)) equal to the Option Consideration, or (ii) the Company shall use its reasonable best efforts to ensure that each holder of a Company Option shall execute an agreement which provides that such Company Option shall be canceled, and in consideration for such cancellation the holder thereof shall become entitled to receive an amount in cash (subject to Section 2.8(f)) equal to the product of (i) the number of shares subject to the Company Options, whether vested or unvested, held by such holder and (ii) the excess of the Merger Consideration per share over the per share exercise price of each such Company Option.

Section 2.7. Company Warrants. The Company shall use its reasonable best efforts to provide that each of the issued and outstanding warrants to purchase shares of Company Common Stock (the "Company Warrants") shall be exercised in full immediately prior to the consummation of the Offer. In addition, the Company shall, by written notice to each holder of Company Warrants, offer to pay such holder upon the consummation of the Offer, in exchange for the cancellation of such holder's Company Warrants (regardless of exercise price) upon the consummation of the Offer, an amount in cash determined by multiplying (A) the excess, if any, of the Offer Price over the applicable exercise price per share of the Company Warrant by (B) the number of shares of Company Common Stock such holder could have purchased had such holder exercised such Company Warrant in full immediately prior to the consummation of the Offer (the "Warrant

**D** = 7

Consideration"). The Company shall be entitled to deduct and withhold from the Warrant Consideration otherwise payable to any holder of Company Warrants such amount that the Company is required to deduct and withhold with respect to the making of such payment under the Code, the rules and regulations promulgated thereunder, or any provision of state, local or foreign tax Law. The Company shall promptly pay or cause to be paid any amounts withheld pursuant to this Section 2.7 for applicable foreign, federal, state and local taxes to the appropriate Governmental Entity on behalf of such holders of Company Warrants.

Section 2.8. Surrender of Shares of Company Common Stock; Stock Transfer Books.

(a) Prior to the Effective Time, Parent shall designate a bank or trust company reasonably acceptable to the Company to act as agent (the "Paying Agent") for the holders of shares of Company Common Stock to receive the funds necessary to make the payments to such holders provided for in Section 2.4 upon surrender of their Certificates and for the holders of Company Options to receive the payments provided for in Section 2.6(b) upon surrender and cancellation of the Company Options. Parent will, on or prior to the Effective Time, deposit with the Paying Agent the Merger Consideration to be paid in respect of the shares of Company Common Stock and Company Options (the "Fund"). The Fund shall be invested by the Paying Agent as directed by Parent; provided that such investments shall be in obligations of, or quaranteed by, the United States of America or any agency thereof and backed by the full faith and credit of the United States of America, in commercial paper obligations rated A-1 or P-1 or better by Moody's Investor Services, Inc., or Standard and Poor's Corporation, respectively, or in deposit accounts, certificates of deposit or banker's acceptances of, or repurchase or reverse repurchase agreements with, or Eurodollar time deposits purchased from, commercial banks with capital, surplus and undivided profits aggregating in excess of US\$100 million (based upon the most recent financial statements of such bank which are then publicly available). Any net profit resulting from, or interest or income produced by, such investments, shall be placed in the Fund and be payable to the Surviving Corporation upon demand of non-disbursed funds pursuant to Section 2.8(d) hereof. Parent shall promptly replace any monies lost through any investment made pursuant to this Section 2.8(a), and Parent shall in any event be liable for the payment of the Merger Consideration, notwithstanding any losses in the Fund. The Fund shall not be used for any purpose except as expressly provided in this Agreement. The Paying Agent shall make the payments provided for in Sections 2.4 and 2.6(b).

(b) Promptly after the Effective Time, the Surviving Corporation shall cause to be mailed to each person who was, at the Effective Time, a holder of record of shares of Company Common Stock entitled to receive the Merger Consideration pursuant to Section 2.4 a form of letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon proper delivery of the Certificates to the Paying Agent) and instructions for use in effecting the surrender of the Certificates pursuant to such letter of transmittal. Upon surrender to the Paying Agent of a Certificate, together with such letter of transmittal, duly completed and validly executed in accordance with the instructions thereto, and such other documents as may be reasonably required pursuant to such instructions, the holder of such Certificate shall be entitled to receive in exchange therefor the Merger Consideration for each share of Company Common Stock formerly evidenced by such Certificate, and such Certificate shall then be canceled. Until so surrendered, each such Certificate shall, at and after the Effective Time, represent for all purposes, only the right to receive such Merger Consideration. No interest shall accrue or be paid to any beneficial owner of shares of Company Common Stock or any holder of any Certificate with respect to the Merger Consideration payable upon the surrender of any Certificate. If payment of the Merger Consideration is to be made to a person other than the person in whose name the surrendered Certificate is registered on the stock transfer books of the Company, it shall be a condition of payment that the Certificate so surrendered shall be endorsed in blank or to the Paying Agent or otherwise be in proper form for transfer and that the person requesting such payment shall have paid all transfer and other taxes required by reason of the payment of the Merger Consideration to a person other than the registered holder of the Certificate surrendered or shall have established to the satisfaction of the Surviving Corporation that such taxes either have been paid or are not applicable. If any Certificate shall have been lost, stolen or destroyed, upon making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation or Parent, the posting by such person of a bond, in such reasonable amount as the Surviving Corporation or Parent may direct, as indemnity against any claim that may be made against it with respect to

A-8

such Certificate, the Paying Agent will issue in exchange for such lost, stolen or destroyed Certificate the applicable Merger Consideration such holder is entitled to receive pursuant to Section 2.4.

- (c) The Parent and the Surviving Corporation shall cause the Paying Agent to pay the amounts payable to each holder of a Company Option pursuant to Section 2.6(b) as soon as reasonably practicable following the Effective Time.
- (d) At any time following the sixth month after the Effective Time, the Surviving Corporation shall be entitled to require the Paying Agent to deliver to it any portion of the Fund which had been made available to the Paying Agent and not disbursed to holders of shares of Company Common Stock (including, without limitation, all interest and other income received by the Paying Agent in respect of all amounts held in the Fund or other funds made available to it), and thereafter each such holder shall be entitled to look only to Parent (subject to abandoned property, escheat and other similar Laws), and only as general creditors thereof, with respect to any Merger Consideration that may be payable upon due surrender of the Certificates held by such holder. If any Certificates representing shares of Company Common Stock shall not have been surrendered immediately prior to such date on which the Merger Consideration in respect of such Certificate would otherwise escheat to or become the property of any Governmental Entity (as defined in Section 4.5), any such cash, shares, dividends or distributions payable in respect of such Certificate shall, to the extent permitted by applicable Law, become the property of the Surviving Corporation, free and clear of all claims or interest of any person previously entitled thereto. Notwithstanding the foregoing, none of the Surviving Corporation, Parent, or the Paying Agent shall be liable to any holder of a share of Company Common Stock for any Merger Consideration delivered in respect of such share of Company Common Stock to a public official pursuant to any abandoned property, escheat or other similar Law.
- (e) At the Effective Time, the stock transfer books of the Company shall be closed and thereafter there shall be no further registration of transfers of shares of Company Common Stock on the records of the Company. From and after the Effective Time, except for Parent, the holders of shares of Company Common Stock outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares of Company Common Stock except as otherwise provided herein or by applicable Law, and all cash paid pursuant to this Article II upon the surrender or exchange of Certificates shall be deemed to have been paid in full satisfaction of all rights pertaining to the shares of Company Common Stock theretofore represented by such Certificate.
- (f) Parent, Sub, the Surviving Corporation and the Paying Agent, as the case may be, shall be entitled to deduct and withhold from the Merger Consideration otherwise payable pursuant to this Agreement to any holder of shares of Company Common Stock or Company Options such amounts that Parent, Sub, the Surviving Corporation or the Paying Agent is required to deduct and withhold with respect to the making of such payment under the Internal Revenue Code of

1986, as amended (the "Code"), the rules and regulations promulgated thereunder or any provision of state, local or foreign tax Law. To the extent that amounts are so withheld by Parent, Sub, the Surviving Corporation or the Paying Agent, such amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the shares of Company Common Stock or Company Options in respect of which such deduction and withholding was made by Parent, Sub, the Surviving Corporation or the Paying Agent. The Surviving Corporation or Parent shall promptly pay or cause to be paid any amounts withheld pursuant to this Section 2.8(f) for applicable foreign, federal, state and local taxes to the appropriate Governmental Entity on behalf of such holders of Company Common Stock or Company Options.

(g) Except as otherwise provided in this Agreement, the Surviving Corporation shall pay all charges and expenses including those of the Paying Agent, in connection with the exchange of cash for shares of Company Common Stock.

A-9

#### Section 2.9. Shareholders Meeting.

- (a) Subject to Section  $2.9\,\mathrm{(d)}$ , if required by applicable Law in order to consummate the Merger, the Company, acting through the Post-Acceptance Board, shall, in accordance with applicable Law:
  - (i) duly call, give notice of, convene and hold a special meeting of the Company Shareholders (the "Shareholders Meeting"), as promptly as practicable following the acceptance for payment and purchase of shares of Company Common Stock by Sub pursuant to the Offer, for the purpose of obtaining the approval of the Merger and this Agreement;
  - (ii) prepare and file with the SEC a preliminary proxy statement or information statement (the "Proxy Statement") relating to the Merger and this Agreement, and use its reasonable best efforts (x) to obtain and furnish the information required to be included by applicable Law in the preliminary Proxy Statement and, after consultation with Parent, to respond promptly to any comments made by the SEC with respect to the Proxy Statement, and (y) to cause the Proxy Statement and any amendment or supplement thereto, to be mailed to the Company Shareholders, provided, that the Company (1) will promptly notify Parent of its receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements of the Proxy Statement or for additional information; (2) will promptly provide Parent with copies of all correspondence between the Company or any of its representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to the Proxy Statement or the Merger and (3) will not amend or supplement the Proxy Statement without first consulting with Parent and its counsel, and (z) to obtain the necessary approvals of the Merger and this Agreement by its shareholders to the extent required by the CGCL;
  - (iii) prepare and revise the Proxy Statement so that, at the date mailed to Company shareholders and at the time of the Shareholders Meeting, the Proxy Statement will (x) not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order that the statements made therein, in light of the circumstances under which they are made, are not misleading (except that the Company shall not be responsible under this clause (iii) with respect to statements made therein based on information supplied by Parent or Sub expressly for inclusion in the Proxy Statement), and (y) comply in all material respects with the provisions of the Exchange Act and the rules and regulations thereunder:
  - (iv) include in the Proxy Statement the recommendation of the Board of Directors of the Company that the Company Shareholders vote in favor of the approval of the Merger and the adoption of this Agreement; and
  - (v) solicit proxies from the Company Shareholders in favor of the approval of the Merger and the adoption of this Agreement and take all other action necessary, or in the reasonable judgment of Parent, helpful to secure the vote or consent of such holders required by the CGCL or this Agreement to effect the Merger.
- (b) Parent shall furnish to the Company such information concerning itself and Sub, for inclusion in the Proxy Statement, as may be requested by the Company and required to be included in the Proxy Statement. Such information provided by Parent and Sub in writing expressly for inclusion in the Proxy Statement will not, at the date the Proxy Statement is filed with the SEC, and mailed to Company shareholders and (including any corrections or modifications made by Parent or Sub to such information) at the time of the Shareholders Meeting, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order that the statements made therein, in light of the circumstances under which they were made, are not misleading.
  - (c) Parent shall vote or cause to be voted all shares of the Company Common

Stock purchased pursuant to the Offer and all other shares beneficially owned by Parent or Sub in favor of adoption of the Merger and the Merger Agreement.

(d) In the event that Parent or Sub, collectively, shall acquire at least 90% of the issued and outstanding shares of Company Common Stock, the Company, Parent and Sub shall take all appropriate and necessary

#### A-10

action to cause the Merger to become effective as soon as practicable after the expiration or termination of the Offer, without a meeting of shareholders of the Company, in accordance with Section 1110 of the CGCL.

Section 2.10. Subsequent Actions. If, at any time after the Effective Time, the Surviving Corporation shall consider or be advised that any deeds, bills of sale, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm of record or otherwise in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of either Sub or the Company acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement and the Merger, the officers and directors of the Surviving Corporation are hereby authorized to execute and deliver, in the name and on behalf of each of Sub or the Company or otherwise, all such deeds, bills of sale, assignments and assurances and to take and do, in the name and on behalf of each of Sub or the Company or otherwise, all such other actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title and interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement and the Merger.

#### ARTICLE III

#### THE SURVIVING CORPORATION

Section 3.1. Articles of Incorporation. The Articles of Incorporation of the Company as in effect immediately prior to the Effective Time shall be the Articles of Incorporation of the Surviving Corporation, until, subject to Section  $6.7\,(a)$ , the same shall thereafter be altered, amended or repealed in accordance with applicable Law or such Articles of Incorporation.

Section 3.2. Bylaws. The Bylaws of the Company as in effect immediately prior to the Effective Time shall be the Bylaws of the Surviving Corporation, until, subject to Section 6.7(a), the same shall thereafter be altered, amended or repealed in accordance with applicable Law, the Articles of Incorporation of the Surviving Corporation or such Bylaws.

Section 3.3. Directors and Officers. From and after the Effective Time, until the earlier of their resignation or removal or until their respective successors are duly elected or appointed and qualified in accordance with applicable Law, (i) the directors of Sub at the Effective Time shall be the directors of the Surviving Corporation, and (ii) the officers of Sub at the Effective Time shall be the officers of the Surviving Corporation.

# ARTICLE IV

# REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Sub (except as set forth in the written disclosure letter (which letter shall in each case specifically identify by reference to Sections of this Agreement any exceptions to each of the representations and warranties contained in this Agreement) delivered by the Company to Parent and Sub in connection with the execution and delivery of this Agreement (the "Company Disclosure Letter")) as follows:

Section 4.1. Organization and Standing. Each of the Company and each Subsidiary (as defined in Section 4.2(c)) (i) is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization, (ii) has the necessary corporate power and authority to own, lease and operate its properties and assets and to conduct its business as presently conducted and (iii) is duly qualified or licensed to do business as a foreign corporation and is in good standing in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its business makes such qualification or licensing necessary, except, in the case of this clause (iii), where the failure to be so qualified or licensed or be in such good standing is not reasonably likely to have a Company Material Adverse Effect (as defined in Section 9.11(a)). The Company has furnished or made available to Parent true and complete copies of its articles of incorporation (including any certificates of designations attached thereto, the "Company Articles of Incorporation") and bylaws (the "Company Bylaws") and the articles of incorporation and bylaws (or

documents are in full force and effect, and neither the Company nor any Subsidiary is in violation of any provision of its articles of incorporation, bylaws or equivalent organizational documents.

## Section 4.2. Capitalization

- (a) The authorized capital stock of the Company consists of 15,000,000 shares of Company Common Stock and 5,000,000 shares of preferred stock, no par value per share (the "Preferred Stock"). As of October 25, 2001, (i) 11,103,321 shares of Company Common Stock are issued and outstanding, all of which are validly issued, fully paid and nonassessable and free of preemptive rights, (ii) no shares of Company Common Stock are held in the treasury of the Company, (iii) Company Options (as defined in Section 1.6) to purchase 1,737,699 shares of Company Common Stock are outstanding pursuant to the Company Stock Option Plans, and an additional 188,886 shares of Company Common Stock are authorized and reserved for future issuance pursuant to the Company Stock Option Plans that have not yet been awarded, (iv) no shares of Preferred Stock are issued or outstanding, and (v) Company Warrants to purchase 31,806 shares of Company Common Stock are outstanding and 31,806 shares of Company Common Stock are reserved for future issuance pursuant to the Company Warrants. Section 4.2 of the Company Disclosure Letter sets forth a true and complete list of the outstanding Company Options and Company Warrants together with the exercise prices thereof. During the period from October 25, 2001 through the date of this Agreement, the Company has not issued any shares of capital stock of the Company or any other securities of the Company or any options or warrants to acquire any such shares or other securities, other than issuances of shares of Company Common Stock pursuant to the exercise of Company Options or Company Warrants outstanding as of October 25, 2001.
- (b) Except as set forth in Section 4.2(a) of this Agreement or in Section 4.2(b) of the Company Disclosure Letter, there are no options, warrants, convertible securities, subscriptions, stock appreciation rights, phantom stock plans or stock equivalents or other similar rights, agreements, arrangements or commitments (contingent or otherwise) of any character issued or authorized by the Company or any Subsidiary relating to the issued or unissued capital stock of the Company or any Subsidiary or obligating the Company or any Subsidiary to issue or sell any shares of capital stock of, or options, warrants, convertible securities, subscriptions or other equity interests in, the Company or any Subsidiary. All shares of Company Common Stock subject to issuance as aforesaid, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid and nonassessable. There are no outstanding contractual obligations of the Company or any Subsidiary to repurchase, redeem or otherwise acquire any shares of Company Common Stock or any capital stock of any Subsidiary or to pay any dividend or make any other distribution in respect thereof or to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any person.
- (c) Section 4.2(c) of the Company Disclosure Letter sets forth a correct and complete list of each corporation, association, subsidiary, partnership, limited liability company or other entity of which the Company owns of record or beneficially, directly or indirectly, 30% or more of the outstanding equity interests (each a "Subsidiary" and collectively, the "Subsidiaries"). The Company owns beneficially and of record all of the issued and outstanding capital stock of each Subsidiary and does not own an equity interest in any other corporation, association, partnership, limited liability company or other entity, other than in the Subsidiaries. Each outstanding share of capital stock of each Subsidiary is duly authorized, validly issued, fully paid and nonassessable and each such share owned by the Company or another Subsidiary is free and clear of all security interests, liens, claims, pledges, options, rights of first refusal, agreements, limitations on the Company's or such other Subsidiary's voting rights, charges and other encumbrances of any nature whatsoever, except for liens for Taxes accrued but not yet payable.

# Section 4.3. Authority for Agreement

(a) The Company has all necessary corporate power and authority to execute and deliver this Agreement and the Option Agreement, to perform its obligations hereunder and thereunder and, subject to obtaining necessary shareholder approval (if required by applicable Law), to consummate the Merger and the

#### A-12

other transactions to which the Company is a party contemplated by this Agreement and the Option Agreement. The execution, delivery and performance by the Company of this Agreement and the Option Agreement, and the consummation by the Company of the Merger and the other transactions to which the Company is a party contemplated by this Agreement and the Option Agreement, have been duly authorized by all necessary corporate action (including, without limitation, the unanimous approval of the Board of Directors of the Company) and no other corporate proceedings on the part of the Company are necessary to authorize this Agreement and the Option Agreement or to consummate the Merger or the other transactions to which the Company is a party contemplated by this Agreement and the Option Agreement (other than, with respect to the Merger, the approval and adoption of this Agreement and the Merger by the affirmative vote of a majority

of the then outstanding shares of Company Common Stock (if required by applicable Law) and the filing and recordation of appropriate merger documents as required by the CGCL). Each of this Agreement and the Option Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by Parent and Sub, constitutes a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created hereby is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general application related to or affecting creditors' rights and to general equity principles. The affirmative vote of a majority of the outstanding shares of Company Common Stock entitled to vote at a duly called and held meeting of shareholders is the only vote of the Company Shareholders necessary to approve this Agreement, the Merger and the other transactions to which the Company is a party contemplated by this Agreement.

- (b) At a meeting duly called and held on October 29, 2001, the Board of Directors of the Company unanimously (i) determined that, as of the date of this Agreement, this Agreement, the Option Agreement and the transactions contemplated hereby and thereby to which the Company is a party, including the Offer and the Merger, are fair to and in the best interests of the Company and the Company Shareholders, (ii) approved, authorized and adopted this Agreement, the Option Agreement, the Offer and the Merger and the other transactions to which the Company is a party contemplated hereby, and (iii) resolved, subject to the provisions of this Agreement, to recommend approval, adoption and acceptance of this Agreement, the Offer, and the Merger by the Company Shareholders. No antitakeover or similar statute in California or New York, or to the knowledge of the Company and its Subsidiaries in any other jurisdiction, is applicable to Parent, Sub, the Company or the Surviving Corporation in connection with the Merger, the Offer, this Agreement, the Option Agreement, the Tender Agreements or any of the transactions to which the Company is a party contemplated hereby or thereby.
- (c) Dresdner Kleinwort Wasserstein, Inc. (the "Independent Advisor") has delivered to the Board of Directors of the Company its opinion, as of the date of this Agreement, that, as of such date and based on the assumptions, qualifications and limitations contained therein, the consideration to be received by the Company Shareholders in the Offer and the Merger is fair to the Company Shareholders from a financial point of view. The Company will provide to Parent and Sub a copy of the written opinion of the Independent Advisor promptly after the date hereof.

Section 4.4. No Conflict. The execution and delivery of this Agreement and the Option Agreement by the Company do not, and the performance of this Agreement and the Option Agreement by the Company and the consummation of the Merger, the Offer and the other transactions to which the Company is a party contemplated by this Agreement and the Option Agreement will not, (i) conflict with or violate the Company Articles of Incorporation or Company Bylaws or equivalent organizational documents of any of its Subsidiaries, (ii) subject to Section 4.5, conflict with or violate any United States federal, state or material local or any material foreign statute, law, rule, regulation, ordinance, code, order, judgment, decree or any other material requirement or rule of law (a "Law") applicable to the Company or any of its Subsidiaries or by which any material property or material group of properties or material asset or material group of assets of the Company or any of its Subsidiaries is bound or affected, or (iii) result in a material breach of or constitute a material default (or an event which with notice or lapse of time or both would become a material default) under, give to others any right of termination, amendment, acceleration or cancellation of, result in triggering any payment or other obligations, or result in the creation of a lien or other encumbrance on any material property or

## A-13

material group of properties or material asset or material group of assets of the Company or any of its Subsidiaries pursuant to, any note, bond, mortgage, indenture, contract, agreement, lease, license, permit, franchise or other instrument or obligation or Material Contract (as defined in Section 4.13) to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries or any property or asset of any of them is bound or affected.

Section 4.5. Required Filings and Consents. The execution and delivery of this Agreement and the Option Agreement by the Company do not, and the performance of this Agreement and the Option Agreement by the Company will not, require any consent, approval, authorization or permit of, or filing with or notification to, any United States federal, state or local or any foreign government or any court, administrative or regulatory agency or commission or other governmental entity or agency, domestic or foreign (a "Governmental Entity"), except (i) for applicable requirements, if any, of the Exchange Act (including relating to the Offering Documents and, if necessary, the Proxy Statement) and state securities or "blue sky" laws ("Blue Sky Laws"), (ii) for those required by the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), (iii) for filings contemplated by Article II hereof, and (iv) for other immaterial approvals, authorizations, consents, orders,

Section 4.6. Compliance. Each of the Company and its Subsidiaries (i) has been operated at all times in compliance in all material respects with all Laws applicable to the Company or any of its Subsidiaries or by which any property, business or asset of the Company or any of its Subsidiaries is bound or affected and (ii) is not in material default or material violation of any notes, bonds, mortgages, indentures, contracts, agreements, leases, licenses, permits, franchises, or other instruments or obligations or Material Contract to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries or any material property or material group of properties or material asset or material group of assets of the Company or any of its Subsidiaries is bound or affected.

#### Section 4.7. SEC Filings, Financial Statements.

- (a) The Company and each Subsidiary, as necessary, has filed all forms, reports, statements and documents required to be filed with the Securities and Exchange Commission ("SEC") since August 28, 1999 (collectively, including all exhibits thereto, the "SEC Reports"), each of which has complied in all material respects with the applicable requirements of the Securities Act of 1933, as amended (the "Securities Act"), and the rules and regulations promulgated thereunder, and the Exchange Act, and the rules and regulations promulgated thereunder, each as in effect on the date so filed. None of the SEC Reports as of their filing dates or, if amended, as of the date of the last such amendment, contained any untrue statement of a material fact or omitted to state a material fact required to be stated or incorporated by reference therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (b) All of the financial statements included in the SEC Reports, in each case, including any related notes thereto, as filed with the SEC (collectively referred to as the "Company Financial Statements") have been prepared in accordance with United States generally accepted accounting principles ("GAAP") applied on a consistent basis throughout the periods involved (except as noted therein, and, in the case of the unaudited statements, as may be permitted by the applicable rules and regulations promulgated under the Exchange Act and the Securities Act and subject, in the case of unaudited statements, to normal year end adjustments permitted by GAAP and consistent with the Company's past practices) and fairly present the consolidated financial position of the Company and its Subsidiaries at the respective dates thereof and the consolidated results of its operations and changes in cash flows for the periods indicated.
- (c) There are no liabilities of the Company or any of its Subsidiaries of any kind whatsoever, whether or not accrued and whether or not contingent or absolute, that are material to the Company and its Subsidiaries, taken as a whole, other than (i) liabilities disclosed or provided for in the consolidated balance sheet of the Company and its Subsidiaries at September 2, 2000, including the notes thereto, (ii) liabilities disclosed in the SEC Reports, and (iii) liabilities incurred in the ordinary course of business consistent with past practice since September 2, 2000, none of which is, individually or in the aggregate, reasonably likely to be material to the Company.

# A-14

- (d) The Company has heretofore furnished or made available to Parent a complete and correct copy of any amendments or modifications which have not yet been filed with the SEC to agreements, documents or other instruments which previously had been filed by the Company with the SEC as exhibits to the SEC Reports pursuant to the Securities Act and the rules and regulations promulgated thereunder or the Exchange Act and the rules and regulations promulgated thereunder.
- Section 4.8. Absence of Certain Changes or Events. Except as contemplated by this Agreement or to the extent specifically disclosed in the SEC Reports filed prior to the date hereof, since September 2, 2000, the Company and its Subsidiaries have conducted their respective businesses only in the ordinary course and consistent with prior practice and there has not been (i) any event or occurrence of any condition that has had or would reasonably be expected to have, individually or in the aggregate with other events, occurrences or conditions, a Company Material Adverse Effect, (ii) any declaration, setting aside or payment of any dividend or any other distribution with respect to any of the capital stock of the Company or any Subsidiary, (iii) any change in accounting methods, principles or practices employed by the Company, except as may have been required by changes in GAAP or applicable Law that are not material and except for any changes to comply with Statement of Financial Accounting Standards No. 141 and No. 142, or (iv) any action of the type described in Sections 6.1(b) (i) or (ii) or 6.1(c) (ii), (iii), (iv), (v), (vi), (vii), (viii), (ix), (xi), (xii), (xiii) or (xiv) which had such action been taken after the date of this Agreement would be in violation of any such Section.
- Section 4.9. Taxes. The Company and each of its Subsidiaries have timely filed all Tax Returns (as defined below) required to be filed by any of them. All such Tax Returns are true, correct and complete in all material respects.

All Taxes (as defined below) of the Company and its Subsidiaries which are (i) shown as due on such Tax Returns, (ii) otherwise due and payable or (iii) claimed or asserted by any taxing authority to be due, have been paid, except for those Taxes being contested in good faith and for which adequate reserves have been established in the financial statements included in the SEC Reports in accordance with GAAP. There are no liens for any Taxes upon the assets of the Company or any of its Subsidiaries, other than statutory liens for Taxes not yet due and payable and liens for real estate Taxes contested in good faith. The Company does not know of any proposed or threatened Tax claims or assessments that, if upheld, individually or in the aggregate, are reasonably likely to have a Company Material Adverse Effect. Neither the Company nor any of its Subsidiaries has made an election under Section 341(f) of the Code. Neither the Company nor any of its Subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, which waiver or extension remains in effect. The Company and each Subsidiary has withheld and paid over to the relevant taxing authority all Taxes required to have been withheld and paid in connection with payments to employees, independent contractors, creditors, shareholders or other third parties. The unpaid Taxes of the Company and its Subsidiaries for the current taxable period (A) did not, as of the most recent Company Financial Statements, exceed the reserve for Tax liability set forth on the face of the balance sheet in the most recent Company Financial Statements and (B) do not exceed that reserve as adjusted for the passage of time through the Closing in accordance with the past custom and practice of the Company and its Subsidiaries in filing their Tax Returns. For purposes of this Agreement, (a) "Tax" (and, with correlative meaning, "Taxes") means any federal, state, local or foreign income, gross receipts, property, sales, use, license, excise, franchise, employment, payroll, premium, withholding, alternative or added minimum, ad valorem, transfer or excise tax, or any other tax, custom, duty, governmental fee or other like assessment or charge of any kind whatsoever, together with any interest or penalty or addition thereto, whether disputed or not, imposed by any Governmental Entity, and (b) "Tax Return" means any return, report or similar statement required to be filed with respect to any Tax (including any attached schedules), including, without limitation, any information return, claim for refund, amended return or declaration of estimated Tax.

Section 4.9 of the Company Disclosure Letter sets forth with reasonable specificity: (i) all jurisdictions in which the Company or any Subsidiary currently has a presence requiring it to pay Taxes measured by income or gross receipts (a "Taxable Presence") and all jurisdictions in which the Company or any Subsidiary has had a Taxable Presence since January 1, 1998, (ii) all Tax Returns for Taxes measured by income or gross receipts filed or due to be filed applicable to the three year period ending on the date hereof and (iii) all

#### A-15

material correspondence with any Tax authorities (including, without limitation, all audits, notices and requests for information from or to taxing authorities) since January 1, 1998.

#### Section 4.10. Assets.

- (a) Except as set forth in the Company's Annual Report on Form 10-K for the fiscal year ended September 2, 2000 (the "10-K"), the Company and each of its Subsidiaries have good and marketable title to, or a valid leasehold interest in, all of their real and personal properties and assets reflected in the 10-K or acquired after September 2, 2000 (other than assets disposed of since September 2, 2000 in the ordinary course of business consistent with past practice or which had an individual value of less than \$200,000 or an aggregate value of less than \$1,000,000), in each case free and clear of all title defects, liens, encumbrances and restrictions, except for (i) liens, encumbrances or restrictions which secure indebtedness which are properly reflected in the 10-K; (ii) liens for Taxes accrued but not yet payable; (iii) liens arising as a matter of law in the ordinary course of business with respect to obligations incurred after September 2, 2000, provided that the obligations secured by such liens are not delinquent; and (iv) liens that do not individually or in the aggregate, materially detract from the value of the assets subject thereto or materially impact the operation of the Company or any Subsidiary. Section 4.10 of the Company Disclosure Letter sets forth a true, correct and complete list of all real property (i) owned or leased by the Company or a Subsidiary, (ii) as to which the Company or a Subsidiary has a written, or to the knowledge of the Company or any Subsidiary any other, license, easement or right of way to use, (iii) as to which the Company or a Subsidiary has the option to purchase, lease, license or acquire an easement or right of way pursuant to any written, or to the knowledge of the Company or any Subsidiary any other, agreement, arrangement or understanding or (iv) in which the Company or a Subsidiary has any other material interest.
- (b) Neither the Company nor any of its Subsidiaries has any legal obligation, absolute or contingent, to any other person to sell or otherwise dispose of any of its assets (other than sales of products and inventory in the ordinary course of business consistent with past practices) with an individual value in excess of \$200,000 or an aggregate value in excess of \$500,000.
  - (c) The buildings, fixtures and equipment of the Company and its

Subsidiaries are in good operating condition and repair (ordinary wear and tear excepted) and are adequate for the uses to which they are being put and for the conduct of the business of the Company and its Subsidiaries as conducted at and prior to the Effective Time.

Section 4.11. Change of Control Agreements. Neither the execution and delivery of this Agreement nor the consummation of the Merger or the other transactions contemplated by this Agreement will (either alone or in conjunction with any other event) result in, cause the accelerated vesting or delivery of, or increase the amount or value of, any payment or benefit to any director, officer or employee of the Company or any Subsidiary. Without limiting the generality of the foregoing, no amount paid or payable by the Company or any Subsidiary in connection with the Merger or the other transactions contemplated by this Agreement, including accelerated vesting of options (either solely as a result thereof or as a result of such transactions in conjunction with any other event) will be an "excess parachute payment" within the meaning of Section 280G of the Code.

Section 4.12. Litigation. Except to the extent specifically disclosed in the SEC Reports filed prior to the date hereof, there are no claims, suits, actions, investigations, indictments or information, or administrative, arbitration or other proceedings ("Litigation") pending or, to the knowledge of the Company and its Subsidiaries, threatened against the Company or any of its Subsidiaries. Except to the extent specifically disclosed in the SEC Reports filed prior to the date hereof, there are no judgments, orders, injunctions, decrees, stipulations or awards (whether rendered by a court, administrative agency, or by arbitration, pursuant to a grievance or other procedure) to which the Company or any of its Subsidiaries is a party, or to the knowledge of the Company or any Subsidiary, is otherwise subject. To the knowledge of the Company and its Subsidiaries, the Company has complied with and is in compliance with the Settlement Agreement with the United States government dated July 23, 1998, and there have been no written, or to the knowledge of the Company or any of its Subsidiaries any other, allegations by any person or entity, including any Governmental Entity, that any of the Company or its Subsidiaries has failed to so comply.

#### A-16

# Section 4.13. Contracts and Commitments.

- (a) Except for any contracts which have prior the date hereof been terminated in accordance with their terms or been terminated by the parties thereto, in each case with no continuing liabilities or obligations on the part of the Company or any Subsidiary, Section 4.13 of the Company Disclosure Letter sets forth a true, correct and complete list of the following contracts to which the Company or a Subsidiary is a party (including every amendment, modification or supplement to the foregoing): (i) any contracts of employment, (ii) any contracts or agreements which limit or restrict the Company or any Subsidiary (or, to the knowledge of the Company and its Subsidiaries, any employee) from engaging in any business in any jurisdiction or limit the freedom of the Company or any Subsidiary to compete in any line of business in any geographic area or requiring the Company or any Subsidiary to share any profits, (iii) agreements or arrangements for the purchase or sale of any assets (other than purchases or sales of products and inventory in the ordinary course of business consistent with past practices) involving payments of more than \$200,000 individually or \$500,000 in the aggregate, (iv) all bonds, debentures, notes, loans, credit or loan agreements or commitments, mortgages, indentures or guarantees or other agreements or contracts relating to the borrowing of money, in each case individually involving remaining principal amounts in excess of \$200,000, (v) agreements with unions, (vi) independent contractor agreements and leased or temporary employee agreements involving in each case individually aggregate annual payments of more than \$75,000 or which are not terminable upon less than 30 days' notice, (vii) leases of any real property involving in each case individually annual rent of \$75,000 or more or leases of personal property involving in each case individually annual rent of \$50,000 or more, (viii) agreements with copackers, manufacturers, suppliers and trade partners, in each case involving individually aggregate annual payments made or received by the Company and its Subsidiaries of more than \$25,000, (ix) all agreements with bottlers and distributors and all franchise and licensing agreements involving in each case individually aggregate annual payments of more than \$25,000, (x) any contract or agreement for the granting or receiving of a license or sublicense or arrangement (other than agreements or licenses for standard, packaged software products and other than licenses to the Company or any Subsidiary involving in each case individually annual payments of less than \$75,000) under which any person is obligated to pay or has the right to receive a royalty, license fee or similar payment, and (xi) all other contracts, agreements or commitments individually involving in each case individually payments made by or to the Company or a Subsidiary of \$250,000 or more (individually, a "Material Contract" and collectively, "Material Contracts"). The Company has delivered or made available to Parent true, correct and complete copies of all Material Contracts, including all amendments thereto.
- (b) The Material Contracts are legal, valid, binding and enforceable in accordance with their respective terms with respect to the Company and to the knowledge of the Company and its Subsidiaries with respect to each other party

to any of such Material Contracts, except, in each case, to the extent that enforcement of rights and remedies created by any Material Contracts are subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general application related to or affecting creditors' rights and to general equity principles. There are no existing material defaults or breaches of the Company under any Material Contract (or events or conditions which, with notice or lapse of time or both would constitute a material default or breach) and, to the knowledge of the Company and its Subsidiaries, there are no such material defaults (or events or conditions which, with notice or lapse of time or both, would constitute a material default or breach) with respect to any third party to any Material Contract. The Company and its Subsidiaries have no knowledge of any pending or threatened bankruptcy, insolvency or similar proceeding with respect to any party to any of such agreements. Section 4.13 of the Company Disclosure Letter identifies each Material Contract set forth therein that requires the consent of or notice to the other party thereto to avoid any breach, default or violation of such contract, agreement or other instrument in connection with the transactions contemplated hereby.

(c) The calculations of allowance for product returns for use in connection with the preparation of the Company Financial Statements contained in the SEC Reports conformed to GAAP. Set forth in Section 4.13 of the Company Disclosure Letter is a description of the Company's policy regarding product returns.

Section 4.14. Employee Benefit Plans. All employee benefit plans, compensation arrangements and other benefit arrangements covering employees of the Company or any of its Subsidiaries (the "Company"

#### A-17

Benefit Plans") and all employee agreements providing for compensation, severance or other benefits to any employee or former employee of the Company or any of its Subsidiaries are listed in Section 4.14 of the Company Disclosure Letter. True, correct and complete copies of the following documents with respect to each of the Company Benefit Plans have been provided or made available by the Company to Parent: (i) any plans and related trust documents and amendments thereto, (ii) summary plan descriptions and material modifications thereto, (iii) written communications made since January 1, 2000 to employees relating to the Company Benefit Plans and (iv) written descriptions of all non-written agreements relating to the Company Benefit Plans. To the extent applicable, the Company Benefit Plans comply in all material respects with the requirements of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and the Code, and any Company Benefit Plan intended to be qualified under Section 401(a) of the Code has received a determination letter or is a model prototype plan and continues to satisfy the requirements for such qualification. Neither the Company nor any of its Subsidiaries nor any ERISA Affiliate of the Company maintains, contributes to or has maintained or contributed in the past six (6) years to any benefit plan which is covered by Title IV of ERISA or Section 412 of the Code. Neither any Company Benefit Plan, nor the Company nor any Subsidiary has incurred any liability or penalty under Section 4975 of the Code or Section 502(i) of ERISA in excess of \$10,000 in the aggregate or engaged in any transactions that are reasonably likely to result in any liabilities or penalties in excess of \$10,000 in the aggregate. Each of the Company and its Subsidiaries and any ERISA Affiliate which maintains a "group health plan" within the meaning of Section 5000(b)(1) of the Code has complied with the notice and continuation requirements of Section 4980B of the Code, Part 6 of Subtitle B of Title I of ERISA and the regulations thereunder (COBRA), and the creditable coverage certification requirements and limitations on pre-existing condition exclusion requirements of Section 9801 of the Code, Part 7 of Subtitle B of Title I of ERISA and the regulations thereunder (HIPAA). Each Company Benefit Plan has been maintained and administered in compliance in all material respects with its terms and with ERISA and the Code to the extent applicable thereto. There is no pending or, to the knowledge of the Company and its Subsidiaries, threatened or anticipated Litigation against or otherwise involving any of the Company Benefit Plans and no Litigation (excluding claims for benefits incurred in the ordinary course of Company Benefit Plan activities) has been brought against or with respect to any such Company Benefit Plan. All contributions required to be made to the Company Benefit Plans have been made or provided for. Neither the Company nor any of its Subsidiaries maintains or contributes to any plan or arrangement which provides or has any liability to provide life insurance or medical or other employee welfare benefits to any employee or former employee upon his retirement or termination of employment, and neither the Company nor any of its Subsidiaries has ever represented, promised or contracted (whether in oral or written form) to any employee or former employee that such benefits would be provided.

Any individual who performs services for the Company or any of its Subsidiaries (other than through a contract with an organization other than such individual) and who is not treated as an employee for federal income tax purposes by the Company or its Subsidiaries is not an employee for such purposes. There are no agreements in effect between the Company or any Subsidiary and any individual retained by the Company or any Subsidiary to provide services as a consultant or independent contractor.

For purposes of this Agreement "ERISA Affiliate" means any business or

entity which is a member of the same "controlled group of corporations," an "affiliated service group" or is under "common control" with an entity within the meanings of Sections 414(b), (c) or (m) of the Code, is required to be aggregated with the entity under Section 414(o) of the Code, or is under "common control" with the entity, within the meaning of Section 4001(a) (14) of ERISA, or any regulations promulgated or proposed under any of the foregoing Sections.

Section 4.15. Labor and Employment Matters.

(a) Neither the Company nor any of its Subsidiaries is a party to, or bound by, any collective bargaining agreement or other contracts, arrangements, agreements or understandings with a labor union or labor organization that was certified by the National Labor Relations Board ("NLRB"). There is no existing, pending or, to the knowledge of the Company and its Subsidiaries, threatened (i) unfair labor practice charge or complaint, labor dispute, labor arbitration proceeding or any other matter before the NLRB or any other

#### A-18

comparable state agency against or involving the Company or any of its Subsidiaries, (ii) activity or proceeding by a labor union or representative thereof to organize any employees of the Company or any of its Subsidiaries, (iii) certification or decertification question relating to collective bargaining units at the premises of the Company or any of its Subsidiaries or (iv) lockout, strike, organized slowdown, work stoppage or work interruption with respect to such employees.

- (b) Neither the Company nor any of its Subsidiaries has taken any action that would constitute a "Mass Layoff" or "Plant Closing" within the meaning of the Worker Adjustment and Retraining Notification ("WARN") Act or would otherwise trigger notice requirements or liability under any state or local plant closing notice Law. No agreement, arbitration or court decision or governmental order to which the Company or any of its Subsidiaries is a party, or to the knowledge of the Company and its Subsidiaries is subject, in any way limits or restricts any of the Company, any of its Subsidiaries or Parent from relocating or closing any of the operations of the Company or any of its Subsidiaries.
- (c) Neither the Company nor any of its Subsidiaries has failed in any material respect to pay when due any wages (including overtime wages), bonuses, commissions, benefits, taxes, penalties or assessments or other monies, owed to, or arising out of the employment of or any relationship or arrangement with, any officer, director, employee, sales representative, contractor, consultant or other agent. The Company and its Subsidiaries are in compliance in all material respects with all applicable Laws relating to employment and the payment of wages and benefits. There are no, and the Company has no reason to believe there would be any, citations, investigations, administrative proceedings or formal complaints of violations of any federal or state wage and hour Laws pending or, to the knowledge of the Company and its Subsidiaries, threatened before the Department of Labor or any federal, state or administrative agency or court against or involving the Company or any of its Subsidiaries.
- (d) The Company and each of its Subsidiaries are in compliance in all material respects with all immigration Laws relating to employment and have properly completed and maintained all applicable forms (including but not limited to I-9 forms) and, to the knowledge of the Company and its Subsidiaries, there are no citations, investigations, administrative proceedings or formal complaints of violations of the immigration Laws pending or, to the knowledge of the Company and its Subsidiaries, threatened before the Immigration and Naturalization Service or any federal, state or administrative agency or court against or involving the Company or any of its Subsidiaries.
- (e) There are no investigations, administrative proceedings, charges or formal complaints of discrimination (including discrimination based upon sex, age, marital status, race, national origin, sexual preference, disability, handicap or veteran status) pending or, to the knowledge of the Company and its Subsidiaries, threatened before the Equal Employment Opportunity Commission or any federal, state or local agency or court against or involving the Company or any of its Subsidiaries. No discrimination, sexual harassment, retaliation and/or wrongful or tortious conduct claim is pending or, to the knowledge of the Company and its Subsidiaries, threatened against the Company or any of its Subsidiaries under the 1866, 1877, 1964 or 1991 Civil Rights Acts, the Equal Pay Act, the Age Discrimination in Employment Act, as amended, the Americans with Disabilities Act, the Family and Medical Leave Act, the Fair Labor Standards Act, ERISA, or any other federal Law relating to employment or any comparable state or local fair employment practices act regulating discrimination in the workplace, and no wrongful discharge, libel, slander, invasion of privacy or other claim (including but not limited to violations of the Fair Credit Reporting Act, as amended, and any applicable whistleblower statutes) under any state or federal law is pending or, to the knowledge of the Company and its Subsidiaries, threatened against the Company or any of its Subsidiaries.
- (f) If the Company or any of its Subsidiaries is a Federal, State or local contractor obligated to develop and maintain an affirmative action plan, no discrimination claim, show-cause notice, conciliation proceeding, sanctions or

debarment proceedings is pending or, to the knowledge of the Company and its Subsidiaries, has been threatened against the Company or any of its Subsidiaries with the Office of Federal Contract Compliance Programs or any other Federal agency or any comparable state or local agency or court and no desk audit or on-site review is in progress.

#### A-19

- (g) There are no citations, investigations, administrative proceedings or formal complaints of violations of local, state or federal occupational safety and health Laws pending or, to the knowledge of the Company and its Subsidiaries, threatened before the Occupational Safety and Health Review Commission or any federal, state or local agency or court against or involving the Company or any of its Subsidiaries.
- (h) No workers' compensation or retaliation claim is pending against the Company or any of its Subsidiaries in excess of \$50,000 in the aggregate and the Company maintains insurance with respect to workers' compensation claims pursuant to insurance policies that are currently in force as is customary for businesses similar to the business of the Company, or has accrued an adequate liability for such obligations, including, without limitation, adequate accruals with respect to accrued but unreported claims and retroactive insurance premiums.

Section 4.16. Environmental Compliance and Disclosure.

(a) As used in this Section 4.16, the following defined terms have the meanings set forth below:

"Action" means any actual or threatened claim, demand, action, lawsuit, hearing, or proceeding of any kind, whether judicial or administrative in nature.

"Environment" means any surface water or groundwater, drinking supply water, land or soil, surface or subsurface strata or medium, the ambient air, or any other environmental medium.

"Environmental Condition" means any condition relating to the Environment, whether on or off site, whether or not yet discovered, that results or may result in any Action, removal or remedial action, liability, damages, expenses, costs (including reasonable attorneys', consulting, engineering, and accounting fees), or losses of any kind whatsoever, including, without limitation, any Release of any Hazardous Materials.

"Environmental Laws" mean any international, regional, national, federal, state, provincial, or local Laws, statutes, treaties, directives, ordinances, rules, or regulations relating to the Environment, or human health and safety, or the Release of any Hazardous Materials.

"Hazardous Materials" mean any waste, pollutant, hazardous substance, toxic, flammable, explosive, reactive, corrosive, infectious, radioactive, carcinogenic or mutagenic substance, hazardous waste, special waste, industrial substance, by-product, process intermediate product or waste, petroleum or petroleum-derived substance or waste, chemical liquids or solids, liquid or gaseous products, or any constituent of any such substance or waste, the generation, use, handling, storage, treatment, transport, or disposal of which is in any way governed by or subject to any Environmental Laws, or any substance that is or contains polychlorinated biphenyls (PCB's), radon gas, urea formaldehyde, asbestos-containing materials (ACM) or lead.

"Release" means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, disposing, or dumping into the Environment.

- (b) All plants, facilities and real and personal property owned, leased, licensed or otherwise used by the Company or any Subsidiary (collectively referred to as the "Facilities"), and the Company's and any Subsidiary's business operations and activities at, on, or under any of the Facilities or otherwise (collectively referred to as the "Operations"), are and have been at all times during the period of such ownership, lease, license, use, operation or activity by the Company or any Subsidiary in compliance in all material respects with all Environmental Laws and with all licenses, permits, consents or authorizations required by Environmental Laws;
- (c) There are not, and have not been at any time in the past, any Actions, settlements, consent orders or decrees, judicial or administrative orders or decrees, or judgments under any Environmental Laws against or relating to the Company or any Subsidiary, any of the Operations or, during the period of ownership, lease, license, use, operation or activity by the Company or any Subsidiary, any of the Facilities;
- (d) Neither the Company nor any Subsidiary has received notice or is aware of any existing or potential liabilities, obligations, fines, costs, fees or expenses relating to or arising out of any violation of any

Environmental Laws or the unauthorized Release of any Hazardous Materials by the Company or any Subsidiary anywhere, including, without limitation, under the Comprehensive Environmental Response, Compensation, and Liability Act, as amended, 42 U.S.C. Section 9601 et seq., (collectively referred to as "Environmental Liabilities"), and to the Company's or any Subsidiary's knowledge, no facts, circumstances or conditions exist that are related to, associated with or attributable to any of the Facilities or the Operations that may result in any Environmental Liabilities;

- (e) No remediation, removal or cleanup of any Hazardous Materials is being conducted or, to the Company's or any Subsidiary's knowledge, has been conducted at any time in the past at, on or under the Facilities, or caused by or associated with any of the Operations;
- (f) To the Company's or any Subsidiary's knowledge, no Environmental Condition of any kind is or has been at any time in the past at, on or under the Facilities, or caused by or associated with any of the Operations that has not been corrected or remediated at least to the extent required under Environmental Laws and that, individually or in the aggregate, is or could reasonably be expected to be material;
- (g) Hazardous Materials of any kind are not being and, to the Company's or any Subsidiary's knowledge, have not been at any time in the past generated, used, stored, treated, disposed of or Released at, on or under any of the Facilities, or as a result of or in association with any of the Operations, except in compliance in all material respects with applicable Environmental Laws; and
- (h) To the Company's or any Subsidiary's knowledge, no underground or aboveground storage tanks, solid waste management units, landfills or other waste disposal areas, dikes, or impoundments of any kind are located at, on, or under any of the Facilities.

Section 4.17. Intellectual Property Rights. Section 4.17 of the Company Disclosure Letter sets forth a complete and accurate list and description of all United States and foreign patents, trademarks, trade names, service marks, trade dress and domain names, whether the foregoing is in the form of registrations or pending applications therefor, and all material common law rights, that are owned by the Company or any Subsidiary (all such rights together with any copyrights that are owned by the Company or any Subsidiary being hereinafter referred to as the "Intellectual Property Rights"). The Company is the sole legal owner of its material common law Intellectual Property Rights, free and clear of any liabilities, liens, encumbrances, restrictions or claims (except for liens for Taxes accrued, but not yet payable or liens arising as a matter of Law in the ordinary course of business, provided that the obliqations secured by such liens are not delinquent ("Permitted Liens")); and is the sole legal owner of, and has good, valid and enforceable title to, all of its other Intellectual Property Rights, free and clear of any liabilities, liens, encumbrances, restrictions, claims or imperfections of title of any kind (except for Permitted Liens). The Intellectual Property Rights have not been adjudged invalid or unenforceable in whole or in part, and any registrations thereof are in full force and effect. To the knowledge of the Company and its Subsidiaries, no facts or circumstances exist that might result in the invalidity or unenforceability of the Intellectual Property Rights. To the knowledge of the Company and its Subsidiaries, the Company's websites are operated in compliance with all applicable privacy Laws, contest and sweepstakes Laws, rights of publicity Laws, and trademark Laws. The Intellectual Property Rights are not subject to any license (royalty bearing or royalty free) and are not subject to any other arrangement requiring any payment to any person or the obligation to grant rights to any person in exchange, and the Intellectual Property Rights are all of the patent, copyright and trademark rights necessary to the conduct of the business of the Company and the Subsidiaries as currently being conducted. The validity of the Intellectual Property Rights and title thereto (i) have not been questioned in any prior litigation; (ii) are not being questioned in any pending litigation; and (iii) to the knowledge of the Company and its Subsidiaries, are not the subject(s) of any threatened or proposed litigation. To the knowledge of the Company and its Subsidiaries, the business of the Company and the Subsidiaries, as now conducted, does not infringe or conflict with and has not been alleged to infringe or conflict with any copyrights, patents, trade secrets, trademarks, trade names, service marks, trade dress or domain names of others, whether or not such rights result from the common law or registration rights of others. The Company, to its knowledge and to the knowledge of its Subsidiaries, has not received any written or oral inquiries, threats or demands by the Lemelson Medical, Education and Research Foundation asserting violation of one or more of its patents by the Company or any of the Subsidiaries. To the

knowledge of the Company and its Subsidiaries, the consummation of the transactions contemplated hereby will not result in the loss or impairment of any of the Intellectual Property Rights. To the knowledge of the Company and its Subsidiaries, no person or entity currently engaging in any commercial activity

Section 4.18. Products. Since August 28, 1999:

- (a) No shipment or other delivery of products made by the Company or any Subsidiary of the Company (the "Products"), as of the date of such shipment or delivery, is or has been (i) to the extent such Product is subject to the Federal Food, Drug and Cosmetic Act, as amended (the "FD&C Act"), adulterated or misbranded within the meaning of Section 342 or Section 343, respectively, of the FD&C Act or is otherwise in violation of the FD&C Act; (ii) an article which may not be introduced into interstate commerce under the provisions of Section 344 of the FD&C Act to the extent such Product has been introduced by the Company or any Subsidiary into interstate commerce; or (iii) "adulterated" or "misbranded" or is otherwise in violation of any applicable food law of any jurisdiction.
- (b) There have been no recalls or withdrawals related to the  $\ensuremath{\operatorname{Products}}$  .
  - (c) There have been no tampering incidents relating to the Products.
- (d) No federal or state regulatory authority has issued any warning letter or other formal notice of violation of any statute, ordinance or regulation concerning the composition, production, storage, labeling or advertising of the Products.
- Section 4.19. Brokers. Except pursuant to the letter agreement dated September 4, 2001 between the Company and the Independent Advisor (the "Independent Advisor Engagement Letter"), no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with this Agreement, the Merger or the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Company. Section 4.19 of the Company Disclosure Letter includes a complete and correct copy of the Independent Advisor Engagement Letter.
- Section 4.20. Trade Secrets. Each of the Company and its Subsidiaries is, or will be on the Closing Date, the sole legal owner of, and has, or will on the Closing Date have, good, valid and enforceable title to, all trade secrets (including, without limitation, any processes entitled to trade secret protection; and formulae) and proprietary information used in connection with the manufacture of the Products, and the conduct of the operations and businesses of the Company and its Subsidiaries (collectively the "Trade Secrets"), free and clear of any liabilities liens, encumbrances, restrictions, claims or imperfections of title of any kind, other than Permitted Liens. To the knowledge of the Company and its Subsidiaries, neither the Company or any Subsidiary nor any predecessor in interest thereof has disclosed, apart from any information that does not extend beyond what is required by applicable Law to be disclosed, any Trade Secrets. The Company has used, and to the knowledge of the Company and its Subsidiaries the Company's predecessors in interest used, commercially reasonable efforts to protect the Company's rights in its Trade Secrets.
- Section 4.21. Insurance Policies. The Company has delivered to Parent prior to the date hereof a complete and accurate list of all insurance policies in force naming the Company, any of its Subsidiaries or employees thereof as an insured or beneficiary or as a loss payable payee or for which the Company or any Subsidiary has paid or is obligated to pay all or part of the premiums. Neither the Company nor any Subsidiary has received notice of any pending or threatened cancellation or premium increase (retroactive or otherwise) with respect thereto, and each of the Company and the Subsidiaries is in compliance in all material respects with all conditions contained therein. There are no material pending claims against such insurance policies by the Company or any Subsidiary as to which insurers are defending under reservation of rights or have denied liability, and there exists no material claim under such insurance policies that has not been properly filed by the Company or any Subsidiary. The insurance policies of the Company and its Subsidiaries are comparable in the aggregate to the insurance policies held by other entities of similar size and engaged in comparable businesses.

A-22

Section 4.22. Notes and Accounts Receivable.

- (a) There are no notes receivable of the Company or any Subsidiary owing by any director, officer, shareholder or employee of the Company or any Subsidiary ("Affiliate Debt").
- (b) All accounts receivable of the Company and any Subsidiary are current or covered by adequate reserves for uncollectability, and there are no material disputes regarding the collectibility of any such accounts receivable.
- Section 4.23. Transactions with Affiliates. Other than compensation and benefits received in the ordinary course of business as an employee or director of the Company or its Subsidiaries (collectively, the "Affiliate Transactions"), no director, officer or other "affiliate" or "associate" (as hereinafter

defined) of the Company or any Subsidiary or any entity in which, to the knowledge of the Company and its Subsidiaries, any such director, officer or other affiliate or associate, owns any beneficial interest (other than a publicly held corporation whose stock is traded on a national securities exchange or in the over-the-counter market and less than 1% of the stock of which is beneficially owned by any such persons) has any interest in: (i) any contract, arrangement or understanding with, or relating to the business or operations of Company or any Subsidiary; (ii) any loan, arrangement, understanding, agreement or contract for or relating to indebtedness of the Company or any Subsidiary; (iii) any property (real, personal or mixed), tangible, or intangible, used or currently intended to be used in, the business or operations of the Company or any Subsidiary; or (iv) any Major Supplier, Major Customer or Major Distributor (each as defined in Section 4.26). Without limiting the generality of the foregoing, there are no amounts due or payable by the Company or any Subsidiary to any of the Principal Shareholders or any of their affiliates or associates in connection with the transactions contemplated by this Agreement or the Tender Agreements or otherwise.

Section 4.24. No Existing Discussions. As of the date hereof, the Company is not engaged, directly or indirectly, in any negotiations or discussions with any other party with respect to a Competing Acquisition Proposal (as defined in Section 6.5).

Section 4.25. Shareholder Rights Agreement. Neither the Company nor any Subsidiary has adopted a shareholder rights agreement or any similar plan or agreement which limits or impairs the ability to purchase, or become the direct or indirect beneficial owner of, shares of Company Common Stock or any other equity or debt securities of the Company or any of its Subsidiaries, other than any shareholder rights plan or shareholder rights agreement that (a) is adopted after the date of this Agreement, (b) does not impair the ability of the parties to consummate the Offer or the Merger in accordance with the terms of this Agreement and (c) otherwise does not have an adverse effect on Parent or Sub or on the rights of Parent or Sub under this Agreement or any of the Tender Agreements.

Section 4.26. Major Suppliers, Customers and Distributors.

- (a) Section 4.26(a) of the Company Disclosure Letter sets forth a list of each supplier of goods or services to Company and the Subsidiaries (i) who is involved in contract manufacturing, copacking arrangements or toll manufacturing (i.e., taking possession but not title to inventory and finished goods for purposes of production activities and receiving a fee in return for such production activities) or a supplier of critical or key ingredients or (ii) to whom the Company and the Subsidiaries paid in the aggregate more than \$25,000 during the 12-month period ended September 1, 2001 (each a "Major Supplier" and, collectively, "Major Suppliers"), together with in each case the amount paid during such period. Neither the Company nor any Subsidiary is engaged in any material dispute with any Major Supplier and, to the knowledge of the Company and its Subsidiaries, no Major Supplier intends to terminate, or materially limit or reduce, its business relations with the Company or any Subsidiary.
- (b) Section 4.26(b) of the Company Disclosure Letter sets forth a list of each customer which accounted for net sales to the Company and the Subsidiaries in the aggregate of more than \$500,000 during the 12-month period ended September 1, 2001 (each a "Major Customer" and, collectively, "Major Customers") together with the amount of net sales produced during such period. Neither the Company nor any Subsidiary is engaged in any material dispute with any Major Customer and, to the knowledge of the

## A-23

Company and its Subsidiaries, no Major Customer intends to terminate, or materially limit or reduce, its business relations with the Company or any Subsidiary.

(c) Section 4.26(c) of the Company Disclosure Letter sets forth a list of each distributor from whom the Company and its Subsidiaries received, or to whom the Company and its Subsidiaries paid, in the aggregate more than \$500,000 during the 12-month period ended September 1, 2001 (each a "Major Distributor" and, collectively, "Major Distributors") together in each case with the amount received or paid during such period. Neither the Company nor any Subsidiary is engaged in any material dispute with any Major Distributor and, to the knowledge of the Company and its Subsidiaries, no Major Distributor intends to terminate, or materially limit or reduce, its business relations with the Company or any Subsidiary.

## ARTICLE V

# REPRESENTATIONS AND WARRANTIES OF PARENT AND SUB

Each of Parent and Sub, jointly and severally, represents and warrants to the Company as follows:

Section 5.1. Organization and Standing. Such person is a corporation duly

organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation and has the corporate power and authority to own, lease and operate its properties and assets and to conduct its business as presently conducted.

Section 5.2. Sub.

- (a) Since the date of its incorporation, Sub has not carried on any business or conducted any operations other than the execution of this Agreement and the performance of its obligations hereunder. Sub was incorporated solely for the consummation of the transactions contemplated hereby.
- (b) The authorized capital stock of Sub consists of 100 shares of common stock, no par value per share, all of which have been validly issued, fully paid and nonassessable and are owned by Parent free and clear of any Liens.
- Section 5.3. Authority for Agreement. Such person has all necessary corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the Merger and the other transactions contemplated by this Agreement. The execution, delivery and performance by such person of this Agreement, and the consummation by each such person of the Merger and the other transactions contemplated by this Agreement, have been duly authorized by all necessary corporate action and no other corporate proceedings on the part of such person are necessary to authorize this Agreement or to consummate the Merger or the other transactions contemplated by this Agreement (other than, with respect to the Merger, the filing and recordation of appropriate merger documents as required by the CGCL). This Agreement has been duly executed and delivered by such person and, assuming due authorization, execution and delivery by the Company, constitutes a legal, valid and binding obligation of each of such person enforceable against such person in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general application related to or affecting creditors' rights and to general equity principles.
- Section 5.4. No Conflict. The execution and delivery of this Agreement by such person do not, and the performance of this Agreement by such person and the consummation of the Merger and the other transactions contemplated by this Agreement will not, (i) conflict with or violate the certificate of incorporation or bylaws of such person, (ii) conflict with or violate any Law applicable to such person or by which any property or asset of such person is bound or affected, or (iii) result in any material breach of or constitute a material default (or an event which with notice or lapse of time or both would become a material default) under, give to others any right of termination, amendment, acceleration or cancellation of, or result in the creation of a lien or other encumbrance on any material property or material group of properties or asset or material group of assets of such person pursuant to, any note, bond, mortgage, indenture, contract, agreement, lease, license, permit, franchise or other instrument or obligation to which such person is a party or by which such person or any property or asset of either of them is bound or affected, except in the case of clauses

## A-24

- (ii) and (iii) for any such conflicts, violations, breaches, defaults or other occurrences which would not, individually or in the aggregate, prevent or materially delay the performance by such person of its respective obligations under this Agreement or the consummation of the Merger or the other transactions contemplated by this Agreement.
- Section 5.5. Required Filings and Consents. The execution and delivery of this Agreement by such person do not, and the performance of this Agreement by such person will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity, except (i) for applicable requirements of the Exchange Act (including relating to the Offering Documents and, if necessary, the Proxy Statement), Blue Sky Laws and filing and recordation of appropriate merger documents as required by the CGCL, (ii) for those required by the HSR Act, (iii) for filings and consents contemplated by Article II and (iv) where failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, prevent or materially delay the performance by such person of any of its respective obligations under this Agreement or the consummation of the Merger or the other transactions contemplated by this Adreement.
- Section 5.6. Brokers. Except for Morgan, Stanley & Co., Incorporated, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission payable by such person in connection with this Agreement, the Merger or the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of such person.
- Section 5.7. Financing. Parent and Sub will have the funds available as is necessary to consummate the transactions contemplated hereby in accordance with the terms hereof.

Section 6.1. Conduct of the Business Pending the Merger.

- (a) The Company covenants and agrees that between the date of this Agreement and the earlier of the Effective Time or the termination of this Agreement in accordance with Section 8.1, unless Parent shall otherwise expressly agree in writing (which agreement shall not be unreasonably withheld or delayed), (i) the business of the Company and its Subsidiaries shall be conducted only in, and the Company and its Subsidiaries shall not take any action except in, the ordinary course of business and in a manner consistent with prior practice, (ii) the Company and its Subsidiaries shall use its reasonable best efforts to preserve intact their business organizations, keep available the services of their current officers and employees and preserve the current relationships of the Company and its Subsidiaries with customers, suppliers, distributors and other persons with which the Company or its Subsidiaries has material business relations, and (iii) the Company and its Subsidiaries will comply in all material respects with all applicable Laws and regulations wherever its business is conducted, including, without limitation, the timely filing of all reports, forms or other documents with the SEC required pursuant to the Securities Act or the Exchange Act.
- (b) The Company covenants and agrees that between the date of this Agreement and the earlier of the Effective Time or the termination of this Agreement in accordance with Section 8.1 hereof, unless Parent shall otherwise expressly agree in writing (which agreement shall not be unreasonably withheld or delayed), the Company shall not, nor shall the Company permit any of its Subsidiaries to, (i) declare or pay any dividends on or make other distributions (whether in cash, stock or property) in respect of any of its capital stock, except for dividends payable to the Company by a wholly owned Subsidiary of the Company, (ii) split, combine or reclassify any of its capital stock or issue or authorize or propose the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; (iii) repurchase, redeem or otherwise acquire any shares of its capital stock; (iv) issue, deliver or sell, or authorize or propose the issuance, delivery or sale of, any shares of its capital stock or any securities convertible into any such shares of its capital stock, or any rights, warrants or options to acquire any such shares or convertible securities or any stock appreciation rights, phantom stock plans or stock equivalents, other than the issuance of shares of Company Common Stock upon (A) the exercise of Company Options outstanding as of the date of this Agreement or awarded in conformity

## A-25

with clause (v) below, and (B) the exercise of warrants outstanding as of the date of this Agreement, (v) award or grant, or authorize or propose the award or grant of, any unvested Company Options, other than the award or grant of Company Options to employees (other than executive officers of the Company) in accordance with the terms of the Company Stock Option Plans and in the ordinary course of business and consistent with past practices for up to no more than 100,000 shares of Company Common Stock in the aggregate for all employees issued to employees who have not prior to the date hereof received option grants in calendar year 2001; (vi) modify or adjust any outstanding options to acquire shares of Company Common Stock, or (vii) take any action that would, or could reasonably be expected to, result in any of the conditions set forth in Article VII not being satisfied.

(c) The Company covenants and agrees that between the date of this Agreement and the earlier of the Effective Time or the termination of this Agreement in accordance with Section 8.1 hereof, unless Parent shall otherwise expressly agree in writing (which agreement shall not be unreasonably withheld or delayed), the Company shall not, nor shall the Company permit any of its Subsidiaries to, (i) amend its articles of incorporation (including any certificate of designations attached thereto) or bylaws or other equivalent organizational documents; (ii) create, assume or incur any indebtedness for borrowed money or quaranty any such indebtedness of another person or mortgage or pledge any of its assets or properties, other than in connection with (A) existing lines of credit, (B) indebtedness owing to, or guaranties of indebtedness owing to, the Company, (C) leasing contracts entered into in the ordinary course of business with payments of less than \$50,000 in the aggregate on a monthly basis, (D) indebtedness incurred to make payments specifically provided under or contemplated by Section 2.6(a), Section 4.19, Section 6.15 and Section 6.19 of this Agreement or (E) accounts payable incurred in the ordinary course of business consistent with past practice; (iii) make any loans or advances to any other person other than loans or advances between any Subsidiaries of the Company or between the Company and any of its Subsidiaries and other than advances of ordinary business expenses or to employees in the ordinary course of business consistent with past practice in principal amounts of not more than \$10,000; (iv) merge or consolidate with any other entity in any transaction, (v) sell any business or assets, other than sales of assets (other than sales of products and inventory in the ordinary course of business consistent with past practices) having a value of less than \$200,000 individually and \$500,000 in the aggregate to the extent such sales are in the ordinary course of business and consistent with past practices; (vi) change its

accounting policies except as required by GAAP or applicable Law; (vii) make any change in employment terms for any of its directors or officers, except as expressly contemplated by this Agreement; (viii) alter, amend or create any obligations with respect to compensation, severance, benefits, change of control payments or any other payments to employees, directors or affiliates of the Company or its Subsidiaries or enter into any new, or amend any existing, employment agreements, except (A) as required by applicable Law or (B) severance agreements containing terms consistent with the Company's policies and practices as of the date hereof in amounts not to exceed \$15,000 individually or \$150,000 in the aggregate other than severance agreements with key employees identified in Section 6.1(c) of the Company Disclosure Letter, (ix) make any change to the Company Benefit Plans, except as required by applicable Law, (x) amend or cancel or agree to the amendment or cancellation of any Material Contract; (xi) pay, loan or advance (other than the payment of compensation, directors' fees or reimbursement of expenses in the ordinary course of business) any amount to, or sell, transfer or lease any properties or assets (real, personal or mixed, tangible or intangible) to, or enter into any agreement with, any of its officers or directors or any "affiliate" or "associate" of any of its officers or directors; (xii) form or commence the operations of any business or any corporation, partnership, joint venture, business association or other business organization or division thereof; (xiii) make any tax election or settle or compromise any tax liability involving amounts in excess of \$25,000 individually or \$200,000 in the aggregate; or (xiv) pay, discharge, settle or satisfy any claims litigation, liabilities or obligations (whether absolute, accrued, asserted or unasserted, contingent or otherwise) involving amounts in excess of \$75,000 individually or \$200,000 in the aggregate. The Company shall not amend, modify, terminate or rescind, or waive any provision of, or otherwise alter in any way the Employment Agreement of even date herewith between the Company and Stephen Williamson prior to the Effective Time without the prior written consent of Parent.

In connection with the continued operation of the Company and its Subsidiaries between the date hereof and the Closing Date, the Company will confer in good faith on a regular and frequent basis with one or more

#### A-26

representatives of Parent designated to the Company regarding operational matters and the general status of ongoing operations promptly and will notify Parent of any event or occurrence that has had or may reasonably be expected to have a Company Material Adverse Effect. The Company acknowledges that Parent does not and will not waive any rights it may have under this Agreement as a result of such consultations. The Company shall not take any action that would, or that could reasonably be expected to, result in any of the representations and warranties of the Company sets forth in this Agreement becoming untrue.

## Section 6.2. Access to Information; Confidentiality.

- (a) From the date hereof to the Effective Time, the Company shall, and shall cause the officers, directors, employees, auditors, attorneys, financial advisors, lenders and other agents (collectively, the "Representatives") of the Company to, afford the Representatives of Parent and Sub reasonable access at all reasonable times and upon reasonable notice to the officers, employees, agents, properties, offices and other facilities, books and records of the Company and its Subsidiaries, and shall furnish Parent and Sub with all financial, operating and other data and information as Parent or Sub, through its Representatives, may reasonably request. Nothing in this Section 6.2 shall require the Company to provide access to or disclose information where such access or disclosure would result in the loss of any attorney-client privilege. The parties hereto shall make appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply. The Company shall furnish to Parent and Sub financial and operating data and information for each four or five week fiscal period in the form currently prepared by the Company within seventeen (17) days following the end of each such period. Parent and Sub will remain subject to the terms of the Confidentiality Agreement.
- (b) No investigation pursuant to this Section 6.2 shall limit or modify in any way or affect any representation or warranty in this Agreement of any party hereto or any condition to the obligations of the parties hereto.
- Section 6.3. Notification of Certain Matters. The Company shall give prompt notice to Parent, and Parent shall give prompt notice to the Company, of (i) the occurrence or non-occurrence of any event the occurrence or non-occurrence of which would be reasonably likely to cause any representation or warranty made by it contained in this Agreement to be materially untrue or inaccurate so as to cause the condition set forth in paragraph 5(f)(A) of Annex I to fail to be satisfied, and (ii) any failure of the Company, Parent or Sub, as the case may be, in any material respect to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it so as to cause the condition set forth in paragraph 5(f)(B) of Annex I to fail to be satisfied. If any event or matter arises after the date of this Agreement which, if existing or occurring at the date of this Agreement, would have been required to be set forth or described in the Company Disclosure Letter or which is necessary to make any representation and warranty true and correct on the

Closing Date, then the Company shall, for informational purposes only, promptly supplement, or amend, and deliver to Parent the Company Disclosure Letter which it has delivered pursuant to this Agreement. The disclosure of any matter in accordance with the provisions of this Section 6.3 shall not in any way limit or otherwise affect the remedies available hereunder to the party receiving such disclosure (including for purposes of determining whether the conditions set forth in paragraph 5(f) of Annex I have been satisfied).

## Section 6.4. Reasonable Efforts; Further Assurances; Cooperation.

(a) Subject to the other provisions of this Agreement, each of the parties hereto will each use its commercially reasonable, good faith efforts to perform its obligations set forth in this Agreement and to take, or cause to be taken, and do, or cause to be done, all things necessary, proper or advisable under applicable Law to obtain all consents required as described in Section 4.4 of the Company Disclosure Letter and all regulatory approvals (including, but not limited to those identified in Sections 4.5 and 5.4) and to satisfy all conditions to their respective obligations under this Agreement and to cause the transactions contemplated in this Agreement to be effected as soon as practicable after the date hereof, in accordance with the terms of this Agreement and will cooperate fully with each other and their respective officers, directors, employees, agents, counsel, accountants and other designees in connection with any steps required to be taken as a part of their respective obligations under this Agreement.

#### A-27

- (b) Each of the parties hereto promptly will make their respective filings and submissions and will use its commercially reasonable, good faith efforts to take all actions necessary, proper or advisable under applicable Laws and regulations to obtain any required approval of any Governmental Entity with jurisdiction over the transactions contemplated by this Agreement. Each of the parties hereto will furnish all information required for any application or other filing to be made pursuant to the rules and regulations of any applicable Law in connection with the transactions contemplated by this Agreement. Notwithstanding the foregoing, in connection with any filing or submission required or action to be taken by Parent to consummate the Merger, (i) neither Parent nor any of its affiliates is required to become subject to any requirement or condition that it divest or "hold separate" any assets or businesses or any similar transaction or restriction, (ii) neither Parent nor any of its affiliates is required to divest or hold separate or otherwise take (or refrain from taking) or commit to take (or refrain from taking) any action that limits its freedom of action with respect to, or its ability to retain, any of the businesses, product lines or assets of Parent or any of its affiliates, and (iii) neither Parent nor any of its affiliates shall be required to litigate or otherwise contest in any judicial proceeding any adverse determination by any Governmental Entity with respect to the Merger or the other transactions contemplated by this Agreement.
- (c) The Company will use its commercially reasonable, good faith efforts to give any notices to third parties and use its commercially reasonable, good faith efforts (in consultation with Parent) to obtain any third party consents (i) necessary, proper or advisable to consummate the transactions contemplated by this Agreement, (ii) disclosed or required to be disclosed in the Company Disclosure Letter, including, without limitation, the consents described in Section 4.4 of the Company Disclosure Letter, (iii) required to avoid a breach of or default under any Material Contracts in connection with the consummation of the transactions contemplated by this Agreement or (iv) required to prevent a Company Material Adverse Effect whether prior to or after the Closing Date.
- (d) The Company and its Board of Directors shall, if any state takeover statute or similar statute is or becomes applicable to this Agreement, the Offer, the Merger, the Tender Agreements or any other transactions contemplated hereby or thereby, take all action reasonably necessary to ensure that the Offer, the Merger and the other transactions contemplated by this Agreement and the Tender Agreements may be consummated as promptly as practicable on the terms contemplated hereby and thereby and otherwise to minimize the effect of such statute or regulation on this Agreement, the Offer, the Merger, the Tender Agreements and the other transactions contemplated hereby and thereby.

## Section 6.5. Board Recommendations.

- (a) In connection with the Merger and the meeting of the Company Shareholders to consider the approval and adoption of this Agreement and the Merger (the "Shareholders Meeting"), the Board of Directors of the Company shall (i) subject to Section 6.5(b), recommend to the Company Shareholders to vote in favor of this Agreement and the Merger and use its reasonable best efforts to obtain the necessary approvals by the Company Shareholders of this Agreement and the Merger and (ii) otherwise comply with all legal requirements applicable to such meeting.
- (b) Neither the Board of Directors of the Company nor any committee thereof shall, except as expressly permitted by this Section 6.5(b), (i) withdraw, qualify or modify, or propose publicly to withdraw, qualify or modify, in any manner adverse to Parent or Sub, the approval or recommendation of such

Board of Directors or such committee of the Offer, the Merger or this Agreement or otherwise seek in any manner to abandon the Merger or this Agreement, (ii) endorse, approve, recommend or submit to the Company Shareholders, or propose publicly to endorse, approve, recommend or submit to the Company Shareholders, any transaction involving a Competing Acquisition Proposal (as defined below), or (iii) cause the Company to enter into any letter of intent, agreement in principle, acquisition agreement, memorandum of understanding or other similar agreement or understanding (each, an "Acquisition Agreement") related to or with respect to any Competing Acquisition Proposal. Notwithstanding the foregoing, if the Company has complied fully with this Section 6.5 and Section 6.9 and the Board of Directors of the Company determines in good faith, after it has received a Superior Proposal (as hereinafter defined) in compliance with Section 6.9 and after taking into account advice from independent outside legal counsel with respect to its fiduciary duties to Company

#### A - 2.8

Shareholders under applicable California Law, that such action is required for the Board of Directors of the Company to comply with its fiduciary obligations to the Company Shareholders under applicable California Law, the Board of Directors of the Company may (subject to this and the following sentences), but only prior to the consummation of the Offer, take any of the actions described in clauses (i) through (iii) of this Section 6.5(b) (a "Subsequent Adverse Determination"), but only at a time that is after the third business day following Parent's receipt of written notice advising Parent that the Board of Directors of the Company intends to make a Subsequent Adverse Determination; provided that the Board of Directors of the Company may not at any time withdraw its approval of the Offer, the Merger or this Agreement or seek to abandon the Offer, the Merger or this Agreement. Such written notice shall specify the material terms and conditions of such Superior Proposal (and include a copy thereof with all accompanying documentation), identify the person making such Superior Proposal and state that the Board of Directors of the Company intends to make a Subsequent Adverse Determination. During such three business day period, the Company shall provide a full opportunity for Parent to propose such adjustments to the terms and conditions of this Agreement as would enable the Company to proceed with its recommendation to its shareholders without a Subsequent Adverse Determination. For purposes of this Agreement, "Competing Acquisition Proposal" means any proposal from a third party with respect to a merger, consolidation, share exchange, tender offer or similar transaction involving the Company or any Subsidiary, or any purchase or other acquisition of 20% or more of the assets of the Company or any Subsidiary or any purchase or other acquisition of any equity interest in the Company or any Subsidiary. For purposes of this Agreement, a "Superior Proposal" means any bona fide proposal with respect to a merger, consolidation, share exchange, tender offer, business combination or similar transaction involving the Company or any Subsidiary, or any purchase or other acquisition of 70% or more of the assets of the Company and its Subsidiaries, taken as a whole, or any purchase or other acquisition of more than 50% of the equity interests in the Company, in each case, which (w) is fully financed without any financing condition, (x) does not contain a "right of first refusal" or "right of first offer" with respect to any proposal that Parent may make, (y) does not contain any "due diligence" condition and (z) the Board of Directors determines in good faith after taking into account advice from the independent financial advisor of the Company is more favorable to the Company Shareholders than the transactions contemplated by this Agreement from a financial point of view taking into account any proposed changes to such transactions that may be proposed by Parent in response to such proposal. Notwithstanding any other provision of this Agreement, the Company shall submit this Agreement to the Company Shareholders whether or not the Board of Directors of the Company makes a Subsequent Adverse Determination; provided that the Company, concurrently with the submission of this Agreement and the Merger may submit to the Company Shareholders other Competing Acquisition Proposals, if the Company has complied with Section 6.5 and Section 6.9 with respect to such other Competing Acquisition Proposals.

Section 6.6. Shareholder Litigation. After consummation of the Offer, the Company shall give Parent the opportunity to participate in the defense or settlement of any shareholder Litigation against the Company and its directors relating to the transactions contemplated by this Agreement or the Merger ("Shareholder Litigation"); provided, however, that no such settlement (whether made before or after the consummation of the Offer) shall be agreed to without Parent's consent which consent will not be unreasonably withheld; provided, further, that no such settlement without Parent's consent shall be deemed to constitute a breach of this Agreement unless such settlement is material or has a material impact on Parent or Sub in relation to the transactions contemplated by this Agreement.

# Section 6.7. Indemnification.

(a) It is understood and agreed that all rights to indemnification by the Company now existing in favor of each present and former director and officer of the Company or its Subsidiaries (the "Indemnified Parties") as provided in the Company Articles of Incorporation or the Company Bylaws, in each case as in effect on the date of this Agreement, or pursuant to any other agreements in effect on the date hereof, copies of which have been provided to Parent, shall survive the Merger. Except as required by applicable Law, Parent shall ensure

that the Surviving Corporation does not prior to the sixth anniversary of the Effective Time rescind or otherwise amend the provisions of Article IV of the Articles of Incorporation of the Surviving Corporation in effect as of the Effective Time with respect to the period prior to the Effective Time or Article X of the Bylaws

#### A-29

of the Surviving Corporation in effect as of the Effective Time with respect to the period prior to the Effective Time. Parent shall (i) cause the Surviving Corporation to continue in full force and effect for a period of at least six years from the Effective Time and (ii) perform, or cause the Surviving Corporation to perform, in a timely manner, the Surviving Corporation's obligations with respect thereto. Parent and Sub agree that any claims for indemnification hereunder as to which they have received written notice prior to the sixth anniversary of the Effective Time shall survive, whether or not such claims shall have been finally adjudicated or settled.

- (b) Parent shall cause the Surviving Corporation to, and the Surviving Corporation shall, maintain in effect for six years from the Effective Time, the current directors' and officers' liability insurance policies ("D&O Insurance") covered by such policies (provided that the Surviving Corporation may substitute therefor policies of at least the same coverage containing terms and conditions which are not less favorable in any material respect) with respect to matters occurring prior to the Effective Time; provided, however, that in no event shall the Surviving Corporation be required to expend pursuant to this Section 6.7(b) more than an amount per year equal to two hundred percent (200%) of current annual premiums paid by the Company for such insurance. In the event that, but for the proviso to the immediately preceding sentence, the Surviving Corporation would be required to expend more than two hundred percent (200%) of current annual premiums, the Surviving Corporation shall obtain the maximum amount of such insurance obtainable by payment of annual premiums equal to two hundred percent (200%) of current annual premiums.
- (c) If the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then, and in each such case, proper provision shall be made so that the successors and assigns of the Surviving Corporation shall assume the obligations set forth in this Section 6.7. (d) The provisions of this Section 6.7 are intended for the benefit of, and may be enforced by, each person entitled to indemnification under this Section 6.7.

Section 6.8. Public Announcements. Parent, Sub and the Company shall consult with each other before issuing any press release or otherwise making any public statements with respect to this Agreement or the Merger and shall not issue any such press release or make any such public statement prior to such consultation, except as may be required by Law or the requirements of any securities exchange or trading system. Promptly following the execution of this Agreement, the Company may issue a press release in the form agreed to by the parties.

Section 6.9. Competing Acquisition Proposals. The Company shall not, nor shall it authorize or permit any of its Subsidiaries or Representatives to, directly or indirectly, (a) solicit, initiate or encourage the submission of any Competing Acquisition Proposal or (b) participate in or encourage any discussion or negotiations regarding, or furnish to any person any non-public information with respect to, enter into any Acquisition Agreement with respect to, or take any other action to facilitate any inquiries or the making of, any proposal that constitutes, or may reasonably be expected to lead to, any Competing Acquisition Proposal; provided, however, that the foregoing shall not prohibit the Board of Directors of the Company, prior to the consummation of the Offer, from furnishing information to, or entering into or participating in discussions or negotiations with, any person or entity that makes an unsolicited Competing Acquisition Proposal which did not result from a breach of Section 6.5 or this Section 6.9 only if, and only to the extent that, (A) the Board of Directors of the Company, after taking into account advice from independent outside legal counsel and financial advisors, determines in good faith that such action is required for the Board of Directors of the Company to comply with its fiduciary obligations to the Company Shareholders under applicable California law, (B) prior to taking such action, the Company receives from such person or entity an executed agreement in reasonably customary form relating to the confidentiality of information to be provided to such person or entity and including standstill provisions no less favorable to the Company than those contained in the Confidentiality Agreement and (C) the Board of Directors of the Company concludes in good faith, after taking into account the advice of its independent financial advisor, that the Competing Acquisition Proposal is a Superior Proposal. The Company shall provide immediate oral and written notice to Parent of (a) the receipt of any such Competing Acquisition Proposal or any inquiry which could reasonably be expected to lead

such Competing Acquisition Proposal or inquiry, (c) the identity of such person or entity making any such Competing Acquisition Proposal or inquiry and (d) the Company's intention to furnish information to, or enter into discussions or negotiations with, such person or entity. The Company shall continue to keep Parent fully informed of the status and details of any such Competing Acquisition Proposal or inquiry.

Nothing contained in this Agreement shall prohibit the Company or any of the officers, directors or employees of the Company from (i) referring a third party to this Section 6.9 or (ii) complying with Rule 14(e)-2 promulgated under the Exchange Act or from making any disclosure to the Company Shareholders if, in the good faith judgment of the Board of Directors of the Company, based on advice from its outside counsel, failure to so disclose would violate applicable Law; provided, however, that neither the Company nor its Board of Directors nor any committee thereof shall, except as specifically permitted by Section 6.5(b), withdraw, qualify, or modify, or propose to withdraw, qualify or modify, its position with respect to the Offer, the Merger or this Agreement or approve or recommend, or propose to approve or recommend a Competing Acquisition Proposal.

Section 6.10. Undertakings of Parent. Parent shall perform, or cause to be performed, when due all obligations of Sub and the Surviving Corporation under this Agreement.

Section 6.11. Director Resignations. The Company shall use its commercially reasonable efforts to cause to be delivered to Parent resignations of all the directors of the Company's Subsidiaries to be effective upon the consummation of the Merger.

Section 6.12. Shareholders' Rights Agreement. The Company shall use its commercially reasonable efforts to terminate, as of immediately prior to the consummation of the Offer, the Shareholders' Rights Agreement dated as of May 22, 2000 between the Company, the Principal Shareholders and the other parties named on the signature pages thereof (the "Shareholders' Rights Agreement"), and the Company shall execute and deliver such agreements or certificates as Parent may reasonably request to further evidence the termination of the Shareholders' Rights Agreement. The Company hereby waives any rights that it may have under the Shareholders' Rights Agreement relating to the execution and delivery of the Merger Agreement and the Tender Agreements.

Section 6.13. Employee Benefits. As of the Effective Time and for a period of at least one year after the Closing Date, Parent shall, or shall cause the Surviving Corporation to, maintain compensation and benefit plans and arrangements for employees of the Company who continue employment with the Surviving Corporation or any subsidiary thereof after the Effective Time ("Continuing Employees") that, in the aggregate, are at least, in the case of compensation plans and arrangements substantially comparable in value, and in the case of benefit plans and arrangements otherwise substantially comparable, to those currently provided by the Company to the Continuing Employees as of the Effective Time (excluding any stock options or other stock-based compensation), except as required by applicable Law (including as required to preserve any favorable tax treatment afforded such benefits as of the Effective Time). Parent shall give each Continuing Employee credit for past service for any benefit plan or scheme (with the exception of any defined benefit pension plan) that may be in place for Continuing Employees within a period of one year after the Closing Date as if such service had been with Parent; provided that nothing in this sentence shall obligate Parent, the Surviving Corporation or any subsidiary thereof to establish or maintain any particular benefit plan or scheme for Continuing Employees. Parent agrees that the Surviving Corporation shall be responsible for providing all legally-mandated continuation coverage for Continuing Employees and their covered dependents who experience a loss of coverage due to a "qualifying event" (within the meaning of Section 603 of ERISA) which occurs at any time on or after the Effective Time. Nothing in this Section 6.13 is intended to create any employment obligation other than as employees at will who may be terminated with or without cause.

Section 6.14. Environmental Assessments. On and after the date of this Agreement and through the Closing Date, Parent and its agents and representatives shall have the right to conduct, at Parent's sole cost, environmental assessments (including reasonable on-site access during normal business hours and at least meeting Phase I environmental assessment requirements) ("Environmental Assessments") of the Facilities located in Dinuba, California and Saco, Maine. The Company and the Subsidiaries shall cooperate in good

A-31

faith with Parent regarding such Environmental Assessments and use their reasonable best efforts to facilitate such Environmental Assessments. Parent shall give forty-eight (48) business hours (two business days) advance notice to the Company of Parent's intention to access the Facilities so that the Company may have the opportunity to have a representative(s) present. Parent shall use its commercially reasonable efforts to minimize interference with the operations or activities of the Facilities. Upon request, Parent shall give the Company a true and accurate copy of any documentation or results of any such Environmental Assessments of the Facilities. Any such documentation or results of the

Environmental Assessments of the Facilities shall be subject to and governed by the Confidentiality Agreement. Parent acknowledges that prior to the signing of this Agreement, Environmental Assessments were conducted by Parent's agents regarding the Facilities located in Dinuba, California and Saco, Maine. Parent agrees that additional Environmental Assessments will be conducted only if Parent has an objective, reasonable basis for requesting the same based on information of which Parent becomes aware after the date hereof and that was not reasonably apparent from the reports prepared in connection with the Environmental Assessments conducted prior to the date hereof.

Section 6.15. Settlement Fee. Prior to the consummation of the Offer, the Company shall cause to be paid in full all unpaid fines, penalties and interest owed to the United States government under the settlement agreement with the United States government dated July 23, 1998 (the "Settlement Fee") and the Company shall provide to the Food and Drug Administration's San Francisco District Office not less than 10 days prior notice of the consummation of the Offer. On or prior to the expiration date of the Offer, the Purchaser shall, or shall cause Sub to, lend to the Company on commercially reasonable terms an amount equal to the amount necessary to pay the Settlement Fee.

Section 6.16. Conveyance Taxes. Parent and the Company shall cooperate in the preparation, execution and filing of all returns, questionnaires, applications or other documents regarding any real property transfer or gains, sales, use, transfer, value added stock transfer and stamp taxes, any transfer, recording, registration and other fees, and any similar Taxes which become payable in connection with the transactions contemplated hereby that are required or permitted to be filed on or before the Effective Time.

Section 6.17. Shareholder Rights Plan. Neither the Company nor any Subsidiary shall adopt any shareholder rights agreement or any similar plan or agreement which limits or impairs the ability to purchase, or become the direct or indirect beneficial owner of, shares of Company Common Stock or any other equity or debt securities of the Company or any of its Subsidiaries, other than any shareholder rights plan or shareholder rights agreement that (a) does not impair the ability of the parties to consummate the Offer or the Merger in accordance with the terms of this Agreement and (b) otherwise does not have an adverse effect on Parent or Sub or on the rights of Parent or Sub under this Agreement or any of the Tender Agreements.

Section 6.18. Actions by Company Board. Prior to the consummation of the Offer, the Board of Directors of the Company shall comply as applicable with the provisions of the SEC's no-action letter dated January 12, 1999 addressed to Skadden, Arps, Slate, Meagher and Flom LLP relating to Section 16(b) of the Exchange Act.

Section 6.19. Loans to Permit Exercise of Company Options. If and only to the extent that Parent requires any of D. Stephen C. Williamson, James R. Steichen and/or Theodore R. Leaman (the "Executives") to exercise Company Options immediately prior to the consummation of the Offer pursuant to Section 1.4 of the Executive's Tender Agreements, the Company will loan to D. Stephen C. Williamson, James R. Steichen and/or Theodore R. Leaman such funds as may be necessary to permit any of such Executives to exercise such Company Options. In the event that the Company becomes obligated to advance funds pursuant to the immediately preceding sentence and the Company fails to advance such funds to any Executive, Parent shall advance such funds to any Executive on commercially reasonable terms.

A-32

## ARTICLE VII

## CONDITIONS

Section 7.1. Conditions to the Obligation of Each Party. The respective obligations of Parent, Sub and the Company to effect the Merger are subject to the satisfaction of the following conditions, unless waived in writing by all parties:

- (a) Sub shall have previously accepted for payment and paid for Shares pursuant to the Offer, except that neither Parent, Sub nor the Company shall be entitled to invoke this condition if, in breach of its obligations under this Agreement, it shall have been the cause of the failure of Sub to purchase pursuant to the Offer Shares validly tendered and not withdrawn;
- (b) This Agreement and the Merger shall have been approved and adopted by the requisite vote of the Company Shareholders, if any, as required by the CGCL, the Company Articles of Incorporation and the Company Bylaws;
- (c) No temporary restraining order, preliminary or permanent injunction or other order issued by any court of competent jurisdiction or other legal prohibition preventing the consummation of the Merger shall be in effect, provided that the party relying on this condition shall have complied in all material respects with its obligations under Section 6.4;

(d) All actions by or in respect of or fillings with any Governmental Entity required to permit the consummation of the Offer and the Merger shall have been obtained or made (including the expiration or termination of any applicable waiting period under the HSR Act).

#### ARTICLE VIII

#### TERMINATION, AMENDMENT AND WAIVER

- Section 8.1. Termination. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after approval of matters presented in connection with the Merger by the Company Shareholders, only:
  - (a) By mutual written consent of duly authorized representatives of Parent and the Company;
  - (b) By any of Parent, Sub or the Company if any court of competent jurisdiction or other Governmental Entity shall have issued an order, decree, ruling or taken any other action permanently restraining, enjoining or otherwise prohibiting the Merger and such order, decree, ruling or other action shall have become final and nonappealable, except if the party relying on such order, decree or ruling or other action has not complied in all material respects with its obligations under Section 6.4;
  - (c) By any of Parent, Sub or the Company if (x) as a result of the failure of any of the Offer Conditions the Offer shall have terminated or expired in accordance with its terms without Sub, in accordance with this Agreement, having accepted for payment any Shares pursuant to the Offer within the time period for acceptance specified by this Agreement, or (y) Sub, in accordance with this Agreement, shall not have accepted for payment any Shares pursuant to the Offer by May 31, 2002; provided, however, that the right to terminate this Agreement under this Section 8.1(c) shall not be available to any party whose failure to fulfill any obligation under this Agreement has been the primary cause of, or resulted in, the expiration or termination of the Offer on or before such date;
  - (d) By Parent or Sub if a Competing Acquisition Proposal shall have been consummated or if the Board of Directors of the Company shall have made a Subsequent Adverse Determination;
  - (e) By the Company, if Parent or Sub shall breach in any material respect any of their respective representations, warranties or obligations hereunder and, within twenty (20) days after written notice of such breach to Parent from the Company, such breach shall not have been cured in all material respects or waived by the Company;

## A-33

- (f) By any of the Company, Parent or Sub, if this Agreement and the Merger shall fail to be approved and adopted by the Company Shareholders at a duly held Shareholders' Meeting or at any adjournment or postponement thereof; provided, that neither Parent nor Sub may terminate this Agreement under this Section 8.1(f) if the Company Common Stock it is entitled to vote (whether by ownership, proxy or otherwise) shall not have been voted in favor of this Agreement and the Merger;
- (g) By Parent or Sub, if the Company shall breach any of its representations, warranties, covenants or other obligations hereunder which would give rise to the failure of a condition set forth in paragraph 5(f) of Annex I and, within twenty (20) days after written notice of such breach to the Company from Parent, such breach shall not have been cured, or waived by Parent or Sub; or
- (h) By Parent or Sub, if, for any reason (i) the representations and warranties of the Company in Section 4.2(a) of this Agreement or (ii) the representations and warranties of any Principal Shareholder in Section 2.1 of any Tender Agreement (clause (i) and clause (ii) collectively being referred to as the "Capital Stock Representations") shall not be true and correct in all respects; provided, that neither Parent nor Sub may terminate this Agreement under this Section 8.1(h) if, after giving effect to the failure of any Capital Stock Representations to be true and correct, the number of shares of Company Common Stock owned by the Principal Shareholders which is subject to the Tender Agreements shall be equal to or greater than 54.0% of the sum of (x) the issued and outstanding shares of Company Common Stock and (y) the number of shares of Company Common Stock issuable upon the exercise of vested Company Options, and the number of shares of Company Common Stock owned by the Principal Shareholders which is subject to the voting provisions contained in the Tender Agreements shall be equal to or greater than 50.1% of the sum of (x) the issued and outstanding shares of Company Common Stock and (y) the number of shares of Company Common Stock issuable upon the exercise of vested Company Options.

- (a) In the event of the termination of this Agreement pursuant to Section 8.1 hereof, this Agreement shall forthwith be terminated and have no further effect except as specifically provided herein and, except as provided in this Section 8.2 and in Section 9.10 and Section 9.12, there shall be no liability on the part of any party hereto, provided that nothing herein shall relieve any party from liability for any willful breach hereof.
- (b) If Parent or Sub exercises its right to terminate this Agreement under Section 8.1(d), the Company shall pay to Parent upon demand \$8,000,000 (the "Termination Fee"), payable in same-day funds, to reimburse Parent for its time, expense, lost opportunity costs and other costs and damages associated with pursuing the Merger.
- (c) If within one year after termination of this Agreement (other than a termination pursuant to Section 8.1(b), Section 8.1(c) (but only if the termination pursuant to Section 8.1(c) results from a failure to satisfy the Offer Conditions set forth in paragraph 2 of Annex I) or Section 8.1(e)) the Company shall enter into any agreement relating to a Competing Acquisition Proposal with a person or entity other than Parent or Sub or a Competing Acquisition Proposal with a person or entity other than Parent or Sub shall otherwise be consummated, then immediately prior to, and as a condition of, the entering into of such agreement or the consummation of such transaction, as the case may be, the Company shall pay to Parent upon demand the Termination Fee, payable in same-day funds, to reimburse Parent for its time, expense, lost opportunity costs and other costs and damages associated with pursuing the Merger; provided that no such amount shall be payable if the Termination Fee shall have been paid in accordance with Section 8.2(b) of this Agreement.
- (d) Notwithstanding anything to the contrary set forth in this Agreement, if the Company fails promptly to pay to Parent any amounts due under this Section 8.2, the Company shall pay the costs and expenses (including reasonable legal fees and expenses) in connection with any action, including the filing of any lawsuit or other legal action, taken to collect payment, together with interest on the amount of any unpaid fee or obligation at the publicly announced prime rate of Citibank, N.A. in effect from time to time from the date such fee or obligation was required to be paid.

#### A-34

- Section 8.3. Amendments. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto; provided, however, that after approval of the Merger by the Company Shareholders, no amendment may be made without the further approval of the Company Shareholders if the effect of such amendment would be to reduce the Merger Consideration or change the form thereof, and no amendment may be made to Section 6.7 without the consent of the third party beneficiaries of such Section.
- Section 8.4. Waiver. At any time prior to the Effective Time, whether before or after the Shareholders Meeting, any party hereto, by action taken or authorized by its board of directors, may (i) extend the time for the performance of any of the covenants, obligations or other acts of any other party hereto or (ii) waive any inaccuracy of any representations or warranties or compliance with any of the agreements, covenants or conditions of any other party or with any conditions to its own obligations. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party by its duly authorized officer. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights. The waiver of any such right with respect to particular facts and other circumstances shall not be deemed a waiver with respect to any other facts and circumstances and each such right shall be deemed an ongoing right that may be asserted at any time and from time to time.

## ARTICLE IX

# GENERAL PROVISIONS

- Section 9.1. No Third Party Beneficiaries. Other than the provisions of Section 6.7 hereof, nothing in this Agreement shall confer any rights or remedies upon any person other than the parties hereto.
- Section 9.2. Entire Agreement. This Agreement, together with the Confidentiality Agreement, constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes any prior understandings, agreements, or representations by or among the parties, written or oral, with respect to the subject matter hereof. No amendment, modification or alteration of the terms or provisions of this Agreement or the Company Disclosure Letter shall be binding unless the same shall be in writing and duly executed by the parties hereto.
- Section 9.3. Succession and Assignment. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties; provided, however, that each of Parent and Sub

may freely assign its rights to another direct or indirect wholly owned subsidiary of Parent or Sub without such prior written approval but no such assignment shall relieve Parent or Sub of any of its obligations hereunder. Any purported assignment without such consent shall be void.

- Section 9.4. Counterparts. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument.
- Section 9.5. Headings. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.
- Section 9.6. Governing Law. This Agreement shall be governed by and construed in accordance with the Laws of the State of New York, without regard to principles of conflicts of Law thereof.
- Section 9.7. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the parties agree that the court making the determination of invalidity or unenforceability shall have the power to reduce the scope, duration, or area of the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with

## A-35

a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified after the expiration of the time within which the judgment may be appealed.

Section 9.8. Specific Performance. Each of the parties acknowledges and agrees that the other party would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each of the parties agrees that the other party shall be entitled to seek an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the parties and the matter, in addition to any other remedy to which it may be entitled, at law or in equity.

Section 9.9. Construction. The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

Section 9.10. Non-Survival of Representations and Warranties and Agreements. The representations, warranties and agreements in this Agreement shall terminate at the Effective Time or upon the termination of this Agreement pursuant to Section 8.1, as the case may be, except that (i) the agreements set forth in Articles II and IX and Sections 6.6 and 6.7 shall survive the Effective Time indefinitely and (ii) the agreements set forth in Sections 6.2 (with respect to confidentiality), 6.6, 6.8 and 8.2 and in Article IX shall survive the termination of this Agreement indefinitely.

## Section 9.11. Certain Definitions.

(a) For purposes of this Agreement, the terms "associate" and "affiliate" shall have the same meaning as set forth in Rule 12b-2 promulgated under the Exchange Act, and the term "person" shall mean any individual, corporation, partnership (general or limited), limited liability company, limited liability partnership, trust, joint venture, joint-stock company, syndicate, association, entity, unincorporated organization or government or any political subdivision, agency or instrumentality thereof or any other entity. For purposes of this Agreement, the phrase "Company Material Adverse Effect" shall mean, with respect to the Company, any change, event or effect that, when taken together with all other adverse changes, events or effects that have occurred, (i) is materially adverse to the business, operations, properties, condition (financial or otherwise), assets, liabilities (including, without limitation, contingent liabilities) of the Company and its Subsidiaries taken as a whole or (ii) prevents or delays beyond May 31, 2002 the consummation of the Offer or the Merger; provided, however, that a Company Material Adverse Effect shall not include (x) in and of itself failure by the Company to meet analysts' earnings forecasts or estimates or changes in the market price of the Company Common Stock in and of itself or (y) any effect resulting from any change in general economic conditions, including any change in general economic conditions due to any act of war, terrorism or threat of war or terrorism, which changes do not disproportionately affect the Company and its Subsidiaries (provided that it is

understood and agreed that the foregoing exclusion shall not exclude any change, event or effect that would otherwise be a Company Material Adverse Effect caused by any act of war or terrorism or threat of war or terrorism directed at the beverage industry which disproportionately affects the Company and its Subsidiaries as compared to the beverage industry as a whole) or (z) any effect resulting solely in and of itself from the public announcement of this Agreement or the transactions contemplated by this Agreement or the consummation of such transactions (other than any Litigation or claims the substance of which is not related to the public announcement of this Agreement or the transactions contemplated by this Agreement or the consummation of such transactions).

Subject to the provisions set forth below, for purposes of determining whether any Litigation arising from a claim that is made after the execution of this Agreement and prior to the date that is the earlier of (i) the consummation of the Offer or (ii) the termination of this Agreement is, or is reasonably likely to constitute or contribute to a "Company Material Adverse Effect," the alleged claim amount by the plaintiffs in such Litigation shall not be taken into account in making such determination, but instead the relevant factors in determining the materiality of such Litigation shall be (w) the relative merits of such litigation, (x) the

## A-36

objective probability of success of such Litigation, (y) the reasonably expected magnitude of any judgment, settlement and/or defense costs associated with such Litigation, and (z) the benefit of any insurance proceeds available to the Company to cover losses relating to such Litigation. In the event of a dispute between the Parent and Sub, on the one hand, and the Company, on the other hand, regarding whether such Litigation constitutes or contributes to or is reasonably likely to constitute or contribute to a Company Material Adverse Effect, upon the request of Parent, Sub or the Company, the parties hereto shall submit such dispute to a mutually agreed upon nationally recognized United States law firm, and the determination of such law firm regarding whether and to what extent such Litigation constitutes or contributes to or is reasonably likely to constitute or contribute to a Company Material Adverse Effect (the "Dispute Decision") shall be final and binding on the parties hereto. The expenses of such determination shall be borne equally by the Parent and the Company. The closing of the Offer shall not take place until the second business day following the rendering of the Dispute Decision, assuming the satisfaction, or waiver by Parent and Sub, of the Offer Conditions. For the avoidance of doubt, any Litigation arising in connection with those claims described in Section 4.12 of the Company Disclosure Letter shall not constitute or contribute to a Company Material Adverse Effect. Notwithstanding anything else contained herein, any Shareholder Litigation shall be excluded from Litigation subject to determination pursuant to this paragraph and shall not in and of itself be taken into account in determining whether a Company Material Adverse Effect has occurred or would be reasonably expected to occur.

(b) For purposes of this Agreement, the phrases "to the knowledge of the Company," "known to the Company," and similar formulations shall mean the actual knowledge of the Company's executive officers and other officers having primary responsibility for such matter.

Section 9.12. Fees and Expenses. Except as provided in Section 8.2, all costs and expenses incurred by the parties hereto in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses.

Section 9.13. Notices. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by telecopy or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses, or at such other address for a party as shall be specified in a notice given in accordance with this Section 9.13:

If to Parent or Sub:

The Coca-Cola Company One Coca-Cola Plaza Atlanta, Georgia 30313 Telecopier: (404) 676-8621 Attention: Chief Financial Officer

with a copy (which does not constitute notice) to:

The Coca-Cola Company One Coca-Cola Plaza Atlanta, Georgia 30313 Telecopier: (404) 676-6209 Attention: General Counsel

and with a copy to:

King & Spalding

191 Peachtree Street Atlanta, Georgia 30303 Telecopier: (404) 572-5100 Attention: C. William Baxley

A-37

If to the Company:

Odwalla, Inc. 120 Stone Pine Road Half Moon Bay, California 94019 Telecopier: (650) 712-5967 Attention: Chief Financial Officer

with a copy to:

Morrison & Foerster LLP 425 Market Street San Francisco, California 94965 Telecopier: (415) 268-7522 Attention: Robert S. Townsend

Section 9.14. Waiver of Jury Trial. EACH OF PARENT, SUB AND THE COMPANY HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM (WHETHER BASED UPON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY.

IN WITNESS WHEREOF, Parent, Sub and the Company have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

THE COCA-COLA COMPANY

By: /s/ DAVID M. TAGGART

-----

Name: David M. Taggart Title: Vice President

PERRY PHILLIP CORP.

By: /s/ PAUL ETCHELLS

-----

Name: Paul Etchells Title: President

ODWALLA, INC.

By: /s/ D. STEPHEN C. WILLIAMSON

\_\_\_\_\_

Name: D. Stephen C. Williamson Title: Chairman of the Board and Chief Executive Officer

A-38

ANNEX I

## CONDITIONS TO THE OFFER

Capitalized terms used in this Annex I shall have the meanings assigned to them in this Agreement.

Notwithstanding any other provision of the Offer, and in addition to (and not in limitation of) Sub's rights to extend and/or amend the Offer at any time in its sole discretion (subject to the provisions of this Agreement), Sub's obligations under this Agreement pursuant to the Offer shall be subject to the following conditions, and if such conditions are not satisfied, Sub may delay the acceptance for payment of or, subject to any applicable rules and regulations of the SEC, including Rule 14e-1(c) under the Exchange Act, the payment for, any tendered Shares, and may terminate or amend the Offer as to any Shares not then paid for:

- (1) at or prior to the expiration date of the Offer, the number of Shares validly tendered and not withdrawn, together with any Shares then owned by Parent or Sub, shall satisfy the Minimum Condition or, if applicable pursuant to the provisions of this Agreement, the Option Exercise Minimum Number or the Revised Minimum Number;
- (2) at or prior to the expiration date of the Offer, (a) any applicable waiting period under the HSR Act shall have expired or been terminated, and (b) all other actions by or in respect of or other filings with any Governmental Entity required to permit the consummation of the Offer and the Merger shall have been obtained or made, other than, in the

case of clause (b), such actions or filings the failure of which to make or obtain would not reasonably be expected to have a Company Material Adverse Effect:

- (3) the Shareholders' Rights Agreement shall have been terminated;
- (4) the Settlement Fee shall have been paid in full immediately prior to the consummation of the Offer, unless such failure to make payment of the Settlement Fee is caused by the failure of Parent to meet its obligation under Section 6.15 of this Agreement; and the Company shall have provided to the Food and Drug Administration's San Francisco District Office not less than 10 days prior notice of the consummation of the Offer; or
- (5) at any time after the date of this Agreement and prior to acceptance for payment of Shares pursuant to the Offer, none of the following events or conditions shall occur or exist:
  - (a) there shall be instituted, pending or threatened any action or proceeding by any Governmental Authority: (i) challenging or seeking to permanently restrain, enjoin or otherwise prohibit the Offer or the Merger; (ii) seeking to restrain or prohibit Parent's or Sub's full rights of ownership or operation of any portion of the business or assets of the Company, or to compel Parent or Sub to dispose of or hold separate all or any portion of the business or assets of the Company; (iii) seeking to impose limitations on the ability of Parent or Sub effectively to exercise full rights of ownership of the Shares acquired pursuant to the Offer and the Merger, including, without limitation, the right to vote any Shares acquired or owned by Parent or Sub on all matters properly presented to the Company's shareholders; (iv) seeking to require divestiture by Parent or Sub of any Shares; or (v) that otherwise would reasonably be expected to have a Company Material Adverse Effect;
  - (b) there shall be enacted, enforced, promulgated, in force or deemed applicable to the Offer or the Merger by any Governmental Authority any statute, rule, regulation, judgment, order or injunction (other than the application of the routine waiting period provisions of the HSR Act) that has, directly or indirectly, resulted, or is reasonably likely to, directly or indirectly, result in any of the consequences referred to in paragraph (a) above;
  - (c) an event shall have occurred that has resulted in, or would be reasonably expected to result in, a Company Material Adverse Effect;
  - (d) any court of competent jurisdiction or other Governmental Entity shall have issued an order, decree, ruling or taken any other action restraining, enjoining or otherwise prohibiting the Offer or the Merger; provided, that with respect to Shareholder Litigation, unless such order, decree,

I-1

ruling or other action shall be final and nonappealable, this condition in and of itself shall not be a ground for termination and shall only have the effect of delaying the consummation of the Offer during the effective period of such order, decree, ruling or other action;

- (e) there shall have occurred (i) any general suspension of trading in, or limitation on prices for, securities on any national securities exchange or in the over-the-counter market, (ii) the declaration of any banking moratorium or any suspension of payments in respect of banks or any material limitation (whether or not mandatory) on the extension of credit by lending institutions in the United States, (iii) the commencement of a war, material armed hostilities or other material international or national calamity directly or indirectly involving the United States that has a significant adverse effect on the functioning of the financial markets in the United States, or (iv) in the case of any of the foregoing existing at the time of execution of the Agreement, a material acceleration or worsening thereof; provided that this condition in and of itself shall not be a ground for termination and shall only have the effect of delaying the consummation of the Offer during the effective period of such declaration or suspension or other event;
- (f) (A) the representations and warranties of the Company in this Agreement shall not be true and correct in all respects as of the date of this Agreement or as of the expiration of the Offer without giving effect to any exceptions for "material" or "materiality" or "Company Material Adverse Effect", except where the failure of such representations and warranties to be true and correct, individually or in the aggregate, does not have and is not reasonably likely to have a Company Material Adverse Effect; (B) the Company shall not have performed in all material respects all covenants, agreements and obligations required to be performed by it under this Agreement; (C) the directors of the Company shall not have resigned as provided in Article

I; or (D) an officer of the Company shall not have delivered to Parent and Buyer a certificate to the effect that each of the foregoing conditions is satisfied in all respects; or

(g) this Agreement shall have been terminated in accordance with its terms or amended in accordance with its terms to provide for such termination or amendment of the Offer.

The foregoing conditions are for the sole benefit of Parent and Sub and may be asserted or waived by Parent or Sub, regardless of the circumstances giving rise to any such condition (including any action or omission by Parent or Sub), in whole or in part at any time and from time to time in their sole discretion. The failure by Parent or Sub at any time to exercise any of the foregoing rights shall not be deemed a waiver of any such right; the waiver of any such right with respect to any particular facts and circumstances shall not be deemed a waiver with respect to any other facts and circumstances; and each such right shall be deemed an ongoing right and may be asserted at any time and from time to time.

I-2

EXHIBIT A

Bain Capital Fund VI, L.P. BCIP Associates II BCIP Trust Associates II BCIP Associates II-B BCIP Trust Associates II-B BCIP Trust Associates II-C PEP Investments Pty. Limited U.S. Equity Partners, L.P. U.S. Equity Partners (Offshore), L.P. Catterton-Simon Partners III, L.P. Stephen C. Williamson James Steichen Theodore Leaman Douglas Levin Michael Carter Julie Carter Robert Carter

A-1

APPENDIX B

## FORM TENDER AGREEMENT WITH VOTING AGREEMENT

## TENDER AGREEMENT

THIS TENDER AGREEMENT (this "Agreement") dated October 29, 2001, is entered into between THE COCA-COLA COMPANY, a Delaware corporation ("Parent"), PERRY PHILLIP CORP., a California corporation and wholly owned subsidiary of Parent ("Sub"), and ("Shareholder"), with respect to the shares of common stock, no par value (the "Company Common Stock"), of Odwalla, Inc., a California corporation (the "Company"), owned by Shareholder.

# WITNESSETH:

WHEREAS, Parent, Sub and the Company have entered into an Agreement and Plan of Merger dated as of the date hereof (the "Merger Agreement") pursuant to which Sub has agreed to make a cash tender offer described therein and thereafter merge with and into the Company (the "Merger") with the result that the Company becomes a wholly owned subsidiary of Parent;

WHEREAS, as of the date hereof, Shareholder beneficially owns and has the power to vote and dispose of shares of Company Common Stock as identified on Schedule I hereto (such shares of Company Common Stock, together with any securities issued or exchanged with respect to such shares of Company Common Stock, and upon any recapitalization, reclassification, merger, consolidation, spin-off, partial or complete liquidation, stock dividend, split-up or combination of the securities of the Company or any other change in the Company's capital structure, are collectively referred to herein as the "Securities");

WHEREAS, Parent and Sub desire to enter into this Agreement in connection with their efforts to consummate the acquisition of the Company, and in consideration of Parent's and Sub's agreements herein and in the Merger Agreement, Shareholder has agreed to cooperate with Parent and Sub with respect to the acquisition of the Company by Parent and Sub upon the terms and subject to the conditions in the Merger Agreement; and

WHEREAS, capitalized terms used in this Agreement and not defined have the meaning given to such terms in the Merger Agreement.

NOW, THEREFORE, in contemplation of the foregoing and in consideration of

the mutual agreements, covenants, representations and warranties contained herein and intending to be legally bound hereby, the parties hereto agree as follows:

## 1. Certain Covenants.

1.1 Lock-Up. Subject to Section 1.4, Shareholder hereby covenants and agrees that during the term of this Agreement, Shareholder will not (a) directly or indirectly, sell, transfer, assign, pledge, hypothecate, tender, encumber or otherwise dispose of or limit its right to vote in any manner any of the Securities, or agree to do any of the foregoing, or (b) take any action which would have the effect of preventing or disabling Shareholder from performing its obligations under this Agreement. Notwithstanding the foregoing, in connection with any transfer not involving or relating to any Competing Acquisition Proposal, Shareholder may transfer any or all of the Securities as follows: (i) in the case of a Shareholder that is an entity, to any subsidiary, partner or member of Shareholder, and (ii) in the case of an individual Shareholder, to Shareholder's spouse, ancestors, descendants or any trust for any of their benefits or to a charitable trust; provided, however, that in any such case, prior to and as a condition to the effectiveness of such transfer, (x) each person or entity to which any of such Securities or any interest in any of such Securities is or may be transferred (a) shall have executed and delivered to Parent and Sub a counterpart to this Agreement pursuant to which such person or entity shall be bound by all of the terms and provisions of this Agreement, and (b) shall have agreed in writing with Parent and Sub to hold such Securities or interest in such Securities subject to all of the terms and provisions of this Agreement, and (y) this Agreement shall be the legal, valid

B-1

and binding agreement of such person, enforceable against such person in accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created by this Agreement is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles.

1.2 No Solicitation. During the term of this Agreement, neither the Shareholder nor any director, officer, agent, representative, employee, affiliate or associate (collectively, "Representatives") of Shareholder shall, directly or indirectly, (a) solicit, initiate or encourage the submission of any Competing Acquisition Proposal (as defined in the Merger Agreement) or any other sale, transfer, pledge or other disposition or conversion of any of the Securities or (b) participate in or encourage any discussion or negotiations regarding, or furnish to any person any non-public information with respect to, enter into any agreement with respect to, or take any other action to facilitate any inquiries or the making of any proposal that constitutes, or may reasonably be expected to lead to, any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Securities, in any case, from, to or with any person other than Parent or Sub. Shareholder will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any such other parties conducted heretofore with respect to any of the foregoing. Shareholder will notify Parent immediately if any party contacts the Shareholder following the date hereof (other than Parent and Sub) concerning any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Securities.

## 1.3 Grant of Proxy; Voting Agreement.

(a) The Shareholder has revoked or terminated any proxies, voting agreements or similar arrangements previously given or entered into with respect to the Securities and hereby irrevocably appoints Parent as proxy for Shareholder to vote the Securities for Shareholder and in Shareholder's name, place and stead, at any annual, special or other meeting or action of the shareholders of the Company, as applicable, or at any adjournment thereof or pursuant to any consent of the shareholders of the Company, in lieu of a meeting or otherwise, whether before or after the closing of the Offer (as defined in the Merger Agreement), in the following manner: (i) for the adoption and approval of the Merger Agreement and the Merger and (ii) in any manner as Parent, in its sole discretion, may see fit with respect to any extraordinary corporate transaction (other than the Merger), such as a merger, consolidation, business combination, tender or exchange offer, reorganization, recapitalization, liquidation, sale or transfer of a material amount of the assets or securities of the Company or any of its subsidiaries (other than pursuant to the Merger) or any other change of control involving the Company or any of its subsidiaries, including, but not limited to, any Competing Acquisition Proposal. THE AUTHORITY GRANTED UNDER THE IRREVOCABLE PROXY SHALL NOT EXPIRE ELEVEN (11) MONTHS AFTER ITS DATE BUT SHALL BE IRREVOCABLE UNTIL THE TERMINATION DATE (AS HEREINAFTER DEFINED) AND DEEMED TO BE COUPLED WITH AN INTEREST SUFFICIENT IN LAW AS REQUIRED BY SECTION 705 OF THE CALIFORNIA GENERAL CORPORATION LAW. The parties acknowledge and agree that neither Parent, nor Parent's successors, assigns, subsidiaries, divisions, employees, officers, directors, shareholders, agents and affiliates shall owe any duty to, whether in law or otherwise, or incur any liability of any kind whatsoever, including

without limitation, with respect to any and all claims, losses, demands, causes of action, costs, expenses (including reasonable attorney's fees) and compensation of any kind or nature whatsoever to the Shareholder in connection with or as a result of any voting (or refrain from voting) by Parent of the Securities subject to the irrevocable proxy hereby granted to Parent at any annual, special or other meeting or action or the execution of any consent of the shareholders of the Company. The parties acknowledge that, pursuant to the authority hereby granted under the irrevocable proxy, Parent may vote the Securities in furtherance of its own interests, and Parent is not acting as a fiduciary for the Shareholder.

(b) Notwithstanding the foregoing grant to Parent of the irrevocable proxy, if Parent elects not to exercise its rights to vote the Securities pursuant to the irrevocable proxy, Shareholder agrees to vote the Securities during the term of this Agreement (i) in favor of or give its consent to, as applicable, a proposal to adopt and approve the Merger Agreement and the Merger as described in clause (i) of

Section 1.3(a), or (ii) in the manner directed by Parent if the issue on which Shareholder is requested to vote is a matter described in clause (ii) of Section 1.3(a), in each case at any annual, special or other meeting or action of the shareholders of the Company, in lieu of a meeting or otherwise.

- (c) This irrevocable proxy shall not be terminated by any act of the Shareholder or by operation of law, whether by the death or incapacity of the Shareholder or by the occurrence of any other event or events (including, without limiting the foregoing, the termination of any trust or estate for which Shareholder is acting as a fiduciary or fiduciaries or the dissolution or liquidation of any corporation or partnership). If between the execution hereof and the Termination Date, Shareholder should die or become incapacitated, or if any trust or estate holding the Securities should be terminated, or if any corporation or partnership holding the Securities should be dissolved or liquidated, or if any other such similar event or events shall occur before the Termination Date, certificates representing the Securities shall be delivered by or on behalf of Shareholder in accordance with the terms and conditions of the Merger Agreement and this Agreement, and actions taken by the Parent hereunder shall be as valid as if such death, incapacity, termination, dissolution, liquidation or other similar event or events had not occurred, regardless of whether or not the Parent has received notice of such death, incapacity, termination, dissolution, liquidation or other event.
- (d) Nothing in this Agreement shall give Parent or Sub the power to vote for the election of directors.
- 1.4 Tender of Securities. Shareholder agrees to tender the Securities to Sub in the Offer as soon as practicable following the commencement of the Offer, and in any event not later ten (10) business days following the commencement of the Offer and Shareholder shall not withdraw any Securities so tendered unless the Offer is terminated or has expired. Subject to the terms and conditions of the Offer and the Merger Agreement, Sub hereby agrees to purchase the shares of Company Common Stock so tendered at a price per share equal to \$15.25 or any higher price that may be paid in the Offer; provided, however, that Sub's obligations to accept for payment and pay for the Securities in the Offer is subject to all the terms and conditions of the Offer set forth in the Merger Agreement and Annex I thereto.
- 1.5 Public Announcement. Shareholder shall consult with Parent before issuing any press releases or otherwise making any public statements with respect to the transactions contemplated herein and shall not issue any such press release or make any such public statement without the approval of Parent, except as may be required by law.
- 1.6 Disclosure. Shareholder hereby authorizes Parent and Sub to publish and disclose in any announcement or disclosure required by the Securities and Exchange Commission (the "SEC") or the New York Stock Exchange (the "NYSE") or any other national securities exchange and in the Offer Documents and, if necessary, the Proxy Statement (each as defined in the Merger Agreement), (including all documents and schedules filed with the SEC in connection with either of the foregoing), its identity and ownership of the Securities and the nature of its commitments, arrangements and understandings under this Agreement. Parent and Sub hereby authorize Shareholder to make such disclosure or filings as may be required by the SEC or the NYSE or any other national securities exchange.

# 1.7 Stop Transfer Instruction; Legend.

(a) Promptly following the date hereof, Shareholder and Sub shall deliver joint written instructions to the Company and to the Company's transfer agent stating that the Securities may not be sold, transferred, pledged, assigned, hypothecated, tendered or otherwise disposed of in any manner without the prior written consent of Sub or except in accordance with the terms and conditions of this Agreement.

(b) Promptly following the date hereof, Shareholder shall cause a legend to be placed on the certificates (to the extent the Securities are certificated) representing the Securities as set forth below:

"The Securities represented by this certificate are subject to restrictions on transfer and may not be sold, transferred, pledged, assigned, hypothecated, tendered or otherwise disposed of except in

B-3

accordance with and subject to the terms and conditions of a Tender Agreement dated October 29, 2001, between the registered holder hereof and The Coca-Cola Company."

The parties hereto agree that the legend set forth above shall be removed only upon delivery to the Company's transfer agent of written notice signed by Sub (which notice shall not be unreasonably withheld or delayed) that this Agreement has terminated and the restrictions set forth in the legend above are of no further force and effect.

- 2. Representations and Warranties of Shareholder. Shareholder hereby represents and warrants to Parent and Sub, as of the date hereof and as of the date Sub purchases shares of Company Common Stock pursuant to the Offer, that:
- 2.1 Ownership. Shareholder has good and marketable title to, and is the sole legal and beneficial owner of the Securities, in each case free and clear of all liabilities, claims, liens, options, proxies, charges, participations and encumbrances of any kind or character whatsoever (collectively, "Liens"). At the time Sub purchases shares of Company Common Stock pursuant to the Offer, Shareholder will transfer and convey to Parent or its designee good and marketable title to the shares of Company Common Stock included in the Securities, free and clear of all Liens created by or arising through Shareholder.
- 2.2 Authorization. Shareholder has all requisite power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby and has sole voting power and sole power of disposition, with respect to the Securities with no restrictions on its voting rights or rights of disposition pertaining thereto. Shareholder has duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of Shareholder, enforceable against Shareholder in accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created hereby is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles. No state anti-takeover or similar statute is applicable to Parent, Sub, the Company or the Surviving Corporation in connection with the Merger, the Merger Agreement or this Agreement or any of the transactions contemplated hereby or thereby.
- 2.3 No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) require the Shareholder to file or register with, or obtain any material permit, authorization, consent or approval of, any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity, or (b) violate, or cause a breach of or default under, any contract, agreement or understanding, any statute or law, or any judgment, decree, order, regulation or rule of any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity or any arbitration award binding upon the Shareholder, except for such violations, breaches or defaults which are not reasonably likely to have a material adverse effect on the Shareholder's ability to satisfy its obligations under this Agreement. No proceedings are pending which, if adversely determined, will have a material adverse effect on any ability to vote or dispose of any of the Securities. The Shareholder has not previously assigned or sold any of the Securities to any third party.
- 2.4 Shareholder Has Adequate Information. Shareholder is a sophisticated seller with respect to the Securities and has adequate information concerning the business and financial condition of the Company to make an informed decision regarding the sale of the Securities and has independently and without reliance upon either Sub or Parent and based on such information as Shareholder has deemed appropriate, made its own analysis and decision to enter into this Agreement. Shareholder acknowledges that neither Sub nor Parent has made and neither makes any representation or warranty, whether express or implied, of any kind or character except as expressly set forth in this Agreement. Shareholder acknowledges that the agreements contained herein with respect to the Securities by Shareholder is irrevocable, and that Shareholder shall have no recourse to the Securities or Parent, except with respect to breaches of representations, warranties, covenants and agreements expressly set forth in this Agreement.

non-public information concerning the Securities and the Company which is not known to Shareholder and which may be material to Shareholder's decision to sell the Securities ("Parent's Excluded Information"), (b) Shareholder has requested not to receive Parent's Excluded Information and has determined to sell the Securities notwithstanding its lack of knowledge of Parent's Excluded Information, and (c) Parent shall have no liability or obligation to Shareholder in connection with, and Shareholder hereby waives and releases Parent from, any claims which Shareholder or its successors and assigns may have against Parent (whether pursuant to applicable securities, laws or otherwise) with respect to the non-disclosure of Parent's Excluded Information; provided, however, nothing contained in this Section 2.5 shall limit Shareholder's right to rely upon the express representations and warranties made by Parent in this Agreement, or Shareholder's remedies in respect of breaches of any such representations and warranties.

- 2.6 No Setoff. The Shareholder has no liability or obligation related to or in connection with the Securities other than the obligations to Parent and Sub as set forth in this Agreement. There are no legal or equitable defenses or counterclaims that have been or may be asserted by or on behalf of the Company, as applicable, to reduce the amount of the Securities or affect the validity or enforceability of the Securities.
- 2.7 No Amounts Payable to Shareholder. Other than with respect to (a) amounts payable to the director(s) of the Company appointed by Shareholder in such person's capacity as a director, including all director fees and reasonable travel expenses to attend Board meetings that have not been reimbursed, and (b) reasonable fees and expenses of legal counsel in connection with the negotiation and execution of this Agreement, the Merger Agreement and the other Tender Agreements which shall be reimbursed by the Company to Shareholder up to an amount in the aggregate for all Principal Shareholders not to exceed \$100,000, there are no amounts due or payable by the Company or any Subsidiary to the Shareholder or any of its affiliates or associates in connection with the transactions contemplated by the Merger Agreement or this Agreement or otherwise.
- 3. Representations and Warranties of Parent and Sub. Parent and Sub hereby represent and warrant to Shareholder, as of the date hereof that:
- 3.1 Authorization. Parent and Sub have all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. Parent and Sub have duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of each of Parent and Sub, enforceable against each of Parent and Sub in accordance with its terms, subject to the qualification however, that enforcement of the rights and remedies created hereby is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles.
- 3.2 No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) require Parent to file or register with, or obtain any material permit, authorization, consent or approval of, any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity, or (b) violate, or cause a breach of or default under, any contract, agreement or understanding, any statute or law, or any judgment, decree, order, regulation or rule of any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity or any arbitration award binding upon Parent or Sub, except for such violations, breaches or defaults which are not reasonably likely to have a material adverse effect on each of Parent or Sub's ability to satisfy its obligations under this Agreement.
- 4. Survival of Representations and Warranties. The respective representations and warranties of Shareholder and Parent contained herein shall not be deemed waived or otherwise affected by any investigation made by the other party hereto, and each representation and warranty contained herein shall survive the closing of the transactions contemplated hereby until the expiration of the applicable statute of limitations, including extensions thereof.

B-5

- 5. No Control. Nothing contained in this Agreement shall give Parent or Sub the right to control or direct the Company or the Company's operations.
- 6. Specific Performance. Shareholder acknowledges that Sub and Parent will be irreparably harmed and that there will be no adequate remedy at law for a violation of any of the covenants or agreements of Shareholder which are contained in this Agreement. It is accordingly agreed that, in addition to any other remedies which may be available to Sub and Parent upon the breach by Shareholder of such covenants and agreements, Sub and Parent shall have the right to obtain injunctive relief to restrain any breach or threatened breach of such covenants or agreements or otherwise to obtain specific performance of any of such covenants or agreements.

7. Termination of Shareholders' Rights Agreement. Shareholder, the Company and the other persons named on the signature pages thereto are parties to that certain Shareholders' Rights Agreement dated as of May 2, 2000 (the "Shareholders' Rights Agreement"). Shareholder acknowledges and agrees that, effective as of the date Sub purchases shares of Company Common Stock pursuant to the Offer, the Shareholders' Rights Agreement shall be terminated in accordance with the terms thereof and shall have no further force or effect, and Shareholder agrees to execute and deliver such agreements or certificates as Parent may reasonably request to further evidence the termination of the Shareholders' Rights Agreement. Shareholder hereby waives any rights that it may have under the Shareholders' Rights Agreement relating to the execution and delivery of this Agreement, the Merger Agreement and the other Tender Agreements and the transactions contemplated hereby and thereby.

#### 8. Miscellaneous.

- 8.1 Term. This Agreement shall terminate upon the earlier of (i) the consummation of the Merger or the termination of the Merger Agreement pursuant to Section 8.1(a), (b), (c), (d) or (e) thereof, or (ii) six (6) months after the termination of the Merger Agreement pursuant to Section 8.1(f), (g) or (h) (the "Termination Date"). At the Termination Date, this Agreement shall thereupon become void and be of no further force and effect, provided that nothing herein shall relieve any party from liability hereof for breaches of this Agreement prior to the Termination Date.
- 8.2 Fiduciary Duties. Notwithstanding anything in this Agreement to the contrary: (a) Shareholder makes no agreement or understanding herein in any capacity other than in Shareholder's capacity as a record holder and beneficial owner of Securities, (b) nothing herein shall be construed to limit or affect any action or inaction by any officer, partner, member or employee, as applicable, of Shareholder serving on the Company's Board of Directors acting in such person's capacity as a director or fiduciary of the Company, and (c) Shareholder shall have no liability to Parent or any of its affiliates under this Agreement or otherwise as a result of any action or inaction by any officer, partner, member or employee, as applicable, of Shareholder serving on the Company's Board of Directors acting in such person's capacity as a director or fiduciary of the Company.
- 8.3 Expenses. Subject to Section 2.7, each of the parties hereto shall pay its own expenses incurred in connection with this Agreement. Each of the parties hereto warrants and covenants to the others that it will bear all claims for brokerage fees attributable to action taken by it.
- 8.4 Binding Effect. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective representatives and permitted successors and assigns.
- 8.5 Entire Agreement. This Agreement contains the entire understanding of the parties and supersedes all prior agreements and understandings between the parties with respect to its subject matter. This Agreement may be amended only by a written instrument duly executed by the parties hereto.
- 8.6 Headings. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Time is of the essence with respect to all provisions of this Agreement.
- $8.7\,$  Assignment. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party may assign either this Agreement or

any of its rights, interests, or obligations hereunder without the prior written approval of the other parties; provided, however, that each of Parent and Sub may freely assign its rights to another direct or indirect wholly owned subsidiary of Parent or Sub without such prior written approval but no such assignment shall relieve Parent or Sub of any of its obligations hereunder. Any purported assignment without such consent shall be void.

- $8.8\,$  Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be an original, but each of which together shall constitute one and the same Agreement.
- 8.9 Notices. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given if so given) by delivery, telegram or telecopy, or by mail (registered or certified mail, postage prepaid, return receipt requested) or by any national courier service, provided that any notice delivered as herein provided shall also be delivered by telecopy at the time of such delivery. All communications hereunder shall be delivered to the respective parties at the following addresses (or at such other address for a party as shall be specified by like notice, provided that notices of a change of address shall be effective only upon receipt thereof):

	<c> If to Parent or Sub:</c>	<c> The Coca-Cola Company One Coca-Cola Plaza Atlanta, Georgia 30313 Attention: Chief Financial Officer Telecopy: (404) 676-6209</c>
	with a copy to:	The Coca-Cola Company One Coca-Cola Plaza Atlanta, Georgia 30313 Attention: General Counsel Telecopy: (404) 676-6209 and
	with a copy to:	King & Spalding 191 Peachtree Street Atlanta, Georgia 30303-1763 Attention: C. William Baxley
(b)	If to Shareholder:	Telecopy: (404) 572-5100
		Attention:
		Telecopy:

		8.10 Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without regard to its principles of conflicts of laws.			
8.11 Enforceability. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.					
8.12 Further Assurances. From time to time at or after the date Sub purchases shares of Company Common Stock pursuant to the Offer, at Parent's request and without further consideration, Shareholder shall execute and deliver to Parent such documents and take such action as Parent may reasonably request in order to consummate more effectively the transactions contemplated hereby and to vest in Parent good, valid and marketable title to the Securities, including, but not limited to, using its best efforts to cause the appropriate transfer agent or registrar to transfer of record the Securities.					
8.13 Remedies Not Exclusive. All rights, powers and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity will be cumulative and not alternative, and the exercise					
		в-7			
		will not preclude the simultaneous or later ht, power or remedy by such party.			
8.14 Waiver of Jury Trial. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.					
		B-8			
IN WITNESS WHEREOF, Parent, Sub and Shareholder have caused this Agreement to be duly executed as of the day and year first above written.					
		THE COCA-COLA COMPANY			
		By:			
		Name:			
		Title:			
		PERRY PHILLIP CORP.			
SHAREHOLDER:

Title:

-----

\_\_\_\_\_

By:
Name:
Title:

B-9

APPENDIX C

#### FORM TENDER AGREEMENT WITHOUT VOTING AGREEMENT

#### TENDER AGREEMENT

THIS TENDER AGREEMENT (this "Agreement") dated October 29, 2001, is entered into between THE COCA-COLA COMPANY, a Delaware corporation ("Parent"), PERRY PHILLIP CORP., a California corporation and wholly owned subsidiary of Parent ("Sub"), and ("Shareholder"), with respect to the shares of common stock, no par value (the "Company Common Stock"), of Odwalla, Inc., a California corporation (the "Company"), owned by Shareholder.

#### WITNESSETH:

WHEREAS, Parent, Sub and the Company have entered into an Agreement and Plan of Merger dated as of the date hereof (the "Merger Agreement") pursuant to which Sub has agreed to make a cash tender offer described therein and thereafter merge with and into the Company (the "Merger") with the result that the Company becomes a wholly owned subsidiary of Parent;

WHEREAS, as of the date hereof, Shareholder beneficially owns and has the power to vote and dispose of shares of Company Common Stock as identified on Schedule I hereto (such shares of Company Common Stock, together with any securities issued or exchanged with respect to such shares of Company Common Stock, and upon any recapitalization, reclassification, merger, consolidation, spin-off, partial or complete liquidation, stock dividend, split-up or combination of the securities of the Company or any other change in the Company's capital structure, are collectively referred to herein as the "Securities");

WHEREAS, Parent and Sub desire to enter into this Agreement in connection with their efforts to consummate the acquisition of the Company, and in consideration of Parent's and Sub's agreements herein and in the Merger Agreement, Shareholder has agreed to cooperate with Parent and Sub with respect to the acquisition of the Company by Parent and Sub upon the terms and subject to the conditions in the Merger Agreement; and

WHEREAS, capitalized terms used in this Agreement and not defined have the meaning given to such terms in the Merger Agreement.

NOW, THEREFORE, in contemplation of the foregoing and in consideration of the mutual agreements, covenants, representations and warranties contained herein and intending to be legally bound hereby, the parties hereto agree as follows:

# 1. Certain Covenants.

1.1 Lock-Up. Subject to Section 1.4, Shareholder hereby covenants and agrees that during the term of this Agreement, Shareholder will not (a) directly or indirectly, sell, transfer, assign, pledge, hypothecate, tender, encumber or otherwise dispose of or limit its right to vote in any manner any of the Securities, or agree to do any of the foregoing, or (b) take any action which would have the effect of preventing or disabling Shareholder from performing its obligations under this Agreement. Notwithstanding the foregoing, in connection with any transfer not involving or relating to any Competing Acquisition Proposal, Shareholder may transfer any or all of the Securities as follows: (i) in the case of a Shareholder that is an entity, to any subsidiary, partner or member of Shareholder, and (ii) in the case of an individual Shareholder, to Shareholder's spouse, ancestors, descendants or any trust for any of their benefits or to a charitable trust; provided, however, that in any such case, prior to and as a condition to the effectiveness of such transfer, (x) each person or entity to which any of such Securities or any interest in any of such Securities is or may be transferred (a) shall have executed and delivered to Parent and Sub a counterpart to this Agreement pursuant to which such person or entity shall be bound by all of the terms and provisions of this Agreement, and (b) shall have agreed in writing with Parent and Sub to hold such Securities or interest in such Securities subject to all of the terms and provisions of this Agreement, and (y) this Agreement shall be the legal, valid

accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created by this Agreement is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles.

 $1.2\,\,$  No Solicitation. During the term of this Agreement, neither the Shareholder nor any director, officer, agent, representative, employee, affiliate or associate (collectively, "Representatives") of Shareholder shall, directly or indirectly, (a) solicit, initiate or encourage the submission of any Competing Acquisition Proposal (as defined in the Merger Agreement) or any other sale, transfer, pledge or other disposition or conversion of any of the Securities or (b) participate in or encourage any discussion or negotiations regarding, or furnish to any person any non-public information with respect to, enter into any agreement with respect to, or take any other action to facilitate any inquiries or the making of any proposal that constitutes, or may reasonably be expected to lead to, any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Securities, in any case, from, to or with any person other than Parent or Sub. Shareholder will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any such other parties conducted heretofore with respect to any of the foregoing. Shareholder will notify Parent immediately if any party contacts the Shareholder following the date hereof (other than Parent and Sub) concerning any Competing Acquisition Proposal or any other sale, transfer, pledge or other disposition or conversion of any of the Securities.

## 1.3 [Intentionally Omitted]

- 1.4 Tender of Securities. Shareholder agrees to tender the Securities to Sub in the Offer as soon as practicable following the commencement of the Offer, and in any event not later ten (10) business days following the commencement of the Offer and Shareholder shall not withdraw any Securities so tendered unless the Offer is terminated or has expired. Subject to the terms and conditions of the Offer and the Merger Agreement, Sub hereby agrees to purchase the shares of Company Common Stock so tendered at a price per share equal to \$15.25 or any higher price that may be paid in the Offer; provided, however, that Sub's obligations to accept for payment and pay for the Securities in the Offer is subject to all the terms and conditions of the Offer set forth in the Merger Agreement and Annex I thereto.
- 1.5 Public Announcement. Shareholder shall consult with Parent before issuing any press releases or otherwise making any public statements with respect to the transactions contemplated herein and shall not issue any such press release or make any such public statement without the approval of Parent, except as may be required by law.
- 1.6 Disclosure. Shareholder hereby authorizes Parent and Sub to publish and disclose in any announcement or disclosure required by the Securities and Exchange Commission (the "SEC") or the New York Stock Exchange (the "NYSE") or any other national securities exchange and in the Offer Documents and, if necessary, the Proxy Statement (each as defined in the Merger Agreement), (including all documents and schedules filed with the SEC in connection with either of the foregoing), its identity and ownership of the Securities and the nature of its commitments, arrangements and understandings under this Agreement. Parent and Sub hereby authorize Shareholder to make such disclosure or filings as may be required by the SEC or the NYSE or any other national securities exchange.

# 1.7 Stop Transfer Instruction; Legend.

- (a) Promptly following the date hereof, Shareholder and Sub shall deliver joint written instructions to the Company and to the Company's transfer agent stating that the Securities may not be sold, transferred, pledged, assigned, hypothecated, tendered or otherwise disposed of in any manner without the prior written consent of Sub or except in accordance with the terms and conditions of this Agreement.
- (b) Promptly following the date hereof, Shareholder shall cause a legend to be placed on the certificates (to the extent the Securities are certificated) representing the Securities as set forth below:

"The Securities represented by this certificate are subject to restrictions on transfer and may not be sold, transferred, pledged, assigned, hypothecated, tendered or otherwise disposed of except in

C-2

accordance with and subject to the terms and conditions of a Tender Agreement dated October 29, 2001, between the registered holder hereof and The Coca-Cola Company."

The parties hereto agree that the legend set forth above shall be removed only upon delivery to the Company's transfer agent of written notice signed by Sub (which notice shall not be unreasonably withheld or delayed) that this Agreement has terminated and the restrictions set forth in the legend

above are of no further force and effect.

- 2. Representations and Warranties of Shareholder. Shareholder hereby represents and warrants to Parent and Sub, as of the date hereof and as of the date Sub purchases shares of Company Common Stock pursuant to the Offer, that:
- 2.1 Ownership. Shareholder has good and marketable title to, and is the sole legal and beneficial owner of the Securities, in each case free and clear of all liabilities, claims, liens, options, proxies, charges, participations and encumbrances of any kind or character whatsoever (collectively, "Liens"). At the time Sub purchases shares of Company Common Stock pursuant to the Offer, Shareholder will transfer and convey to Parent or its designee good and marketable title to the shares of Company Common Stock included in the Securities, free and clear of all Liens created by or arising through Shareholder.
- 2.2 Authorization. Shareholder has all requisite power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby and has sole voting power and sole power of disposition, with respect to the Securities with no restrictions on its voting rights or rights of disposition pertaining thereto. Shareholder has duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of Shareholder, enforceable against Shareholder in accordance with its terms, subject to the qualification, however, that enforcement of the rights and remedies created hereby is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles. No state anti-takeover or similar statute is applicable to Parent, Sub, the Company or the Surviving Corporation in connection with the Merger, the Merger Agreement or this Agreement or any of the transactions contemplated hereby or thereby.
- $2.3\,$  No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) require the Shareholder to file or register with, or obtain any material permit, authorization, consent or approval of, any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity, or (b) violate, or cause a breach of or default under, any contract, agreement or understanding, any statute or law, or any judgment, decree, order, regulation or rule of any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity or any arbitration award binding upon the Shareholder, except for such violations, breaches or defaults which are not reasonably likely to have a material adverse effect on the Shareholder's ability to satisfy its obligations under this Agreement. No proceedings are pending which, if adversely determined, will have a material adverse effect on any ability to vote or dispose of any of the Securities. The Shareholder has not previously assigned or sold any of the Securities to any third party.
- 2.4 Shareholder Has Adequate Information. Shareholder is a sophisticated seller with respect to the Securities and has adequate information concerning the business and financial condition of the Company to make an informed decision regarding the sale of the Securities and has independently and without reliance upon either Sub or Parent and based on such information as Shareholder has deemed appropriate, made its own analysis and decision to enter into this Agreement. Shareholder acknowledges that neither Sub nor Parent has made and neither makes any representation or warranty, whether express or implied, of any kind or character except as expressly set forth in this Agreement. Shareholder acknowledges that the agreements contained herein with respect to the Securities by Shareholder is irrevocable, and that Shareholder shall have no recourse to the Securities or Parent, except with respect to breaches of representations, warranties, covenants and agreements expressly set forth in this Agreement.

C-3

- 2.5 Parent's Excluded Information. Shareholder acknowledges and confirms that (a) Sub or Parent may possess or hereafter come into possession of certain non-public information concerning the Securities and the Company which is not known to Shareholder and which may be material to Shareholder's decision to sell the Securities ("Parent's Excluded Information"), (b) Shareholder has requested not to receive Parent's Excluded Information and has determined to sell the Securities notwithstanding its lack of knowledge of Parent's Excluded Information, and (c) Parent shall have no liability or obligation to Shareholder in connection with, and Shareholder hereby waives and releases Parent from, any claims which Shareholder or its successors and assigns may have against Parent (whether pursuant to applicable securities, laws or otherwise) with respect to the non-disclosure of Parent's Excluded Information; provided, however, nothing contained in this Section 2.5 shall limit Shareholder's right to rely upon the express representations and warranties made by Parent in this Agreement, or Shareholder's remedies in respect of breaches of any such representations and warranties.
- 2.6 No Setoff. The Shareholder has no liability or obligation related to or in connection with the Securities other than the obligations to Parent and Sub as set forth in this Agreement. There are no legal or equitable defenses or counterclaims that have been or may be asserted by or on behalf of the Company,

as applicable, to reduce the amount of the Securities or affect the validity or enforceability of the Securities.

- 2.7 No Amounts Payable to Shareholder. Other than with respect to (a) amounts payable to the director(s) of the Company appointed by Shareholder in such person's capacity as a director, including all director fees and reasonable travel expenses to attend Board meetings that have not been reimbursed, and (b) reasonable fees and expenses of legal counsel in connection with the negotiation and execution of this Agreement, the Merger Agreement and the other Tender Agreements which shall be reimbursed by the Company to Shareholder up to an amount in the aggregate for all Principal Shareholders not to exceed \$100,000, there are no amounts due or payable by the Company or any Subsidiary to the Shareholder or any of its affiliates or associates in connection with the transactions contemplated by the Merger Agreement or this Agreement or otherwise.
- 3. Representations and Warranties of Parent and Sub. Parent and Sub hereby represent and warrant to Shareholder, as of the date hereof that:
- 3.1 Authorization. Parent and Sub have all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. Parent and Sub have duly executed and delivered this Agreement and this Agreement is a legal, valid and binding agreement of each of Parent and Sub, enforceable against each of Parent and Sub in accordance with its terms, subject to the qualification however, that enforcement of the rights and remedies created hereby is subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general application related to or affecting creditors' rights and to general equity principles.
- 3.2 No Violation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (a) require Parent to file or register with, or obtain any material permit, authorization, consent or approval of, any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity, or (b) violate, or cause a breach of or default under, any contract, agreement or understanding, any statute or law, or any judgment, decree, order, regulation or rule of any governmental agency, authority, administrative or regulatory body, court or other tribunal, foreign or domestic, or any other entity or any arbitration award binding upon Parent or Sub, except for such violations, breaches or defaults which are not reasonably likely to have a material adverse effect on each of Parent or Sub's ability to satisfy its obligations under this Agreement.
- 4. Survival of Representations and Warranties. The respective representations and warranties of Shareholder and Parent contained herein shall not be deemed waived or otherwise affected by any investigation made by the other party hereto, and each representation and warranty contained herein shall survive the closing of the transactions contemplated hereby until the expiration of the applicable statute of limitations, including extensions thereof.

C-4

- 5. No Control. Nothing contained in this Agreement shall give Parent or Sub the right to control or direct the Company or the Company's operations.
- 6. Specific Performance. Shareholder acknowledges that Sub and Parent will be irreparably harmed and that there will be no adequate remedy at law for a violation of any of the covenants or agreements of Shareholder which are contained in this Agreement. It is accordingly agreed that, in addition to any other remedies which may be available to Sub and Parent upon the breach by Shareholder of such covenants and agreements, Sub and Parent shall have the right to obtain injunctive relief to restrain any breach or threatened breach of such covenants or agreements or otherwise to obtain specific performance of any of such covenants or agreements.
- 7. Termination of Shareholders' Rights Agreement. Shareholder, the Company and the other persons named on the signature pages thereto are parties to that certain Shareholders' Rights Agreement dated as of May 2, 2000 (the "Shareholders' Rights Agreement"). Shareholder acknowledges and agrees that, effective as of the date Sub purchases shares of Company Common Stock pursuant to the Offer, the Shareholders' Rights Agreement shall be terminated in accordance with the terms thereof and shall have no further force or effect, and Shareholder agrees to execute and deliver such agreements or certificates as Parent may reasonably request to further evidence the termination of the Shareholders' Rights Agreement. Shareholder hereby waives any rights that it may have under the Shareholders' Rights Agreement relating to the execution and delivery of this Agreement, the Merger Agreement and the other Tender Agreements and the transactions contemplated hereby and thereby.

# 8. Miscellaneous.

 $8.1\,$  Term. This Agreement shall terminate upon the earlier of (i) the consummation of the Merger or the termination of the Merger Agreement pursuant to Section 8.1(a), (b), (c), (d) or (e) thereof, or (ii) six (6) months after

the termination of the Merger Agreement pursuant to Section 8.1(f), (g) or (h) (the "Termination Date"). At the Termination Date, this Agreement shall thereupon become void and be of no further force and effect, provided that nothing herein shall relieve any party from liability hereof for breaches of this Agreement prior to the Termination Date.

- 8.2 Fiduciary Duties. Notwithstanding anything in this Agreement to the contrary: (a) Shareholder makes no agreement or understanding herein in any capacity other than in Shareholder's capacity as a record holder and beneficial owner of Securities, (b) nothing herein shall be construed to limit or affect any action or inaction by any officer, partner, member or employee, as applicable, of Shareholder serving on the Company's Board of Directors acting in such person's capacity as a director or fiduciary of the Company, and (c) Shareholder shall have no liability to Parent or any of its affiliates under this Agreement or otherwise as a result of any action or inaction by any officer, partner, member or employee, as applicable, of Shareholder serving on the Company's Board of Directors acting in such person's capacity as a director or fiduciary of the Company.
- 8.3 Expenses. Subject to Section 2.7, each of the parties hereto shall pay its own expenses incurred in connection with this Agreement. Each of the parties hereto warrants and covenants to the others that it will bear all claims for brokerage fees attributable to action taken by it.
- 8.4 Binding Effect. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective representatives and permitted successors and assigns.
- 8.5 Entire Agreement. This Agreement contains the entire understanding of the parties and supersedes all prior agreements and understandings between the parties with respect to its subject matter. This Agreement may be amended only by a written instrument duly executed by the parties hereto.
- 8.6 Headings. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Time is of the essence with respect to all provisions of this Agreement.
- $8.7\,$  Assignment. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party may assign either this Agreement or

any of its rights, interests, or obligations hereunder without the prior written approval of the other parties; provided, however, that each of Parent and Sub may freely assign its rights to another direct or indirect wholly owned subsidiary of Parent or Sub without such prior written approval but no such assignment shall relieve Parent or Sub of any of its obligations hereunder. Any purported assignment without such consent shall be void.

- 8.8 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be an original, but each of which together shall constitute one and the same Agreement.
- 8.9 Notices. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given if so given) by delivery, telegram or telecopy, or by mail (registered or certified mail, postage prepaid, return receipt requested) or by any national courier service, provided that any notice delivered as herein provided shall also be delivered by telecopy at the time of such delivery. All communications hereunder shall be delivered to the respective parties at the following addresses (or at such other address for a party as shall be specified by like notice, provided that notices of a change of address shall be effective only upon receipt thereof):

# <Table> <S> <C>

(a) If to Parent or Sub: The Coca-Cola Company
One-Coca Cola Plaza
Atlanta, Georgia 30313
Attention: Chief Financial Officer
Telecopy: (404) 676-6209
With a copy to: The Coca-Cola Company
One-Coca Cola Plaza
Atlanta, Georgia 30313
Attention: General Counsel
Telecopy: (404) 676-6209
and

<C>

with a copy to: King & Spalding
191 Peachtree Street
Atlanta, Georgia 30303-1763
Attention: C. William Baxley
Telecopy: (404) 572-5100

(b) If to Shareholder:

Attention:
Telecopy:

</Table>

- $8.10\,$  Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without regard to its principles of conflicts of laws.
- 8.11 Enforceability. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.
- 8.12 Further Assurances. From time to time at or after the date Sub purchases shares of Company Common Stock pursuant to the Offer, at Parent's request and without further consideration, Shareholder shall execute and deliver to Parent such documents and take such action as Parent may reasonably request in order to consummate more effectively the transactions contemplated hereby and to vest in Parent good, valid and marketable title to the Securities, including, but not limited to, using its best efforts to cause the appropriate transfer agent or registrar to transfer of record the Securities.
- 8.13 Remedies Not Exclusive. All rights, powers and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity will be cumulative and not alternative, and the exercise of any thereof by either party will not preclude the simultaneous or later exercise of any other such right, power or remedy by such party.

C = 6

8.14 Waiver of Jury Trial. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

C-7

IN WITNESS WHEREOF, Parent, Sub and Shareholder have caused this Agreement to be duly executed as of the day and year first above written.

THE COCA-COLA COMPANY \_\_\_\_\_ Name: \_\_\_\_\_ Title: \_\_\_\_\_ PERRY PHILLIP CORP. Title: SHAREHOLDER: By: ------Name: \_\_\_\_\_ Title: \_\_\_\_\_

C-8

APPENDIX D

# STOCK OPTION AGREEMENT

STOCK OPTION AGREEMENT, dated as of October 29, 2001 (this "Agreement"), is entered into among Odwalla, Inc., a California corporation (the "Company"), The Coca-Cola Company, a Delaware corporation ("Parent"), and Perry Phillip Corp., a California corporation and wholly owned subsidiary of Parent ("Sub").

#### WITNESSETH:

WHEREAS, concurrently with the execution and delivery of this Agreement, the parties hereto are entering into an Agreement and Plan of Merger (the "Merger Agreement") which provides, upon the terms and subject to the conditions set forth therein, for (i) the commencement by Sub of a tender offer (the "Offer") to purchase any and all of the outstanding shares of the common stock, no par value (the "Shares"), of the Company at a price of \$15.25 per Share in cash, net to the seller (such price, or such greater amount which may be paid pursuant to the Offer, the "Offer Price") and (ii) the subsequent merger of Sub with and into the Company (the "Merger"), whereby each Share issued and outstanding (other than Shares owned by Parent, Sub or any other subsidiary of Parent and other than Shares which are held by shareholders exercising dissenters' rights pursuant to Chapter 13 of the California General Corporation Law (the "CGCL")) shall be converted into the right to receive the Offer Price;

WHEREAS, as a condition to the willingness of Parent and Sub to enter into the Merger Agreement, Parent and Sub have required that the Company agree, and in order to induce Parent and Sub to enter into the Merger Agreement, the Company has agreed, to grant to Sub an option to purchase Shares upon the terms and subject to the conditions of this Agreement; and

WHEREAS, capitalized terms used but not defined in this Agreement shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, the parties hereto agree as follows:

#### ARTICLE I

#### THE TOP-UP OPTION

Section 1.1. Grant of Top-Up Stock Option. Subject to the terms and conditions set forth herein, the Company hereby grants to Sub an irrevocable option (the "Top-Up Stock Option") to purchase that number of Shares (the "Top-Up Option Shares") equal to the number of Shares that, when added to the number of Shares owned by Sub, Parent and any other subsidiary of Parent immediately following consummation of the Offer, shall constitute 90.1% of the Shares outstanding on a fully diluted basis (assuming the issuance of the Top-Up Option Shares) and shall permit Sub to effect the Merger pursuant to Section 1110 of the CGCL (assuming the issuance of the Top-Up Option Shares) at a purchase price per Top-Up Option Share equal to the Offer Price; provided, however, that the Top-Up Stock Option shall not be exercisable if the number of Shares subject thereto exceeds the number of authorized shares of common stock of the Company available for issuance. The Company agrees to provide Parent and Sub with information regarding the number of Shares available for issuance on an ongoing basis.

Section 1.2. Exercise of Top-Up Stock Option.

- (a) Subject to the conditions set forth in Section 2.1, Sub may, at its election, exercise the Top-Up Stock Option in whole, but not in part, at any one time after the occurrence of a Top-Up Exercise Event (as defined below) and prior to the Top-Up Termination Date (as defined below).
- (b) A "Top-Up Exercise Event" shall occur for purposes of this Agreement upon Sub's acceptance for payment pursuant to the Offer of Shares constituting, together with Shares owned directly or indirectly by

D-1

Sub, Parent and any other subsidiary of Parent, more than 50% of the Shares then outstanding but less than 90% of the Shares then outstanding on a fully diluted basis.

- (c) Except as provided in the last sentence of this Section 1.2(c), the "Top-Up Termination Date" shall occur for purposes of this Agreement upon the earliest to occur of: (i) the Effective Time; (ii) the date which is 20 business days after the occurrence of a Top-Up Exercise Event; (iii) the termination of the Merger Agreement; and (iv) the date on which Sub reduces the Minimum Condition to the Revised Minimum Number and accepts for payment a number of Shares equal to the Revised Minimum Number.
- (d) Notwithstanding the occurrence of the Top-Up Termination Date, Sub shall be entitled to purchase the Top-Up Option Shares if it has exercised the Top-Up Stock Option in accordance with the terms hereof prior to such occurrence, and the occurrence of the Top-Up Termination Date shall not affect any rights hereunder which by their terms do not terminate or expire prior to or as of such date.
- (e) In the event Sub wishes to exercise the Top-Up Stock Option, Sub shall send to the Company a written notice (a "Top-Up Exercise Notice," the date of which notice is referred to herein as the "Top-Up Notice Date") specifying the denominations of the certificate or certificates evidencing the Top-Up Option Shares which Sub wishes to receive, the place for the closing of the purchase

and sale pursuant to the Top-Up Stock Option (the "Top-Up Closing") and a date not earlier than one day nor later than ten business days after the Top-Up Notice Date for the Top-Up Closing; provided, however, that (i) if the Top-Up Closing cannot be consummated by reason of any applicable Laws or orders, the period of time that otherwise would run pursuant to this sentence shall run instead from the date on which such restriction on consummation has expired or been terminated and (ii) without limiting the foregoing, if prior notification to or approval of any Governmental Entity is required in connection with such purchase, Sub and the Company shall promptly file the required notice or application for approval and shall cooperate in the expeditious filing of such notice or application, and the period of time that otherwise would run pursuant to this sentence shall run instead from the date on which, as the case may be, (A) any required notification period has expired or been terminated or (B) any required approval has been obtained, and in either event, any requisite waiting period has expired or been terminated. The Company shall, promptly after receipt of the Top-Up Exercise Notice, deliver a written notice to Sub confirming the number of Top-Up Option Shares and the aggregate purchase price therefor.

#### ARTICLE II

#### CLOSING

- Section 2.1. Conditions to Closing. The obligation of the Company to deliver Top-Up Option Shares upon the exercise of the Top-Up Stock Option is subject to the following conditions:
  - (a) any applicable waiting period under the HSR Act and any applicable non-United States Laws regulating competition, antitrust, investment or exchange controls relating to the issuance of the Top-Up Option Shares hereunder shall have expired or been terminated;
  - (b) no provision of any applicable Law or regulation and no judgment, injunction, order or decree shall prohibit the exercise of the Top-Up Stock Option or the delivery of the Top-Up Option Shares in respect of any such exercise: and
  - (c) delivery of the Top-Up Option Shares would not violate, or otherwise cause a violation of the rule of the Nasdaq Stock Market set forth in Section  $4350\,(i)\,1\,(D)$  of the NASD Manual.

Section 2.2. Closing.

(a) At the Top-Up Closing (i) the Company shall deliver to Sub a certificate or certificates evidencing the applicable number of Top-Up Option Shares (in the denominations designated by Sub in the Top-Up Exercise Notice) and (ii) Sub shall purchase each Top-Up Option Share from the Company at the Offer Price. Payment by Sub of the purchase price for the Top-Up Option Shares shall be made by delivery of immediately available funds by wire transfer to an account designated by the Company.

D-2

- (b) The Company shall pay all expenses, and any and all federal, state and local taxes and other charges, that may be payable in connection with the preparation, issuance and delivery of stock certificates under this Section 2.2.
- Section 2.3. Securities Law Compliance. Sub understands that the Shares which Sub may acquire hereunder will not be registered under the Securities Act of 1933, as amended (the "Securities Act") in reliance upon an exemption thereunder for transactions not involving a public transaction. Sub is, or will be upon the purchase of the Top-Up Option Shares, an accredited investor for purposes of the federal securities laws. The Top-Up Stock Option and the Top-Up Option Shares to be acquired upon exercise of the Top-Up Stock Option are being and will be acquired by Sub without a view to public distribution thereof otherwise than in compliance with the Securities Act and applicable state securities laws and will not be transferred or otherwise disposed of except in a transaction registered or exempt from registration under the Securities Act and in compliance with applicable state securities laws. Sub will not effect any offer or sale of Top-Up Option Shares which would cause the Company to violate the registration requirements of the Securities Act or the registration or qualification requirements of the securities laws of any jurisdiction. Sub will comply with all obligations under applicable securities law in connection with the receipt of the Top-Up Stock Option or the purchase of the Top-Up Option Shares or otherwise with respect to the consummation of the transactions contemplated by this Agreement.

# ARTICLE III

# ADDITIONAL AGREEMENTS

Section 3.1. Restrictive Legends. Certificates evidencing the Shares to be delivered hereunder may include legends legally required including the legend in substantially the following form:

REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR ANY STATE SECURITIES OR BLUE SKY LAWS, AND MAY BE REOFFERED OR SOLD ONLY IF SO REGISTERED OR IF AN EXEMPTION FROM SUCH REGISTRATION IS AVAILABLE. SUCH SECURITIES ARE ALSO SUBJECT TO ADDITIONAL RESTRICTIONS ON TRANSFER AS SET FORTH IN THE STOCK OPTION AGREEMENT, DATED AS OF OCTOBER 29, 2001, A COPY OF WHICH MAY BE OBTAINED FROM THE ISSUER UPON REQUEST.

It is understood and agreed that (i) the reference to the resale restrictions of the Securities Act and state securities or blue sky laws in the foregoing legend shall be removed by delivery of substitute certificate(s) without such reference if the Company or Sub, as the case may be, shall have delivered to the other party a copy of a letter from the staff of the Securities and Exchange Commission, or an opinion of counsel, in form and substance reasonably satisfactory to the other party, to the effect that such legend is not required for purposes of the Securities Act or such laws; (ii) the reference to the provisions of this Agreement in the foregoing legend shall be removed by delivery of substitute Certificate(s) without such reference if the Shares have been sold or transferred in compliance with the provisions of this Agreement and under circumstances that do not require the retention of such reference; and (iii) the legend shall be removed in its entirety if the conditions in the preceding clauses (i) and (ii) are both satisfied. In addition, such certificates shall bear any other legend as may be required by law.

Section 3.2. Reasonable Best Efforts. Subject to the terms and conditions of this Agreement, Parent, Sub and the Company will use their reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to consummate the transactions contemplated by this Agreement.

Section 3.3. Further Assurances. Each party shall perform such further acts and execute such further documents and instruments as may reasonably be required to vest in Sub and Parent the power to consummate the transactions contemplated by this Agreement. If Sub shall exercise the Top-Up Stock Option granted hereunder in accordance with the terms of this Agreement, each party shall, without additional consideration, execute and deliver all such further documents and instruments and take all such further

D-3

actions as any other party may reasonably request to consummate the transactions contemplated by this Agreement.

#### ARTICLE IV

# MISCELLANEOUS

- Section 4.1. Waiver of Conditions. The conditions to each of the parties' obligations to consummate this Agreement are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable law.
- Section 4.2. Expenses. Each of the parties hereto shall pay its own expenses incurred in connection with this Agreement. Each of the parties hereto warrants and covenants to the others that it will bear all claims for brokerage fees attributable to action taken by it.
- Section 4.3. Binding Effect. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective representatives and permitted successors and assigns.
- Section 4.4. Entire Agreement. This Agreement contains the entire understanding of the parties and supersedes all prior agreements and understandings between the parties with respect to its subject matter. This Agreement may be amended only by a written instrument duly executed by the parties hereto.
- Section 4.5. Headings. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Time is of the essence with respect to all provisions of this Agreement.
- Section 4.6. Assignment. This Agreement shall be binding upon and inure to the benefit of the parties named herein and their respective successors and permitted assigns. No party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties; provided, however, that each of Parent and Sub may freely assign its rights to another direct or indirect wholly owned subsidiary of Parent or Sub without such prior written approval but no such assignment shall relieve Parent or Sub of any of its obligations hereunder. Any purported assignment without such consent shall be void.
- Section 4.7. Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be an original, but each of which together shall constitute one and the same Agreement.
  - Section 4.8. Notices. All notices, requests, claims, demands and other

communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given if so given) by delivery, telegram or telecopy, or by mail (registered or certified mail, postage prepaid, return receipt requested) or by any national courier service, provided that any notice delivered as herein provided shall also be delivered by telecopy at the time of such delivery. All communications hereunder shall be delivered to the respective

D-4

parties at the following addresses (or at such other address for a party as shall be specified by like notice, provided that notices of a change of address shall be effective only upon receipt thereof):

<Table>

<S> <C> <C>

(a) If to Parent or The Coca-Cola Company
Sub: One Coca-Cola Plaza
Atlanta, Georgia 30313

Attention: Chief Financial Officer

Telecopy: (404) 676-8621
With a copy
(which does
not constitute
notice) to:

The Coca-Cola Company
One Coca-Cola Plaza
Atlanta, Georgia 30313
Attention: General Counsel
Telecopy: (404) 676-6209

and with a copy to: King & Spalding
191 Peachtree Street

Atlanta, Georgia 30303-1763 Attention: C. William Baxley Telecopy: (404) 572-5100

(b) If to Odwalla, Inc.
the Company: 120 Stone Pine Road

Half Moon Bay, California 94019

Attention: Chief Financial Officer

Telecopy: (650) 712-5967 with a copy to: Morrison & Foerster LLP

425 Market Street

San Francisco, California 94965 Attention: Robert S. Townsend Telecopy: (415) 268-7522

</Table>

Section 4.9. Governing Law. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of New York, without regard to its principles of conflicts of laws.

Section 4.10. Enforceability. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.

Section 4.11. Remedies Not Exclusive. All rights, powers and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity will be cumulative and not alternative, and the exercise of any thereof by either party will not preclude the simultaneous or later exercise of any other such right, power or remedy by such party.

Section 4.12. Waiver of Jury Trial. EACH PARTY HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

Section 4.13. Specific Performance. The parties hereto agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof and that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof, in addition to any other remedy to which they are entitled at law or in equity.

D-5

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto on the date first hereinabove written.

ODWALLA, INC.

By: /s/ D. STEPHEN C. WILLIAMSON

Name: D. Stephen C. Williamson
Title: Chairman of the Board and
Chief Executive Officer

-----

THE COCA-COLA COMPANY

By: /s/ DAVID M. TAGGART

Name: David M. Taggart Title: Vice President

PERRY PHILLIP CORP.

By: /s/ PAUL ETCHELLS

\_\_\_\_\_

Name: Paul Etchells Title: President

Facsimile copies of the Letter of Transmittal, properly completed and duly executed, will be accepted. The Letter of Transmittal and certificates for Shares and any other required documents should be sent or delivered by each shareholder of Odwalla or such shareholder's broker, dealer, commercial bank, trust company or other nominee to the Depositary at one of the addresses set forth below:

THE DEPOSITARY FOR THE OFFER IS:

EQUISERVE TRUST COMPANY, N.A.

<Table> <S>

> BY MAIL: P.O. Box 43034

Providence, RI 02940-3034

CONFIRM BY TELEPHONE: Toll Free (781) 575-4816 </Table>

<C>

BY HAND: c/o Securities Transfer and Reporting Services, Inc. 100 William Street -- Galleria New York, NY 10038 BY EMAIL:

fctc cocacola@equiserve.com

<C>

BY OVERNIGHT COURIER: 40 Campanelli Drive Braintree, MA 02184

INTERNET: www.equiserve.com

Any questions and requests for assistance may be directed to the Information Agent or the Dealer Manager at their respective telephone numbers and addresses listed below. Additional copies of this Offer to Purchase, the Letter of Transmittal and the Notice of Guaranteed Delivery may also be obtained from the Information Agent. You may also contact your broker, dealer, commercial bank or trust company for assistance concerning the Offer.

THE INFORMATION AGENT FOR THE OFFER IS:

(Innisfree M&A Incorporated LOGO)

501 Madison Avenue, 20th Floor New York, New York 10022 Banks and Brokers Call Collect: 212-750-5833 All Others Call Toll-Free: 1-888-750-5834

THE DEALER MANAGER FOR THE OFFER IS:

MORGAN STANLEY

Morgan Stanley & Co. Incorporated 1585 Broadway New York, New York 10036 212-761-3539

Letter of Transmittal

To Tender Shares of Common Stock of

Odwalla, Inc.

Pursuant to the Offer to Purchase, dated November 6, 2001, by

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT, SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001 UNLESS THE OFFER IS EXTENDED.

THE DEPOSITARY FOR THE OFFER IS:

EQUISERVE TRUST COMPANY, N.A.

<Table>

<S>

BY MAIL:

<C>

EquiServe Trust Company, N.A. P.O. Box 43034 Providence, RI 02940-3034

BY HAND: 100 William Street -- Galleria New York, NY 10038

BY OVERNIGHT COURIER: BY HAND:
EquiServe Trust Company, N.A.
c/o Securities Transfer and
Reporting Services Inc.

BI OVERNIONI CONTENT CONTE

</Table>

DELIVERY OF THIS LETTER OF TRANSMITTAL TO AN ADDRESS OTHER THAN AS SET FORTH ABOVE, DOES NOT CONSTITUTE A VALID DELIVERY TO THE DEPOSITARY. YOU MUST SIGN THIS LETTER OF TRANSMITTAL IN THE APPROPRIATE SPACE THEREFOR PROVIDED BELOW AND COMPLETE THE SUBSTITUTE FORM W-9 SET FORTH BELOW.

THE INSTRUCTIONS ACCOMPANYING THIS LETTER OF TRANSMITTAL SHOULD BE READ CAREFULLY BEFORE THIS LETTER OF TRANSMITTAL IS COMPLETED.

<table> <s></s></table>	<c></c>	<c></c>	<c></c>
DESCRIPTION O	F SHARES TENDERED		
NAME(S) AND ADDRESS(ES) OF REGISTERED HOLDER(S) (PLEASE FILL IN, IF BLANK)		SHARES TENDERED	
OF SHARES	SHARE CERTIFICATE	NUMBER OF SHARES REPRESENTED BY	NUMBER
TENDERED**	NUMBER(S)*	CERTIFICATE(S)*	
	TOTAL SHARES		

<sup>\*</sup> Need not be completed by shareholders tendering by book entry transfer.

<sup>\*\*</sup> Unless otherwise indicated, it will be assumed that all Shares represented by any certificates delivered to the Depositary are being tendered. See Instruction 4. 

# </Table>

This Letter of Transmittal is to be completed by shareholders of Odwalla, Inc., a California corporation ("Odwalla"), if certificates for Shares (as defined below) are to be forwarded herewith or, unless an Agent's Message (as defined in the Offer to Purchase dated November 6, 2001 (the "Offer to Purchase")) is utilized, if delivery of Shares is to be made by book-entry transfer to an account maintained by EquiServe Trust Company, N.A. (the "Depositary"), at The Depositary Trust Company ("DTC") (the "Book-Entry Transfer Facility") pursuant to the procedures set forth in Section 3 of the Offer to Purchase. DELIVERY OF DOCUMENTS TO A BOOK-ENTRY TRANSFER FACILITY DOES NOT CONSTITUTE DELIVERY TO THE DEPOSITARY.

Shareholders whose certificates for Shares are not immediately available or who cannot deliver their Shares and all other documents required hereby to the Depositary prior to the Expiration Date (as defined in the Offer to Purchase), or who cannot comply with the book-entry transfer procedures on a timely basis, must tender their Shares pursuant to the guaranteed delivery procedures set forth in Section 3 of the Offer to Purchase. See Instruction 2.

NOTE: SIGNATURES MUST BE PROVIDED BELOW PLEASE READ THE ACCOMPANYING INSTRUCTIONS CAREFULLY

[ ]	CHECK HERE IF TENDERED SHARES ARE BEING DELIVERED BY BOOK-ENTRY TRANSFER TO THE DEPOSITARY'S ACCOUNT AT THE BOOK-ENTRY TRANSFER FACILITY AND COMPLETE THE FOLLOWING:
	Name of Tendering Institution
	Name of Book-Entry Transfer Facility
	Account No.
	Transaction Code No.
[ ]	CHECK HERE IF TENDERED SHARES ARE BEING DELIVERED PURSUANT TO A NOTICE OF GUARANTEED DELIVERY PREVIOUSLY SENT TO THE DEPOSITARY AND COMPLETE THE FOLLOWING:
	Name(s) of Tendering Shareholder(s)
	Date of Execution of Notice of Guaranteed Delivery
	Window Ticket Number (if any)
	Name of Institution which Guaranteed Delivery
	If Delivery is by Book-Entry Transfer:
	Name of Tendering Institution
	Name of Book-Entry Transfer Facility
	Account No.
	Transaction Code No.
	2

### LADIES AND GENTLEMEN:

The undersigned hereby tenders to TCCC Acquisition Corp. (the "Offeror"), a California corporation and wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"), the above-described shares of common stock, no par value per share (the "Shares"), of Odwalla, Inc., a California corporation ("Odwalla"), pursuant to the Offeror's offer to purchase all of the outstanding Shares at a purchase price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest, upon the terms and subject to the conditions set forth in the Offer to Purchase, receipt of which is hereby acknowledged, and in this Letter of Transmittal (which, together with the Offer to Purchase, and any amendments or supplements hereto or thereto, collectively constitute the "Offer"). The Offer is being made in connection with the

Agreement and Plan of Merger, dated as of October 29, 2001 (the "Merger Agreement"), among TCCC, the Offeror and Odwalla.

Subject to and effective upon acceptance for payment of and payment for the Shares tendered herewith, the undersigned hereby sells, assigns and transfers to or upon the order of the Offeror all right, title and interest in and to all the Shares that are being tendered hereby (and any and all other Shares or other securities issued or issuable in respect thereof) and appoints the Depositary the true and lawful agent and attorney-in-fact of the undersigned with respect to such Shares (and all such other Shares or securities), with full power of substitution (such power of attorney being deemed to be an irrevocable power coupled with an interest), to (a) deliver certificates for such Shares (and all such other Shares or securities), or transfer ownership of such Shares (and all such other Shares or securities) on the account books maintained by any of the Book-Entry Transfer Facilities, together, in any such case, with all accompanying evidences of transfer and authenticity, to or upon the order of the Offeror, (b) present such Shares (and all such other Shares or securities) for transfer on the books of Odwalla and (c) receive all benefits and otherwise exercise all rights of beneficial ownership of such Shares (and all such other Shares or securities), all in accordance with the terms of the Offer.

The undersigned hereby irrevocably appoints each designee of the Offeror as the agent, attorney-in-fact and proxy of the undersigned, each with full power of substitution, to exercise all voting and other rights of the undersigned in such manner as each such attorney and proxy or his substitute shall in his sole judgment deem proper, with respect to all of the Shares tendered hereby which have been accepted for payment by the Offeror prior to the time of any vote or other action (and any and all other Shares or other securities or rights issued or issuable in respect of such Shares) at any meeting of shareholders of Odwalla (whether annual or special and whether or not an adjourned meeting), any actions by written consent in lieu of any such meeting or otherwise. This proxy is irrevocable, is coupled with an interest in the Shares and is granted in consideration of, and is effective upon, the acceptance for payment of such Shares by the Offeror in accordance with the terms of the Offer. Such acceptance for payment shall revoke any other power of attorney, proxy or written consent granted by the undersigned at any time with respect to such Shares (and all such other Shares or other securities or rights), and no subsequent powers of attorney or proxies will be given or written consents will be executed by the undersigned (and if given or executed, will not be deemed effective). The undersigned understands that in order for Shares to be deemed validly tendered, immediately upon the acceptance for payment of such Shares, the Offeror or its designee must be able to exercise full voting rights with respect to such Shares and other securities, including voting at any meeting of shareholders.

The undersigned hereby represents and warrants that the undersigned has full power and authority to tender, sell, assign and transfer the Shares tendered hereby (and any and all other Shares or other securities or rights issued or issuable in respect of such Shares) and that when the same are accepted for payment by the Offeror, the Offeror will acquire good and unencumbered title thereto, free and clear of all liens, restrictions, charges and encumbrances and not subject to any adverse claims. The undersigned, upon request, will execute and deliver any additional documents deemed by the Depositary or the Offeror to be necessary or desirable to complete the sale, assignment and transfer of the Shares tendered hereby (and all such other Shares or other securities or rights).

All authority herein conferred or agreed to be conferred shall survive the death or incapacity of the undersigned, and any obligation of the undersigned hereunder shall be binding upon the heirs, personal representatives, successors and assigns of the undersigned. Except as stated in the Offer, this tender is irrevocable.

The undersigned understands that tenders of Shares pursuant to any one of the procedures described in Section 3 of the Offer to Purchase and in the instructions hereto will constitute an agreement between the undersigned and the Offeror upon the terms and subject to the conditions of the Offer.

Unless otherwise indicated under "Special Payment Instructions," please issue the check for the purchase price of any Shares accepted for payment, and return any Shares not tendered or not accepted for payment, in the name(s) of the undersigned. Similarly, unless otherwise indicated under "Special Delivery Instructions," please mail the check for the purchase price of any Shares accepted for payment and return any certificates for Shares not tendered or not accepted for payment (and accompanying documents, as appropriate) to the undersigned at the address shown below the under-

3

signed's signature(s). If both "Special Payment Instructions" and "Special Delivery Instructions" are completed, please issue the check for the purchase price of any Shares purchased and return any Shares not tendered or not purchased in the name(s) of, and deliver said check and any certificates to, the person(s) so indicated. Shareholders tendering Shares by book-entry transfer may request that any Shares not accepted for payment be returned by crediting such account maintained at such Book-Entry Transfer Facility as such shareholder may

designate by making an appropriate entry under "Special Payment Instructions." The undersigned recognizes that the Offeror has no obligation, pursuant to the "Special Payment Instructions," to transfer any Shares from the name of the registered holder(s) thereof if the Offeror does not accept for payment any of the Shares so tendered.

SPECIAL PAYMENT INSTRUCTIONS (SEE INSTRUCTIONS 1, 5, 6 AND 7)

To be completed ONLY if the check for the purchase price of Shares purchased or certificates for Shares not tendered or not purchased are to be issued in the name of someone other than the undersigned.

Issue [ ] check	[ ] cert	ificate	to:				
Name(s)							
			(PLEASE	PRINT)			
Address							
						(ZIP	CODE)
(	TAXPAYER		CATION O SUBSTITU		Y NUMBER)		

SPECIAL DELIVERY INSTRUCTIONS (SEE INSTRUCTIONS 5 AND 7)

To be completed ONLY if the check for the purchase price of Shares purchased or certificates for Shares not tendered or not purchased are to be mailed to someone other than the undersigned or to the undersigned at an address other than that shown below the undersigned's signature(s).

Mail check and/or certificates to:

Name
(PLEASE PRINT)

Address

(ZIP CODE)

4

# INSTRUCTIONS

### FORMING PART OF THE TERMS AND CONDITIONS OF THE OFFER

- 1. Guarantee of Signatures. Except as otherwise provided below, signatures on all Letters of Transmittal must be guaranteed by a firm that is a bank, broker, dealer, credit union, savings association or other entity which is a member in good standing of the Securities Transfer Agents Medallion Program or by any other bank, broker, dealer, credit union, savings association or other entity which is an "eligible guarantor institution," as such term is defined in Rule 17Ad-15 under the Securities Exchange Act of 1934, as amended (each of the foregoing constituting an "Eligible Institution"), unless the Shares tendered thereby are tendered (i) by a registered holder of Shares who has not completed the box labeled "Special Payment Instructions" on the Letter of Transmittal or (ii) for the account of an Eligible Institution. See Instruction 5. If the certificates are registered in the name of a person or persons other than the signer of this Letter of Transmittal, or if payment is to be made or delivered to, or certificates evidencing unpurchased Shares are to be issued or returned to, a person other than the registered owner or owners, then the tendered certificates must be endorsed or accompanied by duly executed stock powers, in either case signed exactly as the name or names of the registered owner or owners appear on the certificates or stock powers, with the signatures on the certificates or stock powers guaranteed by an Eligible Institution as provided herein. See Instruction 5.
- 2. Delivery of Letter of Transmittal and Shares. This Letter of Transmittal is to be used either if certificates are to be forwarded herewith or, unless an Agent's Message (as defined in the Offer to Purchase) is utilized, if the delivery of Shares is to be made by book-entry transfer pursuant to the procedures set forth in Section 3 of the Offer to Purchase. Certificates for all physically delivered Shares, or a confirmation of a book-entry transfer into the Depositary's account at the Book-Entry Transfer Facility of all Shares delivered electronically, as well as a properly completed and duly executed Letter of Transmittal (or a facsimile thereof) and any other documents required by this

Letter of Transmittal, or an Agent's Message in the case of a book-entry delivery, must be received by the Depositary at one of its addresses set forth on the front page of this Letter of Transmittal prior to the Expiration Date. Shareholders who cannot deliver their Shares and all other required documents to the Depositary prior to the Expiration Date must tender their Shares pursuant to the quaranteed delivery procedures set forth in Section 3 of the Offer to Purchase. Pursuant to such procedures: (a) such tender must be made by or through an Eligible Institution; (b) a properly completed and duly executed Notice of Guaranteed Delivery, substantially in the form provided by the Offeror, must be received by the Depositary prior to the Expiration Date; and (c) the certificates for all tendered Shares, in proper form for tender, or a confirmation of a book-entry transfer into the Depositary's account at the Book-Entry Transfer Facility of all Shares delivered electronically, as well as a properly completed and duly executed Letter of Transmittal (or a facsimile thereof), and any other documents required by this Letter of Transmittal, must be received by the Depositary within three trading days after the date of execution of such Notice of Guaranteed Delivery, all as provided in Section 3 of the Offer to Purchase. The term "trading day" is any day on which the Nasdaq National Market ("Nasdaq") is open for business.

THE METHOD OF DELIVERY OF SHARES, THE LETTER OF TRANSMITTAL AND ALL OTHER REQUIRED DOCUMENTS, INCLUDING DELIVERY THROUGH A BOOK-ENTRY TRANSFER FACILITY, IS AT THE OPTION AND RISK OF THE TENDERING SHAREHOLDER. SHARES WILL BE DEEMED DELIVERED ONLY WHEN ACTUALLY RECEIVED BY THE DEPOSITARY (INCLUDING, IN THE CASE OF A BOOK-ENTRY TRANSFER, BY A CONFIRMATION OF A BOOK-ENTRY TRANSFER). IF DELIVERY IS BY MAIL, REGISTERED MAIL WITH RETURN RECEIPT REQUESTED, PROPERLY INSURED, IS RECOMMENDED. IN ALL CASES, SUFFICIENT TIME SHOULD BE ALLOWED TO ENSURE TIMELY DELIVERY.

No alternative, conditional or contingent tenders will be accepted, and no fractional Shares will be purchased. By executing this Letter of Transmittal (or a facsimile thereof), the tendering shareholder waives any right to receive any notice of the acceptance for payment of the Shares.

- 3. Inadequate Space. If the space provided herein is inadequate, the certificate numbers and/or the number of Shares and any other required information should be listed on a separate schedule attached hereto and separately signed on each page thereof in the same manner as this Letter of Transmittal is signed.
- 4. Partial Tenders (not applicable to shareholders who tender by book-entry transfer). If fewer than all the Shares represented by any certificate delivered to the Depositary are to be tendered, fill in the number of Shares which are to be tendered in the box entitled "Number of Shares Tendered." In such case, a new certificate for the remainder of the Shares represented by the old certificate will be sent to the person(s) signing this Letter of Transmittal unless otherwise provided in the appropriate box marked "Special Payment Instructions" and/or "Special Delivery Instructions" on this

5

Letter of Transmittal, as promptly as practicable following the Expiration Date. All Shares represented by certificates delivered to the Depositary will be deemed to have been tendered unless otherwise indicated.

5. Signatures on Letter of Transmittal; Stock Powers and Endorsements. If this Letter of Transmittal is signed by the registered holder(s) of the Shares tendered hereby, the signature(s) must correspond with the name(s) as written on the face of the certificates without alteration, enlargement or any change whatsoever.

If any of the Shares tendered hereby are held of record by two or more persons, all such persons must sign this Letter of Transmittal.

If any of the Shares tendered hereby are registered in different names on different certificates, it will be necessary to complete, sign and submit as many separate Letters of Transmittal as there are different registrations of certificates.

If this Letter of Transmittal is signed by the registered holder(s) of the Shares tendered hereby, no endorsements of certificates or separate stock powers are required unless payment of the purchase price is to be made, or Shares not tendered or not purchased are to be returned, in the name of any person other than the registered holder(s), in which case the certificate(s) for such Shares tendered hereby must be endorsed, or accompanied by, appropriate stock powers, in either case signed exactly as the name(s) of the registered holder(s) appear(s) on the certificate for such Shares. Signatures on any such certificates or stock powers must be guaranteed by an Eligible Institution.

If this Letter of Transmittal is signed by a person other than the registered holder(s) of the Shares tendered hereby, the certificate must be endorsed or accompanied by appropriate stock powers, in either case signed exactly as the name(s) of the registered holder(s) appear(s) on the certificates for such Shares. Signature(s) on any such certificates or stock powers must be guaranteed by an Eligible Institution.

If this Letter of Transmittal or any certificate or stock power is signed by a trustee, executor, administrator, guardian, attorney-in-fact, officer of a corporation or other person acting in a fiduciary or representative capacity, such person should so indicate when signing, and proper evidence satisfactory to the Offeror of the authority of such person so to act must be submitted.

6. Stock Transfer Taxes. The Offeror will pay or cause to be paid any stock transfer taxes with respect to the sale and transfer of any Shares to it or its order pursuant to the Offer. If, however, payment of the purchase price is to be made to, or Shares not tendered or not purchased are to be returned in the name of, any person other than the registered holder(s), then the amount of any stock transfer taxes (whether imposed on the registered holder(s), such other person or otherwise) payable on account of the transfer to such person will be deducted from the purchase price unless satisfactory evidence of the payment of such taxes, or exemption therefrom, is submitted.

EXCEPT AS PROVIDED IN THIS INSTRUCTION 6, IT WILL NOT BE NECESSARY FOR TRANSFER TAX STAMPS TO BE AFFIXED TO THE CERTIFICATES LISTED IN THIS LETTER OF TRANSMITTAL.

- 7. Special Payment and Delivery Instruction. If the check for the purchase price of any Shares purchased is to be issued, or any Shares not tendered or not purchased are to be returned, in the name of a person other than the person(s) signing this Letter of Transmittal or if the check or any certificates for Shares not tendered or not purchased are to be mailed to someone other than the person(s) signing this Letter of Transmittal or to the person(s) signing this Letter of Transmittal at an address other than that shown above, the appropriate boxes on this Letter of Transmittal should be completed.
- 8. Substitute Form W-9. Under U.S. Federal income tax law, a tendering shareholder that is a United States Person as defined under the U.S. federal income tax law whose Shares are accepted for payment is required to provide the Depositary with such shareholder's correct taxpayer identification number ("TIN") on Substitute Form W-9, which is provided below, unless an exemption applies. Failure to provide the information on the Substitute Form W-9 may subject the tendering shareholder to a \$50 penalty and to 30.5% federal income tax backup withholding on the payment of the purchase price for the Shares.
- 9. Foreign Holders. Foreign holders must submit a completed IRS Form W-8BEN to avoid 30.5% backup withholding.

(

- 10. Requests for Assistance or Additional Copies. Requests for assistance or additional copies of the Offer to Purchase and this Letter of Transmittal may be obtained from the Information Agent at its address or telephone number set forth below.
- 11. Waiver of Conditions. The conditions of the Offer may be waived by the Offeror (subject to certain limitations in the Merger Agreement), in whole or in part, at any time or from time to time, in the Offeror's sole discretion.
- 12. Lost, Destroyed, Mutilated, or Stolen Certificates. If any certificate(s) representing Shares has been lost, destroyed, mutilated, or stolen, the shareholder should promptly notify ChaseMellon Shareholder Services, L.L.C. at (415) 743-1434. The shareholder will then be instructed as to the steps to be taken in order to replace the certificate(s). This Letter of Transmittal and related documents cannot be processed until the procedures for replacing lost, destroyed, mutilated or stolen certificates have been followed.

### IMPORTANT:

THIS LETTER OF TRANSMITTAL OR A FACSIMILE COPY HEREOF (TOGETHER WITH CERTIFICATES OR CONFIRMATION OF BOOK-ENTRY TRANSFER AND ALL OTHER REQUIRED DOCUMENTS) OR A NOTICE OF GUARANTEED DELIVERY MUST BE RECEIVED BY THE DEPOSITARY PRIOR TO THE EXPIRATION DATE (AS DEFINED IN THE OFFER TO PURCHASE).

### IMPORTANT TAX INFORMATION

Under federal income tax law, a shareholder that is a United States Person as defined under U.S. federal income tax law whose tendered Shares are accepted for payment is required to provide the Depositary with such shareholder's correct TIN on the Substitute Form W-9. If such shareholder is an individual, the TIN is such shareholder's Social Security Number. If the Depositary is not provided with the correct TIN, the shareholder may be subject to a \$50 penalty imposed by the Internal Revenue Service. In addition, payments that are made to such shareholder with respect to Shares purchased pursuant to the Offer may be subject to backup withholding.

Certain shareholders (including, among others, all corporations and certain foreign individuals) are not subject to these backup withholding and reporting requirements. In order for a foreign individual to qualify as an exempt recipient, that shareholder must submit a statement, signed under penalties of perjury, attesting to that individual's exempt status. Such statements may be

obtained from the Depositary. All exempt recipients (including foreign persons wishing to qualify as exempt recipients) should see the enclosed Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9 for additional instructions.

If backup withholding applies, the Depositary is required to withhold 30.5% of any payments made to the shareholder. Backup withholding is not an additional tax. Rather, the tax liability of persons subject to backup withholding will be reduced by the amount of tax withheld. If backup withholding results in an overpayment of taxes, a refund may be obtained.

### PURPOSE OF SUBSTITUTE FORM W-9

To prevent backup federal income tax withholding on payments that are made to a shareholder with respect to Shares purchased pursuant to the Offer, the shareholder is required to notify the Depositary of such shareholder's correct TIN by completing the form certifying that the TIN provided on the Substitute Form W-9 is correct.

### WHAT NUMBER TO GIVE THE DEPOSITARY

The shareholder is required to give the Depositary the Social Security Number or Employer Identification Number of the record owner of the Shares. If the Shares are in more than one name or are not in the name of the actual owner, consult the enclosed Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9 for additional guidelines on which number to report.

7

### SIGN HERE (COMPLETE SUBSTITUTE FORM W-9 BELOW)

X
х
SIGNATURE(S) OF OWNER(S)
Name(s)
Capacity (Full Title)
Address
(INCLUDE ZIP CODE)
Daytime Area Code and Telephone Number
Tax Identification or Social Security Number
(SEE SUBSTITUTE FORM W-9)
Dated
(Must be signed by registered holder(s) exactly as name(s) appear(s) on stock certificate(s) or on a security position listing or by the person(s) authorized to become registered holder(s) by certificates and documents transmitted herewith. If signature is by a trustee, executor, administrator, guardian, attorneys-in-fact, officer of a corporation or other person acting in a

GUARANTEE OF SIGNATURE(S)
(SEE INSTRUCTIONS 1 AND 5)
FOR USE BY FINANCIAL INSTITUTIONS ONLY.
PLACE MEDALLION GUARANTEE IN SPACE BELOW.

fiduciary or representative capacity, please set forth full title and see

Authorized Signature(s)	
Name	
Name of Firm	
Address	
Area Code and Telephone Number	(INCLUDE ZIP CODE)

Dated

Instruction 5.)

\_\_\_\_\_\_

8

PAYER'S NAME: EQUISERVE TRUST COMPANY, N.A.

<table> <s></s></table>	<c></c>	<c></c>
SUBSTITUTE	PART 1PLEASE PROVIDE YOUR TIN IN THE BOX AT	
	RIGHT AND CERTIFY BY SIGNING AND DATING	TIN
FORM W-9	BELOW.	
DEPARTMENT OF THE TREASURY, INTERNAL REVENUE SERVICE	PART 2For Payees exempt from backup withholding, see the enclosed Guidelines for	Social Security Number OR
SERVICE PAYER'S REQUEST FOR TAXPAYER IDENTIFICATION NUMBER ("TIN") AND	Certification of Taxpayer Identification Number on Substitute Form W-9 and complete as instructed therein.	Employer Identification Number
CERTIFICATION	CertificationUnder Penalties of Perjury, I certify that:  (1) The number shown on this form is my correct TIN (or I am waiting for a number to be issued to me);  (2) I am not subject to backup withholding because (a) I am exempt from backup	
	withholding or (b) I have not been notified by the Internal Revenue Service ("IRS") that I am subject to backup withholding as a result of failure to report all interest or dividends, or (c) the IRS has notified me that I am no longer subject to backup withholding; and (3) I am a U.S. Person (including a U.S. resident alien).	
	CERTIFICATION INSTRUCTIONSYou must cross ou notified by the IRS that you are currently su	<del>-</del>
because of	underreporting interest or dividends on your	tax return. However, if after
being	notified by the IRS that you were subject to	backup withholding, you received
another	notification from the IRS that you were no lo	nger subject to backup
withholding, do	not cross out item (2). (Also see the instruc	tions in the enclosed
Guidelines.)		
	SIGNATURE	DATE

		NOTE: FAILURE TO COMPLETE AND RETU BACKUP WITHHOLDING OF 30.5% OF ANY PLEASE REVIEW THE ENCLOSED GUIDELI	RN THIS SUBSTITUTE FORM W-9 MAY RESULT IN PAYMENTS MADE TO YOU PURSUANT TO THE OFFER. NES FOR CERTIFICATION OF TAXPAYER E FORM W-9 FOR ADDITIONAL DETAILS.	
YOU MUST COMPLETE THE FOLLOWING CE	RTIFICATE IF YOU ARE AWAITING YOUR TIN.			
CERTIFICATE OF AWAITI	NG TAXPAYER IDENTIFICATION NUMBER			
I certify under penalties of perjueither (1) I have mailed or delive appropriate IRS Center or Social S to mail or deliver an application not provide a TIN by the time of p	ry that a TIN has not been issued to me, and red an application to receive a TIN to the ecurity Administration Office or (2) I intend in the near future. I understand that if I do ayment, 30.5% of all payments pursuant to the e withheld but will be refunded to me if I			
<C>

</Table>

<Table>

provide a certified TIN within 60 days.

Signature

Date

# THIS SECTION TO BE COMPLETED BY NON-UNITED STATES HOLDERS

<table> <caption></caption></table>		
<s></s>	<c></c>	<c></c>
Form W-8BEN	CERTIFICATE OF FOREIGN STATUS OF BENEFICIAL OWNER FOR UNITED STATES TAX WITHHOLDING	
(Rev. December 2000)	- SECTION REFERENCES ARE TO THE INTERNAL REVENUE CODE SI SEPARATE INSTRUCTIONS.	EE OMB No. 1545-1621
Department of the Treasury Internal Revenue Service	- GIVE THIS FORM TO THE WITHHOLDING AGENT OR PAYER. DO NOT SEND TO THE IRS.	

		individual	U.S. person, including a resident alien	
	OF BENEFICIAL OWNER (See instructions.)			
owner	organization that is the beneficial 2 Country of incorpo	-		
	[ ] Complex trust [ ] Estate entity issue [ ] Tax-exempt [ ] Private [ ] Government	[ ] International		
	organization foundation			
<\$>				
ADDRESS.	dress (street, apt. or suite no., or rural route). DO NOT US			
City or town, state or where appropriate.	province. Include postal code Country (do not abbrevia	ate)		
5 Mailing address (if di				
City or town, state or where appropriate.	province. Include postal code Country (do not abbrevia	ate)		
6 U.S. taxpayer identifi instructions) [ ] SSN or ITIN [ ] E	cation number, if required (see 7 Foreign tax identifying TN	ng number, if any (optional)		
8 Reference number(s) (s				
25047Z

PART II CLAIM OF TAX TREATY BENEFITS (if applicable) \_\_\_\_\_\_ <Table> <C> <C> <S> 9 I CERTIFY THAT (CHECK ALL THAT APPLY): a [] The beneficial owner is a resident of ... within the meaning of the income tax treaty between the United States and that country. b [ ] If required, the U.S. taxpayer identification number is stated on line 6 (see instructions). c [] The beneficial owner is not an individual, derives the item (or items) of income for which the treaty benefits are claimed, and, if applicable, meets the requirements of the treaty provision dealing with limitation on benefits (see instructions). d [ ] The beneficial owner is not an individual, is claiming treaty benefits for dividends received from a foreign corporation or interest from a U.S. trade or business of a foreign corporation, and meets qualified resident status (see instructions). e [] The beneficial owner is related to the person obligated to pay the income within the meaning of section 267(b) or 707(b), and will file Form 8833 if the amount subject to withholding received during a calendar year exceeds, in the aggregate, \$500,000. 10 SPECIAL RATES AND CONDITIONS (if applicable -- see instructions): The beneficial owner is claiming the provisions of Article . of the treaty identified on line 9a above to claim a ......% rate of withholding on (specify type of income): ...... Explain the reasons the beneficial owner meets the terms of the treaty article: ..... </Table> PART III NOTIONAL PRINCIPAL CONTRACTS \_\_\_\_\_\_ <Table> <C> <C> <S> 11 [ ] I have provided or will provide a statement that identifies those notional principal contracts from which the income is NOT effectively connected with the conduct of a trade or business in the United States. I agree to update this statement as required. \_\_\_\_\_\_ </Table> PART IV CERTIFICATION \_\_\_\_\_\_ Under penalties of perjury, I declare that I have examined the information on this form and to the best of my knowledge and belief it is true, correct, and complete. I further certify under penalties of perjury that: - I am the beneficial owner (or am authorized to sign for the beneficial owner) of all the income to which this form relates, - The beneficial owner is not a U.S. person, - The income to which this form relates is not effectively connected with the conduct of a trade or business in the United States or is effectively connected but is not subject to tax under an income tax treaty, AND - For broker transactions or barter exchanges, the beneficial owner is an exempt foreign person as defined in the instructions. Furthermore, I authorize this form to be provided to any withholding agent that has control, receipt, or custody of the income of which I am the beneficial owner or any withholding agent that can disburse or make payments of the income of which I am the beneficial owner. <Table> <C> Sign Here X ------Signature of beneficial owner (or individual Date (MM-DD-YYYY) Capacity in which acting authorized to sign for beneficial owner) </Table>

10

FORM W-8BEN (Rev. 12-2000)

FOR PAPERWORK REDUCTION ACT NOTICE, SEE SEPARATE INSTRUCTIONS. Cat. No.

Facsimile copies of the Letter of Transmittal, properly completed and duly executed, will be accepted. The Letter of Transmittal and certificates for Shares and any other required documents should be sent or delivered by each shareholder of Odwalla or such shareholder's broker, dealer, commercial bank, trust company or other nominee to the Depositary at one of the addresses set forth below:

THE DEPOSITARY FOR THE OFFER IS:

EQUISERVE TRUST COMPANY, N.A.

<Table>

BY MAIL: P.O. Box 43034 Providence, RI 02940-3034 <C> C> BY HAND:

c/o Securities Transfer and Reporting Services, Inc. 100 William Street--Galleria New York, NY 10038 BY OVERNIGHT COURIER: 40 Campanelli Drive Braintree, MA 02184

CONFIRM BY TELEPHONE:
Toll Free (781) 575-4816
</Table>

BY EMAIL: fctc\_cocacola@equiserve.com

INTERNET: www.equiserve.com

Any questions and requests for assistance may be directed to the Information Agent or the Dealer Manager at their respective telephone numbers and addresses listed below. Additional copies of this Offer to Purchase, the Letter of Transmittal and the Notice of Guaranteed Delivery may also be obtained from the Information Agent. You may also contact your broker, dealer, commercial bank or trust company for assistance concerning the Offer.

THE INFORMATION AGENT FOR THE OFFER IS:

(Innisfree M&A Incorporated LOGO)

501 Madison Avenue, 20th Floor New York, New York 10022 Banks and Brokers Call Collect: 212-750-5833 All Others Call Toll-Free: 1-888-750-5834

THE DEALER MANAGER FOR THE OFFER IS:

MORGAN STANLEY

Morgan Stanley & Co. Incorporated 1585 Broadway
New York, New York 10036 212-761-3539

Offer To Purchase For Cash

All Outstanding Shares Of Common Stock

Odwalla, Inc. at

\$15.25 Net Per Share by

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT, SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

November 6, 2001

To Brokers, Dealers, Commercial Banks, Trust Companies and Other Nominees:

We have been appointed by TCCC Acquisition Corp., a California corporation (the "Offeror"), a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"), to act as Dealer Manager in connection with the Offeror's offer to purchase all outstanding shares of common stock, no par value per share (the "Shares"), of Odwalla Inc., a California corporation ("Odwalla"), at a purchase price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest, upon the terms and subject to the conditions set forth in the Offer to Purchase, dated November 6, 2001 (the "Offer to Purchase"), and in the related Letter of Transmittal (which, together with any amendments or supplements thereto, collectively constitute the "Offer") enclosed herewith. The Offer is being made in connection with the Agreement and Plan of Merger, dated as of October 29, 2001, among TCCC, the Offeror and Odwalla (the "Merger Agreement"). Holders of Shares whose certificates for such Shares (the "Certificates") are not immediately available or who cannot deliver their Certificates and all other required documents to EquiServe Trust Company, N.A. (the "Depositary") or complete the procedures for book-entry transfer prior to the Expiration Date (as defined in Section 1 of the Offer to Purchase) must tender their Shares according to the quaranteed delivery procedures set forth in Section 3 of the Offer to Purchase.

Please furnish copies of the enclosed materials to those of your clients for whose accounts you hold Shares in your name or in the name of your nominee.

Enclosed herewith for your information and forwarding to your clients are copies of the following documents:

- 1. The Offer to Purchase, dated November 6, 2001.
- The Letter of Transmittal to tender Shares for your use and for the information of your clients. Facsimile copies of the Letter of Transmittal may be used to tender Shares.
- 3. A letter to shareholders of Odwalla from D. Stephen C. Williamson, the Chief Executive Officer of Odwalla, together with a Solicitation/Recommendation Statement on Schedule 14D-9 filed with the Securities and Exchange Commission by Odwalla and mailed to the shareholders of Odwalla.
- 4. The Notice of Guaranteed Delivery for Shares to be used to accept the Offer if Certificates and all other required documents are not immediately available or cannot be delivered to the Depositary prior to the Expiration Date (as defined in the Offer to Purchase) or if the procedure for book-entry transfer cannot be completed prior to the Expiration Date.
- 5. A printed form of letter which may be sent to your clients for whose accounts you hold Shares registered in your name, with space provided for obtaining such clients' instructions with regard to the Offer.
- 6. Guidelines of the Internal Revenue Service for Certification of Taxpayer Identification Number on Substitute Form W-9.
- 7. A return envelope addressed to the Depositary.

YOUR PROMPT ACTION IS REQUESTED. WE URGE YOU TO CONTACT YOUR CLIENTS AS PROMPTLY AS POSSIBLE. PLEASE NOTE THAT THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT, SAN FRANCISCO, CALIFORNIA TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

Please note the following:

- 1. The tender price is \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest.
- 2. The Offer is being made for all of the outstanding Shares.
- 3. The Offer and withdrawal rights will expire at 12:00 Midnight, San Francisco, California Time, on December 6, 2001 unless the Offer is extended.
- 4. The Offer is conditioned upon, among other things, there being validly tendered and not withdrawn prior to the expiration of the Offer such number of Shares that would constitute at least 90.1% of the Shares that are outstanding determined on a fully diluted basis (including for purposes of such calculation all Shares that are issuable upon exercise of vested options and outstanding warrants but excluding Shares that are issuable upon the exercise of options and warrants that are cancelled pursuant to the Merger Agreement).
- 5. Tendering shareholders will not be obligated to pay brokerage fees or commissions or, except as set forth in Instruction 6 of the Letter of Transmittal, stock transfer taxes on the transfer of Shares pursuant to the Offer.

In order to accept the Offer, (i) a duly executed and properly completed Letter of Transmittal (or facsimile thereof) and any required signature guarantees or, in the case of a book-entry transfer, an Agent's Message (as defined in the Offer to Purchase) or other required documents should be sent to the Depositary and (ii) Certificates representing the tendered Shares on a timely Book-Entry Confirmation (as defined in the Offer to Purchase) should be delivered to the Depositary in accordance with the instructions set forth in the Offer.

If holders of Shares wish to tender, but it is impracticable for them to forward their Certificates or other required documents or complete the procedures for book-entry transfer prior to the Expiration Date, a tender must be effected by following the guaranteed delivery procedures specified in Section 3 of the Offer to Purchase.

Neither the Offeror, TCCC nor any officer, director, shareholder, agent or other representative of the Offeror will pay any fees or commissions to any broker, dealer or other person for soliciting tenders of Shares pursuant to the Offer. The Offeror will, however, upon request, reimburse you for customary mailing and handling expenses incurred by you in forwarding any of the enclosed materials to your clients. The Offeror will pay or cause to be paid any transfer taxes payable on the transfer of Shares to it, except as otherwise provided in Instruction 6 of the Letter of Transmittal.

2

Any inquiries you may have with respect to the Offer should be addressed to, and additional copies of the enclosed materials may be obtained from, Morgan Stanley & Co., Incorporated, the Dealer Manager, or Innisfree M&A Incorporated, the Information Agent, at the respective addresses and telephone numbers set forth on the back cover page of the Offer to Purchase.

Very truly yours,

MORGAN STANLEY & CO.

 ${\tt Incorporated}$ 

NOTHING CONTAINED HEREIN OR IN THE ENCLOSED DOCUMENTS WILL CONSTITUTE YOU OR ANY OTHER PERSON THE AGENT OF TCCC, THE OFFEROR, THE DEPOSITARY, THE INFORMATION AGENT OR ANY AFFILIATE OF ANY OF THEM, OR AUTHORIZE YOU OR ANY OTHER PERSON TO MAKE ANY STATEMENT OR USE ANY DOCUMENT ON BEHALF OF ANY OF THEM IN CONNECTION WITH THE OFFER OTHER THAN THE ENCLOSED DOCUMENTS AND THE STATEMENTS CONTAINED THEREIN.

Offer To Purchase For Cash

All Outstanding Shares Of Common Stock

Odwalla, Inc. at

\$15.25 Net Per Share by

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

November 6, 2001

To Our Clients:

Enclosed for your consideration are the Offer to Purchase, dated November 6, 2001 (the "Offer to Purchase"), and the related Letter of Transmittal (which, together with any amendments or supplements thereto, collectively constitute the "Offer") relating to an offer by TCCC Acquisition Corp., a California corporation (the "Offeror") and a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation (the "TCCC"), to purchase all outstanding shares of common stock, no par value per share (the "Shares"), of Odwalla Inc., a California corporation ("Odwalla"), at a purchase price of \$15.25 per Share, net to the seller in cash, less any required withholding taxes and without interest, upon the terms and subject to the conditions set forth in the Offer. The Offer is being made in connection with the Agreement and Plan of Merger, dated as of October 29, 2001, among TCCC, the Offeror and Odwalla (the "Merger Agreement"). This material is being forwarded to you as the beneficial owner of Shares carried by us in your account but not registered in your name.

WE ARE THE HOLDER OF RECORD OF SHARES HELD BY US FOR YOUR ACCOUNT. A TENDER OF SUCH SHARES CAN BE MADE ONLY BY US AS THE HOLDER OF RECORD AND PURSUANT TO YOUR INSTRUCTIONS. THE LETTER OF TRANSMITTAL IS FURNISHED TO YOU FOR YOUR INFORMATION ONLY AND CANNOT BE USED BY YOU TO TENDER SHARES HELD BY US FOR YOUR ACCOUNT.

Accordingly, we request instructions as to whether you wish to tender any or all of the Shares held by us for your account, upon the terms and conditions set forth in the Offer.

Please note the following:

- 1. The tender price is \$15.25 per Share, net to you in cash less any required withholding taxes and without interest.
- 2. The Board of Directors of Odwalla unanimously has determined that the Offer and the Merger (as defined in the Offer to Purchase), are fair to and in the best interests of Odwalla's shareholders, has approved the Offer and adopted the Merger Agreement and recommends acceptance of the Offer by Odwalla's shareholders.
- 3. The Offer is being made for all of the outstanding Shares.
- 4. The Offer and withdrawal rights will expire at 12:00 Midnight, San Francisco, California Time, on Thursday, December 6, 2001, unless the Offer is extended.
- 5. The Offer is conditioned upon, among other things, there being validly tendered and not withdrawn prior to the expiration of the Offer such number of Shares that would constitute at least 90.1% of the Shares that are outstanding determined on a fully diluted basis (including for purposes of such calculation all Shares that are issuable upon exercise of vested options and outstanding warrants but excluding Shares that are issuable upon the exercise of options and warrants that are cancelled pursuant to the Merger Agreement).
- 6. Tendering shareholders will not be obligated to pay brokerage fees or commissions or, except as set forth in Instruction 6 of the Letter of Transmittal, stock transfer taxes on the transfer of Shares pursuant to the Offer.

If you wish to have us tender any or all of the Shares, please so instruct us by completing, executing, detaching and returning to us the instruction form contained in this letter. An envelope to return your instructions to us is enclosed. If you authorize tender of your Shares, all such Shares will be

tendered unless otherwise indicated in such instruction form. PLEASE FORWARD YOUR INSTRUCTIONS TO US AS SOON AS POSSIBLE TO ALLOW US AMPLE TIME TO TENDER YOUR SHARES ON YOUR BEHALF PRIOR TO THE EXPIRATION OF THE OFFER.

The Offer is made solely by the Offer to Purchase and the related Letter of Transmittal and any supplements or amendments thereto. The Offer is not being made to, nor will tenders be accepted from or on behalf of, holders of Shares residing in any jurisdiction in which the making of the Offer or acceptance thereof would not be in compliance with the securities laws of such jurisdiction. In any jurisdiction where the securities, blue sky or other laws require the Offer to be made by a licensed broker or dealer, the Offer will be deemed to be made on behalf of the Offeror by one or more registered brokers or dealers licensed under the laws of such jurisdiction.

EXHIBIT (A) (5)

Notice of Guaranteed Delivery for

Tender of Shares of Common Stock

οf

Odwalla, Inc. to

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company (Not To Be Used For Signature Guarantees)

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

This form, or one substantially equivalent hereto, must be used to accept the Offer (as defined below) if certificates for shares of common stock, no par value per share (the "Shares"), of Odwalla Inc., a California corporation ("Odwalla"), are not immediately available or if the procedure for book-entry transfer cannot be completed on a timely basis or time will not permit all required documents to reach the Depositary prior to the Expiration Date (as defined in the Offer to Purchase). Such form may be delivered by hand, facsimile transmission or mail to the Depositary. See Section 3 of the Offer to Purchase, dated November 6, 2001 (the "Offer to Purchase").

THE DEPOSITARY FOR THE OFFER IS:

EQUISERVE TRUST COMPANY, N.A.

<Table> <S>

BY MAIL:

<C>

BY HAND:

<C>

EquiServe Trust Company, N.A.

EquiServe Trust Company, N.A. P.O. Box 43034 c/o Securities Transfer and Reporting Services Inc.
Providence, RI 02940-3034 100 William Street--Galleria New York, NY 10038

<C>

EquiServe Trust Company, N.A. 40 Campanelli Drive Braintree, MA 02184

BY OVERNIGHT COURIER:

</Table>

<Table>

<S>

FACSIMILE FOR ELIGIBLE INSTITUTIONS ONLY: (781) 575-4826 or (781) 575-4827

</Table>

TO CONFIRM RECEIPT OF NOTICE OF GUARANTEED DELIVERY: (781) 575-4816

If you require additional information, please call Innisfree M&A Incorporated at (888) 750-5834 or Morgan Stanley & Co. Incorporated, the Dealer Manager at (212) 761-3539.

DELIVERY OF THIS NOTICE OF GUARANTEED DELIVERY TO AN ADDRESS, OR TRANSMISSION OF INSTRUMENTS VIA A FACSIMILE, OTHER THAN AS SET FORTH ABOVE, DOES NOT CONSTITUTE A VALID DELIVERY TO THE DEPOSITARY.

THIS NOTICE OF GUARANTEED DELIVERY IS NOT TO BE USED TO GUARANTEE SIGNATURES. IF A SIGNATURE ON A LETTER OF TRANSMITTAL IS REQUIRED TO BE GUARANTEED BY AN "ELIGIBLE INSTITUTION" (AS DEFINED IN THE OFFER TO PURCHASE) UNDER THE INSTRUCTIONS THERETO, SUCH SIGNATURE GUARANTEE MUST APPEAR IN THE APPLICABLE SPACE PROVIDED IN THE SIGNATURE BOX ON THE LETTER OF TRANSMITTAL.

THE ELIGIBLE INSTITUTION THAT COMPLETES THIS FORM MUST COMMUNICATE THE GUARANTEE TO THE DEPOSITARY AND MUST DELIVER THE LETTER OF TRANSMITTAL OR AN AGENT'S MESSAGE AND CERTIFICATES FOR SHARES TO THE DEPOSITARY WITHIN THE TIME PERIOD SHOWN HEREIN. FAILURE TO DO SO COULD RESULT IN A FINANCIAL LOSS TO SUCH ELIGIBLE INSTITUTION.

THE GUARANTEE ON THE REVERSE SIDE MUST BE COMPLETED.

# LADIES AND GENTLEMEN:

The undersigned hereby tender(s) to TCCC Acquisition Corp., a California corporation, upon the terms and subject to the conditions set forth in the Offer to Purchase, and the related Letter of Transmittal, receipt of which is hereby acknowledged, the number of Shares of Odwalla indicated below, pursuant to the guaranteed delivery procedure set forth in Section 3 of the Offer to Purchase.

<Table> <S>

Number of Shares:

<C>

SIGN HERE

	Name(s) of Record Holder(s):
Certificate No(s) (if available):	
If Securities will be tendered by book-entry transfer:	(PLEASE PRINT) Address(es):
Name of Tendering Institution: CODE)	(ZIP
Name of Book Entry Transfer Facility:	Area Code and Telephone No(s):
Account No.: at	Signature(s):
Dated:	

	GUARANTEE	
(NOT TO BE USED FOR SIGNATURE GUARANTE:		
The undersigned, a bank, broker, dealer, credit union or other entity which is a member in good standing of the Agents Medallion Program or a bank, broker, dealer, credit association or other entity which is an "eligible guaranto such term is defined in Rule 17Ad-15 under the Securities is as amended, guarantees the delivery to the Depositary of the hereby, together with a properly completed and duly execut Transmittal (or manually signed facsimile(s) thereof) and documents, or an Agent's Message (as defined in the Offer case of a book-entry delivery of Shares, all within three date hereof. A "trading day" is any day on which the Nasdacopen for business.	Securities Transfer union, savings r institution," as Exchange Act of 1934, he Shares tendered ed Letter of any other required to Purchase) in the trading days of the	
Name of Firm:	Authorized Signature:	
- Address:		
-	(DIFACE DDING OD GVDE)	
(7TD CODE)	(PLEASE PRINT OR TYPE) Title:	
(ZIP CODE)	·	
Area Code and Telephone No.:	Dated:	
DO NOT SEND CERTIFICATES FOR SHARES WITH THIS FORM -- CERTIFICATES SHOULD ONLY BE SENT WITH THE LETTER OF TRANSMITTAL.

</Table>

### GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION

### NUMBER ON SUBSTITUTE FORM W-9

Guidelines for Determining the Proper Identification Number to Give The Payer.--Social Security numbers have nine digits separated by two hyphens: i.e. 000-00-0000. Employer identification numbers have nine digits separated by only one hyphen: i.e. 00-0000000. The table below will help determine the number to give the payer.

\_\_\_\_\_\_

<tab <c></c></tab 	ole> <s></s>	<c></c>
FOR	THIS TYPE OF ACCOUNT:	GIVE THE NAME AND SOCIAL SECURITY NUMBER OF
	An individual's account Two or more individuals (joint account)	The individual The actual owner of the account or, if combined funds, the first individual on the account(1)
3.	Husband and wife (joint account)	The actual owner of the account or, if joint funds, the first individual on the account(1)
4.	Custodian account of a minor (Uniform Gift to Minors Act)	The minor(2)
5.	Adult and minor (joint account)	The adult, or if the minor is the only contributor, the minor(1)
6.	Account in the name of guardian or committee for a designated ward, minor or incompetent person	The ward, minor or incompetent person(3)
7.	a. A revocable savings trust account (in which grantor is also trustee)	The grantor-trustee(1)
	b. Any "trust" account that is not a legal or valid trust under State law	The actual owner(1)
_		
	Sole proprietorship account	The owner(4)
FOR	THIS TYPE OF ACCOUNT:	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF
FOR		GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in
FOR9.	THIS TYPE OF ACCOUNT:  A valid trust, estate or pension trust  Corporate account	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title.) (5) The corporation
FOR	THIS TYPE OF ACCOUNT:  A valid trust, estate or pension trust  Corporate account	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title.) (5)
FOR9.	THIS TYPE OF ACCOUNT:  A valid trust, estate or pension trust  Corporate account Religious, charitable or educational organization account	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title.) (5) The corporation
FOR9.	THIS TYPE OF ACCOUNT:  A valid trust, estate or pension trust  Corporate account Religious, charitable or educational organization account Partnership account held in the name	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title.) (5) The corporation The organization
FOR 9.  10. 11. 12.	THIS TYPE OF ACCOUNT:  A valid trust, estate or pension trust  Corporate account Religious, charitable or educational organization account Partnership account held in the name of the business Association, club, or other tax-exempt	GIVE THE EMPLOYER IDENTIFICATION NUMBER OF  The legal entity (Do not furnish the identifying number of the personal representative or trustee unless the legal entity itself is not designated in the account title.) (5) The corporation The partnership

- \_\_\_\_\_
- (1) List first and circle the name of the person whose number you furnish.(2) Circle the minor's name and furnish the minor's social security number.
- (3) Circle the ward's, minor's or incompetent person's name and furnish such person's social security number.
- (4) Show the name of the owner, but you may also enter your business or "DBA" name. You may either use your SSN or EIN (if you have one).
- (5) List first and circle the name of the legal trust, estate, or pension trust.

NOTE: If no name is circled when there is more than one name, the number will be considered to be that of the first name listed.

### OBTAINING A NUMBER

</Table>

If you don't have a taxpayer identification number or you don't know your number, obtain Form SS-5, Application for a Social Security Card, or Form SS-4,

Application for Employer Identification Number (for businesses and all other entities), or Form W-7 for Individual Taxpayer Identification Number (for alien individuals not eligible to get an SSN), at an office of the Social Security Administration or the Internal Revenue Service.

### PAYEES EXEMPT FROM BACKUP WITHHOLDING

Payees specifically exempted from backup withholding on certain payments include the following:

- A corporation.
- A financial institution.
- An organization exempt from tax under section 501(a), or an individual retirement plan, or a custodial account under section 403(b)(7) if the account satisfies the requirements of Section 401(f)(2).
- The United States or any agency or instrumentality thereof.
- A State, the District of Columbia, a possession of the United States, or any political subdivision or instrumentality thereof.
- A foreign government, a political subdivision of a foreign government, or any agency or instrumentality thereof.
- An international organization or any agency or instrumentality thereof.
- A registered dealer in securities or commodities required to register in the U.S., the District of Columbia or a possession of the U.S.
- A real estate investment trust.
- A common trust fund operated by a bank under section 584(a).
- An entity registered at all times during the tax year under the Investment Company Act of 1940.
- A foreign central bank of issue.
- A middleman known in the investment community as a nominee or custodian.

Payments of dividends and patronage dividends not generally subject to backup withholding include the following:

- Payments to nonresident aliens subject to withholding under section 1441.
- Payments to partnerships not engaged in a trade or business in the U.S. and which have at least one nonresident alien partner.
- Payments of patronage dividends where the amount received is not paid in money.
- Payments made by certain foreign organizations.

Payments of interest not generally subject to backup withholding include the following:

- Payments of interest on obligations issued by individuals. NOTE: You may be subject to backup withholding if this interest is \$600 or more and is paid in the course of the payer's trade or business and you have not provided your correct taxpayer identification number to the payer.
- Payments of tax-exempt interest (including exempt-interest dividends under section 852).
- Payments described in section 6049(b)(5) to nonresident aliens.
- Payments on tax-free covenant bonds under section 1451.
- Payments made by certain foreign organizations.

Exempt payees described above, except foreign exempt payees should file a Substitute Form W-9 to avoid possible erroneous backup withholding. FILE THIS FORM WITH THE PAYER, FURNISH YOUR TAXPAYER IDENTIFICATION NUMBER, WRITE "EXEMPT" ON THE FACE OF THE FORM, AND RETURN IT TO THE PAYER. IF THE PAYMENTS ARE INTEREST, DIVIDENDS, OR PATRONAGE DIVIDENDS, ALSO SIGN AND DATE THE FORM.

Foreign exempt payees shall file the appropriate IRS Form W-8 with the payer.

Certain payments other than interest, dividends and patronage dividends that are not subject to information reporting are also not subject to backup withholding. For details, see the regulations under sections 6041, 6041A(a), 6045, and 6050A.

PRIVACY ACT NOTICE--Section 6109 requires most recipients of dividend, interest, or other payments to give taxpayer identification numbers to payers who must report the payments to the IRS. The IRS uses the numbers for identification purposes and to help verify the accuracy of your tax return. Payers must be given the numbers whether or not recipients are required to file tax returns. Payers must generally withhold 30.5% of taxable interest, dividend and certain other payments to a payee who does not furnish a taxpayer identification number to a payer. Certain penalties may also apply.

### PENALTIES

- (1) PENALTY FOR FAILURE TO FURNISH TAXPAYER IDENTIFICATION NUMBER.--If you fail to furnish your taxpayer identification number to a payer, you are subject to a penalty of \$50 for each such failure unless your failure is due to reasonable cause and not to willful neglect.
- (2) CIVIL PENALTY FOR FALSE INFORMATION WITH RESPECT TO WITHHOLDING.--If you make a false statement with no reasonable basis which results in no imposition of backup withholding, you are subject to a penalty of \$500.

(3) CRIMINAL PENALTY FOR FALSIFYING INFORMATION.—Willfully falsifying certifications or affirmations may subject you to criminal penalties including fines and/or imprisonment.

FOR ADDITIONAL INFORMATION CONTACT YOUR TAX CONSULTANT OR THE INTERNAL REVENUE SERVICE.

Department of the Treasury INTERNAL REVENUE SERVICE

# INSTRUCTIONS FOR FORM W-8BEN (Rev. August 2001)

(Use with the December 2000 revision of Form W-8BEN.) CERTIFICATE OF FOREIGN STATUS OF BENEFICIAL OWNER FOR UNITED STATES TAX WITHHOLDING

Section references are to the Internal Revenue Code unless otherwise noted.

GENERAL INSTRUCTIONS

NOTE: For definitions of terms used throughout these instructions, see Definitions on pages 2 and 3.

PURPOSE OF FORM. Foreign persons are subject to U.S. tax at a 30% rate on income they receive from U.S. sources that consists of:

- Interest (including certain original issue discount (OID));
- Dividends;
- Rents:
- Royalties;
- Premiums;
- Annuities:
- Compensation for, or in expectation of, services performed;
- Substitute payments in a securities lending transaction;

or

- Other fixed or determinable annual or periodical gains, profits, or income.

This tax is imposed on the gross amount paid and is generally collected by withholding on that amount. A payment is considered to have been made whether it is made directly to the beneficial owner or to another person, such as an intermediary, agent, or partnership, for the benefit of the beneficial owner.

If you receive certain types of income, you must provide Form W-8BEN to:

- Establish that you are not a U.S. person;
- Claim that you are the beneficial owner of the income for which Form W-8BEN is being provided; and
- If applicable, claim a reduced rate of, or exemption from, withholding as a resident of a foreign country with which the United States has an income tax treaty.

You may also be required to submit Form W-8BEN to claim an exception from domestic information reporting and backup withholding for certain types of income that are not subject to foreign-person withholding. Such income includes:

- Broker proceeds.
- Short-term (183 days or less) original issue discount (OID).
- Bank deposit interest.
- Foreign source interest, dividends, rents, or royalties.
- Proceeds from a wager placed by a nonresident alien individual in the games of blackjack, baccarat, craps, roulette, or big-6 wheel.

You may also use Form W-8BEN to certify that income from a notional principal contract is not effectively connected with the conduct of a trade or business in the United States.

A withholding agent or payer of the income may rely on a properly completed Form W-8BEN to treat a payment associated with the Form W-8BEN as a payment to a foreign person who beneficially owns the amounts paid. If applicable, the withholding agent may rely on the Form W-8BEN to apply a reduced rate of withholding at source.

Provide Form W-8BEN to the withholding agent or payer before income is paid or credited to you. Failure to provide a Form W-8BEN when requested may lead to withholding at a 30% rate (foreign-person withholding) or the backup withholding rate.

NOTE: For additional information and instructions for the withholding agent, see the INSTRUCTIONS FOR THE REQUESTER OF FORMS W-8BEN, W-8ECI, W-8EXP, AND W-STMV

WHO MUST FILE. You must give Form W-8BEN to the withholding agent or payer if you are a foreign person and you are the beneficial owner of an amount subject to withholding. Submit Form W-8BEN when requested by the withholding agent or payer whether or not you are claiming a reduced rate of, or exemption from, withholding.

- You are a U.S. citizen (even if you reside outside the United States) or other U.S. person (including a resident alien individual). Instead, use FORM W-9, Request for Taxpayer Identification Number and Certification.
- You are a disregarded entity with a single owner that is a U.S. person and you are not a hybrid entity claiming treaty benefits. Instead, provide Form W-9.
- You are a nonresident alien individual who claims exemption from withholding on compensation for independent or dependent personal services performed in the United States. Instead, provide FORM 8233, Exemption from Withholding on Compensation for Independent (and Certain Dependent) Personal Services of a Nonresident Alien Individual, or FORM W-4, Employee's Withholding Allowance Certificate.
- You are receiving income that is effectively connected with the conduct of a trade or business in the United States. Instead, provide FORM W-8ECI, Certificate of Foreign Person's Claim for Exemption From Withholding on Income Effectively Connected With the Conduct of a Trade or Business in the United States. If any of the income for which you have provided a Form W-8BEN becomes effectively connected, this is a change in circumstances and Form W-8BEN is no longer valid. You must file Form W-8ECI. See CHANGE IN CIRCUMSTANCES BELOW.
- You are filing for a foreign government, international organization, foreign central bank of issue, foreign tax-exempt organization, foreign private foundation, or government of a U.S. possession claiming the applicability of section 115(2), 501(c), 892, 895, or 1443(b). Instead, provide FORM W-8EXP, Certificate of Foreign Government or Other Foreign Organization for United States Tax Withholding. However, you should use Form W-8BEN if you are claiming treaty benefits or are providing the form only to claim you are a foreign person exempt from backup withholding. You should use Form W-8ECI if you received effectively connected income (for example, income from commercial activities).
- You are a foreign flow-through entity, other than a hybrid entity, claiming treaty benefits. Instead, provide FORM W-8IMY, Certificate of Foreign Intermediary, Foreign Flow-Through Entity, or Certain U.S. Branches for United States Tax Withholding. However, if you are a partner, beneficiary, or owner of a flow-through entity and you are not yourself a flow-through entity, you may be required to furnish a Form W-8BEN to the flow-through entity.
- You are a reverse hybrid entity transmitting beneficial owner documentation provided by your interest holders to claim treaty benefits on their behalf. Instead, provide Form W-8IMY.
- You are a withholding foreign partnership or a withholding foreign trust. A withholding foreign partnership or a withholding foreign trust is a foreign partnership or trust that has entered into a withholding agreement with the IRS under which it agrees to assume primary withholding responsibility for each partner's, beneficiary's, or owner's distributive share of income subject to withholding that is paid to the partnership or trust. Instead, provide Form W-8TMY.
- You are acting as an intermediary (that is, acting not for your own account, but for the account of others as an agent, nominee, or custodian). Instead, provide Form W-8IMY.

GIVING FORM W-8BEN TO THE WITHHOLDING AGENT. DO NOT send Form W-8BEN to the IRS. Instead, give it to the person who is requesting it from you. Generally, this will be the person from whom you receive the payment or who credits your account. Give Form W-8BEN to the person requesting it before the payment is made to you or credited to your account. If you do not provide this form, the withholding agent may have to withhold at a 30% rate (foreign-person withholding) or backup withholding rate. If you receive more than one type of income from a single withholding agent for which you claim different benefits, the withholding agent may, at its option, require you to submit a Form W-8BEN for each different type of income. Generally, a separate Form W-8BEN must be given to each withholding agent.

NOTE: If you own the income or account jointly with one or more other persons, the income or account will be treated by the withholding agent as owned by a foreign person if Forms W-8BEN are provided by all of the owners. If the withholding agent receives a Form W-9 from any of the joint owners, the payment must be treated as made to a U.S. person.

# Cat. No. 25576H

CHANGE IN CIRCUMSTANCES. If a change in circumstances makes any information on the Form W-8BEN you have submitted incorrect, you must notify the withholding agent or payer within 30 days of the change in circumstances and you must file a new Form W-8BEN or other appropriate form.

If you use Form W-8BEN to certify that you are a foreign person, a change of address to an address in the United States is a change in circumstances. Generally, a change of address within the same foreign country or to another foreign country is not a change in circumstances. However, if you use Form W-8BEN to claim treaty benefits, a move to the United States or outside the country where you have been claiming treaty benefits is a change in circumstances. In that case, you must notify the withholding agent or payer within 30 days of the move.

If you become a U.S. citizen or resident after you submit Form W-8BEN, you are no longer subject to the 30% foreign-person withholding rate. You must notify the withholding agent or payer within 30 days of becoming a U.S. citizen or

resident. You may be required to provide a Form W-9. For more information, see Form W-9 and instructions.

EXPIRATION OF FORM W-8BEN. Generally, a Form W-8BEN provided without a U.S. taxpayer identification number (TIN) will remain in effect for a period starting on the date the form is signed and ending on the last day of the third succeeding calendar year, unless a change in circumstances makes any information on the form incorrect. For example, a Form W-8BEN signed on September 30, 2001, remains valid through December 31, 2004. A Form W-8BEN furnished with a U.S. TIN will remain in effect until a change in circumstances makes any information on the form incorrect, provided that the withholding agent reports on Form 1042-S at least one payment annually to the beneficial owner who provided the Form W-8BEN. See Line 6 on page 4 for circumstances under which you must provide a U.S. TIN.

### DEFINITIONS

BENEFICIAL OWNER. For payments other than those for which a reduced rate of withholding is claimed under an income tax treaty, the beneficial owner of income is generally the person who is required under U.S. tax principles to include the income in gross income on a tax return. A person is not a beneficial owner of income, however, to the extent that person is receiving the income as a nominee, agent, or custodian, or to the extent the person is a conduit whose participation in a transaction is disregarded. In the case of amounts paid that do not constitute income, beneficial ownership is determined as if the payment were income.

Foreign partnerships, foreign simple trusts, and foreign grantor trusts are not the beneficial owners of income paid to the partnership or trust. The beneficial owners of income paid to a foreign partnership are generally the partners in the partnership, provided that the partner is not itself a partnership, foreign simple or grantor trust, nominee or other agent. The beneficial owners of income paid to a foreign simple trust (that is, a foreign trust that is described in section 651(a)) are generally the beneficiaries of the trust, if the beneficiary is not a foreign partnership, foreign simple or grantor trust, nominee or other agent. The beneficial owners of a foreign grantor trust (that is, a foreign trust to the extent that all or a portion of the income of the trust is treated as owned by the grantor or another person under sections 671 through 679) are the persons treated as the owners of the trust. The beneficial owners of income paid to a foreign complex trust (that is, a foreign trust that is not a foreign simple trust or foreign grantor trust) is the trust itself.

The beneficial owner of income paid to a foreign estate is the estate itself.

NOTE: A payment to a U.S. partnership, U.S. trust, or U.S. estate is treated as a payment to a U.S. payee that is not subject to 30% foreign-person withholding. A U.S. partnership, trust, or estate should provide the withholding agent with a Form W-9.

FOREIGN PERSON. A foreign person includes a nonresident alien individual, a foreign corporation, a foreign partnership, a foreign trust, a foreign estate, and any other person that is not a U.S. person. It also includes a foreign branch or office of a U.S. financial institution or U.S. clearing organization if the foreign branch is a qualified intermediary. Generally, a payment to a U.S. branch of a foreign person is a payment to a foreign person.

NONRESIDENT ALIEN INDIVIDUAL. Any individual who is not a citizen or resident of the United States is a nonresident alien individual. An alien individual meeting either the "green card test" or the "substantial presence test" for the calendar year is a resident alien. Any person not meeting either test is a nonresident alien individual. Additionally, an alien individual who is a resident of a foreign country under the residence article of an income tax treaty, or an alien individual who is a resident of Puerto Rico, Guam, the Commonwealth of the Northern Mariana Islands, the U.S. Virgin Islands, or American Samoa is a nonresident alien individual. SEE PUB. 519, U.S. Tax Guide for Aliens, for more information on resident and nonresident alien status.

NOTE: Even though a nonresident alien individual married to a U.S. citizen or resident alien may choose to be treated as a resident alien for certain purposes (for example, filing a joint income tax return), such individual is still treated as a nonresident alien for withholding tax purposes on all income except wages.

FLOW-THROUGH ENTITY. A flow-through entity is a foreign partnership (other than a withholding foreign partnership), a foreign simple or foreign grantor trust (other than a withholding foreign trust), or, for payments for which a reduced rate of withholding is claimed under an income tax treaty, any entity to the extent the entity is considered to be fiscally transparent (see below) with respect to the payment by an interest holder's jurisdiction.

HYBRID ENTITY. A hybrid entity is any person (other than an individual) that is treated as fiscally transparent (see below) in the United States but is not treated as fiscally transparent by a country with which the United States has an income tax treaty. Hybrid entity status is relevant for claiming treaty benefits. See LINE 9C on page 5.

REVERSE HYBRID ENTITY. A reverse hybrid entity is any person (other than an individual) that is not fiscally transparent under U.S. tax law principles but that is fiscally transparent under the laws of a jurisdiction with which the United States has an income tax treaty. See Line 9c on page 5.

FISCALLY TRANSPARENT ENTITY. An entity is treated as fiscally transparent with respect to an item of income for which treaty benefits are claimed to the extent that the interest holders in the entity must, on a current basis, take into account separately their shares of an item of income paid to the entity, whether or not distributed, and must determine the character of the items of income as if they were realized directly from the sources from which realized by the entity. For example, partnerships, common trust funds, and simple trusts or grantor trusts are generally considered to be fiscally transparent with respect to items of income received by them.

DISREGARDED ENTITY. A business entity that has a single owner and is not a corporation under Regulations section 301.7701-2 (b) is disregarded as an entity separate from its owner.

AMOUNTS SUBJECT TO WITHHOLDING. Generally, an amount subject to withholding is an amount from sources within the United States that is fixed or determinable annual or periodical (FDAP) income. FDAP income is all income included in gross income, including interest (as well as OID), dividends, rents, royalties, and compensation. FDAP income does not include most gains from the sale of property (including market discount and option premiums).

WITHHOLDING AGENT. Any person, U.S. or foreign, that has control, receipt, or custody of an amount subject to withholding or who can disburse or make payments of an amount subject to withholding is a withholding agent. The withholding agent may be an individual, corporation, partnership, trust, association, or any other entity, including (but not limited to) any foreign intermediary, foreign partnership, and U.S. branches of certain foreign banks and insurance companies. Generally, the person who pays (or causes to be paid) the amount subject to withholding to the foreign person (or to its agent) must withhold.

-----

#### SPECIFIC INSTRUCTIONS

NOTE: A hybrid entity should give Form W-8BEN to a withholding agent only for income for which it is claiming a reduced rate of withholding under an income tax treaty. A reverse hybrid entity should give Form W-8BEN to a withholding agent only for income for which no treaty benefit is being claimed.

# PART I

- LINE 1. Enter your name. If you are a disregarded entity with a single owner who is a foreign person and you are not claiming treaty benefits as a hybrid entity, this form should be completed and signed by your foreign single owner. If the account to which a payment is made or credited is in the name of the disregarded entity, the foreign single owner should inform the withholding agent of this fact. This may be done by including the name and account number of the disregarded entity on line 8 (reference number) of the form. However, if you are a disregarded entity that is claiming treaty benefits as a hybrid entity, this form should be completed and signed by you.
- LINE 2. If you are a corporation, enter the country of incorporation. If you are another type of entity, enter the country under whose laws you are created, organized, or governed. If you are an individual, enter N/A (for "not applicable").
- LINE 3. Check the ONE box that applies. By checking a box, you are representing that you qualify for this classification. You must check the box that represents your classification (for example, corporation, partnership, trust, estate, etc.) under U.S. tax principles. DO NOT check the box that describes your status under the law of the treaty country. If you are a partnership or

disregarded entity receiving a payment for which treaty benefits are being claimed, you MUST check the "Partnership" or "Disregarded entity" box. If you are a sole proprietor, check the "Individual" box, not the "Disregarded entity" box.

# [Caution Icon]

Only entities that are tax-exempt under section 501 should check the "Tax-exempt organizations" box. Such organizations should use Form W-8BEN only if they are claiming a reduced rate of withholding under an income tax treaty or some code exception other than section 501. Use Form W-8EXP if you are claiming an exemption from withholding under section 501.

LINE 4. Your permanent residence address is the address in the country where you claim to be a resident for purposes of that country's income tax. If you are giving Form W-8BEN to claim a reduced rate of withholding under an income tax treaty, you must determine your residency in the manner required by the treaty. Do not show the address of a financial institution, a post office box, or an

address used solely for mailing purposes. If you are an individual who does not have a tax residence in any country, your permanent residence is where you normally reside. If you are not an individual and you do not have a tax residence in any country, the permanent residence address is where you maintain your principal office.

- LINE 5. Enter your mailing address only if it is different from the address you show on line  $4\,.$
- LINE 6. If you are an individual, you are generally required to enter your social security number (SSN). To apply for an SSN, get FORM SS-5 from a Social Security Administration (SSA) office. Fill in Form SS-5 and return it to the SSA.

If you do not have an SSN and are not eligible to get one, you must get an individual taxpayer identification number (ITIN). TO APPLY FOR AN ITIN, file FORM W-7 with the IRS. It usually takes 4-6 weeks to get an ITIN.

If you are not an individual (for example, a foreign estate or trust), or you are an individual who is an employer or who is engaged in a U.S. trade or business as a sole proprietor, use FORM SS-4, Application for Employer Identification Number, to obtain an EIN. If you are a disregarded entity claiming treaty benefits as a hybrid entity, enter YOUR EIN.

You MUST provide a U.S. taxpayer identification number (TIN) if you are:

- 1. Claiming an exemption from withholding under section 871(f) for certain annuities received under qualified plans, or
- 2. A foreign grantor trust with 5 or fewer grantors, or
- 3. Claiming benefits under an income tax treaty.

However, a U.S. TIN is not required to be shown in order to claim treaty benefits on the following items of income:

- Dividends and interest from stocks and debt obligations that are actively traded;
- Dividends from any redeemable security issued by an investment company registered under the Investment Company Act of 1940 (mutual fund);
- Dividends, interest, or royalties from units of beneficial interest in a unit investment trust that are (or were upon issuance) publicly offered and are registered with the SEC under the Securities Act of 1933; and
- Income related to loans of any of the above securities.

NOTE: You may want to obtain and provide a U.S. TIN on Form W-8BEN even though it is not required. A Form W-8BEN containing a U.S. TIN remains valid for as long as your status and the information relevant to the certifications you make on the form remain unchanged provided at least one payment is reported to you annually on Form 1042-S.

LINE 7. If your country of residence for tax purposes has issued you a tax identifying number, enter it here. For example, if you are a resident of Canada, enter your Social Insurance Number.

LINE 8. This line may be used by the filer of Form W-8BEN or by the withholding agent to whom it is provided to include any referencing information that is useful to the withholding agent in carrying out its obligations. For example, withholding agents who are required to associate the Form W-8BEN with a particular Form W-8IMY may want to use line 8 for a referencing number or code that will make the association clear. A beneficial owner may use line 8 to include the number of the account for which he or she is providing the form. A foreign single owner of a disregarded entity may use line 8 to inform the withholding agent that the account to which a payment is made or credited is in the name of the disregarded entity (see instructions for line 1 on page 3).

# PART II

LINE 9A. Enter the country where you claim to be a resident for income tax treaty purposes. For treaty purposes, a person is a resident of a treaty country if the person is a resident of that country under the terms of the treaty.

LINE 9B. If you are claiming benefits under an income tax treaty, you must have a U.S. TIN unless one of the exceptions listed under Line 6 above applies.

LINE 9C. An entity (but not an individual) that is claiming a reduced rate of withholding under an income tax treaty must represent that it (A) derives the item of income for which the treaty benefit is claimed and (B) meets the limitation on benefits provisions contained in the treaty, if any.

An item of income may be derived by either the entity receiving the item of income or by the interest holders in the entity or, in certain circumstances, both. An item of income paid to an entity is considered to be derived by the entity only if the entity is not fiscally transparent under the laws of the entity's jurisdiction with respect to the item of income. An item of income paid

to an entity shall be considered to be derived by the interest holder in the entity only if (A) the interest holder is not fiscally transparent in its jurisdiction with respect to the item of income and (B) the entity is considered to be fiscally transparent under the laws of the interest holder's jurisdiction with respect to the item of income. An item of income paid directly to a type of entity specifically identified in a treaty as a resident of a treaty jurisdiction is treated as derived by a resident of that treaty jurisdiction.

If an entity is claiming treaty benefits on its own behalf, it should complete Form W-8BEN. If an interest holder in an entity that is considered fiscally transparent in the interest holder's jurisdiction is claiming a treaty benefit, the interest holder should complete Form W-8BEN on its own behalf and the fiscally transparent entity should associate the interest holder's Form W-8BEN with a Form W-8IMY completed by the entity.

NOTE: An income tax treaty may not apply to reduce the amount of any tax on an item of income received by an entity that is treated as a domestic corporation for U.S. tax purposes. Therefore, neither the domestic corporation nor its shareholders are entitled to the benefits of a reduction of U.S. income tax on an item of income received from U.S. sources by the corporation.

To determine whether an entity meets the limitation on benefits provisions of a treaty, you must consult the specific provisions or articles under the treaties. Income tax treaties are available on the IRS Web Site at www.irs.gov/in info/treaties.html.

NOTE: If you are an entity that derives the income as a resident of a treaty country, you may check this box if the applicable income tax treaty does not contain a "limitation on benefits" provision.

LINE 9D. If you are a foreign corporation claiming treaty benefits under an income tax treaty that entered into force before January 1, 1987 (and has not been renegotiated) on (A) U.S. source dividends paid to you by another foreign corporation or (B) U.S. source interest paid to you by a U.S. trade or business of another foreign corporation, you must generally be a "qualified resident" of a treaty country. See section 884 for the definition of interest paid by a U.S. trade or business of a foreign corporation ("branch interest") and other applicable rules.

In general, a foreign corporation is a qualified resident of a country if one or more of the following applies:

- It meets a 50% ownership and base erosion test.
- It is primarily and regularly traded on an established securities market in its country of residence or the United States.
- It carries on an active trade or business in its country of residence.
- It gets a ruling from the IRS that it is a qualified resident.

See Regulations section 1.884-5 for the requirements that must be met to satisfy each of these tests.

### [Caution Icon]

If you are claiming treaty benefits under an income tax treaty entered into force after December 31, 1986, do not check box 9c. Instead, check box 9c.

LINE 9E. Check this box if you are related to the withholding agent within the meaning of section 267(b) or 707(b) and the aggregate amount subject to withholding received during the calendar year exceeds \$500,000. Additionally, you must file FORM 8833, Treaty-Based Return Position Disclosure Under Section 6114 or 7701(b).

LINE 10. Line 10 must be used ONLY if you are claiming treaty benefits that require that you meet conditions not covered by the representations you make in lines 9a through 9e. However, this line should always be completed by foreign students and researchers claiming treaty benefits. See SCHOLARSHIP AND FELLOWSHIP GRANTS below for more information.

Additional examples of persons who should complete this line are:

- 1. Exempt organizations claiming treaty benefits under the exempt organization articles of the treaties with Canada, Mexico, Germany, and the Netherlands.
- 2. Persons claiming an exemption under a personal services article that contains a monetary threshold.
- Foreign corporations that are claiming a preferential rate applicable to dividends based on ownership of a specific percentage of stock.
   Persons claiming treaty benefits on royalties if the treaty contains different withholding rates for different types of royalties.

This line is generally not applicable to claiming treaty benefits under an interest or dividends (other than dividends subject to a preferential rate based on ownership) article of a treaty.

SCHOLARSHIP AND FELLOWSHIP GRANTS. A nonresident alien student (including a trainee or business apprentice) or researcher who receives scholarship or fellowship grant income may use Form W-8BEN to claim benefits under a tax treaty that apply to reduce or eliminate U.S. tax on such income. NO FORM W-8BEN IS REQUIRED UNLESS A TREATY BENEFIT IS BEING CLAIMED. A nonresident alien student or researcher who receives compensation for personal services should use Form 8233 to claim any benefits of a tax treaty that apply to such compensation if the compensation is included in, or is in addition to, the individual's scholarship or fellowship grant income.

NOTE: If you are a nonresident alien individual who received noncompensatory scholarship or fellowship income and personal services income FROM THE SAME WITHHOLDING AGENT, you may use Form 8233 to claim a tax treaty withholding exemption for part or all of both types of income.

Generally, only a nonresident alien individual may use the terms of a tax treaty to reduce or eliminate U.S. tax on income from a scholarship or fellowship grant. However, most tax treaties contain a provision known as a "saving clause." Exceptions specified in the saving clause may permit an exemption from tax to continue for scholarship or fellowship grant income even after the recipient has otherwise become a U.S. resident alien for tax purposes. Thus, a student or researcher may continue to use Form W-8BEN to claim a tax treaty benefit if the withholding agent has otherwise indicated an intention to withhold on a scholarship or fellowship grant.

EXAMPLE. Article 20 of the U.S.-China income tax treaty allows an exemption from tax for scholarship income received by a Chinese student temporarily present in the United States. Under U.S. law, this student will become a resident alien for tax purposes if his or her stay in the United States exceeds 5 calendar years. However, paragraph 2 of the first Protocol to the U.S.-China treaty (dated April 30, 1984) allows the provisions of Article 20 to continue to apply even after the Chinese student becomes a resident alien of the United States.

COMPLETING LINES 4 AND 9A. Most tax treaties that contain an article exempting scholarship or fellowship grant income from taxation require that the recipient be a resident of the other treaty country at the time of, or immediately prior to, entry into the United States. Thus, a student or researcher may claim the exemption even if he or she no longer has a permanent address in the other treaty country after entry into the United States. If this is the case, you may provide a U.S. address on line 4 and still be eligible for the exemption if all other conditions required by the tax treaty are met. You must also identify on line 9a the tax treaty country of which you were a resident at the time of, or immediately prior to, your entry into the United States.

COMPLETING LINE 10. You must complete line 10 if you are a student or researcher claiming an exemption from taxation on your scholarship or fellowship grant income under a tax treaty. You must identify the applicable treaty article. Additionally, if you are a U.S. resident alien and are relying on an exception contained in the saving clause of a tax treaty to claim exemption from taxation on your scholarship or fellowship income, you must specify the article number (or location) in the tax treaty that contains the saving clause and its exceptions.

# PART III

If you check this box, you must provide the withholding agent with the required statement for income from a notional principal contract that is to be treated as income not effectively connected with the conduct of a trade or business in the United States. You should update this statement as often as necessary. A new Form W-8BEN is not required for each update provided the form otherwise remains valid.

### PART IV

Form W-8BEN must be signed and dated by the beneficial owner of the income, or, if the beneficial owner is not an individual, by an authorized representative or officer of the beneficial owner. If Form W-8BEN is completed by an agent acting under a duly authorized power of attorney, the form must be accompanied by the power of attorney in proper form or a copy thereof specifically authorizing the agent to represent the principal in making, executing, and presenting the form. FORM 2848, Power of Attorney and Declaration of Representative, may be used for this purpose. The agent, as well as the beneficial owner, may incur liability for the penalties provided for an erroneous, false, or fraudulent form.

BROKER TRANSACTIONS OR BARTER EXCHANGES. Income from transactions with a broker, or barter exchanges, is subject to reporting rules and backup withholding unless Form W-8BEN or a substitute form is filed to notify the broker or barter exchange that you are an exempt foreign person.

You are an exempt foreign person for a calendar year in which: (A) you are a nonresident alien individual or a foreign corporation, partnership, estate, or trust; (B) you are an individual who has not been, and does not plan to be, present in the United States for a total of 183 days or more during the calendar year; and (C) you are neither engaged, nor plan to be engaged during the year,

in a U.S. trade or business that has effectively connected gains from transactions with a broker or barter exchange.

\_\_\_\_\_

PAPERWORK REDUCTION ACT NOTICE. We ask for the information on this form to carry out the Internal Revenue laws of the United States. You are required to provide the information. We need it to ensure that you are complying with these laws and to allow us to figure and collect the right amount of tax.

You are not required to provide the information requested on a form that is subject to the Paperwork Reduction Act unless the form displays a valid OMB control number. Books or records relating to a form or its instructions must be retained as long as their contents may become material in the administration of any Internal Revenue law. Generally, tax returns and return information are confidential, as required by section 6103.

The time needed to complete and file this form will vary depending on individual circumstances. The estimated average time is: RECORDKEEPING,  $5\ hr.$ ,  $58\ min.$ ; LEARNING ABOUT THE LAW OR THE FORM,  $3\ hr.$ ,  $46\ min.$ ; PREPARING AND SENDING THE FORM TO IRS,  $4\ hr.$ ,  $2\ min.$ 

If you have comments concerning the accuracy of these time estimates or suggestions for making this form simpler, we would be happy to hear from you. You can write to the Tax Forms Committee, Western Area Distribution Center, Rancho Cordova, CA 95743-0001. Do not send Form W-8BEN to this office. Instead, give it to your withholding agent.

-----

THIS ANNOUNCEMENT IS NEITHER AN OFFER TO PURCHASE NOR A SOLICITATION OF AN OFFER TO SELL SHARES (AS DEFINED BELOW. THE OFFER (AS DEFINED BELOW) IS MADE SOLELY BY THE OFFER TO PURCHASE, DATED NOVEMBER 6, 2001, AND THE RELATED LETTER OF TRANSMITTAL, AND ANY AMENDMENTS OR SUPPLEMENTS THERETO, AND IS BEING MADE TO ALL HOLDERS OF SHARES. THE OFFER, HOWEVER, IS NOT BEING MADE TO (NOR WILL TENDERS BE ACCEPTED FROM OR ON BEHALF OF) HOLDERS OF SHARES IN ANY JURISDICTION IN WHICH THE OFFER OR ACCEPTANCE THEREOF WOULD NOT BE IN COMPLIANCE WITH THE SECURITIES, BLUE SKY OR OTHER LAWS OF SUCH JURISDICTION. HOWEVER, THE OFFEROR (AS DEFINED BELOW) MAY, IN ITS DISCRETION, TAKE SUCH ACTION AS IT MAY DEEM NECESSARY TO MAKE THE OFFER IN ANY JURISDICTION AND EXTEND THE OFFER TO HOLDERS OF SHARES IN SUCH JURISDICTION. IN THOSE JURISDICTIONS WHERE THE SECURITIES, BLUE SKY OR OTHER LAWS REQUIRE THE OFFER TO BE MADE BY A LICENSED BROKER OR DEALER, THE OFFER SHALL BE DEEMED TO BE MADE ON BEHALF OF THE OFFEROR BY MORGAN STANLEY & CO. INCORPORATED OR ONE OR MORE REGISTERED BROKERS OR DEALERS LICENSED UNDER THE LAWS OF SUCH JURISDICTIONS.

NOTICE OF OFFER TO PURCHASE FOR CASH

All Outstanding Shares of Common Stock

οf

Odwalla, Inc. at

\$15.25 Net Per Share by

TCCC Acquisition Corp. a wholly owned subsidiary of

The Coca-Cola Company

TCCC Acquisition Corp., a California corporation (the "Offeror") and a wholly owned subsidiary of The Coca-Cola Company, a Delaware corporation ("TCCC"), is offering to purchase all of the outstanding shares of common stock, no par value per share ("Shares"), of Odwalla, Inc., a California corporation ("Odwalla"), at a purchase price of \$15.25 per Share, net to the seller in cash less any required withholding taxes and without interest thereon (The "Offer Price"), upon the terms and subject to the conditions set forth in the Offer to Purchase, dated November 6, 2001 ("Offer to Purchase"), and in the related Letter of Transmittal (which, together with any amendments or supplements thereto, collectively constitute the "Offer").

THE OFFER AND WITHDRAWAL RIGHTS WILL EXPIRE AT 12:00 MIDNIGHT, SAN FRANCISCO TIME, ON THURSDAY, DECEMBER 6, 2001, UNLESS THE OFFER IS EXTENDED.

THE OFFER IS CONDITIONED UPON, AMONG OTHER THINGS, (I) THERE BEING VALIDLY TENDERED AND NOT WITHDRAWN PRIOR TO THE EXPIRATION OF THE OFFER SUCH NUMBER OF SHARES THAT WOULD CONSTITUTE AT LEAST NINETY AND ONE-TENTH PERCENT (90.1%) OF SHARES THAT ARE OUTSTANDING DETERMINED ON A FULLY DILUTED BASIS (INCLUDING FOR PURPOSES OF SUCH CALCULATION ALL SHARES THAT ARE ISSUABLE UPON EXERCISE OF VESTED OPTIONS AND OUTSTANDING WARRANTS BUT EXCLUDING SHARES THAT ARE ISSUABLE UPON THE EXERCISE OF OPTIONS AND WARRANTS THAT ARE CANCELLED PURSUANT TO THE MERGER AGREEMENT) (THE "MINIMUM CONDITION), (II) ANY WAITING PERIOD UNDER THE HSR ACT (AS DEFINED IN THE OFFER TO PURCHASE) APPLICABLE TO THE PURCHASE OF SHARES PURSUANT TO THE OFFER HAVING EXPIRED OR HAVING BEEN TERMINATED PRIOR TO THE EXPIRATION OF THE OFFER, AND (III) THE SATISFACTION OR WAIVER OF CERTAIN OTHER TERMS AND CONDITIONS.

The Offer is being made pursuant to the Agreement and Plan of Merger, dated as of October 29, 2001 (the "Merger Agreement"), among TCCC, the Offeror and Odwalla. The Merger Agreement provides that, among other things, after the purchase of Shares pursuant to the Offer and the satisfaction of the other conditions set forth in the Merger Agreement and in accordance with the relevant provisions of the California General Corporation Law, as amended (the "CGCL"), the Offeror will be merged with and into Odwalla (the "Merger"). Following consummation of the Merger, Odwalla will continue as the surviving corporation (the "Surviving Corporation") and will be a wholly owned subsidiary of TCCC. At the effective time of the Merger (the "Effective Time"), each Share that is issued and outstanding (other than Shares owned by TCCC, the Offeror, any other subsidiary of TCCC or the Offeror or Shares owned by Odwalla as treasury stock) will be converted into the right to receive from the Surviving Corporation \$15.25 in cash or greater amount as may be paid in the Offer, less any required withholding taxes and without interest thereon. If, however, after consummation of the Offer, the Offeror owns less than 90% of the then outstanding Shares, a vote of Odwalla's shareholders will be required under Section 1101 of the CGCL to approve the Merger, and a significantly longer period of time will be required to effect the Merger.

In connection with the Merger Agreement, the Offeror and TCCC entered into Tender Agreements each dated as of October 29, 2001 (the "Tender Agreements"), with certain shareholders (the "Tendering Shareholders"). Pursuant to the Tender

Agreements, the Tendering Shareholders have agreed to tender the 6,280,594 Shares owned by them (the "Committed Shares") pursuant to the Offer. The Committed Shares represent approximately 57% of Shares that, as of October 25, 2001, were issued and outstanding on a fully diluted basis.

THE BOARD OF DIRECTORS OF ODWALLA HAS ADOPTED THE MERGER AGREEMENT AND APPROVED THE OFFER AND THE MERGER, HAS DETERMINED THAT THE TERMS OF THE OFFER AND THE MERGER ARE FAIR TO, AND IN THE BEST INTERESTS OF, ODWALLA'S SHAREHOLDERS, AND RECOMMENDS THAT ODWALLA'S SHAREHOLDERS ACCEPT THE OFFER AND TENDER ALL THEIR SHARES PURSUANT TO THE OFFER.

For purposes of the Offer, the Offeror will be deemed to have accepted for payment, and thereby purchased Shares validly tendered and not withdrawn, if and when the Offeror gives oral followed by written notice to EquiServe Trust Company, N.A. (the "Depositary"), of the Offeror's acceptance of such Shares for payment. In all cases, payment for Shares purchased pursuant to the Offer will be made by deposit of the purchase price with the Depositary, which shall act as agent for tendering shareholders for the purpose of receiving payment from the Offeror and transmitting payment to the tendering shareholders. Payment for Shares accepted for payment pursuant to the Offer will be made only after timely receipt by the Depositary of (i) certificates for such Shares or timely confirmation of a book-entry transfer of such Shares into the Depositary's account at the Book-Entry Transfer Facility (as defined in the Offer to Purchase) pursuant to the procedures set forth in the Offer to Purchase, (ii) a properly completed and duly executed Letter of Transmittal (or facsimile thereof), with any required guarantees, or, in the case of a book-entry transfer, an Agent's Message (as defined in the Offer to Purchase) and (iii) any other documents required by the Letter of Transmittal.

If any of the conditions set forth in the Offer to Purchase that relate to the Offeror's obligations to purchase Shares are not satisfied by 12:00 Midnight, San Francisco Time, on December 6, 2001 (the "Expiration Date") (or any other time then set as the Expiration Date), the Offeror may, subject to the terms of the Merger Agreement, (i) terminate the Offer and return all tendered Shares to tendering shareholders,

(ii) waive such unsatisfied conditions and purchase all Shares validly tendered, or (iii) extend the Offer and, subject to the terms of the Offer (including the rights of shareholders to withdraw their shares), retain the shares which have been tendered, until the termination of the Offer, as extended.

Under the terms of the Merger Agreement, the Offeror may not, without the consent of Odwalla, (i) reduce the number of Shares subject to the Offer, (ii) reduce the Offer Price, (iii) impose any other conditions to the Offer other than the conditions set forth in Annex I to the Merger Agreement (the "Offer Conditions") or modify the Offer Conditions (other than to waive any Offer Conditions to the extent permitted by the Merger Agreement) in a manner adverse to the holders of Shares, (iv) except as provided in the Merger Agreement, extend the Offer, (v) change the form of consideration payable in the Offer, or (vi) amend any other term of the Offer in a manner adverse to the Odwalla shareholders. Notwithstanding the foregoing, the Offeror may, without the consent of Odwalla, (i) extend the Offer, if at the scheduled or extended expiration date of the Offer any of the Offer Conditions shall not be satisfied or waived, until such time as such conditions are satisfied or waived, subject in each case to any right of TCCC, the Offeror or Odwalla to terminate the Merger Agreement pursuant to the terms thereof or (ii) extend the Offer for any period required by any rule, regulation, interpretation or position of the Commission or the staff thereof applicable to the Offer. Further, if the Minimum Condition is not satisfied on any scheduled expiration date of the Offer, at the request of Odwalla, the Offeror shall, and TCCC shall cause the Offeror to take any of the actions set forth in clauses (x), (y), or (z) (provided that the Offeror shall have complete discretion concerning which action to take): (x) extend the Offer pursuant to the provisions of the Merger Agreement, (y) amend the Offer in contemplation of the exercise of the Option Agreement (as defined in the Offer to Purchase) (to the extent the Option Agreement is exercisable at such time) to reduce the Minimum Condition to that number of shares (the "Option Exercise Minimum Number") equal to the number of shares which, when combined with the number of shares issued upon exercise of the Option Agreement, equals 90.1% of Shares on a fully diluted basis (including for purposes of such calculation all Shares that are issuable upon exercise of vested options and outstanding warrants but excluding Shares that are issuable upon the exercise of options and warrants that are cancelled pursuant to the Merger Agreement) or (z) amend the Offer to provide that, if (i) the Minimum Condition is not satisfied at the next scheduled expiration date of the Offer (after giving effect to the issuance of any Shares theretofore acquired by TCCC or the Offeror) and (ii) the number of Shares tendered pursuant to the Offer and not withdrawn as of such next scheduled expiration date is more than 50% of the then outstanding Shares, the Offeror shall waive the Minimum Condition and amend the Offer to reduce the number of Shares subject to the Offer to 49.9% of Shares then outstanding (the "Revised Minimum Number") and, subject to the prior satisfaction or waiver of the other conditions of the Offer, purchase, on a pro rata basis, the Revised Minimum Number of shares (it being understood that the Offeror shall not in any event be required to accept for payment, or pay for, any Shares less than the Revised Minimum Number of Shares that are tendered pursuant to the Offer and not withdrawn at the expiration date).

Subject to the limitations set forth in the Offer and the Merger Agreement, the Offeror reserves the right (but will not be obligated), at any time or from time to time in its sole discretion, to extend the period during which the Offer is open by giving oral followed by written notice of such extension to the Depositary and by making a public announcement of such extension. There can be no assurance that the Offeror will exercise its right to extend the Offer. Any extension of the period during which the Offer is open will be followed, as promptly as practicable, by public announcement thereof, such announcement to be issued not later than 9:00 a.m., New York Time, on the next business day after the previously scheduled Expiration Date. During any such extension, all Shares previously tendered and not withdrawn will remain subject to the Offer, subject to the rights of a tendering shareholder to withdraw such shareholder's Shares.

Except as otherwise provided in Section 4 of the Offer to Purchase, tenders of Shares made pursuant to the Offer are irrevocable, except that Shares tendered pursuant to the Offer may be withdrawn at any time prior to the Expiration Date, and, unless theretofore accepted for payment, may also be withdrawn at any time after January 5, 2002. For a withdrawal to be effective, a written or facsimile transmission notice of withdrawal must be timely received by the Depositary at one of its addresses set forth on the back cover of the Offer to Purchase. Any such notice of withdrawal must specify the name of the person who tendered Shares to be withdrawn, the number of Shares to be withdrawn and the name of the registered holder if different from

the name of the person who tendered Shares. If certificates for Shares to be withdrawn have been delivered or otherwise identified to the Depositary, then prior to the physical release of such certificates, the serial numbers shown on such certificates must be submitted to the Depositary and, unless such Shares have been tendered for the account of an Eligible Institution (as defined in the Offer to Purchase), the signature on the notice of withdrawal must be Medallion guaranteed by an Eligible Institution. If Shares have been tendered pursuant to the procedures for book-entry transfer set forth in Section 3 of the Offer to Purchase, the notice of withdrawal must specify the name and number of the account at the Book-Entry Transfer Facility to be credited with withdrawn Shares, in which case a notice of withdrawal will be effective if delivered to the Depositary by any method of delivery described in this paragraph. All questions as to the form and validity (including time of receipt) of a notice of withdrawal will be determined by the Offeror, in its sole discretion, and its determination shall be final and binding on all parties.

The information required to be disclosed by paragraph (d) (1) of Rule 14d-6 of the General Rules and Regulations under the Securities Exchange Act of 1934, as amended, is contained in the Offer to Purchase and is incorporated herein by reference.

Odwalla has provided to the Offeror its list of shareholders and security position listings for the purpose of disseminating the Offer to holders of Shares. The Offer to Purchase, the related Letter of Transmittal and other related materials are being mailed to record holders of Shares and will be mailed to brokers, dealers, commercial banks, trust companies and similar persons whose names, or the names of whose nominees, appear on the shareholder lists or, if applicable, who are listed as participants in a clearing agency's security position listing, for subsequent transmittal to beneficial owners of Shares.

THE OFFER TO PURCHASE AND THE RELATED LETTER OF TRANSMITTAL CONTAIN IMPORTANT INFORMATION THAT SHOULD BE READ BEFORE ANY DECISION IS MADE WITH RESPECT TO THE OFFER.

Any questions or requests for assistance may be directed to the Information Agent or the Dealer Manager as set forth below. Requests for copies of the Offer to Purchase and the related Letter of Transmittal and other tender offer materials may be directed to the Information Agent as set forth below, and copies will be furnished promptly at the Offeror's expense. No fees or commissions will be payable to brokers, dealers or other persons (other than the Dealer Manager and the Information Agent) for soliciting tenders of Shares pursuant to the Offer.

THE INFORMATION AGENT FOR THE OFFER IS:

[INNISFREE M&A INCORPORATED LOGO]
501 Madison Avenue, 20th Floor
New York, New York 10022
Banks and Brokers Call Collect: 212-750-5833
All Others Call Toll-Free: 1-888-750-5834

THE DEALER MANAGER FOR THE OFFER IS:

[MORGAN STANLEY LOGO]

MORGAN STANLEY & CO. INCORPORATED

1585 Broadway

New York, New York 10036

(212) 761-3539

#### ODWALLA GROWTH OPPORTUNITIES MULTIPLY WITH COCA-COLA

Category Leading Brands Odwalla and Samantha will Reach New Markets Nationwide

Odwalla will Continue to be Led by Current Management Team

HALF MOON BAY, CA (October 30, 2001) -- Odwalla, Inc. (NASDAQ:ODWA) today announced it has signed a definitive agreement with The Coca-Cola Company (NYSE:KO) to become a separate business unit of The Minute Maid Company, The Coca-Cola Company's juice division. Under the terms of the transaction, approved today by the Boards of Directors of both companies, The Coca-Cola Company has agreed to acquire all outstanding shares of Odwalla common stock for \$15.25 per share in an all-cash tender offer.

After completion of the transaction, Odwalla will continue to be led by its current management.

Stephen Williamson, who will remain chief executive officer of Odwalla, said, "Our business is to provide people with easy access to great-tasting nourishment. The Coca-Cola Company believes in our vision, our products and our people. This opportunity is about sharing that vision and our products everywhere as we grow this company together. I know the entrepreneurial spirit of Odwalla will be nurtured by the opportunity for growth that this new relationship represents."

Don Short, president and chief executive officer of The Minute Maid Company, said, "Under the leadership of Stephen Williamson and his team, Odwalla's talented and proven people have built unique brands with loyal followings. Odwalla, as the North American leader in the super premium juice category, is a key component of Coca-Cola's strategy for building category leaders and profitable growth in new beverage categories. The innovation and expertise of the Odwalla team coupled with our innovation and logistics network are key to expanding the brands they have created and nurtured."

Short said that Williamson will report to Don Knauss, president and chief executive officer of Minute Maid Americas.

The super premium juice category is one of the fastest growing beverage categories. Odwalla offers an appealing range of juices, vegetable drinks, fortified beverages, smoothies, shakes, soy and lactic beverages; impressive chilled direct store distribution capabilities; and a culture of innovation.

Under the terms of the agreement, The Coca-Cola Company will shortly commence a tender offer for all outstanding shares of Odwalla common stock for \$15.25 per share in cash, or approximately \$181 million on a fully diluted basis. The companies said that certain principal shareholders of Odwalla, representing approximately 57 percent of the shares outstanding, have committed to tender their shares into the offer.

The tender offer will be followed by a merger, in which any remaining shares will be converted into the right to receive \$15.25 per share in cash. The closing of the tender offer is conditioned upon receiving at least 90% of the fully diluted shares of Odwalla's outstanding stock in the tender offer. However, if more than 50% but fewer than 90% of such shares are tendered, Coca-Cola may, under certain circumstances, reduce the number of shares subject to the offer to 49.9% of the outstanding shares and subsequently pursue a merger with Odwalla. The offer is also subject to the receipt of customary regulatory approvals and the satisfaction or waiver of other customary closing conditions.

Upon closing, the transaction is expected to be earnings-neutral in year one, and accretive thereafter as a result of realizing growth opportunities for Minute Maid and Odwalla.

This announcement is neither an offer to purchase nor a solicitation of an offer to sell shares of Odwalla. At the time the offer is commenced, Coca-Cola will file a tender offer statement with the Securities and Exchange Commission ("SEC") and Odwalla will file a solicitation/recommendation statement with respect to the offer. Odwalla shareholders are advised to read the tender offer statement regarding the acquisition of Odwalla referenced in this press release, and the related solicitation/recommendation statement. The tender offer statement (including an offer to purchase, letter of transmittal and related tender documents) and the solicitation/recommendation statement will contain important information which should be read carefully before any decision is made with respect to the offer. These documents will be made available to all shareholders of Odwalla at no expense to them. These documents will also be available to all Odwalla shareholders at no charge on the SEC's website at www.sec.gov.

Dresdner Kleinwort Wasserstein acted as financial advisor to Odwalla,

### ABOUT ODWALLA

Odwalla, Inc., the nation's leading branded super-premium beverage company, delivers nourishment coast to coast with the Odwalla and Samantha lines of all-natural juices, smoothies, dairy-free shakes, pure spring water and natural food bars. Its products are sold and distributed in over 5,000 retail locations, including supermarkets, specialty retailers, natural food stores, warehouse outlets, convenience stores, on-line grocers and food service operators through a direct-store-delivery system. Principal shareholders include Bain Capital, Catterton-Simon Partners and U.S. Equity Partners LP, a private equity fund controlled by Wasserstein & Co. To learn more about the Odwalla and Samantha brands, visit www.odwalla.com or www.freshsamantha.com.

#### ABOUT THE MINUTE MAID COMPANY

The Minute Maid Company is an operating unit of The Coca-Cola Company, the world's leading marketer of juices and juice drinks. The Company sells products under a number of brands, including Minute Maid, Simply Orange, Hi-C and Disney Beverages. To learn more about The Minute Maid Company and its products, visit www.minutemaid.com or www.simplyorangejuice.com

### ABOUT THE COCA-COLA COMPANY

The Coca-Cola Company is the world's largest beverage company and is the leading producer and marketer of soft drinks. Along with Coca-Cola, recognized as the world's best-known brand, the Company markets four of the world's top five soft drink brands, including diet Coke, Fanta and Sprite. Through the world's largest distribution system, consumers in nearly 200 countries enjoy the Company's products at a rate of more than 1 billion servings each day. For more information about The Coca-Cola Company, please visit its website at www.coca-cola.com

This press release contains statements, estimates or projections, not historical in nature, that may constitute "forward-looking statements" as defined under U.S. federal securities laws. These statements, which speak only as of the date given, are subject to certain risks and uncertainties that could cause actual results to differ materially from Coca-Cola's or Odwalla's historical experience and their present expectations or projections. These risks include, but are not limited to, their ability to finance expansion plans, share repurchase programs and general operating activities; changes in the non-alcoholic beverages business environment, including actions of competitors and changes in consumer preferences; regulatory and legal changes; fluctuations in the cost and availability of raw materials; interest rate and currency fluctuations; changes in economic and political conditions; their ability to penetrate developing and emerging markets; the effectiveness of their advertising and marketing programs; litigation uncertainties; adverse weather conditions; and other risks discussed in Coca-Cola's and Odwalla's filings with the Securities and Exchange Commission (the "SEC"), including their Annual Reports on Form 10-K, which filings are available from the SEC. Neither Coca-Cola nor Odwalla undertakes any obligation to publicly update or revise any forward-looking statements.

# # #

FOR MORE INFORMATION, CONTACT:

Ben Deutsch The Coca-Cola Company 404-676-2683

THE COCA-COLA COMPANY LAUNCHES TENDER OFFER FOR ODWALLA, INC.

ATLANTA (November 6, 2001) -- The Coca-Cola Company (NYSE: KO) today announced that its wholly owned subsidiary has commenced a previously announced tender offer for all the outstanding shares of common stock of Odwalla, Inc. at \$15.25 per share, net to the seller, in cash. The tender offer is being made pursuant to an Agreement and Plan of Merger, dated as of October 29, 2001. The tender offer will expire at 12:00 midnight, San Francisco time, on Thursday, December 6, 2001, unless extended. The offer is conditioned upon, among other things, there being validly tendered and not withdrawn a number of shares which equals at least 90.1% of the outstanding shares of Odwalla on a fully diluted basis. If more than 50% of the outstanding shares but fewer than 90.1% of the outstanding shares on a fully diluted basis are tendered, The Coca-Cola Company may, under certain circumstances, reduce the number of shares subject to the offer to 49.9% of the outstanding shares and subsequently pursue a merger with Odwalla. The offer is also subject to the receipt of customary regulatory approvals and the satisfaction or waiver of other customary closing conditions. EquiServe Trust Company, N.A. is the Depositary for the tender offer, Morgan Stanley is the Dealer Manager and Innisfree M&A Incorporated is the Information Agent.

-----

This announcement is neither an offer to purchase nor a solicitation of an offer to sell shares of Odwalla, Inc. The Coca-Cola Company is filing a tender offer statement today with the Securities and Exchange Commission (SEC) and Odwalla is filing a solicitation/recommendation statement with respect to the offer. Odwalla shareholders are advised to read the tender offer statement regarding the acquisition of Odwalla referenced in this press release, and the related solicitation/recommendation statement. The tender offer statement (including an offer to purchase, letter of transmittal and related tender documents) and the solicitation/recommendation statement contain important information which should be read carefully before any decision is made with respect to the offer. These documents will be made available to all stockholders of Odwalla at no expense to them. These documents will also be available at no charge on the SEC's web site at www.sec.gov.

### ABOUT ODWALLA

Odwalla, Inc., the nation's leading branded super-premium beverage company, delivers nourishment coast to coast with the Odwalla and Samantha lines of all-natural juices, smoothies, dairy-free shakes, pure spring water and natural food bars. Its products are sold and distributed in over 5,000 retail locations, including supermarkets, specialty retailers, natural food stores, warehouse outlets, convenience stores, on-line grocers and food service operators through a direct-store-delivery system. Principal shareholders include Bain Capital, Catterton-Simon Partners and U.S. Equity Partners LP, a private equity fund controlled by Wasserstein & Co. To learn more about the Odwalla and Samantha brands, visit www.odwalla.com or www.freshsamantha.com.

### ABOUT THE MINUTE MAID COMPANY

The Minute Maid Company is an operating unit of The Coca-Cola Company, the world's leading marketer of juices and juice drinks. The Company sells products under a number of brands, including Minute

Maid, Simply Orange, Hi-C and Disney Beverages. To learn more about The Minute Maid Company and its products, visit www.minutemaid.com or www.simplyorangejuice.com.

### ABOUT THE COCA-COLA COMPANY

The Coca-Cola Company is the world's largest beverage company and is the leading producer and marketer of soft drinks. Along with Coca-Cola, recognized as the world's best-known brand, the Company markets four of the world's top five soft drink brands, including diet Coke, Fanta and Sprite. Through the world's largest distribution system, consumers in nearly 200 countries enjoy the Company's products at a rate of more than 1 billion servings each day. For more information about The Coca-Cola Company, please visit its website at www.coca-cola.com.

This press release contains statements, estimates or projections, not historical in nature, that may constitute "forward-looking statements" as defined under U.S. federal securities laws. These statements, which speak only as of the date given, are subject to certain risks and uncertainties that could cause actual

results to differ materially from Coca-Cola's or Odwalla's historical experience and their present expectations or projections. These risks include, but are not limited to, their ability to finance expansion plans, share repurchase programs and general operating activities; changes in the non-alcoholic beverages business environment, including actions of competitors and changes in consumer preferences; regulatory and legal changes; fluctuations in the cost and availability of raw materials; interest rate and currency fluctuations; changes in economic and political conditions; their ability to penetrate developing and emerging markets; the effectiveness of their advertising and marketing programs; litigation uncertainties; adverse weather conditions; and other risks discussed in Coca-Cola's and Odwalla's filings with the SEC, including their Annual Reports on Form 10-K, which filings are available from the SEC. Neither Coca-Cola nor Odwalla undertakes any obligation to publicly update or revise any forward-looking statements.

#### EMPLOYMENT AGREEMENT

THIS AGREEMENT made this 29th day of October, 2001, by and between Odwalla, Inc. ("Company"), and Stephen C. Williamson ("Employee"),

### WITNESSETH:

WHEREAS, Company acknowledges that Employee has provided valuable service to Company prior to its anticipated merger with Perry Phillip Corp.; and,

WHEREAS, Company desires to retain Employee at the Effective Time, as defined in the Agreement and Plan of Merger dated October 29, 2001 among The Coca-Cola Company, Perry Phillip Corp. and Company (the "Close"); and,

WHEREAS, Employee is willing to be employed by Company; and

WHEREAS, upon the Close, Employee and Company wish to replace the existing employment agreement between Employee and Odwalla, Inc. dated December 21, 1999.

NOW, THEREFORE, in consideration of the mutual covenants and promises contained herein, the parties agree as follows:

- 1. Employment. Contingent upon the Close, Company hereby employs Employee as Chief Executive Officer and Employee hereby accepts such employment and agrees to perform such duties as are customarily performed by one holding such position and to render these or any other services and duties consistent with such position as may be assigned from time to time by Company.
- 2. Performance of Employee's Duties. The Employee agrees to devote substantially all of his business time to the faithful performance of his duties for Company and to render service to Company to the best of his ability, experience and talent to the reasonable satisfaction of Company. Such duties shall be rendered at such place or places as Company shall require in accordance with the best interests, needs, business and opportunities of Company. Employee's office will be located in the San Francisco Bay Area and his duties shall be primarily performed there.
- 3. Term of Employment. The term of employment shall commence upon the Close, and terminate on December 31, 2002, unless renewed by the parties. The term of employment will be renewed automatically for an additional one (1) year period on December 31, 2002 and on each anniversary thereafter, unless one party gives the other written notice sixty (60) days in advance of any renewal period. In no event shall the term of employment under this Agreement extend beyond December 31, 2005.
- 4. Compensation. All compensation under this Agreement is subject to withholding required by law.
  - a. Salary. Company agrees to pay Employee, in consideration for Employee's services hereunder, a salary at the rate of four-hundred fifty thousand dollars (\$450,000) annually, payable in equal semi-monthly installments or in accordance with Company's normal pay practices as may be altered from time to time by Company. Company shall review annually Employee's compensation and shall determine, in its sole discretion, whether and how much the existing compensation shall be adjusted.
  - b. Special Performance Award. Employee shall be eligible for a Special Performance Award, payable in stock options of The Coca-Cola Company, depending on performance against full year 2002 revenue dollar growth and operating income rate targets. The performance targets will be agreed upon shortly after the Close. For achieving the revenue and operating income targets, the stock option award will equal 20,000 stock options. The award will be granted in 1st Q 2003. Employee must have been an employee of Company during the entirety of calendar year 2002 to be eligible for the award payment.

### 5. Other Benefits.

- a. Company shall provide Employee with such fringe benefits as are normally provided to employees of Company holding positions and performing duties substantially similar to those performed by Employee. Employee will be entitled to five weeks of paid vacation per year. Employee will be covered under Company's directors and officer insurance program and will be indemnified to the fullest extent permitted by applicable law for Employee's actions taken on behalf of Company
- b. Employee shall be eligible to participate in The Coca-Cola Company's stock option program, as it exists from time-to-time, as if an employee at job grade level 17.

- 6. Expenses. Company shall reimburse Employee for reasonable travel and other business expenses, including business class air travel, incurred by Employee in the performance of his duties, in accordance with Company policies, as Company may amend them.
  - 7. Death, Incapacity or Illness of Employee.
- a. If Employee dies during the term of this Agreement, Company shall pay to the estate of the Employee only the compensation then due and owing. Except as herein provided, the death of Employee shall terminate this Agreement and discharge Company from any further liability for the payments provided herein.
- b. In the event Employee becomes disabled, ill or unable to perform his duties under this Agreement, his salary compensation shall continue for a period of one hundred and eighty (180) days or until the expiration of this Agreement, whichever comes first, and then will be suspended until the Employee can resume performance of his duties during the remaining term of this Agreement, if any.
- 8. Expiration. At the expiration of Employee's term of employment (in accordance with Paragraph 3 hereof), provided neither Company nor Employee has previously terminated this Agreement as described in Paragraphs 9, 10, 11, or 12 hereof, Company shall:
  - a. Pay Employee his base salary then in effect for a period of one year following the date of expiration (which amount shall be offset against any severance payments to which Employee might be entitled under any severance pay plan or policy of Company); and
  - b. Reimburse Employee for the cost of acquiring health benefits for Employee and his family through COBRA for a period of one year.
- 9. Termination for Cause. Company may terminate this Agreement immediately without liability or further obligation hereunder upon written notice to Employee if the Employee commits any one or more of the following acts:
  - a. Willful damaging of Company's property, business, reputation or  $\ensuremath{\mathtt{goodwill}}$ ;
    - b. Commission of a felony;
    - c. Death, theft, dishonesty, fraud or embezzlement;
  - d. Inattention to or neglect of the duties to be performed by Employee that is not the result of illness or accident and that is materially harmful to Company;
  - e. The use of alcohol, narcotics or other controlled substances to the extent that it prevents the Employee from efficiently performing services for Company;
    - f. Willfully injuring of any other employee of Company;
  - g. Willfully injuring any person in the course of performance of services for Company;
  - h. Disclosing to a competitor or other unauthorized persons confidential or proprietary information or secrets of Company;
  - i. Solicitation of business on behalf of a competitor or a potential competitor;
  - j. Harassment of any other employee of Company or the commission of any act which otherwise creates an offensive work environment for other employees of Company;
  - k. Failure of Employee for any reason within five days after receipt by Employee of written notice thereof from Company, to correct, cease or otherwise alter any insubordination, failure to comply with instructions or other act or omission to act that in the opinion of Company does or may materially adversely affect its business or operations; or
    - 1. Breach by Employee of any material term of this Agreement;

No act shall be willful if it is performed in good faith and with the belief that it is in Company's best interests. Company shall not be limited to termination as a remedy for any improper or illegal act of Employee, but may also seek damages, injunction or such other remedy as it may deem appropriate under the circumstances.

10. Termination Not for Cause. Company may terminate this Agreement at any time upon written notice to Employee for any reason not stated in Paragraph 9 hereof, provided Company:

- a. Pays Employee his base salary then in effect for a period of one year following the date of termination (which amount shall be offset against any severance payments to which Employee might be entitled under any severance pay plan or policy of Company); and
- b. Reimburses Employee for the cost of acquiring health benefits for Employee and his family through COBRA for a period of one year.
- c. In the event of a termination not for cause, the parties shall negotiate the treatment of the Special Performance Award.
- 11. Termination by Employee for Good Reason. Employee may terminate this Agreement for Good Reason at any time upon written notice to Company. "Good Reason" shall mean either: (i) Company causes Employee to cease to be Chief Executive Officer of Company and have a position, duties and responsibilities commensurate with such a title; (ii) Company reduces Employee's base salary; or (iii) Company requires Employee to relocate outside the San Francisco Bay area. In the event Employee terminates his employment for Good Reason, Company shall:
  - a. Pay Employee his base salary then in effect for a period of one year following the date of termination (which amount shall be offset against any severance payments to which Employee might be entitled under any severance pay plan or policy of Company); and
  - b. Reimburse Employee for the cost of acquiring health benefits for Employee and his family through COBRA for a period of one year.
  - c. In the event of a termination for Good Reason, the parties shall negotiate the treatment of the Special Performance Award.
- 12. Termination by Employee Not for Good Reason. In the event Employee terminates this Agreement not for Good Reason, Company shall have no further obligation to Employee under this Agreement.
- 13. Limitation on Payments. In the event that the severance and other benefits provided for in this Agreement or otherwise payable to Employee (i) constitute "parachute payments" within the meaning of Section 280G of the Internal Revenue Code of 1986 as amended (the "Code"), and (ii) would be subject to the excise tax imposed by Section 4999 of the Code (the "Excise Tax"), then Employee's benefits under this Agreement shall be either (a) delivered in full, or (b) delivered as to such lesser extent as would result in no portion of such benefits being subject to the Excise Tax, whichever of the foregoing amounts, taking into account the applicable federal, state and local income taxes and the Excise Tax, results in the receipt by Employee on an after-tax basis, of the greatest amount of benefits, notwithstanding that all or some portion of such benefits may be taxable under Section 4999 of the Code.

Unless Company and Employee otherwise agree in writing, any determination required under this Paragraph shall be made in writing by Company's independent public accountants (the "Accountants"),

whose determination shall be conclusive and binding upon Employee and Company for all purposes. For purposes of making the calculations required by this Paragraph 13, the Accountants may make reasonable assumptions and approximations concerning applicable taxes and may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. Company and Employee shall furnish to the Accountants such information and documents as the Accountants may reasonably request in order to make a determination under this Paragraph 13. Company shall bear all costs that the Accountants may reasonably incur in connection with any calculations contemplated by this Paragraph 13.

- 14. Limitations on Other Employment. During the term hereof, Employee shall not enter into the services of or be employed in any capacity or for any purposes whatsoever, whether directly or indirectly, by any person, firm, Company or entity other than Company, and will not, during said period of time, be engaged in any business, enterprise or undertaking other than employment by Company, with the exception that Employee may serve on the board of directors of charitable organizations, and may continue his business relationships with Avenal Land and Oil Company, Dalraddy Vineyards, and Kransco Inc., to the extent doing so will not violate Paragraph 15 hereof.
  - 15. Nonsolicitation and Noncompetition.
  - a. Employee agrees that he will not at any time while employed by Company and for a period of two years following the expiration or termination of his employment (the "Nonsolicitation Period"), whether voluntarily or involuntarily, directly or indirectly for himself or any other person or entity solicit, interfere with or endeavor to entice away from The Coca-Cola Company or any of its direct or indirect subsidiaries (collectively, "KO") any other Employee of KO. Additionally, Employee agrees that during the Nonsolicitation Period any employment by Employee or any entity in which he has an interest, directly or indirectly (other than

a publicly traded company in which he does not have a controlling interest) of any person who was in the employ of KO within the preceding year, shall be a violation of this paragraph.

- b. Employee agrees that he will not at any time while employed by Company and for a period of one year following expiration or termination of his employment (the "Noncompetition Period"), for himself or on behalf of any person, partnership, trust, company or other entity other than KO for whatever reason engage, directly or indirectly (either as an employee, officer, director, partner, shareholder, consultant or independent contractor), in the manufacture, sale, or distribution of non-alcoholic beverages in the United States. Employee further agrees that during the Noncompetition Period he will not be connected, directly or indirectly, with any person, the portion of any firm or company engaged in the manufacture, sale, or distribution of non-alcoholic beverages in the United States and that he will not directly or indirectly be employed or become a partner, officer or stockholder of the portion of any firm or company engaged in such business within the United States.
- c. If in any judicial proceeding, a court shall refuse to enforce this Agreement, whether because the time limit is too long or because the restrictions contained herein are more extensive (whether as to geographic area, scope of business or otherwise) than is necessary to protect the business and goodwill of KO, it is expressly understood and agreed between the parties hereto that this Agreement is deemed modified to the extent necessary to permit this Agreement to be enforced in any such proceedings.
- d. If Company or its successors in interest shall make application to a court of competent jurisdiction for injunctive relief, then the Nonsolicitation and Noncompetition Periods specified herein shall be tolled from the time of application for injunctive relief until the date of final adjudication of the claim for injunctive relief. Additionally, Employee waives, to the greatest extent permissible, any requirement that Company post bond or other security as a precondition to an injunction, whether temporary or permanent.
- e. Employee acknowledges that compliance with this paragraph is necessary to protect the goodwill and other proprietary interests of Company and that a breach of this paragraph will give rise to irreparable and continuing injury to Company that is not adequately compensable in monetary damages or at law. Accordingly Employee agrees that Company, its successors and assigns may obtain injunctive relief against the breach or threaten breach of the foregoing provisions, in addition to any other legal remedies which may be available to

it under this Agreement. Employee further acknowledges that in the event of his termination or expiration of employment with Company, his knowledge, experience and capabilities are such that Employee can obtain employment in business activities which are of a different or noncompeting nature than those performed in the course of employment with Company; and that the enforcement of a remedy hereunder by way of injunction will not prevent Employee from earning a reasonable livelihood.

# 16. CONFIDENTIAL INFORMATION, TRADE SECRETS, AND INTELLECTUAL PROPERTY.

- a. During the term of this Agreement and at all times thereafter, Employee will keep in confidence and will not publish, use or disclose to others, without the prior written consent of Company, any Trade Secrets or other confidential information related to KO or KO's business. As used herein, the phrase "Trade Secret" is to be considered as used in accordance with the definition of Trade Secret found under Georgia law in effect at the time of the execution of this Agreement. Without limiting this definition, and for information purposes only, a Trade Secret is the whole or any portion of any technical or non-technical information, including a formula, pattern, compilation, program, device, method, technique, drawing, process, financial data, financial plan, product plan or customer or supplier information that is actually or potentially valuable because it is not generally known to others and that is subject to reasonable efforts by KO to maintain its secrecy.
- b. Upon leaving the employ of Company, Employee will not take with him any written, printed or electronically stored Trade Secret, or other confidential information or any other property of KO obtained by him as the result of his employment, or any reproductions thereof. All such property and all copies thereof shall be surrendered by Employee to Company on termination of employment or at any time on request by Company.
- c. Employee's obligation not to use, publish or disclose confidential information of KO will cease five (5) years after termination or expiration of employment. Employee's obligation not to use, publish or disclose any Trade Secret of KO has no time limitation.
- d. Employee shall disclose to Company and agrees to and does hereby assign to Company, without charge, all his right, title and interest in and to any and all inventions and discoveries that he may make, solely or jointly with others, while in the employ of Company, that relate to or are useful or may be useful in connection with business of the character carried on or contemplated by KO, and

all his right, title and interest in and to any and all domestic and foreign applications for patents as well as any divisions or continuations thereof covering such inventions and discoveries and any and all patents granted for such inventions and discoveries and any and all reissues, extensions and revivals of such patents; and upon request of Company, whether during or subsequent to this employment, Employee shall do any and all acts and execute and deliver such instruments as may be deemed by Company necessary or proper to vest all Employee's right, title and interest in and to said inventions, discoveries, applications and patents in Company and to secure or maintain such applications, patents, reissues, extensions and/or revivals thereof. All necessary and proper expenses in connection with the foregoing shall be borne by Company, and if services in connection therewith are performed at Company's request after termination of employment, Company will pay reasonable compensation for such services. Any inventions and discoveries relating to KO's business made by Employee within one year after termination of employment with Company shall be deemed to be within this provision, unless Employee can prove that the same were conceived and made following said termination and such conception or invention is not based upon or related to any Trade Secret, as defined herein, received pursuant to Employee's employment with Company.

- e. Employee hereby assigns to Company, without charge, all his right, title and interest in and to all original works of authorship filed in any tangible form, prepared by him, solely or jointly with others, within the scope of his employment by Company. In addition, Company and Employee hereby agree that any such original work of authorship that qualifies as a "work made for hire" under the U.S. copyright laws shall be a "work made for hire" and shall be owned by Company.
- f. Employee acknowledges and agrees that in the event he breaches, threatens in any way to breach, or it is inevitable that he will breach, any of the provisions of this Paragraph, damages shall be an inadequate

remedy and Company shall be entitled, without bond, to injunctive or other equitable relief. Company's rights in this respect are in addition to all rights otherwise available at law or in equity.

- 17. Waiver or Modification. No waiver or modification of this Agreement or of any covenant, condition, or limitation herein contained shall be valid unless in writing and duly executed by the party to be charged therewith. Furthermore, no evidence of any modification or waiver shall be offered or received as evidence in any proceeding, arbitration or litigation between the parties arising out of or affecting this Agreement or the rights or obligations of any party hereunder, unless such waiver or modification is in writing, duly executed as aforesaid. The provisions of this paragraph may not be waived except as herein set forth.
- 18. Complete Agreement. This written Agreement contains the sole and entire agreement between the parties as to the matters contained herein and, effective upon the Close, supersedes any and all other agreements between them, including in particular the employment agreement dated December 21, 1999 between Company and Employee. The parties acknowledge and agree that neither of them has made any representation with respect to such matters of this Agreement or any representations except as are specifically set forth herein, and each party acknowledges that he or it has relied on his or its own judgment in entering into this Agreement.
- 19. Choice of Law. This Agreement and the performance hereunder and all suits and special proceedings hereunder shall be construed in accordance with the laws of the State of Georgia.
- 20. Invalid Provision. The invalidity or unenforceability of a particular provision of this Agreement shall not effect the other provisions hereto, and this Agreement shall be construed in all respects as if such invalid or unenforceable provisions were omitted.
- 21. Binding Effect of Agreement. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors, assigns and legal representatives.
- $22\,.$  Assignment. This Agreement shall be construed as a contract for personal services by Employee to Company and shall not be assignable by Employee.
- 23. Survival. The provisions of Paragraphs 15, 16, 17, 18, 19, 20, 21, 22 and 23 hereof shall survive the termination or expiration of this Agreement.
- 24. Notices. All notices, requests, demands and other communications required or permitted hereunder shall be in writing and shall be deemed to have been duly given when delivered by hand or when mailed by certified registered mail, return receipt requested, with postage prepaid to their current address or to such other address as they request in writing.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement the day and year first written above.

ODWALLA, INC.

/s/ JAMES R. STEICHEN

By: James R. Steichen

As its: SR VP, CFO

/s/ D. STEPHEN C. WILLIAMSON

D. Stephen C. Williamson